



DCN 87-212-027-27-01

AD-A190 44

INSTALLATION RESTORATION PROGRAM PHASE II - CONFIRMATION/QUANTIFICATION STAGE 1

MP FILE C

44" 44" 41" 41" 44" 44" 46" 46" 46" 46" 46" 46" 47" 47" 48" 48

FEB 0 4 1986

 $\mathbf{28}$

88

1

FINAL REPORT FOR AIR FORCE PLANT 4 FORT WORTH, TEXAS

VOLUME 1. REPORT TEXT

HEADQUARTERS AERONAUTICAL SYSTEMS LIVISION FACILITIES MANAGEMENT DIVISION (ASD/PMDA) WRIGHT-PATTERSON AIR FORCE BASE, OHIO 45433-6503

AND

HEADQUARTERS, AIR FORCE SYSTEMS COMMAND COMMAND BIOENVIRONMENTAL ENGINEER (AFSC/SGPB) ANDREWS AIR FORCE BASE, DC 20334-5000

DECEMBER 1987

PREPARED BY RADIAN CORPORATION 8501 MO-PAC BOULEVARD POST OFFICE BOX 201088 AUSTIN, TEXAS 78720-1088

USAF CONTRACT NO. F33615-83-D-4001 DELIVERY ORDER 27 RADIAN CONTRACT NO. 212-027-27

> APPROVED FOR PUBLIC RELEASE DISTRIBUTION UNLIMITED

USAFOEHL TECHNICAL PROGRAM MANAGERS MAJOR GEORGE R. NEW CAPTAIN ARTHUR S. KAMINSKI

UNITED STATES AIR FORCE OCCUPATIONAL & ENVIRONMENTAL HEALTH LABORATORY (USAFOEHL) BROOKS AIR FORCE BASE, TEXAS 78235-5501



NOTICE

This report has been prepared for the United States Air Force by Radian Corporation, for the purpose of aiding in the implementation of the Air Force Installation Restoration Program. It is not an endorsement of any product. The views expressed herein are those of the contractor and do not necessarily reflect the official views of the publishing agency, the United States Air Force, nor the Department of Defense.

Copies of this report may be purchased from:

National Technical Information Service 5285 Port Royal Road Springfield, Virginia 22161

Federal Government agencies and their contractors registered with Defense Technical Information Center should direct requests for copies of this report to:

> Defense Technical Information Center Cameron Station Alexandria, Virginia 22314

1	~	2	12	 د	/
6-	<u>د_</u> .	<i>'</i>			

Unclassified

In the second second

Sec. 35.

1

~

ዀ፝ቘዀ፝ዄቘዀኯቘቔኯጜቔዀጜቘዀኯቘዀኯቘዀኯቜዀኇቜዀኇዸቘኯ

SECURITY CLASSIFICATION OF THIS PAGE						
	REPORT DOCUM	ENTATION PAG	E			
18. REPORT SECURITY CLASSIFICATION Unclassified		16. RESTRICTIVE	TD-AI	190 44	7	
2. SECURITY CLASSIFICATION AUTHORITY N/A		3. DISTRIBUTION/A	VAILABILITY O	F REPORT		
26. DECLASSIFICATION/DOWNGRADING SCHEE N/A	DULE	Distributio	on is unlim	ited.		
4. PERFORMING ORGANIZATION REPORT NUM	BER(S)	5. MONITORING OR N/A	GANIZATION R	EPORT NUMBERIS)	
6. NAME OF PERFORMING ORGANIZATION Radian Corporation	66. OFFICE SYMBOL (If applicable)	7. NAME OF MONI		IZATION		
6c. ADDRESS (City, State and ZIP Code) 8501 Mo-Pac Blvd.	<u></u>	76. ADDRESS (City, Brooks AFB		1e) 78235-5501		
Austin, Texas 78759			,			
S. NAME OF FUNDING/SPONSORING ORGANIZATION USAFOEHL	86. OFFICE SYMBOL (<i>If applicable</i>) TS	9. PROCUREMENT I F33615-83-I			MBER	
Bc. ADDRESS (City, State and ZIP Code)	<u> </u>	10. SOURCE OF FUN	CING NOS.			
Brooks AFB, Texas 78235-550)1	PROGRAM ELEMENT NO.	PROJECT NO.	TASK NO.	WORK UNIT	
11. TITLE (Include Security Classification) IRP Phase II Stage 1, AF Plan	nt 4, Texas					
12. PERSONAL AUTHOR(S) Radian Corporation						
13. TYPE OF REPORT 13b. TIME C Final FROM Se	ED B5 TO Sep 86	14. DATE OF REPOR	at (Yr., Mo., Dey) mber 1987	15. PAGE CO	TUNT	
16. SUPPLEMENTARY NOTATION					×	
17. COSATI CODES	18. SUBJECT TERMS (C	ontinue on reverse if ne	cessary and identi	fy by block number)		
FIELD GROUP SUB. GR.	Installation R					
	Organics 🧲		0 ,			
19. ABSTRACT (Concrue on reverse if necessary and	identify by block number					
The Department of Defense's Installation Restoration Program (IRP) is a four-phase program to identify past waste disposal practices, evaluate the environmental impacts, propose mitigation measures, and remediate environ- mental problems. An IRP investigation is currently underway at AF Plant 4, Texas. Phase I, an initial assessment and records search to locate potential environmental problems associated with waste disposal practices, was completed in 1984. Phase II, a field program to investigate potential environmental contamination, has been under way since 1985 under the direction of the U.S. Air Force Occupational and Environmental Health Laboratory (USAFOEHL). Radian Corporation of Austin, Texas, has performed Stage 1 of the Phase II investi- gation which is reported in this document. The Phase II investigation has						
UNCLASSIFIED/UNLIMITED I SAME AS APT			•			

UNCLASSIFIED/UNLIMITED 🔯 SAME AS APT. 🗋	Unclassified	
22. NAME OF RESPONSIBLE INDIVIDUAL	22b. TELEPHONE NUMBER (Include Area Code)	22c. OFFICE SYMBOL
Maj. George R. New	512/536-2158	USAFOEHL/TSS
00 FORM 1473, 83 APR ED	ITION OF 1 JAN 73 IS OBSOLETE.	Unclassified

SECURITY CLASSIFICATION OF THIS PAGE

EXECUTIVE SUMMARY

INTRODUCTION

The Department of Defense (DOD) is conducting a nation-wide program to evaluate past waste disposal practices on DOD property, to control the migration of hazardous contaminants, and to control hazards that may result from these waste disposal practices. This program, the Installation Restoration Program (IRP), consists of four phases: Phase I, Initial Assessment/Records Search; Phase II, Problem Confirmation/Quantification; Phase III, Technology Base Development; and Phase IV, Operations. The United States Air Force is conducting an IRP investigation at Air Force Plant 4 (AF Plant 4) near Fort Worth, Texas. Radian Corporation has performed this Phase II Stage 1 Field Evaluation under USAF Contract No. F33615-83-D-4001, Delivery Order 27. Authorization to proceed on this study was given on 13 September 1985. Field activities were performed from November 1985 to August 1986.

PURPOSE OF THE INVESTIGATION

The purpose of the AF Plant 4 investigation was to define the presence, magnitude, extent, direction and rate of movement of any identified contaminants. In addition, the body of geological and chemical data developed by General Dynamics, operator of the plant, was to be used in the investigation. Finally, additional investigations required beyond this stage, including an estimate of costs (provided separately) were to be determined.

A contaminant source investigation at AF Plant 4 was conducted to determine: (1) the presence or absence of contamination within the specified areas of the field survey; (2) if contamination exists, the potential for migration in the various environmental media; (3) the extent/magnitude of contamination on AF Plant 4 property; and (4) the potential environmental

ዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀዀ



consequences and health risks of migrating contaminants, based on state or federal standards for these contaminants.

Operations at AF Plant 4 center around the production of military aircraft and associated equipment. Industrial operations have been continuous since AF Plant 4 went into production in 1942. The manufacturing of aircraft and associated equipment generates varying quantities of waste oils (including lubricating oils, hydraulic fluids, and coolants), recoverable fuels, solvents (including degreasers and paint thinners), paint residues (including waste liquid paints and paint booth sludges), and spent process chemicals (including acids, caustics, chromium, and cyanide solutions). The total quantity of these wastes currently ranges from about 5,500 to 6,000 tons/year. However, waste quantities depend on the level of contractor activity, which has varied with time.

ENVIRONMENTAL SETTING

AF Plant 4 is located six miles west of the center of Fort Worth in Tarrant County, Texas (Figure ES-1). The facility is bordered by Lake Worth on the north, Carswell AFB on the east, and the community of White Settlement on the west and south.

AF Plant 4 is situated in an area of primarily residential, recreational, and industrial/commercial land use. The topography of the plant is fairly flat except for areas near the creek parallel to Meandering Road. This creek discharges to Lake Worth, Farmer's Branch, and the Trinity River. Land surface slopes gently northeast toward Lake Worth and east toward the West Fork of the Trinity River. Elevations range from a high of approximately 627 feet MSL at the southwest corner of the plant to a low of approximately 600 feet MSL north of the plant at Lake Worth. Surface drainage at AF Plant 4 is intercepted by storm drains and culverts and is discharged to Lake Worth, the Meandering Road Creek, or a tributary of Farmer's Branch.



222222 1033333491 600352220 FEESSON PRIZZER 6.0.000 Sec. 50 1000 CO. 1000

Unclassified

ECURITY CLASSIFICATION OF THIS PAGE

19. Abstract (continued)

focused on closing existing data gaps and evaluating other study topics using the results of previous hydrogeologic investigations sponsored by General Dynamics Corporation, the operating contractor of AF Plant 4.

The Phase II field study was conducted between November 1985 and August 1986. The investigation focused on continued monitoring of existing wells at known waste disposal sites, developing new field data at previously uninvestigated sites, evaluating possible trends in groundwater development and implications for groundwater movement at AF Plant 4, and recommendations concerning future monitoring, additional field programs, and planning for remedial actions. The field program involved terrain conductivity surveys at 3 sites; drilling of 12 soil borings and 6 hand auger holes; installation of 9 shallow monitor wells and 4 Paluxy monitor wells; and the collection and chemical analysis of surface water samples at 11 locations, and groundwater samples at 113 wells.

Results of the Phase II field investigation have confirmed and supported earlier hydrogeologic studies conducted at AF Plant 4. A variety of organic compounds and heavy metals, some at high levels, occur as shallow groundwater contaminants associated with waste disposal sites. Groundwater in the Paluxy aquifer, separated from upper zone contamination by shale and limestone, is generally free of contamination except in a few isolated areas. Water quality in Meandering Road Creek, which forms the western boundary of AF Plant 4 and is a potential receptor of shallow groundwater contamination, has low to nondetectable levels of contamination.

Each of the 21 sites were categorized according to Air Force criteria: 1) no further investigation required (8 sites), 2) additional work needed (1 site), or 3) institute planning for remedial actions (12 sites).

SECURITY CLASSIFICATION OF THIS PAGE

PREFACE

Radian Corporation is the contractor for the Installation Restoration Program (IRP) Phase II, Stage 1 investigation at Air Force Plant 4, Texas. The work was performed for the USAF Occupational and Environmental Health Laboratory (USAFOEHL) under USAF Contract No. F33615-83-D-4001, Delivery Order Number 27.

Field investigation activities included: installation of nine upper zone and four Paluxy aquifer monitor wells; drilling of nine soil borings; completion of six hand-auger borings; conduct of two surface geophysical surveys and a soil gas survey; and collection of soil, groundwater and surface water samples for chemical analysis. Additionally, the effects of continued pumping of water supply wells on the Palauxy aquifer were evaluated and the existing monitor wells were evaluated for inclusion in a long-term monitoring program.

Key Radian project personnel were:

- Francis J. Smith Contract Program Manager
- o Lawrence N. French and William M. Little Delivery Order Managers
- Tobin K. Walters and Peter A. Waterreus Supervising Geologists and Report Authors
- Wendy J. Johnson Groundwater Sampling Team Leader and Report Author
- o Ann E. St. Clair Senior Technical Reviewer
- o Fred L. Shore Senior Chemistry Reviewer

Radian would like to acknowledge the cooperation and assistance of the USAF Plant 4 representative, Maj. Lussier. In addition, Radian acknowledges the assistance of W. C. Rosenthal, G. Long, and M. D. Stuckey of the General Dynamics staff.

The work presented herein was accomplished between November 1985 and September 1986. Maj. George R. New and Capt. Arthur S. Kaminski, Technical Services Division, USAFOEHL, were the Technical Program Managers.

Approved:

Francis J. Smith, P. E. Contract Program Manager



[This page intentionally left blank.]



ľ.

Ę

TABLE OF CONTENTS (Volume 1)

			Page
	PREFACE		v
	EXECUTIVE SUMM	IARY	ES-1
1.0			
	-	of the Investigation	
		of the Program	
		sposal Practices	
		and Site Descriptions	
		al Study Topics	
		and Analytical Program	
	1.7 Investig	ative Personnel	1-16
2.0	ENVIRONMENTAL	SETTING	2-1
	2.1 General	Geographic Setting and Land Use	2-1
		aphic and Topographic Features	
		and Hydrogeologic Conditions	
		criptions	
	2.4.1	Site 1, Landfill 1	
	2.4.2	Site 3, Landfill 3	
	2.4.3	Site 12, Chrome Pit 3	
	2.4.4	Site 17, Former Fuel Storage Site	2-28
	2.4.5	Site 2, Landfill 2	
	2.4.6	Site 4, Landfill 4	2-28
	2.4.7	Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2;	
		Site 8, Fire Department Training Area 5)	
	2.4.8	Site 15, Fuel Saturation Area 2	
	2.4.9	Site 20, Wastewater Collection Basins	
	2.4.10	Site 16, Fuel Saturation Area 3	
	2.4.11	Site 9, Fire Department Training Area 6	
	2.4.12	Site 6, Fire Department Training Area 3	
	2.4.13	Site 7, Fire Department Training Area 4	
	2.4.14	Site 18, Solvent Lines	
	2.4.15	Site 10, Chrome Pit 1	
	2.4.16	Site 5, Fire Department Training Area 2	
	2.4.17	Site 14, Fuel Saturation Area 1	2-36
	2.4.18	Site 19, Nuclear Aerospace Research Facility (NARF)	
	_	Area	
	2.4.19	Jet Engine Test Stand, Building 21	2-36

10

TABLE OF CONTENTS (Continued)

********************		1		₦₷₦₽₦₽₦₽₩₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽₽	1.00
RØ					
88					
	RA	DIA	N		
242	CORP!	DRATIC	i N		
KS .					
<u>85</u>				TABLE OF CONTENTS	
				(Continued)	
				(00/12/1002)	
					Deee
					Page
N.					~ ~
	3.0				
		3.1		echniques	
207 ⁻			3.1.1	Terrain Conductivity (Geophysical) Surveys	3-1
			3.1.2	Soil Gas Survey	3-2
			3.1.3	Drilling Techniques	3-4
			3.1.4	Monitor Well Installation and Development	
			3.1.5	Surveying	
			3.1.6	Environmental Sampling	
			3.1.7	Numbering System for Wells and Sampling Locations	
355					
			3.1.8	Field Safety	
		3.2		tivities	
N)			3.2.1	Site 1, Landfill 1	
			3.2.2	Site 3, Landfill 3	3-26
S.Z.			3.2.3	Site 12, Chrome Pit 3	3-28
			3.2.4	Site 17, Former Fuel Storage Site	3-28
			3.2.5	Site 2, Landfill 2	
			3.2.6	Site 4, Landfill 4	
			3.2.7	Zone 1: Site 13, Die Pits; Site 11, Chrome Pit 2;	
			5.2.7	Site 8; Fire Department Training Area 5	3-33
			3.2.8	Site 15, Fuel Saturation Area 2	
55 C			3.2.9	Site 20, Wastewater Collection Basins	
			3.2.10	Site 16, Fuel Saturation Area 3	
			3.2.11	Site 9, Fire Department Training Area 6	
			3.2.12	Site 6, Fire Department Training Area 3	
			3.2.13	Site 7, Fire Department Training Area 4	3-40
A.S.			3.2.14	Site 18, Solvent Lines	3-41
Ŧ			3.2.15	Site 10, Chrome Pit 1	3-42
			3.2.16	Site 5, Fire Department Training Area 2	
				Site 14, Fuel Saturation Area 1	
				Site 19, Nuclear Aerospace Research Facility (NARF)	5 45
ÊÇ -			J. Z. 10	Area	2.44
			3.2.19	Jet Engine Test Stand, Bldg. 21	
F.s		3.3		nal Study Topics	
			3.3.1	Ambient Monitoring	
			3.3.2	East Parking Lot	
			3.3.3	Fuel Storage Tank	3-50
			3.3.4	Lake Worth Monitor Well	3-50
		3.4	Hazardo	us Constituents in Soil and Drilling Fluids	3-50
				~	-
	4.0	DISCUS	SION OF	RESULTS AND SIGNIFICANCE OF FINDINGS	4-1
<u>K</u>	*• U	4.1		ologic Considerations	
C > 2		₩• L			
				Geologic Data	
				Hydrogeologic Data	
E.		4.2		cal Chemical Considerations	
E-s				Regulatory and Human Health Criteria	
			4.2.2	Quality Confirmation of Organic Chemical Data	4-24



TABLE OF CONTENTS (Continued)

	<u>Render an Reinfahran Reinfahra</u> n K		₽₽₩₽₩₽₽₩₽₽₩₽₽₩₽₩₽₩₽₩₽₩₽₩₽₩₽₩₽₽₩₽₽₩₽₽₩₽₽	1.11.11.11
	RADIA	NN .		
	C 0 E P 0 E A T	1 O N		
			TABLE OF CONTENTS	
			(Continued)	
			(0000-2000-2)	
				Page
02	4.3		ion of Results and Significance of Findings	
Ç.		4.3.1 4.3.2	Site 1, Landfill 1 Site 3, Landfill 3	
<u>.</u>		4.3.3	Site 12, Chrome Pit 3	
		4.3.4	Site 17, Former Fuel Storage Site	
		4.3.5	Site 2, Landfill 2	
<u>S</u> .		4.3.6	Site 4, Landfill 4	4-84
		4.3.7	Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2;	
13		4 2 0	Site 8, Fire Department Training Area 5)	
		4.3.8 4.3.9	Site 15, Fuel Saturation Area 2 Site 20, Wastewater Collection Basins	
		4.3.10	Site 16, Fuel Saturation Area 3	
		4.3.11	Site 9, Fire Department Training Area 6	
		4.3.12	Site 6, Fire Department Training Area 3	4-129
		4.3.13	Site 7, Fire Department Training Area 4	
		4.3.14	Site 18, Solvent Lines	
		4.3.15 4.3.16	Site 10, Chrome Pit 1 Site 5, Fire Department Training Area 2	
		4.3.17	Site 14, Fuel Saturation Area 1	
		4.3.18	Site 19, NARF	
		4.3.19	Jet Engine Test Stand, Building 21	
		4.3.20	Ambient Monitoring	
		4.3.21	East Parking Lot Wells	
		4.3.22 4.3.23	Fuel Storage Tank Lake Worth Monitor Well	
		4.3.24	White Settlement Groundwater Pumping Effects	
C 7.		4.3.25	Well Abandonment	
		4.3.26	Well Inventory	4-230
		4.3.27	Well Network Selection for Future Monitoring	
	, <i>,</i>	a 11.	Efforts	
<u>bal</u>	4.4	Quality 4.4.1	Assurance and Quality Control	
<u>кс</u> .		4.4.1	QA/QC Results for Water Sample Analyses	
		4.4.3	QA/QC Results for Soil Sample Analyses	
		4.4.4	QA/QC Results for Oil and Grease and Total	
			Hydrocarbons Analyses in Water and Soil Samples	
			by IR	
ra.		4.4.5	Radioactivity Analyses	
		4.4.6	Second Column Confirmation for GC Analyses	4-297
	5.0 ALTER	NATIVE M	EASURES	5-1
	5.1		Considerations	
	5.2		y 1 Sites	
	5.3	-	y 2 Site	
Ň		5.3.1	Additional Phase II Activities	
		5.3.2	Remedial Alternatives	5-4



tol [

TABLE OF CONTENTS (Continued)

	5.4	Category	y 3 Sites	j-4
	5.5	Remedial	l Alternatives	i-5
		5.5.1	Site 1, Landfill 1 5	
		5.5.2	Site 3, Landfill 3 5	
		5.5.3	Site 3, Chrome Pit 3 5	-8
		5.5.4	Zone (Site 13, Die Pits; Site 11, Chrome Pit 2;	
			Site 8, Fire Department Training Area 5) 5	-9
		5.5.5	Site 20, Wastewater Collection Basins 5	-10
		5.5.6	Site 16, Fuel Saturation Area 3 5	-11
		5.5.7	Site 9, Fire Department Training Area 6 5	-12
		5.5.8	Site 10, Chrome Pit 1 5	
		5.5.9	Site 5, Fire Department Training Area 2 5	-14
		5.5.10	Site, 14 Fuel Saturation Area 1 5	-15
6.0	RECOM	ENDATION	NS	-1
	6.1		7 1 Sites 6	
	6.2	Category	7 2 Site 6	-1
	6.3	Category	7 3 Sites 6	-2

20

1.0

 د راد راد را

· · · · ·



0.140.010.010.010.010.010.010.010.01

B

TABLE OF CONTENTS (Continued)

Appendix	<u>Title</u>	Volume
A-1	Water Analytical Data	2
A-1	Water Analytical Data (continued)	3
A-1	Water Analytical Data (Continued)	4
A-2	Water Quality Assurance/Quality Control Data	5
A-2	Water Quality Assurance/Quality Control Data	6
A-3	Soil Analytical Data	7
A-4	Soil Quality Assurance/Quality Control Data	
В	Definitions, Nomenclatures, and Units	8
С	Statement of Work	
D	Well Numbering System	
Е	Lithologic and Well Completion Logs	
F	Sampling and Analytical Procedures	9
G	References	
н	Biographies of Key Personnel	
I	Results of Terrain Conductivity and Soil Gas Surveys	
J	Health and Safety Plan	
ĸ	Technical Operations Plan	
L	Hargis & Associates, Inc. Water Quality Data	10



PERSONAL PROPERTY OF THE PERSON OF THE PERSO

[This page intentionally left blank.]

RADIAN

`\$<u>.#`</u>\$<u>.#`\$.#`\$.#`\$.#`\$.#`\$.</u>#`\$.

LIST OF FIGURES

48. "a 48." a 48." a 48." a

EXERCISE.

ESSERT ESSERT TREES

ŝ

Figure		Page
ES-1	General Location of AF Plant 4, Texas	
es-2 es-3	Phase II Stage 1 Study Sites, AF Plant 4, Texas Potentiometric Surface Map of the Upper Zone, AF Plant 4,	ES-/
5-5	Texas	ES-20
1-1	General Location of AF Plant 4, Texas	
1-2	Phase II Stage 1 Study Sites, AF Plant 4, Texas	1-7
2-1	Location Map of AF Plant 4, Texas	2-2
2-2	General Layout and Physical Features of AF Plant 4, Texas	2-3
2-3	Surface Drainage at AF Plant 4, Texas	
2-4	General Soil Association Map, AF Plant 4, Texas	
2-5	Generalized Stratigraphic Column at AF Plant 4, Texas	
26 27	General Geologic Map of AF Plant 4, Texas Generalized North-South Cross-Section through AF Plant 4,	2-11
2-1	Texas	2-12
2-8	Areal Extent of the Paluxy Aquifer, North Texas	
2-9	IRP Phase II Stage 1 Waste/Contamination Sites,	2 10
	AF Plant 4, Texas	2-22
2-10	Index Map for Individual Area Maps, AF Plant 4, Texas	
2-11	Location Map for Sites 1 and 3, Landfills 1 and 3;	
	Site 17, Fuel Saturation Area 1; and Site 5, Fire	
•	Department Training Area 2, AF Plant 4, Texas	2-24
2-12	Location Map for Sites 10 and 12, Chrome Pits 1 and 3;	
	Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; and	
	Site 8, Fire Department Training Area 5); and Site 20,	0 07
0 10	Wastewater Collection Basins, AF Plant 4, Texas	2-21
2–13	Location Map for Sites 2 and 4, Landfills 2 and 4; Site 17, Former Fuel Storage Site; and Site 6, Fire	
	Department Training Area 3, AF Plant 4, Texas	2-20
2-14	Location Map for Site 18, Solvent Lines and Sites	2-27
6 14	15 and 16, Fuel Saturation Areas 2 and 3, AF Plant 4,	
	Texas	2-32
2-15	Location Map for Building 21, Jet Engine Test Stand;	
	Sites 7 and 9, Fire Department Training Areas 4 and 6;	
	and Site 19, NARF Area, AF Plant 4, Texas	2-34
3-1	Index Map for Individual Area Maps, AF Plant 4, Texas	3-17
3-2	Phase II Activities at Sites 1 and 3, Landfills 1 and 3;	
	Site 17, Fuel Saturation Area 1; and Site 5, Fire	
	Department Training Area 2, AF Plant 4, Texas	3-25
3-3	Locations of Terrain Conductivity Surveys,	
	AF Plant 4, Texas	
3-4	Phase II Activities at Sites 10 and 12, Chrome Pits 1 and 3	5;
	Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; and Site 8, Bigs Department Training Area 5); and Site 20	
	Site 8, Fire Department Training Area 5); and Site 20, Wastewater Collection Basins, AF Plant 4, Texas	2_20
	wastewater contection dasins, AF fiant 4, lexas	3-23

xiii



RAD	AN	
	LIST OF FIGURES	
	(Continued)	
Figure		Page
3-5	Phase II Activities at Sites 2 and 4, Landfills 2 and 4;	
	Site 17, Former Fuel Storage Site; and Site 6, Fire	
2.6	Department Training Area 3, AF Plant 4, Texas	3-31
3-6	Phase II Activities at Site 18, Solvent Lines and Sites 15 and 16, Fuel Saturation Areas 2 and 3, AF Plant 4,	
		3-35
3-7	Phase II Activities at Building 21, Jet Engine Test Stand;	
	Sites 7 and 9, Fire Department Training Areas 4 and 6;	• • •
3_0	and Site 19, NARF Area, AF Plant 4, Texas	3-38
3-8	Hand Auger Sampling Locations, Site 9, Fire Department Training Area 6, AF Plant 4, Texas	3-39
3-9	Phase II Activities - Ambient Monitoring and Lake Worth	
	Monitor Well, AF Plant 4, Texas	3-48
3-10	Phase II Activities - East Parking Lot Monitor Wells,	a (a
4.1.1-1	AF Plant 4, Texas Geologic Cross Section C-C', AF Plant 4, Texas	
4.1.2-1	Potentiometric Surface Map of the Upper Zone,	45
	AF Plant 4, Texas	4~5
4.1.2-2	Potentiometric Surface Map of the Upper Paluxy Aquifer,	
4 1 2 2	AF Plant 4, Texas	4-7
4.1.2-3	Potentiometric Surface Map of the Middle Paluxy Aquifer, AF Plant 4, Texas	4-9
4.1.2-4	Potentiometric Surface Map of the Upper Zone in Area 1	- y
	(August, 1986), AF Plant 4, Texas	4-11
4.1.2-5	Potentiometric Surface Map of the Upper Zone in Area 2	
4.1.2-6	(February, 1986), AF Plant 4, Texas Potentiometric Surface Map of the Upper Zone in Area 3	4-12
	(February, 1986), AF Plant 4, Texas	4-14
4.1.2-7	Potentiometric Surface Map of the Upper Zone in Area 4	
	(February, 1986), AF Plant 4, Texas	4-15
4.1.2-8	Potentiometric Surface Map of the Upper Zone in Area 5 (February, 1986), AF Plant 4, Texas	4-17
4.1.2-9	Potentiometric Surface Map of the Upper Zone in the East	4-1)
	Parking Lot Area (February, 1986), AF Plant 4, Texas	4-19
4.3.1-1	Phase II Activities at Site 1, Landfill 1, AF Plant 4,	
4 2 1 2	Texas	
4.3.1-2 4.3.2-1	Surface Water Sampling Points, AF Plant 4, Texas Phase II Activities at Site 3, Landfill 3, AF Plant 4,	4~34
~ , J , 4 1	Texas	449
4.3.2-2	Location of Geophysical Survey, Site 3, Landfill 3,	
	AF Plant 4, Texas	4-53
4.3.3-1	Phase II Activities at Site 12, Chrome Pit 3, AF	4-61
4.3.4-1	Plant 4, Texas Phase II Activities at Site 17, Former Fuel Storage Site,	4-01
∽╻╻╻╼╴╻	AF Plant 4, Texas	4–69
	-	
	xiv	



LIST OF FIGURES (Continued)

Page

a mar and and har and mar har har har har

1 . Pa " . Pa" . Pa ' . Pa ' . Pa ' .

		ĨŸŀĨĔŇĨŶĸĔĬĸĔĬĸĔĬĸĔĬĸĔĬĸĔĬĸĔĬĸĔĬĸŎſĸŎſĸŎſĸŎſĸŎſĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸŎĬĸ	
	RADI	AN	
		LIST OF FIGURES	
88		(Continued)	
	Figure		Page
85			
500 52X	4.3.5-1	Phase II Activities at Site 2, Landfill 2, AF Plant 4, Texas	4-75
	4.3.5-2	Location of Geophysical Survey, Site 2, Landfill 2, AF Plant 4, Texas	4-79
83) 	4.3.6-1	Phase II Activities at Site 4, Landfill 4, AF Plant 4, Texas	4-85
200 200	4.3.7-1	Phase II Activities at Zone 1 (Site 8, Fire Department Training Area 5; Site 11, Chrome Pit 2; and Site 13,	4 05
	4.3.8-1	Die Pits), AF Plant 4, Texas Phase II Activities at Site 15, Fuel Saturation Area 2,	4-93
	4.3.9-1	AF Plant 4, Texas Phase II Activities at Site 20, Wastewater Collection	4-103
		Basins, AF Plant 4, Texas	4-109
	4.3.10-1	Phase II Activities at Site 16, Fuel Saturation Area 3, AF Plant 4, Texas	4-117
	4.3.11-1	Phase II Activities at Site 9, Fire Department Training Area 6, AF Plant 4, Texas	4-122
	4.3.11-2	Phase II Hand Augering Activities at Site 9, Fire	
	4.3.12-1	Department Training Area 6, AF Plant 4, Texas Phase II Activities at Site 6, Fire Department Training	
	4.3.13-1	Area 3, AF Plant 4, Texas Phase II Activities at Site 7, Fire Department Training	
	4.3.14-1	Area 4, AF Plant 4, Texas Phase II Activities at Site 18, Solvent Lines, AF Plant 4,	4-135
0535		Texas Phase II Activities at Site 10, Chrome Pit 1, AF Plant 4,	4-137
88 -		Texas	4-144
	4.3.16-1	Phase II Activities at Site 5, Fire Department Training Area 2, AF Plant 4, Texas	4-152
	4.3.16-2	Location of the Geophysical Survey, Site 5, Fire Department Training Area 2, AF Plant 4, Texas	
	4.3.17-1	Phase II Activities at Site 14, Fuel Saturation Area 1,	
	4.3.18-1	AF Plant 4, Texas Phase II Activities at Site 19, NARF Area, AF Plant 4,	
	4.3.19-1	Texas Phase II Activities at Jet Engine Test Stand, Building 21,	4–168
		AF Plant 4, Texas	
		Location of Outfalls 1 and 5, AF Plant 4, Texas Phase II Activities at Ambient Monitoring Wells,	4-1/4
		AF Plant 4, Texas	4-184
	4.3.21-1	Phase II Activities at East Parking Lot Wells, AF Plant 4, Texas	4-194
	4.3.22-1	Phase II Activities at Fuel Storage Tank, AF Plant 4, Texas	
	4.3.23-1	Location of Lake Worth Monitor Well (P-23U),	
Ges -		AF Plant 4, Texas	4-207

RAPIAN

LIST OF FIGURES (Continued)

1.44 . 44 . 44 . 44 . 4

Figure

Page

4.3.24-1	Simulated Drawdowns in the Paluxy Aquifer from White	
	Settlement Municipal Wells - Year 1	4-220
4.3.24-2	Simulated Drawdowns in the Paluxy Aquifer from White	
	Settlement Municipal Wells - Year 3	4-222
4.3.24-3	Simulated Drawdowns in the Paluxy Aquifer from White	
	Settlement Municipal Wells - Year 5	4-223
4.3.24-4	Simulated Drawdowns in the Paluxy Aquifer from White	
	Settlement Municipal Wells - Year 10	4-224
4.3.24-5	Simulated Drawdowns in the Paluxy Aquifer from White	
	Settlement Municipal Wells - Year 15	4-225
4.3.24-6	Simulated Drawdowns in the Paluxy Aquifer from White	
	Settlement Municipal Wells - Year 20	4-226
4.3.27-1	TCE Concentration Contour Map using Complete Data	
	Set, AF Plant 4, Texas	4-239
4.3.27-2	TCE Concentration Contour Map using Reduced Data	
	Set, AF Plant 4, Texas	4-247
4.3.27-3	Difference Contour Map for TCE Concentration,	
	AF Plant 4, Texas	4-248



1.8" B.8" B. 1" B. 8" B.

r

2.1

LIST OF TABLES

LUCKA BUND

Table		Page
ES-1	IRP Phase II Stage 1 Sites and Study Topics at AF Plant 4,	
ES-2	Texas	
ES-3	AF Plant 4, Texas Analytical Schedule for Soil and Water Samples, IRP Phase	ES-12
	II, Stage 1, AF Plant 4, Texas	ES-18
ES-4	Comparison of Groundwater Quality at Site 1, Landfill 1, AF Plant 4, Texas, with Applicable Water Quality	
	Criteria	ES-23
ES-5	Comparison of Surface Water Quality at Site 1, Landfill 1 AF Plant 4, Texas, with Applicable Water Quality	
	Criteria	ES-24
es-6	Comparison of Water Quality at Site 3, Landfill 3, AF	80.95
es-7	Plant 4, Texas, with Applicable Water Quality Criteria Comparison of Water Quality at Site 12, Chrome Pit 3, AF	E9-72
_	Plant 4, Texas, with Applicable Water Quality Criteria	ES-26
ES-8	Comparison of Water Quality at Site 2, Landfill 2, AF Plant 4, Texas, with Applicable Water Quality Criteria	PC_27
es-9	Comparison of Water Quality at Site 4, Landfill 4, AF	69-71
	Plant 4, Texas, with Applicable Water Quality Criteria	ES-28
ES-10	Comparison of Water Quality at Sites 13, 11, and 8	
	(Zone 1), AF Plant 4, Texas, with Applicable Water Quality Criteria	ES-29
ES-11	Comparison of Water Quality at Site 20, Wastewater	
	Collection Basins, AF Plant 4, Texas, with Applicable	Ra 20
ES-12	Water Quality Criteria Comparison of Water Quality at Site 16, Fuel Saturation	E2-30
	Area 3, AF Plant 4, Texas, with Applicable Water Quality	
DG 10	Criteria	ES-31
ES-13	Comparison of Water Quality at Site 10, Chrome Pit 1, AF Plant 4, Texas, with Applicable Water Quality Criteria	ES-32
ES-14	Comparison of Water Quality at Site 5, FDTA 2, AF	
70.15	Plant 4, Texas, with Applicable Water Quality Criteria	ES-33
ES-15	Comparison of Water Quality, Ambient Monitoring, AF Plant 4, Texas, with Applicable Water Quality Criteria	ES-34
ES-16	Comparison of Water Quality, East Parking Lot, AF	20 34
76 17	Plant 4, Texas, with Applicable Water Quality Criteria	
ES-17	Summary of Analytical Results of Soil Sample Analyses, IRP Phase II Stage 1, AF Plant 4, Texas	
ES-18	Summary of Recommendations from IRP Phase II Stage 1,	-0 0.
1_1		ES-40
1-1	IRP Phase II Stage 1 Sites and Study Topics at AF Plant 4, Texas	1-6
1-2	Analytical Schedule for Soil and Water Samples, IRP Phase	
1 0	II, Stage 1, AF Plant 4, Texas	1-17
1-3	AF Plant 4 IRP Phase II Study Team	1-18



LIST OF TABLES (Continued)

Table

Ō

Page

2222

LALANG MALANG BRANKI JANUNG BERERE DERERE

12222322

1.1.1.1.1.1.1

1222222222

2-1	Soil Associations for AF Plant 4, Texas	2-6
2-2	Geologic Formations Beneath AF Plant 4, Texas	2-10
2-3	Range of Constituents in Groundwater from Selected Wells	
	in the Paluxy Formation, Tarrant County	
2-4	Phase II Stage 1 Sites at AF Plant 4	2-21
3-1	Monitor Well Construction Specifications, IRP Phase II	
	Stage 1, AF Plant 4, Texas	3-8
3-2	General Specifications for Monitor Wells Installed During	
	IRP Phase II Stage 1, AF Plant 4, Texas	3-11
3-3	Collection and Preservation of Water Samples, IRP Phase II	
• •	Stage 1, AF Plant 4, Texas	3-13
3-4	Summary of IRP Phase II Stage 1 Investigation Activities,	
	AF Plant 4, Texas	3-19
3-5	Soil Samples Collected for Hazardous Constituents, AF	
	Plant 4, Texas	3-52
3-6	Maximum Concentration of Contaminants for Characterization	
	of EP Toxicity	3-54
4.2.1-1	Regulatory Standards for Inorganic Groundwater	
	Contaminants	
4.2.1-2	Guidelines for Organic Compounds Detected in Groundwater .	
4.2.1-3	Normal Ranges of Heavy Metal Concentrations Found in Soils	4-25
4.2.2-1	Action Limits for Second Column Confirmation on EPA 601	
	and 602 Organic Compounds	4-26
4.2.2-2	Analytes and Detection Limits for Methods 601, 8010, 624,	
	and 8240	
4.2.2-3	Analytes and Detection Limits for Methods 602 and 8020	
4.2.2-4	Analytes and Detection Limits for Methods 625 and 8270	4-30
4.3.1-1	Groundwater Elevations at Site 1, Landfill 1, AF Plant 4,	
	Texas (January through March 1986)	
4.3.1-2	Results of Groundwater Sample Analyses, Site 1, Landfill 1,	
	AF Plant 4, Texas	4-39
4.3.1-3	Results of Surface Water Sample Analyses, Site 1, Landfill	
	1, AF Plant 4, Texas	4-41
4.3.1-4	Comparison of Groundwater Quality at Site 1, Landfill 1,	
	AF Plant 4, Texas with Applicable Water Quality Criteria	4-44
4.3.1-5	Comparison of Surface Water Quality at Site 1, Landfill 1,	
	AF Plant 4, Texas with Applicable Water Quality Criteria.	4–45
4.3.2-1	Groundwater Elevations at Site 3, Landfill 3, AF Plant 4,	
	Texas (January Through February 1986)	4-51
4.3.2-2	Results of Groundwater Sample Analyses, Site 3, Landfill 3,	,
	AF Plant 4, Texas	
4.3.2-3	Comparison of Water Quality at Landfill 3, Landfill 3, AF	
	Plant 4, Texas with Applicable Water Quality Criteria	4-57
4.3.3-1	Groundwater Elevations at Site 12, Chrome Pit 3,	
	AF Plant 4, Texas (February 1986)	4-63
4.3.3-2	Results of Groundwater Sample Analyses, Site 12, Chrome	
	Pit 3, AF Plant 4, Texas	4-64

RADIAN

Table

LIST OF TABLES (Continued)

4.3.3-3	Comparison of Water Quality at Site 12, Chrome Pit 3, AF	1. 66
4.3.4-1	Plant 4, Texas with Applicable Water Quality Criteria Groundwater Elevations at Site 17, Former Fuel Storage	4-00
		4-71
4.3.4-2	Site, AF Plant 4, Texas (February through April 1986) Results of Soils and Groundwater Sample Analyses, Site 17,	4-/1
7.3.7 2	Former Fuel Storage Site, AF Plant 4, Texas	4.72
		4-72
4.3.5-1	Groundwater Elevations at Site 2, Landfill 2, AF Plant 4,	
	Texas (February through April 1986)	4-77
4.3.5-2	Results of Groundwater Sample Analyses, Site 2, Landfill	
	2, AF Plant 4, Texas	4-81
4.3.5-3	Comparison of Water Quality at Site 2, Landfill 2, AF	
	Plant 4, Texas With Applicable Water Quality Criteria	4-82
4.3.6-1	Groundwater Elevations at Site 4, Landfill 4, AF Plant 4,	
4.3.0 1		4-87
	Texas (February through April 1986)	4-0/
4.3.6-2	Results of Groundwater Sample Analyses, Site 4, Landfill	
	4, AF Plant 4, Texas	4–89
4.3.6-3	Comparison of Water Quality at Site 4, Landfill 4, AF	
	Plant 4, Texas, With Applicable Water Quality Criteria	4-90
4.3.7-1	Groundwater Elevations at Zone 1 (Sites 13, 11, and 8),	
	AF Plant 4, Texas, (February 1986)	4-96
4.3.7-2	Results of Groundwater Sample Analyses, Zone 1, (Sites 13,	
	11, and 8), AF Plant 4, Texas,	4-98
4.3.7-3	Comparison of Water Quality at Zone 1 (Sites 13, 11, and	4 90
4.3./-3	8), AF Plant, Texas, With Applicable Water Quality	
		4 100
	Criteria	4-100
4.3.8-1	Results of Groundwater Sample Analyses, Site 15, Fuel	
	Saturation Area 2, AF Plant 4, Texas	4-105
4.3.8-2	Results of Soil Sample Analyses, Site 15, Fuel Saturation	
	Area 2, AF Plant 4, Texas	4-107
4.3.9-1	Groundwater Elevations at Site 20, Wastewater Collection	
	Basins, AF Plant 4, Texas (February through April 1986)	4-111
4.3.9-2	Results of Groundwater Sample Analyses, Site 20, Waste-	
	water Collection Basins, AF Plant 4, Texas	4-113
4.3.9-3	Comparison of Water Quality at Site 20, Wastewater	
	Collection Basins, AF Plant 4, Texas, With Applicable	
		4-114
4 2 10 1		4-114
4.3.10-1	Results of Groundwater Sample Analyses, Site 16, Fuel	/ 110
	Saturation Area 3, AF Plant 4, Texas	4-119
4.3.11-1		
	AF Plant 4, Texas	4-124
4.3,11-2	Results of Soil Sample Analyses, Site 9, FDTA 6, AF Plant	
	4, Texas	4-127
4.3.12-1	Results of Groundwater Sample Analyses, Site 6, FDTA 3,	
	AF Plant 4, Texas	4-132
4.3.14-1		
	4, Texas (February 1986)	4-139
4.3.14-2	Results of Groundwater Sample Analyses, Site 18, Solvent	
	Lines, AF Plant 4, Texas	4-140

Page

No. Soloria

1.22

RADIAN

LIST OF TABLES (Continued)

Table

Page

4.3.14-3	Results of Soil Sample Analyses, Site 18, Solvent Lines,	
	AF Plant 4, Texas	4~142
4.3.15-1	Groundwater Elevations at Site 10, Chrome Pit 1, AF Plant	
	4, Texas (February through April 1986)	4-146
4.3.15-2	Results of Groundwater Sample Analyses, Site 10, Chrome	
4.5.15 2	Pit 1, AF Plant 4, Texas	4-147
1 3 15 3		4-14/
4.3.15-3	Comparison of Water Quality at Site 10, Chrome Pit 1, AF	
	Plant 4, Texas, With Applicable Water Quality Criteria	4-148
4.3.15-4	Results of Soil Sample Analyses, Site 10, Chrome Pit 1,	
	AF Plant 4, Texas	4-150
4.3.16-1	Groundwater Elevations at Site 5, FDTA 2, AF Plant 4,	
	Texas (February through March 1986)	4-154
4.3.16-2	Results of Groundwater Sample Analyses, Site 5, FDTA 2,	
	AF Plant 4, Texas	4-157
4.3.16-3	Comparison of Water Quality at Site 5, FDTA 2, AF Plant 4,	4 157
4.3.10-3		4 150
	Texas, With Applicable Water Quality Criteria	4-158
4.3.17-1	Groundwater Elevations at Site 14, Fuel Saturation 1,	
	AF Plant 4, Texas (February 1986)	4-163
4.3.17-2	Results of Groundwater Sample Analyses, Site 14, Fuel	
	Saturation Area 1, AF Plant 4, Texas	4-164
4.3.17-3	Results of Soil Sample Analyses, Site 14, Fuel Saturation	
	Area 1, AF Plant 4, Texas	4-166
4.3.18-1	Results of Soil Sample Analyses, Site 19, NARF Area, AF	. 100
4.5.10-1	Plant 4, Texas	4-171
4 2 10 1		4-1/1
4.3.19-1	Groundwater Elevations at Jet Engine Test Stand, Building	1
	21, AF Plant 4, Texas (August 1986)	4-176
4.3.19-2	Results of Surface Water Sample Analyses, Outfall 1, Jet	
	Engine Test Stand, Building 21, AF Plant 4, Texas	4-177
4.3.19-3	Results of Surface Water Sample Analyses, Outfall 5, Jet	
	Engine Test Stand, Building 21, AF Plant 4, Texas	4-178
4.3.19-4	Results of Groundwater Analyses, Jet Engine Test Stand,	
	Building 21, AF Plant 4, Texas	4-179
4.3.19-5	Results of Soil Sample Analyses, Jet Engine Test Stand,	
4.5.17 5	Building 21, AF Plant 4, Texas	4–181
4.3.20-1	Groundwater Elevations at Ambient Monitoring Locations,	4-101
4.3.20-1		/ *05
	AF Plant 4, Texas (January through August 1986)	4–185
4.3.20-2	Results of Groundwater Sample Analyses, Ambient Monitor-	
	ing, AF Plant 4, Texas	4-187
4.3.20-3	Comparison of Water Quality, Ambient Monitoring, AF Plant	
	4, Texas, With Applicable Water Quality Criteria	4-190
4.3.21-1	Groundwater Elevations at East Parking Lot Wells, AF	
	Plant 4, Texas, (February 1986)	4-196
4.3.21-2	Results of Groundwater Sample Analyses, East Parking Lot,	
	AF Plant 4, Texas	4-197
4.3.21-3		
4.3.21-3	Comparison of Water Quality, East Parking Lot, AF Plant 4,	
	Texas, With Applicable Water Quality Criteria	4-199
4.3.22-1	Results of Groundwater Sample Analyses, Fuel Storage	
	Tank, AF Plant 4, Texas	4-205

xx

RAPIAN

Table

LIST OF TABLES (Continued)

Page

4.3.23-1	Results of Groundwater Sample Analyses, Lake Worth Well	
	(P-23), AF Plant 4, Texas	4-209
4.3.24-1	City of White Settlement Well Data	4-212
4.3.24-2	Model Input Parameters	4-218
4.3.25-1	Excerpt from Texas Water Well Drillers Act Related Rules,	
	Texas Administrative Code Sections 319.49 and 50	4-229
4.3.26-1	Complete Well Inventory, AF Plant 4	4-231
4.3.26-2	Well Inventory of AF Plant 4, Arranged by Site	4-232
4.3.27-1	TCE Concentration Data Used in Well Network Selection:	
4.J.2/ I	Initial Data Set	4-240
4.3.27-2	TCE Concentration Data Used in Well Network Selection:	
4.3.21-2		4-244
	Reduced Data Set For Provide Manifesting of AF	4-244
4.3.27-3	Well Network Selection for Future Monitoring at AF	1 940
		4-249
4.4-1	Laboratory Blank Data Reported Above the Method Detection	
	Limit for Water Samples Analyzed for Organic Parameters	4-256
4.4-2	Laboratory Blank Data Reported Above Detection Limits	
	for Metals Analyses in Water Samples	4-257
4.4-3	Field Blank Data Reported Above the Method Detection	
	Limit for Groundwater Samples Analyzed for Organic	
	Parameters	4-258
4.4-4	Trip Blank Data Reported Above the Method Detection Limit	
	for Water Samples Analyzed for Organic Parameters	4-259
4.4-5	Summary of Surrogate Spiked Sample Recoveries for Water	
	Samples Analyzed for Purgeable Halocarbons and Aromatics .	4-263
4.4-6	Summary of EPA Method 625 Surrogate Spike Recovery	. 200
4.4-0	Results for Water Samples	4-264
4.4-7	Summary of Matrix Spiked Sample Recoveries for Water	4 204
4.4-/		4-265
	Samples Analyzed for Purgeable Halocarbons and Aromatics .	4-205
4.4-8	Summary of EPA Method 625 Matrix-Spiked Recovery Results	1 966
	for Water Samples	4-266
4.4-9	Summary of Matrix-Spiked Sample Recoveries for Metals	
	Analyses in Water Samples	4-269
4.4-10	Summary of QCCS Recoveries for Metals Analyses in Water	
	Samples	4-271
4.4-11	Analyses of Field Duplicate Water Samples for	
	Organic Parameters	4-273
4.4-12	Laboratory Duplicate Analyses of Water Samples for	
	Method 601 and 602 Organic Parameters	4-275
4.4-13	Laboratory Duplicate Analyses of Water Samples for	
	Method 625 Organic Parameters	4-277
4.4-14	Summary of Precision Estimates Based on Duplicate Water	
·•- •	Samples and Analyses for Metals	4~278
4.4-15	Reagent Blank Data Reported Above the Method Detection	
7.9-13	Limit for Soil Samples Analyzed for Organic Parameters	4-280
1. 1. 16		
4.4-16	Laboratory Blank Data Reported Above Detection Limits for	4-202
	Metals Analyses for Soil Samples	4-282

RADIAN

1.1

2.2

5

52.00.04389.0149.004

LIST OF TABLES (Continued)

ير ج

<u>Table</u>		Page
4.4-17	Summary of Surrogate-Spiked Sample Recoveries for Soil	
	Samples Analyzed for Purgeable Halocarbons and Aromatics .	4-283
4.4-18	Summary of EPA Method 8270 Surrogate-Spiked Recovery	
	Results for Soil Samples	4-284
4.4-19	Summary of EPA Method 8270 Matrix-Spiked Recovery	
	Results for Soil Samples	4-286
4.4-20	Summary of Matrix-Spiked Sample Recoveries for Metals	
	Analyses for Soil Samples	4-288
4.4-21	Summary of QCCS Recoveries for Metals Analyses for Soil	
	Samples	4-290
4.4-22	Duplicate Analyses for Soil Samples for Organic Analyses .	4-291
4.4-23	Summary of Precision Estimates Based on Duplicate Soil	
	Sample Analyses for Metals	4-292
4.4-24	Second Column Confirmation Results for Methods 601 and 602	4-298

xxíí

· . . .

. .

. .



Geologic units underlying AF Plant 4, from youngest to oldest, are as follows: 1) Quaternary Alluvium, 2) Cretaceous Goodland Limestone, 3) Cretaceous Walnut Formation, 4) Cretaceous Paluxy Formation. 5) Cretaceous Glen Rose Formation, and 6) Cretaceous Twin Mountains Formation. Only the first four are exposed at the surface at the plant. Most of the plant is covered by alluvium deposited by the Trinity River during flood stages. The alluvium is composed of gravel, sand, silt, and clay of varying thicknesses and lateral extents.

AF Plant 4 is located in a structurally stable setting. No major faults or fracture zones have been mapped near the base. The regional dip of the rocks beneath AF Plant 4 is between 35 and 40 feet per mile in an easterly to southeasterly direction.

On the basis of their water-bearing properties, the geologic units at AF Plant 4 may be divided into five hydrogeologic units, listed in order of increasing depth: 1) an upper perched-water zone occurring in the alluvial terrace deposits left by the Trinity River; 2) an aquitard of predominantly dry limestone of the Goodland and Walnut Formations; 3) an aquifer in the Paluxy Sand; 4) an aquitard of relatively impermeable limestone in the Glen Rose Formation; and 5) a major aquifer in the sandstone of the Twin Mountains Formation.

Groundwater present in the upper zone occurs as a perched system and, in general, is not hydraulically connected to the underlying aquifers. For the most part, groundwater in the upper zone is not developed because of the water's limited distribution and susceptibility to surface pollution. The perched water present in the upper zone is separated from the aquifers below by the low permeability limestones and shales of the Goodland Limestone and Walnut Formation.

The Paluxy aquifer, the shallowest bedrock aquifer occurring beneath AF Plant 4, is divided into upper and lower water-bearing zones. Most wells are completed in the lower Paluxy. Regional groundwater flow within the

ES-4



Paluxy is southeastward, in the direction of the regional dip. At Plant 4, groundwater flow is influenced by recharge from Lake Worth, which creates a potentiometric high, and by groundwater withdrawals by the community of White Settlement. This drawdown results locally in a more southerly flow direction within the Paluxy aquifer.

Below the Paluxy aquifer is the Glen Rose Formation, an aquitard restricting water movement between the Paluxy aquifer above and the Twin Mountains aquifer below. The Twin Mountains aquifer is the principal aquifer in Tarrant County.

SITE DESCRIPTIONS

Phase II Stage 1 work at AF Plant 4 has focused on the 21 waste sites and four study topics listed in Table ES-1 and shown in Figure ES-2. Most of the site information was obtained from the Installation Restoration Program Phase I IRP records search report (CH2M Hill, 1984). These IRP sites are described briefly in the following paragraphs.

Site 1, Landfill 1

From 1942 until about 1966, several types of hazardous and nonhazardous wastes were reportedly disposed of in Landfill 1. These wastes included drums of liquid and contents of tanks and bousers loaded with chemical wastes (solvents, thinners, paints, etc.). Contaminated oils, fuels, and hazardous wastes containing metals, chromate sludges, and cyanide are suspected to be present at this site.

Since it was closed during the 1960s, extensive remedial action has been conducted at this site. The major action was the removal of 11,000 cubic yards of earth and the installation of a drain field (French Drain No. 2).

Site 3, Landfill 3

From about 1942 to 1945, this site was reportedly used for the disposal of wastes, including hazardous wastes consisting of mixed oils and solvents. At least one pit (possibly more) in this area was used for holding and burning some of the liquid wastes during the 1940s. Other wastes are suspected to have been disposed of on the ground and later buried. From 1966 until 1967, fill dirt and rubble were used to finish filling and grading this site.



8.8.9.8.9.9.8.9.1.9.0.9.8.9.1.9.

TABLE ES-1. IRP PHASE II STAGE 1 SITES AND STUDY TOPICS AT AF PLANT 4, TEXAS

SITES:

Site 1, Landfill 1 Site 3, Landfill 3 Site 12, Chrome Pit 3 Site 17, Former Fuel Storage Site Site 2, Landfill 2 Site 4, Landfill 4 Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; Site 8, Fire Department Training Area 5) Site 15, Fuel Saturation Area 2 Site 20, Wastewater Collection Basins Site 16, Fuel Saturation Area 3 Site 9, Fire Department Training Area 6 Site 6, Fire Department Training Area 3 Site 7, Fire Department Training Area 4 Site 18, Solvent Lines Site 10, Chrome Pit 1 Site 5, Fire Department Training Area 2 Site 14, Fuel Saturation Area 1 Site 19, Nuclear Aerospace Research Facility (NARF) Area Jet Engine Test Stand, Building 21

ADDITIONAL STUDY TOPICS:

East Parking Lot Fuel Storage Tank Lake Worth Monitor Well Ambient Monitoring



THE REPORT OF THE PROPERTY OF

 ~ 100

ES-7

Site 12, Chrome Pit 3

Chrome Pit 3 was used for the disposal of chromate and other chemical wastes from about 1957 until 1973. Barium chromate sludge, dilute metal solutions, and drums of unidentified liquids were disposed of in this pit.

Site 17, Former Fuel Storage Site

A 100,000-gallon above-ground JP-4 storage tank existed at this site from the early 1940s until it was relocated from here in 1962. Sampling at this site in 1982 confirmed that soils and underlying groundwater are contaminated by fuels and other organic compounds.

Site 2, Landfill 2

The site originally consisted of some low areas and a livestock watering hole. Most of this site was reportedly filled with construction rubble, plaster, and fill dirt during the early 1940s. However, 1962 aerial photographs show some unspecified activity at the stock watering hole. This area was reportedly used for the disposal of lumber and tires that were assumed to be periodically burned. No reports exist of hazardous substances being deposited at this site. The Phase I report provided no additional data on the nature of the activity at this site.

Site 4, Landfill 4

Landfill 4 was reportedly used for the disposal of clean construction rubble from 1956 until the early 1980s. Aerial photographs and a 1973 memo indicate that other types of wastes may have been disposed of at this site from 1966 until at least 1973. This evidence suggests that small quantities of hazardous wastes (solvents, oils, fuels, thinners, etc.) are present in this landfill.

Zone 1

Three sites were combined into one zone for study purposes. Descriptions of these sites are given below.

Site 13, Die Pits: These pits were used for the disposal of chromate sludges, metal solutions, and other chemical wastes until 1962, when the site was graded and the entire die yard was paved. One Phase I interviewee reported that some of the contaminated soils at this site were spread around the die yard during the grading and leveling activities. The site of the original pits was excavated in 1983-84. Soils from other parts of the die yard were not analyzed at that time.

Site 11, Chrome Pit 2: According to the Phase I study, miscellaneous liquid and solid wastes, in addition to chromate solutions, were probably disposed of here. Neither interviews nor aerial photographs could confirm the exact location of this site.

Site 8, Fire Department Training Area (FDTA) 5: This site consisted of a shallow pit in which waste fuels, oils, and chemicals were burned and extinguished for training exercises. This site, which has been graded and paved, is located in the die yard area south of Warehouse 1.

Site 15, Fuel Saturation Area (FSA) 2

This site reportedly became saturated with fuels from leaks in buried fuel lines between the 1970s and early 1980s.

Site 20, Wastewater Collection Basins

Two concrete-lined waste basins, each with a capacity of approximately 85,000 gallons, are used to collect and settle suspended solids from chemical wastewaters before discharge to the City of Fort Worth sanitary sewage system. The basins have been in use since about 1966. Evidence indicates that several spills from the vapor degreaser tanks in the Process Building have occurred since installation of these tanks. Some of the spilled chemicals (primarily trichloroethylene) have flowed to the basins via floor drains. These basins are suspected of being a possible source of organic heavy and metal groundwater contamination.

Site 16, Fuel Saturation Area 3

This site reportedly became saturated with fuels from leaks in buried fuel lines between the mid-1970s and the early 1980s. This site is the subject of current remedial action planning by Intellus.

Site 9, Fire Department Training Area 6

Before 1970, training exercises were conducted twice a year at this site. After 1970, exercises were conducted at monthly intervals. Approximately 250 gallons of waste fuels and oils were reportedly used for each exercise. In addition, the Phase I report indicates that larger quantities of contaminated fuels and oils were probably deposited in the FDTA between exercises.

Site 6, Fire Department Training Area 3

Training exercises at this site also used about 250 gallons of waste fuels and oils per exercise. The location and current condition of FDTA 3 could not be accurately determined because it is not visible on historical aerial photographs.



Site 7, Fire Department Training Area 4

Fire training exercises were reportedly conducted here. Since this site is not visible on historical aerial photographs, its location and current condition are not accurately known. The area is also believed to have received fill material originating from a foundation excavation at the Administration Building.

Site 18, Solvent Lines

These lines reportedly leaked during the 1940s before they were drained, capped and abandoned in 1944. The Phase I study could not determine the actual locations of the leaks. These solvent lines reportedly contained xylene, methyl ethyl ketone, and kerosene.

Site 10, Chrome Pit 1

Miscellaneous liquid and solid chemical wastes, in addition to chrome wastes, were probably disposed of at this site. The actual location of this site is thought to be somewhere beneath the Process Building, but the Phase I report could not accurately confirm this. No monitor wells exist in the immediate vicinity of this site.

Site 5, Fire Department Training Area 2

Exercises were held infrequently (twice a year) at this site. However, disposal of waste oils and fuels and uncontrolled burns may have been more frequent. This site is located under the pavement in the west employee parking area.

Site 14, Fuel Saturation Area 1

The ground at this site, located just west of the Parts Plant, reportedly became saturated by fuels from leaking fuel lines from the mid-1970s to the early 1980s. This site is the subject of Phase IV Remedial Action planning by Intellus.

Site 19, Nuclear Aerospace Research Facility (NARF) Area

The Nuclear Aerospace Research Facility (NARF), formerly located at the north end of AF Plant 4, housed several experimental atomic reactors between 1953 and 1974. The facility was decommissioned and disposed of by contractor in 1974.

Jet Engine Test Stand, Building 21

This site was not identified in Phase I. The jet engine test building is located north of a fuels test area, previously identified as Site 16, a known area of fuel contamination. The sump downslope of Building 21. constructed in 1975, collects water used for cooling, noise suppression, and clean-up in Building 21. This water is pumped into the industrial waste line. The area immediately south of Building 21 contains two underground tanks once used for fuel storage. These tanks are being investigated as potential sources of contamination under a separate underground tank investigation being conducted as part of IRP activities by the operating contractor, General Dynamics. Outfall 3 is downgradient of Building 21 and is regularly sampled as part of IRP and NPDES monitoring activities.

SCOPE OF PHASE II STAGE 1 INVESTIGATION

Table ES-2 summarizes the field activities performed at each site in the IRP Phase II Stage 1 investigation at AF Plant 4. Table ES-3 is an analytical schedule for all samples, by site.

In addition to the studies associated with the waste sites, additional study topics were identified for inclusion in the Phase II Stage 1 field investigation. These included:

- Ambient monitoring of several wells at AF Plant 4 that are not associated with a particular waste disposal site or suspected spill area;
- Sampling and analysis of five wells located in the vicinity of the East Parking Lot to test for the presence of contaminants in the vicinity of a suspected zone of communication between the upper zone and the Paluxy aquifer;
- Sampling an existing upper zone monitor well for the presence of fuel contamination near an above-ground JP-4 storage tank; and
- o Installation, sampling, and analysis of a well in the northern part of AF Plant 4 that borders Lake Worth.

Site	Investigation Activities	Purpose	Map Location
Landfill 1	 Sample surface water at 5 loca- tions on Meandering Road Creek and the creek seep. 	1.	Figure 3-2.
	2. Sampling French drains J and 2 and the drain pipe.	 2. Unaracterize leacnate discharge. and 3. Evaluate possible effect of landfill on groundwater. 	
	 Sample groundwater from 6 existing upper zone and 2 Paluxy wells. 		
Landfill 3	1. Conduct a terrain conductivity survey.	 Determine presence and possible configuration of the landfill. Use data to determine if 	Figure 3-2.
	 Install Paluxy monitor well screened in upper sand member. 		
	 Sample groundwater from 9 existing upper zone wells and 1 new Paluxy well. 	2. Evaluate directly landfill.	
		 Evaluate possible effect of landfill on groundwater. 	
Chrome Pit 3	 Sample groundwater from 8 existing upper zone wells and 1 Paluxy well. 	ing 1. Deterwine post-closure groundwater quality.	Figure 3-4.

F

La (4.6) 4.8 (4.6) 4.6 (4.8) 4.8 (4.6) 4.6 (4.6) 4.6 (4.6) 4.6 (4.6) 4.6 (4.6) 4.6 (4.6) 4.6

ES-12
Site	Investigation Activities	Purpose	Map Location
Former Fuel 1 Storage	1. Install 1 upper zone well.	1. Provide data on groundwater flow and quality in Radar Range area.	Figure 3-5.
0	 Sample groundwater from 2 existing and wells 1 new upper zone well. 	 Evaluate possible effect of former fuel storage site on groundwater. 	
Landfill 2 1	1. Conduct a terrain conductivity survey.	 Determine landfill limits and horizontal extent of contamination, if any. 	Figure 3-5.
7	 Install Paluxy monitor well in upper sand member. 	2. Evaluate Paluxy groundwater directly downgradient of	
m	 Sample groundwater from 7 upper zone wells and 1 new Paluxy well. 	landfill. 3. Evaluate possible effects of landfill on groundwater.	
Landfill 4 1	 Install 1 upper zone well and 1 Paluxy well screened in middle sand unit (upper unit was dry). 	<pre>1.2. Determine possible effects of landfill on groundwater.</pre>	Figure 3~5.
2	 Sample groundwater from 3 upper zone wells and 1 Paluxy well. 		

KKKKKK

والمتعظمات

S. ₹

Ŗ

TABLE ES-2. (Continued)

ES-13

olle	Investigation Activities	Purpose	Location
Zone 1	1. Drill borehole at FDTA 5. 2. Sample groundwater from 7 existing	 Determine soil conditions and presence of water underlying FDTA 5. 	Figure 3-4.
	upper zone wells and 1 faluxy well.	 Evaluate possible effects of disposal pits and fire training area on groundwater. 	
Fuel Saturation Area 2	1. Drill 3 boreholes along the length of the buried fuel line.	 Determine soil conditions and presence of fuel in the subsurface. 	Figure 3-6.
	 Sample groundwater and measure fuel lens at 1 upper zone well. 	 Evaluate possible effect of fuel line on groundwater. 	
Wastewater Collection Basins	 Install 1 upper zone monitor well (located south of site because of access restrictions). 		Figure 3-4.
	 Sample groundwater from 5 upper zone wells. 	 Evaluate effect of lines, if any, on movement of contamination in subsurface. 	
	 Determine location and depth of buried sanitary and industrial waste lines. Evaluate need for additional sampling and analysis of soil. 		

. . .

No. No.

2222

ł,

ES-14

Site	Investigation Activities	Purpose	Map Location
Fuel 1. So Saturation we Area 3	Sample groundwater at 1 upper zone well.	 Evaluate possible effect of buried fuel line on groundwater. 	Figure 3-7.
Fire 1. Hd Department so Training 2. So Area 6 2. So we	Hand-auger and collect shallow soil samples for analysis. Sample groundwater from 1 Paluxy well.	 Determine possible effects of fire training exercises on shallow soil. Evaluate groundwater quality of Paluxy upgradient of AF Plant 4. 	Figures 3-7.8.
Fire 1. I Department we Training a: Area 3 2. S	Install 1 upper zone monitor well west of FDTA 3 (monitor well already exists at site). Sample groundwater from 2 upper zone wells.	 Determine groundwater flow conditions west of FDTA 3. Evaluate possible effects of FDTA 3 on groundwater. 	Figure 3-5.
Fire 1. G Department Training Area 4	Conduct a soil-gas survey.	 Determine location of FDTA 4 and possible effects on shallow soil. 	Figure 3-7.
Solvent 1. I. Lines 2. S. 2. 2.	Install 1 upper zone monitor well. Sample groundwater from 4 upper zone monitor wells.	 Improve monitoring coverage of groundwater conditions. Evaluate possible effect of solvent lines on groundwater. 	Figure 3-6.

\mathbf{S}
1
Ŕ.
R.
X 2
R
2
5
52222
PC -
6
16. 1
1.5
R.
F
M
E.
.
\mathbb{N}
F
E
ra.
\mathbb{R}
N.
1. V

Site	Investigation Activities	Purpose	Map Location
Chrome Pit 1	 Install upper zone monitor well east of site (outside of building). 	 Evaluate groundwater conditions east of Chrome Pit 1. 	Figure 3-4.
	2. Sample groundwater from 2 upper zone wells.	 Evaluate possible effect of Chrome Pit 1 on groundwater. 	
nt	 Conduct a terrain conductivity survey. 	 Determine the extent of a hydrocarbon plume, if any. 	Figure 3-2.
Iraining Area 2	 Sample groundwater from 7 upper zone wells. 	 Evaluate possible effect of FDTA on groundwater. 	
Fuel Saturation	1. Drill 1 borehole near FSA 1.	 Determine soil conditions and presence of fuel in subsurface. 	Figure 3-2.
	 Sample groundwater from 2 upper zone wells. 1 upper zone borehole. and 2 Paluxy wells. 	2. Evaluate effects of FSA 1 on groundwater.	
NARF	1. Drill 4 boreholes in the upper zone.	 Determine shallow soil conditions and presence of RAM in soil. 	Figure 3-7.
	 Sample groundwater from 3 upper zone wells. 	 Evaluate possible effect of NARF on groundwater. 	
	 Conduct review of NARF and White Settlement records for radionuclide analyses. 	 Determine past operations and exact location of NARF. 	

(continued)

ES-16

TABLE ES-2. (Continued)

6

Site		Investigation Activities	Purpose	Map Location
Jet Engine Test Stand	1 •]	1. Install 3 upper zone wells and drill 2 boreholes.	 Collect and analyze soil samples and evaluate groundwater flow and chemistry. 	Figure 3-7.
	2.	 Sample groundwater from 3 upper zone monitor wells. 	 Evaluate possible effect of Jet Engine Test Stand on groundwater. 	
		3. Sample surface water at Outfalls 1 and 5.	3. Evaluate effect of discharge to surface water near facility.	
Ambient Monitoring	•	 Sample groundwater from 11 upper zone wells and 8 Paluxy wells. 	 Evaluate groundwater quality in areas away from waste disposal spill sites. 	Figure 3-9.
East Parking Lot	₹ « C	 Sample groundwater from 3 upper zone monitor wells and 2 Paluxy wells. 	 Evaluate groundwater quality at east portion of AF Plant 4. 	Figure 3-10.
Fuel Storage Tank		1. Sample groundwater from upper zone well HM-23.	 Evaluate possible effect of the Fuel Storage Tank on groundwater. 	Figure 3-5.
Lake Worth Monitor Well	 	1. Install 1 Paluxy well near Lake Worth.	 Evaluate groundwater quality of Paluxy aquifer near recharge source, Lake Worth. 	Figure 3-9.

RADIAN

ES-17



TABLE ES-3. ANALYTICAL SCHEDULE FOR SOIL AND WATER SAMPLES, IRP PHASE II, STAGE 1, AF PLANT 4, TEXAS

Site	Location	Sample Type	voc	BNA	0&G	Heavy Metals	Fuel HC	Xylene	Cr	MEK	RAM
		-71-									
1	Landfill 1	G	X	Х		Х					
		W	X	Х		Х					
3	Landfill 3	G	X	Х		Х					
12	Chrome Pit 3	G	Х	X		Х					
17	Former Fuel Storage Site	G					Х				
		S					Х				
2	Landfill 2	G	X	X		Х					
4	Landfill 4	G	Х	X		X					
		S	X	X		X					
	Zone 1	G	X	X	Х	Х					
		S	X	X		Х					
15	Fuel Saturation Area 2	G	Х				Х				
		S	Х				Х				
20	Wastewater Collection Basins	G	X		Х	X	Х				
		S	Х		Х	Х	Х				
16	Fuel Saturation Area 3	G	X				Х				
9	Fire Dept. Training Area (FDTA) 6	G	х	Х	х		Х				
		S	X	X	х		Х				
6	FDTA 3	G	X		х		Х				
18	Solvent Lines	G			х			Х		х	
		S			X			Х		х	
10	Chrome Pit 1	G	X						Х		
		S	Х						Х		
5	FDTA 2	G			X		х				
14	Fuel Saturation Area 1	G					Х				
		S					х				
19	NARF Area	G	X	Х	Х	X	Х				х
		S									X
	Jet Engine Test Stand,	G			X		х				
	Building 21	S			X		х				
	5	W		X	Х		x				
	Ambient Monitoring	G	X	X	X	Х	х		х		
	East Parking Lot	G	X	Х	X	X	х		х		
	Fuel Storage Tank	G			X		Х				
	Lake Worth Monitor Well	G	Х	Х		X					

Samples: G = Groundwater; W = Surface Water; S = Soil

VOC = Volatile Organic Compounds

BNA = Base Neutral and Acid Extractable Compounds

O&G = Oil and Grease

Fuel HC = Hydrocarbon Fuels

Cr = Chromium

MEK = Methyl Ethyl Ketone

RAM = Radioactive Materials (Alpha, beta, and gamma radionuclides)

In addition to the field activities identified above, the following tasks were performed in this study:

- Evaluation of the long-term effects of pumping by the City of
 White Settlement on water levels and flow directions in the
 Paluxy aquifer;
- o A literature search of local hydrogeologic conditions;
- o Inventory of wells existing on AF Plant 4 at the time of this investigation and categorization of the wells by waste site;
- Review of the existing system of monitor wells at AF Plant 4 and recommendation of an optimal well network for future monitoring efforts; and
- Review and recommendation of available methods of well abandonment.

DISCUSSION OF RESULTS AND SIGNIFICANCE OF FINDINGS

Hydrogeology

The IRP Phase II Stage 1 investigation conducted at AF Plant 4 has been one of several efforts to define the nature and extent of contamination at the facility. The geologic setting and the occurrence and character of the groundwater have been previously described, so that the emphasis of investigation has been to fill in gaps in the data and focus on investigating areas not previously studied. Generally, the results of the IRP Phase II Stage 1 investigation confirm hydrogeologic interpretations of the previous work.

Hydrogeologic data gathered for this investigation consisted of water-level readings at existing and new upper zone and Paluxy monitor wells. Figure ES-3 depicts the potentiometric surface in the upper zone, based on Phase II Stage 1 monitoring. The occurrence and flow of groundwater is



ES-20

 \sim



related closely to the distribution and thickness of alluvial material, the bedrock topography, and surface elevations.

Groundwater flow in the Paluxy aquifer is generally to the southeast at AF Plant 4. Potentiometric surface maps of both the upper and middle members of the Paluxy, based on Phase II Stage 1 data, support earlier interpretations of the potentiometric surface and groundwater flow in the Paluxy.

Results and Significance of Chemical Analyses

To determine possible water quality effects on the local groundwater systems, concentrations of organic and inorganic compounds detected in groundwater samples were compared to various water quality criteria. These criteria, from federal drinking water regulations, standards, and guidelines, include final and proposed Maximum Contaminant Levels (MCLs) and proposed Recommended Maximum Contaminant Levels (RMCLs) above zero (0), established by the EPA as part of national drinking water regulations. The RMCLs are nonenforceable health goals set, with an adequate margin of safety, at levels that would result in no known or anticipated adverse health effects. The MCLs are enforceable standards set at levels as close to the RMCLs as feasible.

In the absence of regulatory standards for some compounds, other human health criteria have been used for the interpretation of IRP data. Although these criteria do not now have the force of standards, they do provide a valid means of assessing the relative degree of contamination. Using human health criteria and standards is a stringent way to evaluate groundwater contamination at AF Plant 4. Since the shallow aquifer is not used as a drinking water supply source, contaminants in-situ in this unit have neither human health nor environmental consequences. Groundwater in the Paluxy Formation, however, is used directly as a drinking water source.

No guidelines exist concerning the maximum allowable or recommended pollutant concentrations in soils for protection of human health and the environment.

ES-21

Tables ES-4 through ES-16 present the results of groundwater and surface water sampling and analysis for sites or study topics where applicable water quality criteria were exceeded in the IRP Phase II Stage 1 program. These include:

- o Site 1, Landfill 1;
- o Site 3, Landfill 3;
- o Site 12, Chrome Pit 3;
- o Site 2, Landfill 2;
- o Site 4, Landfill 4;
- o Zone 1 (Sites 13, 11, and 8);
- o Site 20, Wastewater Collection Basins;
- o Site 10, Chrome Pit 1;
- o Site 16, FSA 3;
- o Site 5, FDTA 2;
- o Ambient monitor wells; and
- o East Parking Lot wells.

The water samples from the following sites or study topics had no concentrations of contaminants detected in excess of applicable water quality criteria:

o Site 17, Former Fuel Storage Site;

- o Site 9, FDTA 6;
- o Site 6, FDTA 3;
- o Site 18, Solvent Lines;
- o Jet Engine Test Stand;
- o Fuel Storage Tank; and
- o Lake Worth monitor well.

Results of analyses of soil samples are summarized in Table ES-17. No regulatory criteria exist for comparison of these results. However, the COMPARISON OF GROUNDWATER QUALITY AT SITE 1, LANDFILL 1, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA TABLE ES-4.

			Samp.	Sampling Location	ion	
Parameter	Criteria*	HM-7	HM-10	HM-20	HM-62	HM63
METALS (mg/L)						
Arsenic Chromium	0.05 0.05			0.14	0.069	0.075
ORGANIC COMPOUNDS (ug/L)						
Vinyl Chloride	1	56				
1,1,1-Trichloroethane Trichloroethylono	200 5		37	240	α	4500
it third verify terms Benzene	רי נ	61	5		•	
Acenaphthene	0.028	2				

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.

RADIAN

 \mathcal{D}

ES-23



TABLE ES-5. COMPARISON OF SURFACE WATER QUALITY AT SITE 1, LANDFILL 1, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

			Sampling	Location	
Parameter	Criteria*	FRDRN#1	FRDRN#2	CRSEEP	DRNPIPE
METALS (mg/L)		<u> </u>		<u> </u>	
Chromium	0.05			0.067	
ORGANIC COMPOUNDS (ug/L	.)				
Vinyl Chloride	1	1200	33		970
Trichloroethylene	5	6700	2300	20	2300
Tetrachloroethylene	8				150
Fluoranthene	0.028				2

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.

TABLE ES-6. COMPARISON OF WATER QUALITY AT SITE 3, LANDFILL 3, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

12.01

A STATEMENT

i.

					Sam	Sampling Location	tion			
Parameter	Criteria#	HM-26	HM-27	HM-34	HM-35	HM-36	HM-37	HM-38	HM-39	P-220
METALS (mg/L)										
Arsenic Chromium Lead	0.05 0.05 0.05		0.13	0.062 0.088	0,065	0.055			0.13 0.14 0.25	
ORGANIC COMPOUNDS (ug/L)										
Vinyl Chlorid e	1									20
Methylene Chloride								160000		
								00047		
trans-1,2-Dichloroethylene		300						120000		1.6
		2000	390				400	00001		}
	80							00061		
	2000							000000		
	750							4400		
	620							000005		
Acensphthene	0.028							007		
	620							1400		
Fluorene	0.028							190		
Phenanthrene	0.028							290		

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.

TATER DATABATE TO ALL STORED AND ALL AND ALL DATABATE DATABATE DATABATE DATABATE DATABATE DATABATE

ES-25

4, COMPARISON OF WATER QUALITY AT SITE 12, CHROME PIT 3, AF PLANT TEXAS WITH APPLICABLE WATER QUALITY CRITERIA TABLE ES-7.

			Samp	Sampling Location	ion	
Parameter	Criteria*	HM-15	HM-16	HM-17	HM~30	HM-45
METALS (mg/L)						
Chromium Lead	0.05 0.05		0.21 0.1			
ORGANIC COMPOUNDS (ug/L)						
1.1-Dichloroethylene Trichloroethylene	7	5 900	10000	15000	180	32 130
* See Tables 4.2.1-1 and	4.2.1-1 and 4.2.1-2 for source of criteria.	urce of cri	teria.			

RADIAN

LANDFILL 2, AF PLANT 4, TEXAS	
8. COMPARISON OF WATER QUALITY AT SITE 2, LANDFILL 2, AF PLANT 4, TH	WITH APPLICABLE WATER QUALITY CRITERIA
TABLE ES-8.	

Ċ.

				Sampi ing	LOCALION		
Parameter	Criteria*	HM-22	HM-42	HM-43	HM-43 HM-44	94-MH	P21U
METALS (mg/L)					; ;		
Chromium	0.05	0.054	0,069			0.093	
Lead Selenium	0.05 0.01					0,0,0	0.03

* See Table 4.2.1-1 for source of criteria.

RADIAN

PARTY NAMES PRESS

.....



TABLE ES-9. COMPARISON OF WATER QUALITY AT SITE 4, LANDFILL 4, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

RAPIAN			
TABLE ES-			AT SITE 4, LANDFILL 4, QUALITY CRITERIA
Param	eter	Criteria*	Sampling Location HM-101
METALS (mg/)	L)		
Chromium Silver		0.05 0.05	0.2 0.13

a a seconda de la companya de la com

Sampling Location HM-11 HM-24 HM-25 HM-28 HM-60		0.14 0.12			26 7	4	870	source of criteria.	
Parameter Criteria*.	METALS (mg/L)	Arsenic 0.05 Chromium 0.05	ORGANIC COMPOUNDS (ug/L)	Vinyl Chloride 1.2-Dichloronronane 6	Trichloroethylene	Chlorobenzene 60 Benzene 5	1.4-Dichlorobenzene 750	* See Tables 4.2.1-1 and 4.2.1-2 for a	



TABLE ES-11. COMPARISON OF WATER QUALITY AT SITE 20, WASTEWATER COLLECTION BASINS, AF PLANT 4, WITH APPLICABLE WATER QUALITY CRITERIA

		Sam	pling Loca	tion
Parameter	Criteria*	HM-31	HM-47	HM-70
METALS (mg/L)				
Arsenic	0.05			0.062
Chromium	0.05			0.075
Lead	0.05			0.068
ORGANIC COMPOUNDS (ug/L))			
Trichloroethylene	5	2400	12000	11000
Chlorobenzene	60		770	
Tetrachloroethylene	8			6500

* See Tables 4.2.2-1 and 4.2.2-2 for source of criteria.



ĺ

M

TABLE ES-12. COMPARISON OF WATER QUALITY AT SITE 16, FUEL SATURATION AREA 3, AF PLANT 4, TEXAS, WITH APPLICABLE WATER QUALITY CRITERIA

Parameter	Criteria [*]	Sampling Location HM-78
ORGANIC COMPOUNDS (ug/L)		
Benzene	5	8400

*See Tables 4.2.2-1 and 4.2.2-2 for source of criteria.

1.1



12.12.12.1. S.S.S.S.

TABLE ES-13. COMPARISON OF WATER QUALITY AT SITE 10, CHROME PIT 1, AF PLANT 4, TEXAS, WITH APPLICABLE WATER QUALITY CRITERIA

Parameter	Criteria*	<u>Sampling Location</u> HM-48 HM-103
ORGANIC COMPOUNDS	(ug/L)	
Trichloroethylene	5	20 5500

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.



TABLE ES-14. COMPA				
TEXAS	WITH APPLICABLE	WATER QUAL	ITY CRITER.	IA
Parameter	Criteria*	Sam HM-19	pling Loca HM-51	tion HM-66
ORGANIC COMPOUNDS (ug/	'L)			
1,2-Dichloroethane Trichloroethylene Toluene	5 5 2000	9	27000 420000 96000	29
* See Tables 4.2.1-1 a	nd 4.2.1-2 for so	urce of cr	iteria.	

ING. AF PLANT 4. TEXAS	
Y, AMBIENT MONITORI	CRITERIA
COMPARISON OF WATER QUALITY, AMBIENT MONITORING, AF PLANT	WITH APPLICABLE WATER QUALITY CRITERI
TABLE ES-15.	

			Š	Sampling Location	ation		
Parameter	Criteria*	HM29	HM52	HM-54	HM-56	HM-58	HM-59
METALS (mg/L)							
Arsenic Chromium Lead	0.05 0.05 0.05	0.14	0.27			0.075	
ORGANIC COMPOUNDS (ug/L)							
Trichloroethylene Tetrachloroethylene Benzene	ഗയഗ	5600		6.4	110		6.6 43

(Continued)



1949 (B)

0.110.010

(Continued) TABLE ES-15.

				Sampling	Sampling Location		
Parameter	Criteria*	HM-61	HM-64	HM-79	HM81	HM-83	HM-85
METALS (mg/L)							
Arsenic 0.11		0.11				0.28	0.086
Chromium 0.17		0.17			0.05		
Lead 0.08		0.08			0.2	0.05	
ORGANIC COMPOUNDS (ug/L)							
Trichloroethylene	'n		75	170		7.4	
Tetrachloroethylene Benzene	ωv			13			
ables 4.2.1-1	and 4.2.1-2 for source of criteria.	trce of cri	iteria.				

ELECTRON STATES IN STATES

- Y. 190.0

RADIAN

OX.

ES-35



TABLE ES-16. COMPARISON OF WATER QUALITY, EAST PARKING LOT, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

		·	Samplin	g Location	
Parameter	Criteria*	HM-68	HM-71	HM-82	P-8U
METALS (mg/L)					
Chromium	0.05		0.14	0.33	
ORGANIC COMPOUNDS (ug/L)					
Trichloroethylene	5	32	1300	19000	7200
Tetrachloroethylene	8		23		

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.



TABLE ES-17. SUMMARY OF ANALYTICAL RESULTS OF SOIL SAMPLE ANALYSES, IRP PHASE II STAGE 1, AF PLANT 4, TEXAS

Site	Parameter	Range of Observed Concentrations
Site 17, Former Fuel Storage Site	Hydrocarbon fuels	ND
Site 15, FSA 2	Hydrocarbon fuels 1,1-Dichloroethylene	ND-4600 mg/kg 0.012 mg/kg
Site 9, FDTA 6	1,1,1-Trichloroethane Phthalate esters Phenanthrene	ND-0.022 mg/kg ND-3.1 mg/kg ND-0.83 mg/kg
Site 18, Solvent Lines	Oil and grease Xylene Methyl ethyl ketone	nd Nd Nd
Site 10, Chrome Pit 1	Chromium	7.1-7.4 mg/kg
	trans-1,2-Dichloroethylen Trichloroethylene	e 0.039-0.0096 mg/kg 0.065-0.17 mg/kg
Site 14, FSA 1	Hydrocarbon fuels	ND-59,000 mg/kg
Site 19, NARF Area	alpha radiation beta radiation gamma radiation	6.7-12.4 pCi/g 10.0-23.1 pCi/g ND
Jet Engine Test Stand	Hydrocarbon fuels Oil and grease	ND-1700 mg/kg ND-2000 mg/kg

8 THEFT SAFESS PROVEN 2222



following sites appear to exhibit levels of soil contamination of potential concern:

o Site 15, FSA 2;
o Site 9, FDTA 6; and
o Site 14, FSA 1.

White Settlement Groundwater Pumping Effects

An investigation of the pumpage effects of the City of White Settlement's municipal wells was performed on the groundwater to determine the zone of influence, long-term drawdowns (if any) and the influence on the direction of groundwater flow in the Paluxy aquifer beneath the Plant 4 area. Groundwater flows and drawdowns were determined by reviewing existing geologic and hydrologic information and current pumpage records for the City of White Settlement. These data were supported by comparing actual water level data to simulated results from a simple finite-difference steady-state groundwater model (Koch, 1984). Drawdowns were predicted after years 1, 3, 5, 10, 15 and 20, assuming that wells continue pumping at the current capacity for 20 years.

Results of the simulation, as well as existing information related to groundwater flow in the Paluxy aquifer, indicate that groundwater beneath the Plant 4 area is influenced to some degree by the withdrawal of water by nearby municipal wells. The analysis also suggests that fairly significant drawdowns can be expected near pumping wells, which in turn could direct groundwater to flow from Lake Worth and AF Plant 4 toward these wells. Considering water quality, there appears to be a logical concern about the potential for contaminants originating at AF Plant 4 to migrate toward these wells.

Well Network Selection for Future Monitoring Efforts

The total system of wells at AF Plant 4 was evaluated to design an optimal well network for future monitoring efforts. The objective of designing an optimal monitor well network was to decrease the number of

samples needed to adequately represent groundwater quality and the hydrogeologic conditions in the aquifer. The wells evaluated consisted of the 87 upper zone monitor wells installed by Hargis and Montgomery and Hargis and Associates and the nine upper zone wells installed by Radian. In addition to the upper zone monitor wells, the Paluxy Formation monitor wells have been evaluated with respect to future monitoring efforts at AF Plant 4. The selected well network for Plant 4 includes 64 upper zone monitor wells (HM-series) and all of the currently active Paluxy Formation monitor wells.

ALTERNATIVE MEASURES

Alternatives were considered for further IRP activities at each site investigated in the Phase II Stage 1 study. Types of alternatives considered include continued monitoring, additional field investigations, no further action, and advancement to Phase IV Remedial Action planning.

For those sites anticipated to require Phase IV remedial actions, remedial options were identified at a general, conceptual level. However, since General Dynamics is initiating complete, plant-wide remedial action planning at the present time, the examination of alternatives presented in this study should be considered to be a limited and preliminary overview of general alternatives. Recommendations for selection of alternatives should be based on the detailed planning currently underway.

RECOMMENDATIONS

Table ES-18 summarizes the recommendations from the Phase II Stage 1 investigation at AF Plant 4. The sites are categorized as follows:

Category 1 Sites - No Further IRP Activities

The following sites were investigated during this and previous studies and found to currently contain little or no hazardous material. On the basis of this finding, no further action is considered or recommended for:

Metals only slightly exceed Few organic compounds detected and at Concentrations of Cr, Pb, and organic compounds (primarily TCE) exceed levels below applicable water quality cable water quality criteria in upper above MCLs in shallow groundwater may other organic compounds exceed appli-No hydrocarbon fuels detected in soil applicable water quality criteria in and metals in upper zone groundwater Concentrations of organic compounds and surface water exceed applicable Concentrations of metals, TCE, and Metals detected at levels slightly compounds detected at levels above zone and upper Paluxy groundwater. No organic upper zone groundwater. water quality criteria. or groundwater samples. be from Chrome Pit 3. Rationale applicable criteria. criteria. STAGE 1, AF PLANT 4, TEXAS. Proceed with Phase IV remedial Proceed with Phase IV remedial Proceed with Phase IV remedial No further IRP activities No further IRP activities No further IRP activities Recommended Action action planning action planning action planning recommended recommended recommended Site 12, Chrome Pit 3 Storage Site Site 17, Former Fuel Site 4. Landfill 4 Site 3, Landfill 3 Site 2. Landfill 2 Site 1, Landfill 1 Site

criteria in samples from one well.

(Continued)

1. 8552332 TASS

RADIAN

SUMMARY OF RECOMMENDATIONS FROM IRP PHASE II

TABLE ES-18.

ES-40

SUMMARY OF RECOMMENDATIONS FROM IRP PHASE II STAGE 1, AF PLANT 4, TEXAS. (Continued) TABLE ES-18.

Rationale	Concentrations of metals and organic compounds in upper zone groundwater exceed applicable water quality criteria.	Soil sample results indicate fuel contamination but existing upper zone wells appear to be upgradient of the site. Therefore, existence and extent of groundwater contamination at the site is unknown.	Concentrations of metals in upper zone groundwater slightly above MCLs. Con- centrations of TCE and chlorobenzene also exceed applicable water quality criteria in upper zone groundwater.	Low levels of organic compounds (below proposed MCLs) detected in ground- water. Hydrocarbon fuels detected in soils. Remedial action planning already underway.	Hydrocarbon fuels detected at moderate levels in soil samples. No organic
Recommended Action	Proceed with Phase IV remedial action planning	Additional Phase II study re- quired. Install 2 upper zone monitor wells downgradient of the site (west and north) to determine quality of upper zone groundwater exiting the site	Proceed with Phase IV remedial action planning	Proceed with Phase IV remedial action planning	Proceed with Phase IV remedial action planning
Site	Zone 1. Site 13. Die Pits; Site 11. Chrome Pit 2; Site 8. FDTA 5	Site 15. Fuel Saturation Area 2	Site 20, Wastewater Collection Basins	Site 16, Fuel Saturation Area 3	Site 9, FDTA 6

(Continued)

compounds detected in groundwater at

levels above applicable criteria.

art ann an annaich beaceal beaceal beacail beacai

2002020 12220

2222222

ES-41

TABLE ES-18. SUMMARY OF RECOMMENDATIONS FROM IRP PHASE II STAGE 1. AF PLANT 4. TEXAS. (Continued)

التحديث وللملالية

AS SISE

Concernance

Site	Recommended Action	Rationale
Site 6. FDTA 3	No further IRP activities recommended	No organic compounds detected in groundwater at levels above applicable criteria.
Site 7, FDTA 4	No further IRP activities recommended	Site could not be located by soil-gas survey. Plant personnel report that no FDTA ever existed at this location.
Site 18, Solvent Lines	No further IRP activities recommended	No organic compounds detected in soil samples or groundwater.
Site 10. Chrome Pit 1	Proceed with Phase IV remedial action planning	TCE in upper zone groundwater exceeds proposed MCL. Volatile organic com- pounds detected in soil samples.
Site 5, FDTA 2	Proceed with Phase IV remedial action planning	Concentrations of organic compounds in upper zone groundwater exceed appli- cable water quality criteria.
Site 14, Fuel Saturation Area 1	Proceed with Phase IV remedial action planning	Hydrocarbon fuels detected in soil and in groundwater grab sample from soil boring.
Site 19, NARF Area	No further IRP activities recommended	No radiation detected above acceptable levels in soil samples.

(Continued)

sang beered becare broden bos

222

RADIAN

and a second

11.

SUMMARY OF RECOMMENDATIONS FROM IRP PHASE II	Continued)	
INDATIONS FR	1. AF PLANT 4. TEXAS. (Continued)	
OF RECOMME	. AF PLANT	
.8. SUMMARY	STAGE	
TABLE ES-18.		

Recommended Action

a	
Ĩ.	
<u>S</u>	

Building 21

No further IRP activities recommended Jet Engine Test Stand,

compounds detected in surface water at No organic levels above applicable water quality criteria. No hydrocarbons or oil and Low levels of hydrocarbon fuels grease detected in groundwater. detected in soil samples.

RADIAN

Rationale

RI FREED FREED SERVER SURVEY STREET STREET STREET STREET SERVER SERVER SERVER SERVER SERVERSE SE

RADIAN

o Site 17, Former Fuel Storage Site;

- o Site 2, Landfill 2;
- o Site 4, Landfill 4;
- o Site 6, FDTA 3;
- o Site 7, FDTA 4;
- o Site 18, Solvent Lines;
- o NARF Area; and
- o Jet Engine Test Stand, Building 21.

Individual wells at the various sites should continue to be considered for inclusion in a plant-wide monitoring program.

Category 2 Site - Additional Phase II Effort Required

Site 15, Fuel Saturation Area 2, was studied during this program but insufficient data exist to fully characterize the nature and extent of contamination. Additional Phase II investigations will be required. The following action is recommended.

> Installation of two upper zone monitor wells downgradient (west and north) of the site to determine the character of upper zone groundwater exiting the site.

Category 3 Sites - Ready for Phase IV Actions

The following sites were investigated during this and previous studies and found to contain varying levels of hazardous material. All of these well-characterized sites should be released for Phase IV planning, as appropriate.

o Site 1, Landfill 1;
o Site 3, Landfill 3;
o Site 12, Chrome Pit 3;
o Zone 1 (Sites 13, 11 and 8);

ES-44



- o Site 20, Wastewater Collection Basins;
- o Site 10, Chrome Pit 1;
- o Site 16, FSA 3;
- o Site 9, FDTA 6;
- o Site 5, FDTA 2; and
- o Site 14, FSA 1.

Individual wells at the various sites should also continue to be considered for a plant-wide monitoring program.

الالالاللاللال

[This page intentionally left blank.]

. Na kalendar da kalendar da

RADIAN

1.0 INTRODUCTION

The Department of Defense (DOD) is conducting a nation-wide program to evaluate past waste disposal practices on DOD property, to control the migration of hazardous contaminants, and to control hazards that may result from these waste disposal practices. This program, the Installation Restoration Program (IRP), consists of four phases: Phase I, Initial Assessment/Records Search; Phase II, Problem Confirmation/Quantification; Phase III, Technology Base Development and Phase IV, Operations. The United States Air Force is conducting an IRP investigation at Air Force Plant 4 near Fort Worth, Texas. Radian Corporation has performed this Phase II Stage 1 Field Evaluation under USAF Contract No. F33615-83-D-4001, Delivery Order 27.

1.1 Purpose of the Investigation

The purpose of this investigation was to define the presence, magnitude, extent, direction and rate of movement of any identified contaminants. In addition, the body of geologic and chemical data developed by General Dynamics, operator of the plant, was to be used in the investigation. Finally, additional investigations required beyond this stage, including an estimate of costs (provided separately) were to be determined.

A contaminant source investigation at AF Plant 4 was conducted to determine: (1) the presence or absence of contamination within the specified areas of the field survey; (2) if contamination exists, the potential for migration in the various environmental media; (3) the extent/magnitude of contamination on AF Plant 4 property; (4) the potential environmental consequences and health risks of migrating contaminants, based on state or federal standards for these contaminants.

RADIAN

1.2 Duration of the Program

Authorization to proceed on this study was given on 13 September 1985. Field activities began in November 1985. The field work consisted of geophysical surveys, a soil gas survey, coring and sampling of near-surface soil at various locations, installation of upper zone and Paluxy groundwater monitor wells, surface water sampling and sampling of groundwater from completed wells. Field work was completed in August 1986.

1.3 Waste Disposal Practices

The following summation of the waste disposal history of AF Plant 4 is from the USAF Phase I IRP report (CH2M Hill, 1984).

Industrial operations have been continuous since Air Force Plant 4 went into production in 1942. Plant operations center around the production of military aircraft and associated equipment. Manufacturing of aircraft and associated equipment results in the generation of varying quantities of waste oils (including lubricating oils, hydraulic fluids, and coolants), recoverable fuels, solvents (including degreasers and paint thinners), paint residues (including waste liquid paints and paint booth sludges), and spent process chemicals (including acids, caustics, chromium, and cyanide solutions). The total quantity of these wastes currently ranges from about 5,500 to 6,000 tons/year. Development of this information was based on interviews with General Dynamics personnel, previous reports, and Monthly Waste Shipment Reports from 1981 through 1983. Since waste quantities depend on the levels of contractor activity and vary from one time period to the next, the total quantities may have been higher in the past during periods of heavier production.

Martin States and the states of the states o
RAPIAN

Past (based on information obtained from files and on the best recollection of interviewees) and present waste disposal practices are summa-rized below:

- <u>1942-1955</u>: In general, waste oil, solvents, recoverable fuels, paint wastes, and spent process chemicals were disposed of at on-site landfills. Some of these materials, e.g., oils, solvents, and fuels, were burned at the landfill in surface pits. Wastewaters containing process chemicals, including chromium and other metals, were discharged to the sanitary sewer system, which went to the City of Fort Worth treatment system. Washwater from paint booths and water containing residual oils were also discharged to the sanitary sewage system.
- o <u>1955-1966</u>: Wastes were handled in the same manner as in the previous period, with few changes. Waste oils, recoverable fuels, and spent solvents were disposed of in the landfills and also burned in fire department training exercises at designated fire department training areas.
- <u>1966-1970</u>: Waste oils and fuels continued to be disposed of at on-site landfills and burned in fire department training exercises. Solvents, waste paints, and process cyanide solutions were disposed of by contractor. Other wastes continued to be discharged to the sanitary sewer system.
- <u>1970-1975</u>: Waste oils and fuels, solvents, process cyanide chemicals, and paint wastes continued to be disposed of in the same manner. Other process chemical solutions, rinse waters, and paint booth wastewater were treated by a newly installed



chemical waste treatment system that removed metals and neutralized acids and bases.

- o <u>1975-1983</u>: Waste oils were disposed of by a contractor and, to a lesser extent, by burning with recoverable fuels at fire department training exercises. Solvents, process cyanide chemicals, and paint wastes continued to be disposed of by contractor. Paint booth washwaters and some waters containing residual oils were discharged to the chemical waste treatment system and also to the industrial wastewater collection system, which dishcarged, untreated, to the sanitary sewer system. Other process chemical solutions continued to be discharged to the chemical waste treatment system.
- <u>1983 to present</u>: Wastes continued to be disposed of in the same manner as in the previous period, except that all waste oils and recoverable fuels were disposed of contractor, and none were burned in fire department training exercises.

1.4 Location and Site Descriptions

AF Plant 4 is located six miles west of the center of Fort Worth in Tarrant County, Texas (Figure 1-1). Lake Worth borders the facility on the north, Carswell AFB on the east, and the community of White Settlement on the west and south.

Phase II Stage 1 work at AF Plant 4 has focused on the 21 waste sites and four study topics listed in Table 1-1. Brief descriptions of the locations and features of the Phase II sites and study topics are provided below. The 21 waste sites (including the three Zone 1 sites) and the fuel storage tank are shown in Figure 1-2. Sites are presented and discussed in the order in which they appear in the Delivery Order. The additional study topics are not plotted because the areas they cover are too large and diverse.



1-5

<u>ͺ</u>

TABLE 1-1. IRP PHASE II STAGE 1 SITES AND STUDY TOPICS AT AF PLANT 4, TEXAS

SITES:

Site 1, Landfill 1
Site 3, Landfill 3
Site 12, Chrome Pit 3
Site 17, Former Fuel Storage Site
Site 2, Landfill 2
Site 4, Landfill 4
Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; Site 8,
Fire Department Training Area 5)
Site 15, Fuel Saturation Area 2
Site 20, Wastewater Collection Basins
Site 16, Fuel Saturation Area 3
Site 9, Fire Department Training Area 6
Site 6, Fire Department Training Area 3
Site 7, Fire Department Training Area 4
Site 18, Solvent Lines
Site 10, Chrome Pit 1
Site 5, Fire Department Training Area 2
Site 14, Fuel Saturation Area 1
Site 19, Nuclear Aerospace Research Facility (NARF) Area
Jet Engine Test Stand, Building 21

ADDITIONAL STUDY TOPICS:

East Parking Lot Fuel Storage Tank Lake Worth Monitor Well Ambient Monitoring



للتشكيك فللملاسة

1-7

Most of the site information was obtained from the Installation Restoration Program Phase I IRP records search report (CH2M-Hill, 1984).

Site 1, Landfill 1

From 1942 until about 1966, several types of hazardous and nonhazardous wastes were reportedly disposed of in Landfill 1. These wastes included drums of liquid and contents of tanks and bousers loaded with chemical wastes (solvents, thinners, paints, etc.). Contaminated oils, fuels, and hazardous wastes containing metals, chromate sludges, and cyanide are suspected to be present at this site.

Since it was closed during the 1960s, extensive remedial action has been conducted at this site. The major action was the removal of 11,000 cubic yards of earth and the installation of a drain field (French Drain No. 2).

Site 3, Landfill 3

From about 1942 to 1945, this site was reportedly was for disposal of wastes, including hazardous wastes consisting of mixed oils and solvents. At least one pit (possibly more) present in this area during the 1940s was used for holding and burning some of the liquid wastes. Other wastes are suspected to have been disposed of on the ground and later buried. From 1966 until 1967, fill dirt and rubble were used to finish filling and grading this site.

Site 12, Chrome Pit 3

Chrome Pit 3 was used for the disposal of chromate and other chemical wastes from about 1957 until 1973. Barium chromate sludge, dilute metal solutions, and drums of unidentified liquids were disposed of in this pit.

Site 17, Former Fuel Storage Site

A 100,000-gallon above-ground JP-4 storage tank from the early 1940s was relocated from here in 1962. Sampling at this site in 1982 confirmed that soils and underlying groundwater are contaminated by fuels and other organic compounds.

Site 2. Landfill 2

The site originally consisted of some low areas and a livestock watering hole. Most of this site was reportedly filled with construction rubble, plaster, and fill dirt during the early 1940s. However, 1962 aerial photographs show some activity at the stock watering hole at this site. This area was reportedly used for the disposal of lumber and tires and was assumed to be periodically burned. No reports exist of hazardous substances being deposited at this site.

Site 4, Landfill 4

Landfill 4 was reportedly used for the disposal of clean construction rubble from 1956 until the early 1980s. Aerial photographs and a 1973 memo indicate that other types of wastes may have been disposed of at this site from 1966 until at least 1973. This evidence seems to show that small quantities of hazardous wastes (solvents, oils, fuels, thinners, etc.) are present in this landfill.

Zone 1

Three sites were combined into one zone for study purposes. Descriptions of these sites are given below.

Site 13. Die Pits: These pits were used for the disposal of chromate sludges, metal solutions, and other chemical wastes until 1962, when the site was graded and the entire die yard was paved. One Phase I interviewee reported that some of the contaminated soils at this site were spread around the die yard during the grading and leveling activities. The site of the original pits was excavated in 1983-84. Soils from other parts of the die yard were not analyzed at that time.

<u>Site 11. Chrome Pit 2</u>: According to the Phase I study, miscellaneous liquid and solid wastes, in addition to chromate solutions, were probably disposed of here. Neither interviews nor aerial photographs could confirm the exact location of this site.

<u>Site 8, Fire Department Training Area (FDTA) 5</u>: This site consisted of a shallow pit in which waste fuels, oils, or chemicals were deposited for training exercises. This site, which has been graded and paved, is located in the die yard area south of Warehouse 1.

Site 15, Fuel Saturation Area 2

This site reportedly became saturated with fuels from leaks in buried fuel lines between the 1970s and early 1980s.

Site 20, Wastewater Collection Basins

Two concrete-lined waste basins, each with a capacity of approximately 85,000 gallons, are used to collect and settle suspended solids from chemical wastewaters before discharge to the City of Fort Worth sanitary sewage system. The basins have been in use since about 1966. Evidence indicated that several spills of vapor degreaser tanks in the Process Building have occurred since installation of these tanks. Some of the spilled chemicals (primarily trichloroethylene) have flowed to the basins via floor drains. These basins are suspected of being a possible source of organic chemical and metal groundwater contamination.



Site 16, Fuel Saturation Area 3

This site reportedly became saturated with fuels from leaks in buried fuel lines between the mid-1970s and the early 1980s. This site is the subject of current remedial action planning by Intellus.

Site 9, Fire Department Training Area 6

Before 1970, training exercises were conducted twice a year at this site. After 1970, exercises were conducted at monthly intervals. Approximately 250 gallons of waste fuels and oils were reportedly used for each exercise. In addition, the Phase I report indicated that larger quantities of contaminated fuels and oils were probably deposited in the FDTA between exercises.

Site 6, Fire Department Training Area 3

Training exercises at this site also used about 250 gallons of waste fuels and oils per exercise. Its location and current condition could not be accurately determined because it is not visible on historical aerial photographs.

Site 7, Fire Department Training Area 4

The same type of training exercises were conducted here. This site is not visible on historical aerial photographs, so that its location and current condition are not accurately known. The area is also believed to have received fill material originating from a foundation excavation at the Administration Building.

Site 18, Solvent Lines

These lines reportedly leaked during the 1940s before they were drained, capped and abandoned in 1944. The Phase I study could not determine the actual locations of the leaks. These solvent lines reportedly contained xylene, methyl ethyl ketone, and kerosene.

Site 10, Chrome Pit 1

Miscellaneous liquid and solid chemical wastes, in addition to chrome wastes, were probably disposed of at this site. The actual location of this site is thought to be somewhere beneath the Process Building, but the Phase I report could not accurately confirm this. No monitor wells exist in the immediate vicinity of this site.

Site 5, Fire Department Training Area 2

Exercises were held infrequently (twice a year) at this site. However, disposal of waste oils and fuels and uncontrolled burns may have been more frequent. This site is located under the pavement in the west employee parking area.

Site 14, Fuel Saturation Area 1

The ground at this site, located just west of the Parts Plant, reportedly became saturated by fuels from leaking fuel lines from the mid-1970s to the early 1980s. This site is the subject of Phase IV Remedial Action planning by Intellus.

Site 19, Nuclear Aerospace Research Facility (NARF) Area

The Nuclear Aerospace Research Facility (NARF), formerly located at the north end of AF Plant 4, housed several experimental atomic reactors between 1953 and 1974. The facility was decommissioned and disposed of by contractor in 1974.

Jet Engine Test Stand, Building 21

This site was not previously identified in Phase I. The jet engine test building is located north of a fuels test area, previously identified as Site 16, a known area of fuel contamination. The sump downslope of Building 21 was constructed in 1975 and collects water used for cooling, noise suppression, and clean-up in Building 21. This water is pumped into the industrial waste line. The area immediately south of Building 21 contains two underground tanks that were once used for fuel storage. These tanks are being investigated as potential sources of contamination under a separate underground tank investigation being conducted as part of IRP activities by the operating contractor, General Dynamics. Outfall 3 is downgradient of Building 21, and is regularly sampled as part of IRP and NPDES monitoring activities.

1.5 <u>Additional Study Topics</u>

In addition to the studies associated with the waste sites discussed above, additional study topics were identified for inclusion in the Phase II investigation. These topics are discussed below.

Ambient Monitoring

Several wells installed and distributed throughout the plant area of AF Plant 4 are not associated with a particular waste disposal site or suspected spill area. Groundwater from these wells was sampled and analyzed to gain a complete view of groundwater quality at AF Plant 4.

RAPIAN

East Parking Lot

Five wells located in the vicinity of the East Parking lot were sampled during the Phase II Stage 1 investigation. The purpose of this investigation was to test for the presence of contaminants in the vicinity of a suspected zone of communication between the upper zone and the Paluxy aquifer.

Fuel Storage Tank

This area is the location of a 100,000-gallon above-ground JP-4 storage tank created in 1962. An existing upper zone monitor well was sampled for the presence of fuel contamination.

Lake Worth Monitor Well

This area is in the northern part of AF Plant 4 that borders Lake Worth. The well was sampled to determine the character of the groundwater in the vicinity of Lake Worth.

White Settlement Groundwater Pumping Effects

This study evaluated the long-term effects of pumping by the city of White Settlement on water levels and flow directions in the Paluxy aquifer. The purpose of the evaluation was to predict the likely direction of movement of any contaminants in the Paluxy aquifer, as well as the effects of pumping on this movement.

Literature Search

A literature search of local hydrogeologic conditions was conducted and documented in the report. The results of that search are contained in Section 2.



Well Inventory

All of the wells existing on AF Plant 4 at the time of this investigation were inventoried and grouped into subsets associated with a particular site, zone, or ambient area.

Well Network Selection

The total system of wells was evaluated, and an optimal well network redesign was recommended for future monitoring efforts at AF Plant 4. This recommendation can be found in Section 4.3.27. Wells not considered necessary to the integrity of the chemical and hydrogeologic data base were identified as candidates for abandonment.

Well Abandonment Procedures

As part of this investigation, available methods of well abandonment were evaluated and a candidate method or technique was recommended. This recommendation can be found in Section 4.3.25.

1.6 <u>Sampling and Analytical Program</u>

The sampling program at AF Plant 4 involved collecting soils, surface water, and groundwater. Soil samples were collected with a Shelby tube or split-spoon sampler during drilling activities or by use of a hand auger. All soil samples were placed in individual glass jars and frozen. Surface water was collected as grab samples. Permanently installed electric pumps were used to collect groundwater from the Paluxy monitor wells. The remaining wells were sampled using a "Tri-loc" portable submersible pump or a Teflon bailer.

All water samples were chilled to 4°C. In addition, samples collected for purgeable aromatics and xylene analysis were preserved with hydrochloric acid (HCl) to pH<2. Samples collected for oil and grease and



hydrocarbon fuels analysis were preserved with sulfuric acid (H_2SO_4) to pH<2. Samples for metals analysis were preserved with nitric acid (HNO_3) to pH<2. All samples were shipped to Radian Analytical Services for analysis. The schedule of analyses is summarized in Table 1-2.

1.7 Investigative Personnel

The AF Plant 4 Phase II (Stage 1) investigation was conducted by staff from the Austin, McLean and Sacramento offices of Radian Corporation. Francis J. Smith, Contract Program Manager, was responsible for the contractual administration of the program. The overall technical program was directed by Lawrence N. French, Senior Geologist and Certified Professional Geological Scientist. Mr. French directed all activities of the program and participated directly with USAF personnel.

The geophysical surveys and soil gas investigations were supervised by Mr. Peter A. Waterreus. Soil sampling activities and monitor well installation were supervised by Mr. French, Mr. Waterreus, and Tobin K. Walters. Monitor well sampling activities were conducted by Neil A. Robinson, Arthur H. Morrill, Fred R. Snyder, Wendy J. Johnson, Wallace J. Hise, and Mr. Walters. Mr. Walters evaluated the effects of White Settlement pumpage and Gary D. Henderson evaluated the long-term monitor well network. Mr. French and Ms. Johnson were the principal authors of the draft report. Cartographic and technical illustrations were prepared by Jill P. Rossi. Ann E. St.Clair and William M. Little provided senior technical staff review and editing. Dr. F. L. Shore provided review of chemistry data and interpretation. Table 1-3 summarizes the project individuals and their functions/roles. Appendix J contains resumes for the primary individuals.



È.

TABLE 1-2. ANALYTICAL SCHEDULE FOR SOIL AND WATER SAMPLES, IRP PHASE II, STAGE 1, AF PLANT 4, TEXAS

Site	Location	Sample Type	VOC	BNA	OSG	Heavy Metals	Fuel HC	Xylene	Cr	MEK	RAM
1	Landfill 1	G	X	X		x					
		W	X	X		X					
3	Landfill 3	G	X	X		X					
12	Chrome Pit 3	G	X	X		X					
17	Former Fuel Storage Site	G					X				
	-	S					X				
2	Landfill 2	G	X	X		X					
4	Lendfill 4	G	X	X		X					
		S	X	X		X					
	Zone 1	Ğ	X	X	X	X					
		S	X	X		X					
15	Fuel Saturation Area 2	G	X			••	X				
-		S	x				X				
20	Wastewater Collection Basins	Ğ	x		X	x	X				
200		Š	x		x	x	x				
16	Fuel Saturation Area 3	Ğ	X		••	41	X				
9	Fire Dept. Training Area (FDTA) 6	Ğ	X	X	X		x				
	•	S	X	X	X		X				
6	EDIA 3	G	Х		X		х				
18	Solvent Lines	G			X			X		X	
		S			X			X		X	
10	Chrome Pit 1	G	X						Х		
		S	X						X		
5	FDTA 2	Ğ			X		х				
14	Fuel Saturation Area 1	G					X				
		S					x				
19	NARF Area	G	X	x	X	х	x				x
		Š									x
	Jet Engine Test Stand,	G			X		X				
	Building 21	S			X		X				
		Ŵ		x	x		X				
	Ambient Monitoring	Ğ	X	X	x	x	x		x		
	East Parking Lot	G	x	X	X	X	X		X		
	Fuel Storage Tank	G	n	•	x	n	x		A		
	Lake Worth Monitor Well	G	x	х	n	x	Λ				
			<u>^</u>			^					

Samples: G = Groundwater; W = Surface Water; S = Soil
VOC = Volatile Organic Compounds
BNA = Base Neutral and Acid Extractable Compounds
O&G = Oil and Grease
Fuel HC = Hydrocarbon Fuels
Cr = Chromium
MEY = Methyl Ethyl Kerone
PAM = Fadicactive Materials (Alpha, beta, and gamma radicnuclides)

TABLE 1-3. AF	
TABLE 1-3. AF	PLANT 4 IRP PHASE II STUDY TEAM
Individual	Function/Role
F. J. Smith	Contract Program Manager
L. N. French	Technical Director
P. A. Waterreus	Geophysical and soil gas surveys, monitor well installation, soil sampling, report preparation
T. K. Walters	Monitor well installation, White Settlement pumping effects
N. A. Robinson	Water sampling
A. H. Morrill	Water sampling
F. R. Snyder	Water sampling
W. J. Johnson	Water sampling, report preparation
W. J. Hise	Water sampling
G. D. Henderson	Long-term monitor well evaluation
J. P. Rossi	Cartography, technical illustrations
F. L. Shore	Chemistry review
W. M. Little	Technical review
A. E. St. Clair	Technical review
ubcontractors:	
Delta Geophysical Service	es - Geophysical survey
Tracer Research Corporati	on - Soil gas study
Southwestern Laboratories sampling	- Upper zone monitor well installation and soil
Underground Resources Mar	agement, Inc Paluxy monitor well installation
Sempco, Inc Well locat	ion and elevation surveying

RAPIAN

2.0 ENVIRONMENTAL SETTING

The following discussion of the basic features and history of the AF Plant 4 environmental setting, which sets the context for evaluating the results of the current study, is derived primarily from the USAF Installation Restoration Program Phase I Records Search Report (CH2M Hill, 1984). Information from that report is supplemented by information from the literature, from the general findings of this study, and by other information supplied by the USAF. Section 4 contains a detailed discussion of the results of the current study.

2.1 <u>General Geographic Setting and Land Use</u>

AF Plant 4 is located six miles west of the center of Fort Worth in Tarrant County, Texas (Figure 2-1). The plant is bordered by Lake Worth on the north, the community of White Settlement on the south and west and Carswell AFB on the east. The major facilities and physical features of the AF Plant 4 site are shown in Figure 2-2.

The plant is situated in an area of primarily residential, recreational, and industrial/commercial land use. AF Plant 4 is the principal industrial user of the area. Recreational land use includes the Y.M.C.A.'s Camp Carter and various parks on the shores of Lake Worth.

2.2 Physiographic and Topographic Features

AF Plant 4 is located within the Grand Prairie section of the Central Lowlands Physiographic Province. This area is characterized by broad terrace surfaces sloping gently eastward, interrupted by westward-facing escarpments. The land is typically grass-covered and barren of trees, except for isolated stands of upland timber. The northwestern part of AF Plant 4 is within the Western Cross Timbers Physiographic Province, which is



الم" والو" هذار " هأل "هوال أوف الوق" الولار الو

7

2-2



•

COLOCITA CALCO

•••••••••••



characterized by rolling topography and a heavy growth of post and blackjack oaks.

The topography of the plant is fairly flat except for areas near the creek parallel to Meandering Road, which discharges to Lake Worth, Farmer's Branch and the Trinity River. Land surface slopes gently northeast toward Lake Worth and east toward the West Fork of the Trinity River. Elevations range from a high of approximately 627 feet MSL at the southwest corner of the plant to a low of approximately 600 feet MSL north of the plant at Lake Worth.

Surface drainage at AF Plant 4 is intercepted by storm drains and culverts and is discharged to Lake Worth, the Meandering Road Creek, or a tributary of Farmer's Branch. AF Plant 4 is partially drained by Farmer's Branch, which in turn discharges into the West Fork Trinity River east of Carswell Air Force Base. Figure 2-3 shows the direction of surface water flow at AF Plant 4.

2.3 Geologic and Hydrogeologic Conditions

Surficial Soils

The U.S.D.A. Soil Conservation Service has identified two similar soil associations at AF Plant 4. Most of the surficial soils of the installation area are nearly level or gently sloping clay soils of the Sanger-Purves-Slidell and Aledo-Bolar-Sanger associations. Soils of the Sanger-Purves-Slidell association occupy most of the area at AF Plant 4. The Aledo-Bolar-Sanger association exists only in a thin strip along the northern boundary of the plant. These soils are described in Table 2-1 and shown on Figure 2-4.

Lithology

A geologic section showing the geologic formations beneath AF Plant w is presented in Figure 2.1. Descriptions and properties of units pertinent



Sec. Sec. Sec.

A MARCHART CONTRACTOR

Association	Description	Thickness (inches)	Permeability (cm/sec)
Sanger-Purves-Slidell: Clay soils of nearly level to gently sloping uplands.	Clay loam Clay over bedrock Silty clay	8-80	4.2×10^{-5} to 3 x 10^{-4}
Aledo-Bolar-Sanger: Loamy and clayey soils of gently sloping to moderately steep up- lands.	Clay loam over bedrock Clay loam	8–70	$(4.2 \times 10^{-5} \text{ to})$ 9 x 10

TABLE 2-1. SOIL ASSOCIATIONS FOR AF PLANT 4, TEXAS

SOURCE: U.S. Department of Agriculture, 1981, Soil Survey of Tarrant County: Soil Conservation Service, 218 pp.

5

1.





<u>eneral en la serie de la s</u>



to this study are summarized in Table 2-2. From youngest to oldest, the geologic units of interest to AF Plant 4 are as follows: 1) Quaternary Alluvium, 2) Cretaceous Goodland Limestone, 3) Cretaceous Walnut Formation, 4) Cretaceous Paluxy Formation, 5) Cretaceous Glen Rose Formation, and 6) Cretaceous Twin Mountains Formation. Only the first four are exposed at the surface at Plant 4. Except for a thin strip along Meandering Road Creek and Lake Worth, the surface is blanketed with alluvium and man-made fill. The locations and extent of the bedrock units at the plant are shown on a geologic map, Figure 2-6.

The majority of the plant is covered by alluvium deposited by the Trinity River during flood stages. The alluvium is composed of gravel, sand, silt, and clay of varying thicknesses and lateral extents.

The Goodland Limestone and Paluxy Formation are exposed in very small sections along the north and northwest boundary of the AF Plant 4, along Meandering Road Creek and Lake Worth. The Goodland is a chalky-white, fossiliferous limestone and marl. The Walnut Formation is a shell-agglomerate limestone with varying amounts of clay and shale. Although the Glen Rose Limestone and sandstones of the Twin Mountains Formation are not exposed at AF Plant 4, these formations are important in understanding the hydrogeology of the AF Plant 4 area.

Structure

AF Plant 4 is located in a structurally stable setting (the Texas craton), west of the faults that lie along the Ouachita Structural Belt. No major faults or fracture zones have been mapped near the base. The regional dip of the rocks beneath AF Plant 4 is between 35 and 40 feet per mile in an easterly to southeasterly direction. The stratigraphic and structural relationships of the uppermost geologic units at AF Plant 4 are illustrated in Figure 2-6, which shows a cross-section from south to north across the plant site. The location of the cross section, taken from Hargis and Associates

B 11 As	Bertes and Group	Formation and Number	[11]	Character of Rocks	Topographic Expression	Water-Bearing Properties
	Recent and Plaistocene	Allwin	3 2	Sand, gravel, cley, and silt.	Terrace and flood-plain deposits.	Smell to moderate yields. Water unextisfactory for use unless trasted.
C. • 15 00 us	Comenche Beries Veshite Group	Duck Creek Formation	8	Mapure Limmatone and mart, which is blue when fresh and atter-colored when weathered. Fossiliferous with distinctive mmonites.	Bench topography produced by Lomer Limestone unit. Upper mari forma slope separating the Duck Creak from Fort Worth Limestone.	Small to moderate yields. Water unmetiafactory for use unless treated.
	Comenche Series Frederickeburg Group	Kiamichi Formation	Q	Blue and brownish-yellow warl, thin limeatone and mendatone flage.	Greesy alope meperating scarps of Goodland and Duck Creek formations.	Small to moderate yielde. Water unestiefactory for use unless treated.
		Goodland Limestone	0-130	Chalky-white fosailifarous liaestone, and blum to yel- lowish brown sarl.	Prominent glering-white escerpeent slong stremes.	Bmeil to moderate yielda. Mater unestiefectory for use unlass treated.
 		Malnut Clay		Rhell agglomerate fossilifer- ous city and limestone, sendy city, and black shale. UNCONFORMITY	Forms conspicuous ascarpment and waterfalls in western Cross Timbers beit.	Not known to yield weter to welle in Terrent County.
Cr e La ce oua	Commuche Series Trinity Group	Pand Sand	140-150	- 2	dy soll, humasocky topography wity wooded etth oaks.	a of supply for sost Malds, emailer cities, came industries,
		Glen Rose Limestone	250-450	Fine-grafmad Limestone, chale, marl, and sandstone.	Mat exposed in Terrant County.	Sende yield muell supplies to wells in Fort Worth and western Tarrant County. Water too highly winnesitzed east of Fort Worth.
		Twin Mountaine Forme- tion (Formarly Travia Peak Formation)	2%0-450	- jeć p	st exposed in Terrant County.	
Panney Lyani an	Und!fferent!eted		6,000-7,000	recon uncurronning	Not exposed in Terrant County.	Not tested. Probably would not yield fresh water.
	E.A. Laggat.			e s'alle e s'alle entre de la constante de la c		







525522

2-11

1 H K



755.22

- RASSELLE

RECENT NEWSFILM

2020142

XXXXXXX

2222222

122225252

Truess

2-12

EXPLANATION
P-1 WELL IDENTIFIER Image: Destified on removement of material Image: Destified on removement of models retain Image: Destified on removement of models on body and the retain of the restain models on the restain of the re
Figure 2-7. Continued

滋

8

0.000000000

J

an ballet ba

(1985), is shown in Figure 2-6. Borings completed for the current study substantiated the data shown. These are discussed further in Section 4.

Hydrogeology

On the basis of their water-bearing properties, the geologic units at AF Plant 4 may be divided into the following five hydrogeologic units, listed from most shallow to deepest: 1) an upper perched-water zone occurring in the alluvial terrace deposits left by the Trinity River; 2) an aquitard of predominantly dry limestone of the Goodland and Walnut Formations; 3) an aquifer in the Paluxy sand; 4) an aquitard of relatively impermeable limestone in the Glen Rose Formation; and 5) a major aquifer in the sandstone of the Twin Mountains Formation. Each of these units, shown in Figure 2-5, is examined in more detail below.

Upper Zone - Groundwater occurs within fluvial sediments deposited by the Trinity River, but these deposits are usually limited in areal extent. Recharge to the water-bearing deposits is local, from rainfall and infiltration from stream channels and drainage ditches. The direction of groundwater flow is geneally controlled by the bedrock topography of the Walnut Formation.

In parts of Tarrant County, generally close to the Trinity River, water in the alluvium is developed for irrigation and residential use. The community of River Oaks, immediately east of Carswell AFB, had supply wells that developed water from the alluvial deposits at a location near the USAF Hospital. The wells were abandoned when Carswell AFB purchased the property. Groundwater present in the upper zone occurs as a perched water table and, in general, it is not hydraulically connected to the underlying aquifers (CH2M Hill, 1984). For the most part, it is not economical to develop groundwater from the alluvium because of the water's limited distribution and susceptibility to surface pollution.

2-14

Goodland/Walnut Aquitard - The perched water present in the alluvium is separated from the aquifers below by the low permeability limestones and shales of the Goodland Limestone and Walnut Formation. The aquitard is composed of moist clay and shale layers interbedded with dry limestone beds. Though primarily dry, drillers in the area report that small amounts of water enter the borehole while drilling through the Walnut Formation, suggesting that groundwater may move through the Walnut along bedding planes (Hargis and Associates, 1985). The thickness of the Goodland/Walnut aquitard is approximately 25 feet or greater beneath most of AF Plant 4. However, the top of the aquitard is an erosional surface and weathering may have reduced the thickness of the limestone in isolated areas. A soil boring at AF Plant 4, beneath Building 189 along Grants Lane, revealed that the Goodland Limestone there had been completely eroded and only three feet of the Walnut Formation remained (Hargis and Associates, 1985). It is also reported that the upper zone and the Paluxy Formation are in contact at the eastern boundary of AF Plant 4, where both the Goodland and Walnut Formations have eroded away (Hargis and Associates, 1985). In areas of similarly extensive erosion, water in the upper zone could come in contact with water in the Paluxy aquifer.

<u>Paluxy Aquifer</u> - The Paluxy aquifer, whose areal extent is shown in Figure 2-8, is the most shallow aquifer occurring beneath AF Plant 4. In the plant area, water in the Paluxy would naturally occur under confined conditions beneath the Goodland/Walnut aquitard (except where the aquitard has eroded away, as discussed above). However, extensive pumping in the Fort Worth area has lowered the Paluxy potentiometric surface below the top of the formation, resulting in unconfined conditions beneath the base. The Paluxy Formation is divided into upper and lower sand members and the aquifer is likewise divided into upper and lower sand members the upper sand is finegrained and shaley, the lower sand is more coarse and permeable; therefore, most wells are completed in the lower section.

Recharge to the Paluxy aquifer occurs where the formation outcrops west of AF Plant 4. The Paluxy also outcrops north of the plant in the bed of

2-15



ACCOUNT SERVICE SERVICE

ZZI ZKERI N

Values Balling Sales

.

2-16

Lake Worth. The lake represents a significant recharge point for the aquifer and creates a potentiometric high in its vicinity. Regional groundwater flow within the Paluxy is southeastward, in the direction of the regional dip. At Plant 4. groundwater flow is influenced by recharge from Lake Worth, which creates a potentiometric high, and by groundwater withdrawals by the community of White Settlement. This drawdown results locally in a more southerly flow direction within the Paluxy aquifer.

The Paluxy aquifer is an important source of potable groundwater in the Fort Worth area. Communities surrounding AF Plant 4, especially White Settlement, develop municipal water supplies from the Paluxy and, to a greater extent from the Twin Mountains aquifer below. Because of its extensive use as a potable and municipal water supply, water levels in the Paluxy aquifer have declined significantly over the years. Water levels in the immediate AF Plant 4 vicinity have not decreased as much as in the Fort Worth area in general because of the proximity of the Lake Worth recharge area and because the base does not develop water from the Paluxy.

Transmissivities in the Paluxy aquifer range from 1,263 to 13,808 gallons per day per foot (gpd/ft), an average of 3,700 gpd/ft (CH2M Hill, 1984). The Paluxy Formation thickness ranges from 140 to 190 feet, averaging 160 feet in Tarrant County. The actual water-bearing thickness in the AF Plant 4 area probably approximates the formation thickness, but the aquifer is separated into two distinct water-bearing zones, denoted as the upper and middle/lower Paluxy. In some cases, the middle and lower Paluxy are also separated by low-permeability layers. In the vicinity of AF Plant 4, hydraulic conductivities range from 13 to 140 gpd/ft² (based on an approximate sand thickness for the aquifer of 100 ft.). Well yields within the Paluxy aquifer range from 10 to 480 gallons per minute (gpm), averaging approximately 100 gpm.

Water quality in the Paluxy aquifer is generally good and is satisfactory for potable use. Table 2-3 shows the range of chemical constituents in Paluxy groundwater reported by the Texas Department of Water Resources.

2-17

haddalar a sharar a s



TABLE 2-3.	RANGE OF CONSTITUENTS IN GROUNDWATER FROM SELECTED WELLS
	IN THE PALUXY FORMATION, TARRANT COUNTY

RAPIAN	
TABLE 2-3. RANGE OF CONSTITUENTS IN GRO IN THE PALUXY FORMATION, TAI	
Constituent or Property	Concentration
Bicarbonate (HCO ₃)	177–689
Boron (B)	0.1-0.6
Calcium (Ca)	0-120
Chloride (C1)	5-117
Fluoride (F)	0-4.5
Iron (Fe)	0-9.9
Magnesium (Mg)	0-43
Nitrate, (NO ₃) [as Nitrogen]	0-10.0
Silica (SiO ₂)	1-30
Sodium (Na)	11-740
Sulfate (SO ₄)	6-1,080
Dissolved Solids	264-2,176
Total Hardness (CaCO ₃)	2-401
pH	7.1-9.2
Sodium-Absorption Ratio (SAR)	0.2-68.8
Residual Sodium Carbonate (RSC)	0-10.0
Specific Conductance (umhos at 25°C)	427-3,193
WOTE: Analyses given are in milligrams per conductance, pH, SAR, and RSC. SOURCE: Texas Department of Water Resources	
<u>Glen Rose Aquitard</u> - Below the Paluxy Aquifer are the fine-grained limestone, shale, marl, and sandstone beds of the Glen Rose Formation. The thickness of the formation ranges from 250 to 450 feet. Although the sands in the Glen Rose Formation yield small amounts of water to wells in Fort Worth and western Tarrant County, the relatively impermeable limestone is an aquitard restricting water movement between the Paluxy aquifer above and the Twin Mountains aquifer below.

<u>Twin Mountains Aquifer</u> - The Twin Mountains Formation is the oldest formation used for water supply in the AF Plant 4 area. The formation consists of a basal conglomerate of chert and quartz, grading upward into coarseto fine-grained sand interbedded with shale. The formation occurs approximately 600 feet below AF Plant 4. The thickness of the formation ranges from 250 to 430 feet.

Recharge to the Twin Mountains aquifer occurs west of AF Plant 4, where the formation crops out. Water movement is eastward in the downdip direction. Like water in the Paluxy aquifer, Twin Mountains water occurs in water table conditions in the recharge area and becomes confined as it moves downdip.

The Twin Mountains aquifer is the principal aquifer in Tarrant County. The formation yields large water supplies for municipal and industrial purposes. Transmissivities in the Twin Mountains aquifer range from 1,950 to 29,700 gpd/ft and average 8,450 gpd/ft in Tarrant County. Hydraulic conductivities range from 8 to 165 gpd/ft² and average 68 gpd/ft² in Tarrant County (CH2M Hill, 1984).

Groundwater withdrawals from the Twin Mountains aquifer, primarily for municipal water supply, have reduced the water levels. Between 1955 and 1976, the potentiometric surface of the aquifer dropped approximately 250 feet.

Sala a contrata a contrata con

Water suitable for potable use occurs in the Twin Mountains aquifer throughout the Fort Worth area; however, water in the upper sands in the aquifer east of Fort Worth may be too mineralized for human consumption.

2.4 Site Descriptions

Phase I studies for the AF Plant 4 Installation Restoration Frogram were completed by CH2M Hill in August 1984. The purpose of the Phase I study was to identify by means of a records search past waste management activities that may have caused groundwater contamination and the migration of contaminants.

Twenty sites were identified in the Phase I investigation. These 20 and one additional site (listed in Table 2-4) were investigated in the Phase II Stage 1 study. Locations of the waste sites are shown in Figure 2-9. Figure 2-10 is an index map, showing the areas covered by the larger scale, sitespecific maps presented in the following subsections. The general features of each site are discussed below. Figures associated with each site also show the locations of the monitor wells and soil borings discussed in Sections 3 and 4 below. Most of the information provided was obtained from the Installation Restoration Program Phase I records search report (CH2M Hill, 1984).

2.4.1 Site 1, Landfill 1

Landfill 1 was used to dispose of much of the facilities' wastes from 1942 until about 1966. This site, located west of Warehouse 3 (Facility 14) between the fence and Meandering Road (Figure 2-11), encompasses approximately six acres. Currently, the site is completely covered by an employee parking area.

Several types of hazardous and nonhazardous wastes were reportedly disposed of in Landfill 1. Drums of liquid wastes were buried in the

RAPIAN

TABLE 2-4. PHASE II STAGE 1 SITES AT AF PLANT 4

Site 1, Landfill 1 Site 3, Landfill 3 Site 12, Chrome Pit 3 Site 17, Former Fuel Storage Site Site 2, Landfill 2 Site 4, Landfill 4 Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; Site 8, FDTA 5) Site 15, Fuel Saturation Area (FSA) 2 Site 20, Wastewater Collection Basins Site 16, Fire Saturation Area 3 Site 9, Fire Department Training Area (FDTA) 6 Site 6, Fire Department Training Area 3 Site 7, Fire Department Training Area 4 Site 18, Solvent Lines Site 10, Chrome Pit 1 Site 5, Fire Department Training Area 2 Site 14, Fire Saturation Area 1 Site 19, Nuclear Aerospace Research Facility (NARF) Area Jet Engine Test Stand, Building 21



12227455

<u> aaaaa seesaa saadaa maxaa xaxaaa seesee</u> haadaa

DXXXX

2-22





2-24

MINUND

RAPIAN

5

landfill, and tanks and bousers loaded with chemical wastes (solvents, thinners, paints, etc.) were emptied into the landfill or into shallow pits at various locations throughout the landfill area. Miscellaneous contaminated oils and fuels were also disposed of in pits and were routinely burned. Sludge from these pits was periodically dredged out and deposited in the landfill area. At least five separate or overlapping pits are discernible on historical aerial photographs of this site. Other hazardous wastes suspected to be present at this site include metals (e.g., mercury, magnesium, etc.), chromate sludges, and cyanide. Although some hazardous wastes are probably scattered throughout much of the landfill, most of the fill material probably consists of trash, rubble, plasters (aircraft moldings), lumber, and fill dirt.

Extensive remedial action has been carried out here since the site was closed during the 1960s. Before the parking lot was graded for paving, two 6-inch perforated pipes were laid on bedrock to channel leachate drainage to a stormwater outfall to Meandering Road Creek. In 1982 and 1983, following identification of hazardous substances in the storm drain, the original drain lines were connected to an additional 20 feet of 4-inch perforated drain line (French drain), which was placed on bedrock upgradient from the existing storm sewer to intercept flows and to allow collection and disposal. The existing 36-inch storm sewer and catch basins were lined with PVC material to eliminate infiltration of percolate.

In 1983, additional remedial action was carried out. Since the main source of residual contamination was suspected to be the former waste oil pits, a portion of the site was excavated to bedrock and removed from the AF Plant 4 facility to an approved hazardous waste facility. To make this excavation, a portion of the parking lot was removed and approximately 11,000 cubic yards of earth were removed. Six 24-inch drain lines were placed in the bottom of this pit and connected to a collector box for surface collection of percolate. The pit was backfilled and has been reconverted to a parking area.

RADIAN

Percolate from the three collection points is periodically pumped, analyzed for pollutants, and disposed of off-site as hazardous waste when necessary.

2.4.2 Site 3, Landfill 3

Landfill 3 is located west of Landfill No 1. between Meandering Road and Meandering Road Creek (Figure 2-11). This landfill covers approximately 3 acres of land that was formerly a low area bordering the creek. Interviewees reported that from about 1942 to 1945 this site was used for disposing of miscellaneous wastes, including hazardous liquid wastes consisting of mixed oils and solvents. At least one pit (possibly more) present in this area during the 1940s was used for holding and burning some of the liquid wastes. Other wastes appear to have been disposed of on the ground and later buried. From about 1945 until 1966, there is no evidence of significant activity. From 1966 until 1967, fill dirt and rubble were used to finish filling and grading the site. Water quality analyses from upper zone monitoring wells (HM-26 and HM-27) have indicated the presence of elevated levels of volatile organic compounds in the groundwater. CH2M Hill (1984) considered these data to be indirect evidence for contaminant migration within the upper zone at Site 3.

2.4.3 Site 12, Chrome Pit 3

Chrome Pit 3 is located west of Warehouse 1 in the radar range area (Figure 2-12). The approximate dimensions of this pit are 66 feet by 165 feet. The depth is 10 to 15 feet. From about 1957 until 1973, barium chromate sludge, dilute metal solutions, and drums of unidentified liquids were disposed of in this pit.

Soil borings and shallow groundwater sampling conducted in 1982 confirmed contamination at this site. Approximately 8,900 cubic yards of contaminated soils were excavated during December 1983 and January 1984, and



RAPIAN

disposed of off site at an approved hazardous waste landfill. Soil testing conducted during the excavation indicated that most significantly contaminated soils had been removed from the site.

2.4.4 Site 17, Former Fuel Storage Site

The site of the former fuel storage tank is located at the southwest corner of AF Plant 4, near the center of the radar range area (Figure 2-13). A 100,000-gallon JP-4 aboveground fuel storage tank was located here from the early 1940s until 1962. Sampling at this site in 1982 confirmed that soils and underlying groundwater are contaminated by fuels and other organic compounds.

2.4.5 <u>Site 2, Landfill 2</u>

Landfill 2 is located in the northern portion of the present radar range, west of Warehouse 2 (Figure 2-13). This site, south of the employee parking area, occupies approximately 7.5 acres bordering Meandering Road. It originally consisted of some low areas and a livestock watering hole. Most of this site was reportedly filled with construction rubble, plasters, and fill dirt during the early 1940s; however, 1962 and earlier aerial photographs show some inspecified activity at the stock watering hole. This area, reportedly used for disposing of lumber and tires, was assumed to be periodically burned. No reports exist of hazardous substances being deposited at this site. The Phase I report provided no additional data on the nature of the activity at the site.

2.4.6 <u>Site 4, Landfill 4</u>

Landfill 4 is located near the southwest boundary of the AF Plant 4 facility (Figure 2-13). This landfill, formerly part of the low area bordering the adjacent creek, occupies approximately two acres of land west of Meandering Road. Landfill 4 was reportedly used for disposing of clean



- ELLERGY BOORD - TELEGON

2555554 255555

12222255

RADIAN

construction rubble from 1956 until the early 1980s. Aerial photographs and a 1973 memo suggest that other types of wastes may have been disposed of from 1966 until at least 1973. Because of this evidence, small quantities of highhazard wastes (solvents, oils, fuels, thinners, etc.) are thought to be present in the landfill. VOCs and other organic compounds have been found in two upper zone monitoring wells (HM-5 and HM-9) installed here in 1982. CH2M Hill (1984) considered these data to be indirect evidence of upper zone migration at this site. Recent data from the wells, however, give no indication of the presence of these compounds.

2.4.7 Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; Site 8, Fire Department Training Area 5)

Three sites have been combined into one zone, shown on Figure 2-12. The site 13 die pits are located on the west side of the present die yard near the south fence of AF Plant 4. Three pits approximately 20 feet by 90 feet were constructed here in 1956. The pits were used for disposing of chromate sludges, metal solutions, and other chemical wastes until 1962, when the site was graded and the entire die yard was paved. One Phase I interviewee reported that some of the contaminated soils at this site were spread around the die yard during grading and levelling activities. The site of the original pits was excavated in 1983-84, and approximately 1,100 cubic yards of contaminated soils were disposed of at an approved off-site hazardous waste landfill. No quantitative analyses of soils from other parts of the die yard were made at that time.

Site 11, Chrome Pit 2, located in the present die yard area, was used during the mid-1940s. Miscellaneous liquid and solid chemical wastes, in addition to chromate solutions, were probably disposed of at this site. According to the Phase I report, the actual location could not be accurately confirmed.

Site 8, the Fire Department Training Area 5, was reportedly used for fire extinguisher training during the mid-1960s. This site, consisting of a shallow pit about 10 feet by 20 feet, in which waste fuels, oils, or chemicals were deposited for training exercises, is located in the die yard area south of Warehouse 1. It has been graded and paved.

2.4.8 <u>Site 15, Fuel Saturation Area 2</u>

Fuel Saturation Area 2 is located just northwest of the Paint Shop. Facility 176 (Figure 2-14). This site reportedly became saturated by fuels from leaking buried fuel lines between the 1970s and the early 1980s.

2.4.9 Site 20, Wastewater Collection Basins

The Wastewater Collection Basins site is located just south of the Process Building, Facility 181 (Figure 2-12). Two concrete-lined waste basins, each with a capacity of 85,000 gallons, are used to collect and settle suspended solids from chemical wastewaters before discharge to the City of Forth Worth sanitary sewer system. Supernatant from these basins is analyzed for pH and chromium before discharge to the sewer. Settled sludge is periodically removed from these basins, dewatered, and disposed of off site. The basins have been used from about 1966 until the present. Evidence has shown that several spills from vapor degreaser tanks in the Process Building have occurred since installation of these tanks. Some of the spilled chemicals (primarily trichloroethylene) have flowed to the basins via floor drains. They are suspected to be a possible source of organic chemical and metal groundwater contamination (from cracks or leaking drains).

2.4.10 Site 16, Fuel Saturation Area 3

Fuel Saturation Area 3 is located just southwest of Facility 142 (Figure 2-14). This site reportedly became saturated by fuels from leaking



.

100

2-32



buried fuel lines between the mid-1970s and the early 1980s. This site is the subject of current remedial action planning by Intellus.

2.4.11 Site 9, Fire Department Training Area 6

Fire Department Training Area (FDTA) 6 was the designated FDTA from the late 1950s until 1980, when it was closed. This site consisted of a 50-foot diameter gravel-lined ring surrounded by a low, earthen berm. FDTA 6 is located north and adjacent to the DYNAFORM facility, Facility 175 (Figure 2-15). Before 1970, training exercises were conducted here twice a year. After 1970, exercises were conducted at monthly intervals. Approximately 250 gallons of waste fuels and oils were reportedly used for each exercise. In addition, the Phase I report indicated that larger quantities of contaminated fuels and oils were probably deposited in the FDTA between exercises.

In 1983, FDTA 6 was excavated and removed as part of the hazardous waste remedial action being carried out at AF Plant 4. The excavated material was analyzed and disposed of at approved off-site hazardous waste landfills. No monitoring wells have been installed to determine the status of groundwater contamination.

2.4.12 Site 6, Fire Department Training Area 3

FDTA 3, used during the mid-1960s for routine fire department training exercises, is located north of Landfill 4 between Meandering Road and the adjacent creek (Figure 2-13). Training exercises here also used about 250 gallons of waste fuels and oils per exercise. Since the site is not readily visible on historical aerial photographs, its location and present condition have not been accurately determined.

2.4.13 Site 7, Fire Department Training Area 4

FDTA 4 was also used for fire department training exercises during the late 1960s. This site is thought to be located north of the north



2-34



employee parking area, at a location known as "Tater Hill" (Figure 2-15). Training exercises used the same amount of waste fuels and oils. Since the site is not readily visible on historical aerial photographs of AF Plant 4, its location and present condition have not been accurately determined.

2.4.14 Site 18, Solvent Lines

The Solvent Lines site is located along the route of the former buried solvent lines from the northeast corner of Facility 15 to the test lab, Facility 80 (Figure 2-13). These lines reportedly leaked during the 1940s before they were drained, capped, and abandoned in place in 1944. According to the Phase I report, the actual locations of the leaks could not be determined. The contents of these solvent lines reportedly contained xylene, methyl ethyl ketone, and kerosene.

2.4.15 Site 10, Chrome Pit 1

Chrome Pit 1, located under the present process building, Facility 181, was used during the early 1940s (Figure 2-12). Miscellaneous liquid and solid chemical wastes, in addition to chrome wastes, were probably disposed of at this site. The actual location could not be accurately confirmed from Phase I interviews or aerial photographs.

2.4.16 Site 5, Fire Department Training Area 2

Fire Department Training Area (FDTA) 2 was used between 1955 and 1956. This site consisted of a 50-foot diameter earthen ring located just north of Landfill 1 (Figure 2-11). Exercises were held infrequently (twice a year); however, disposal of waste oils and fuels and uncontrolled burns may have been more frequent. This site is currently located under the pavement in the west employee parking area.



2.4.17 Site 14, Fuel Saturation Area 1

Fuel Saturation Area 1 is located west of, and adjacent to, the Parts Plant, Facility 5 (Figure 2-11). This area is just north of the fuel tank truck unloading and pumping station. From the 1970s up until the early 1980s, the ground at this site reportedly became saturated by fuels from leaking buried fuel lines. This site is the subject of current remedial action planning by Intellus.

2.4.18 Site 19, Nuclear Aerospace Research Facility (NARF) Area

The Nuclear Aerospace Research Facility (NARF) formerly located at the north end of AF Plant 4 (Figure 2-15), housed several experimental atomic reactors between 1953 and 1974. The facility was decommissioned and disposed of by contractor in 1974.

2.4.19 Jet Engine Test Stand, Building 21

This site, a jet engine test building, (Figure 2-15) was not previously identified in Phase I. It is located north of a fuels test area, previously identified as Site 16 and a known area of fuel contamination. The sump downslope of Building 21, constructed in 1975, collects water used for cooling, noise suppression, and clean up in Building 21. This water is pumped into the industrial waste line. The area immediately south of Building 21 contains two underground tanks once used for fuel storage. These tanks, along with all others, are being investigated as potential sources of contamination by the operating contractor, General Dynamics. Outfall 3, downgradient of Building 21, is regularly sampled as part of IRP and NPDES monitoring activities.

RADIAN

3.0 FIELD PROGRAM

Radian performed the following activities at AF Plant 4 as part of the IRP Phase II Stage 1 investigation: 1) Geophysical surveys in the vicinity of two landfill areas and a fire department training area; 2) Installation of nine shallow (upper zone) groundwater monitor wells; 3) Installation of four monitor wells into the Paluxy aquifer; 4) Drilling of nine soil borings; 5) Hand-augered borings at one site; 6) Soil gas survey; and 7) Collection and analysis of groundwater, surface water, and soil samples. The period of performance of the field activities was from November 1985 through August 1986.

3.1 Field Techniques

The following sections contain descriptions of the field techniques used in this investigation. The techniques discussed include geophysical surveys, soil survey, hollow-stem auger and rotary drilling, monitor well installation, hand augering, and soil and water sampling. The Technical Operations Plan (TOP) presented in Appendix K describes field procedures in detail.

3.1.1 Terrain Conductivity (Geophysical) Surveys

Terrain conductivity surveys using electromagnetic profiling (EMP) were performed to assess the vertical and lateral extent of wastes and to gain a better understanding of subsurface conditions. The EMP technique measures conductance of subsurface materials. Anomalous conductance values represent the presence of materials with different conductivity than the surrounding materials. In addition to natural variations in conductivity, anomalies may be due to factors such as the presence of metal objects, fill material, or chemical contamination in the subsurface.

Survey results were used to evaluate the extent of contamination and the need for and location of soil borings. EMP surveys were conducted at Landfill 2, Landfill 3, and Fire Department Training Area (FDTA) 2. A Geonics



EM-34 terrain conductivity meter was used to conduct the electromagnetic profiling. Delta Geophysical, a Radian subcontractor, performed the geophysical surveys.

Before conducting the surveys, a control grid was established using appropriate reference points (buildings, roads, etc.). Within each survey area, baselines were established and gridpoints were staked at 30-foot intervals along the baseline. Conductivity data were collected at stations at 30-foot spacings perpendicular to the baselines. This provided a 30 by 30-foot grid for each area.

The depth measured by the EM-34 depends on the coil separation and orientation of the instrument, on the applied frequency, and, to some extent, on the conductivity profile of the subsurface. To evaluate boundary conditions and types of subsurface material present beneath each site, conductivity data were gathered from an effective depth of 45 feet. To identify plumes of contamination migrating from each site area, conductivity data were collected from two effective depths: 45 and 90 feet.

3.1.2 Soil Gas Survey

The presence of volatile organic chemicals (VOCs) in shallow soil gas indicates contamination from the observed compounds either in the vadose zone near the probe or in groundwater below the probe. The technology is most effective in mapping low molecular weight halogenated solvent chemicals and petroleum hydrocarbons possessing high vapor pressures and low aqueous solubilities. These compounds readily partition out of the groundwater and into the soil gas as a result of their high gas/liquid partitioning coefficients. Once in the soil gas, VOCs diffuse vertically in response to a chemical concentration upgradient where the above-ground atmosphere acts as a sink and the groundwater acts as a source for the compounds. RADIAN

nooraan baaaaaa kaanaa kuu kaanaa kuu kaanaa kaasaa kaasaa kaasaada kaasaada kaasaada kaasaada kaasaa kaasaa ka

A field van equipped with two Varian 3300 gas chromatographs and two Spectra-Physics SP4270 computing integrators was used in the survey. Soil gas samples were collected by driving a hollow steel probe from 2 to 12 feet into the ground and evacuating 5 to 10 liters of gas with a vacuum pump. During the soil gas evacuation, samples were collected by inserting a syringe needle through the silicone evacuation line and down into the steel probe. Ten milliliters of gas were collected for immediate analysis in the analytical field van. Soil gas was subsampled (duplicate injections) in volumes ranging from 1 uL to 2 mL, depending on the VOC concentration at any particular location. Detection limits for the halogenated hydrocarbons ranged from 0.00002 to 0.00007 ug/L in soil gas using the electron capture detector, while the limits for petroleum hydrocarbons ranged from 0.01 to 6.0 ug/L using the flameionization detector. Detection limits are a function of the injection volume as well as of the detector sensitivity for individual compounds. The minimum detectable quantity of halogenated hydrocarbons (e.g., TCA) is about 2.5 x 10⁻¹¹ grams, while the minimal detectable quantity of petroleum hydrocarbons (e.g., benzene) is approximately 5 x 10⁻⁹ grams.

Quality assurance procedures were followed to ensure that crosscontamination of soil gas samples was not occuring. Before sampling syringes were purged with nitrogen (i.e., carrier gas) and checked for contamination by injection into the gas chromatograph. System blanks were run periodically to confirm that the probes, adaptors, and 10 mL syringes were not contaminated. Soil gas probes were used only once during the course of a working day and then thoroughly cleaned before use on the following day. Analytical instruments were continuously checked for calibration using standards prepared in water from commercially available pure chemicals. Soil gas pumping was monitored by a vacuum gauge to ensure that an adequate gas flow from the vadose zone was maintained.

Soil gas was collected from a total of 28 sampling points at approximately 100-foot intervals at FDTA 4. An area approximately 600 by 600-feet was surveyed. The target depth for soil gas sampling was 5 to 6 feet below

3-3

RAPIAN

the ground surface at the majority of probe locations. At 8 of the sampling locations, probes were driven to the 8-12 foot depth interval to determine if VOCs were being biodegraded in the shallow soil gas. Since petroleum hydrocarbons (e.g., aviation fuel components) are particularly susceptible to oxidative biodegradation in the shallow soil, where aerobic conditions prevail, sampling for petroleum hydrocarbons is usually conducted at greater depths than for halocarbons.

Soil gas samples were analyzed for methane, total petroleum hydrocarbons, carbon tetrachloride (CCl_4) , 1,1,1-trichloroethane (TCA), trichloroethylene (TCE) and perchloroethylene (PCE). Total petroleum hydrocarbons were defined as C_4 to C_{10} aliphatic and alicyclic compounds, as well as benzene, toluene and xylene isomers. These petroleum compounds are the major constituents of aviation fuels. Methane is produced as a by-product of subsurface biodegradation, and the other four compounds (i.e., CCl_4 , TCA, TCE and PCE) are the most common industrial solvents.

3.1.3 Drilling Techniques

Drilling at AF Plant 4 was performed to directly determine subsurface conditions by collecting and describing soil samples and to install groundwater monitor wells. Before the start of drilling, General Dynamics personnel were informed of monitor well and borehole locations for utility clearances.

Drilling at AF Plant 4 was accomplished using a hollow-stem auger rig for the upper zone monitor wells and soil borings. A rotary drilling rig (both mud and air) was used for the monitor wells installed into the Paluxy aquifer. Selection of these methods was based on the anticipated depth of completion, the need for water-level observations, and the expected geologic conditions. The following paragraphs describe the procedures used for drilling activities.



Hollow-Stem Augering

A hollow-stem auger drilling rig, the CME-75, was used to perform shallow soil borings and to install the upper zone monitor wells. The hollow-stem method allowed an accurate examination of soil conditions, identification of the position of the water table, and recovery of soil samples. The holes were drilled dry; no drilling fluids or additives were used. After each borehole was completed, the drilling rig and equipment were thoroughly cleaned to prevent cross-contamination.

Samples of soil were collected with either a Shelby-tube or a splitspoon sampler. The samplers were driven in advance of the auger flights. Samples were generally taken at five foot intervals or where lithologic changes were noted. The samples were recovered at the surface, described in terms of lithology and moisture, and retained. At certain sites, a pre-determined number of soil samples were selected for chemical analysis. At least one sample was from the water-table interface; selection of other samples was based on visible contamination of the soil, color or texture, or real-time organic vapor analyzer readings. The samples selected for chemical analysis were cooled to 4°C and shipped to Radian's laboratory. The parameters analyzed are listed in Table 1-2.

Cuttings suspected of contamination (on the basis of OVA readings and visual evidence of contamination) were placed in steel 55-gallon drums. Samples of the cuttings were collected and submitted for chemical analysis. Disposal of the cuttings was based on the results of the chemical analyses. Section 3.4 discusses the chemical analysis of drill cuttings and fluids produced in the field program.

Rotary Drilling

Rotary drilling techniques were used for the installation of monitor wells into the Paluxy Formation. The rotary drilling was performed with a Gardner-Denver 1500 CD truck-mounted rig. Using the air rotary method, a 6.75-inch diameter bit was used to advance a pilot borehole through the upper zone alluvial material and/or the Goodland Formation to a depth of approximately five feet into the underlying Walnut Formation. The borehole was then reamed with a 20-inch bit using mud rotary. All drilling operations in the upper zone material were continuously monitored with an OVA. A 16-inch diameter steel casing was installed from the surface to the full depth of the borehole and the annular space was grouted. After achieving a positive seal and allowing the grout to cure for 24 hours, the borehole was advanced using a 15-inch bit and mud rotary method through the Walnut Formation and approximately five feet into the upper sand member of the Paluxy Formation. A 10.75inch diameter steel casing was installed from the surface to the full depth and the annular space was pressure-grouted to the surface. After achieving a positive seal and allowing the grout to cure for 24 hours, the borehole was advanced using an 8-inch bit and air rotary (if possible) or mud rotary to the final depth at the shale/clay unit dividing the upper and middle sand members of the Paluxy Formation. One Paluxy well (P-20M) was completed in the middle sand member because the upper sand member was dry.

The drilling fluid used during mud rotary operations consisted of potable water and bentonite. As the borehole was advanced, the cuttings discharged at the surface were examined for lithology, moisture, and other features useful in describing stratigraphic conditions. All cuttings were retained in a portable, above-ground, mud pit and transfered to steel 55-gallon drums pending the results of chemical analyses. Drilling conditions, such as relative rate and ease of penetration, were noted by the driller. Water encountered during drilling was noted with respect to depth of occurrence. When drilling was completed, a sample of the drilling fluids was placed into a clean, one quart glass mason jar with a Teflon-lined lid and chilled for shipment to Radian's laboratory. The drilling mud was analyzed for purgeable halocarbons, as well as for EP Toxicity and Ignitability.

3-6

Radian

3.1.4 Monitor Well Installation and Development

Groundwater monitor wells were installed either immediately after completion of the drilling operations or after the borehole was noted to produce sufficient water to warrant a well. Monitor well construction specifications are summarized in Table 3-1. Construction methods were consistent with the specifications provided in the Statement of Work. These methods were outlined in the Technical Operations Plan, which was provided to the USAFOEHL on 14 November 1985 for outside distribution for review and comment. All well materials were cleaned before installation in the borehole. Where necessary, appropriate changes were made on a site-by-site basis and approved by the USAFOEHL Technical Program Manager. Decisions relating to the setting of screen and casing, the length of screen, and the amount of sand pack were made on the basis of observed water seepage and static water level. If appropriate, the borehole was allowed to remain open overnight. There were no difficulties associated with caving or borehole integrity.

Upper Zone Monitor Wells

All upper zone monitor wells were completed (screened) in the alluvium or man-made fill overlying the plant site. Wells were installed in the following manner: screen and casing sections were steam-cleaned and assembled on the ground, then lowered carefully into the borehole. As the string of screen and casing was lowered, additional sections of casing were added until the bottom of the screen reached the bottom of the borehole. The top of the casing was capped during construction to prevent downhole completion materials (sand and grout) from entering the casing. Clean sand (Texas Blastsand No. 1A) was poured slowly down the annular space until the top of the sand pack was one to two feet above the top of the screen, as measured by the supervising geologist. Bentonite pellets were then poured from the surface on top of the sand pack and hydrated to form a two foot thick, impermeable seal. Following emplacement of the bentonite seal, neat cement grout



TABLE 3-1. MONITOR WELL CONSTRUCTION SPECIFICATIONS, IRP PHASE II STAGE I, AF PLANT 4, TEXAS

Casing:
Upper Zone - Two inch diameter, threaded and flush joint, Schedule 40 PVC.

- Paluxy 16 inch steel casing, into top of Walnut Formation and grouted; 10 3/4 inch steel casing through Walnut Formation and grouted; 5 inch, Schedule 80 PVC to total depth.
- 2. Screen:
 - Upper Zone Two inch diameter, threaded and flush-joint factory-slotted, Schedule 40 PVC, 0.020 inch slot. Normal screen length 10 feet.
 - Paluxy- 5 -inch diameter, threaded flush-joint factory-slotted Schedule 80 PVC, 0.010 inch slot.
- 3. Sand/gravel pack: Washed, bagged and rounded sand/gravel with grain size compatible with screen slot and formation (Texas Blastsand No. 1A). A sand pack was emplaced from bottom of the borehole to 2 feet above the top of the well screen. Sand was emplaced at a controlled rate to avoid bridging.
- 4. Bentonite seal: Two feet of granulated or pelletized bentonite was placed above the sand section.
- 5. Grout: Neat cement (Type I Portland cement) grout tremied from the top of the bentonite seal to land surface. A grout mixture of 6:1 Portland cement and bentonite was used. The grout was allowed to set for at least 24 hours before any well development activities.
- 6. Surface completion: The PVC casing was cut off to provide a two to three foot stickup and a solid cap was placed on the casing. A 5 inch square protective steel well protector, 4 feet in length, was placed over the exposed PVC casing, and seated in the cement. A locking cap is incorporated in the well cover.
- 7. Alternate flush completion: The PVC casing was cut off after installation about 3 inches below ground level. An end plug or casing cap was provided for each well. A steel flush utility type box was placed over the exposed casing and seated in the cement. A locking well cap lid or box was used. The utility box cover was painted for corrosion control and visibility.
- 8. Guard pipes or posts: Three 3-inch diameter steel posts, 6 feet in length, with a minimum of 2 feet below ground, installed radially 4 feet from the wellhead (Not emplaced for flush surface completion).



was prepared (6-gallons of potable water per 100-1b sack of Type I Portland cement) and tremied from the top of the bentonite seal to the land surface. While the cement was still wet, surface well covers were installed. The covers consisted of either a protective locking metal meter box or a 4-inch diameter steel casing with lockable lid. To protect the well from vehicular damage, three steel guard posts were positioned radially 2-3 feet from each above-ground well completion.

After allowing the cement grout to cure in the well for a minimum of 24 hours, the wells were developed using a hand-operated pump or by bailing using a ball-valve bailer. Some of the wells were slow to recover and were bailed dry several times and allowed to recover before development was considered complete. Other wells produced sufficient water to be developed completely without a recovery period. Water was removed from the well until the water was virtually sediment-free. Appendix E contains the records of well development operations.

Southwestern Laboratories of Dallas, Texas, a Radian subcontractor, performed the upper zone monitor well drilling, monitor well installation, and development.

Paluxy Formation Wells

The Paluxy Formation wells were installed through a steel surface casing that was grouted in place through the Walnut Formation, as discussed above. The completion activities and materials for the Paluxy wells are as follows. Screen and casing, consisting of 5-inch diameter Schedule 80 PVC, was installed into the 8-inch borehole. Clean sand (Texas Blastsand No. 1A) was slowly poured down the annular space to a level of two feet above the top of the screen. Bentonite pellets were then placed on top of the sand pack to a thickness of 2-feet to form a seal. Neat cement was then tremied down the annular space until the grout was within the 10.75-inch steel casing to a thickness of at least 5 feet. The remainder of the annular space was filled



with clean sand to 1 foot from the surface, after which cement was then emplaced. The well was developed by bailing until a virtually sediment-free discharge was produced. A 1/2 horsepower stainless steel submersible pump was installed after development. A protective metal box, surface electrical connections, steel guard posts, and a concrete well pad were placed after the pump was installed.

Underground Resource Management, Inc. of Austin, Texas, a Radian subcontractor, performed the drilling, monitor well installation, and development of the Paluxy wells.

3.1.5 <u>Surveying</u>

After all new monitor wells were installed, wellhead elevations were determined to the nearest 0.05 foot and horizontal locations of the monitor wells were determined to an accuracy of 1 foot. Monitor well locations were then drafted onto a map of AF Plant 4. A local surveying firm, Sempco, Inc., performed this task. Table 3-2 presents results of the survey of monitor wells, including depths and screened intervals.

3.1.6 Environmental Sampling

Environmental samples collected as part of this investigation included soil, surface water, and groundwater. The following subsections describe the techniques used in acquiring samples.

3.1.6.1 Soil Sampling

Soil samples were collected by hand augering at one site and by Shelby tubes or split-spoon samplers at boreholes advanced by hollow-stem augering. GENERAL SPECIFICATIONS FOR MONITOR WELLS INSTALLED DURING IRP PHASE II STAGE 1, AF PLANT 4, TEXAS **TABLE 3-2.**

	Measuring Point				Screen	Total,
Monitor Well	Elevation (feet MSL)	(East) Range	(North) Range	Screened ₁ Interval	Elevations (feet MSL)	Depth ¹
HM-100	670.10	33+55.16	5+88.55	33.5-48.5	636.9-621.9	49
HM-101	664.99	41+47.58	13+34.64	19-34	646.3-631.3	34
HM-102	654.65	37+63.73	16+03.47	14-44	641.0-611.0	44
HM-103	648.28	10+91.76	07+26.61	29–39	619.6-609.6	39
HM-104	649.71	18+34.45	00+04.28	23.5-38.5	626.5-611.5	39
HM-105	641.92	20+58.27	59+54.99	5.5-20.5	636.8-621.8	21
HM-106	649.76	22+60.57	49+66.64	15.1-30.1	635.0-620.0	34
HM-107	642.33	22+58.27	60+22.15	7.5-22.5	631.2-616.2	23
HM-108	639.76	22+56.65	60+96.95	5-15	631.9-621.9	15
P-20M	673.23	46+35.42	9+48.13	125-155	547.9-517.9	155
P-21U	660,99	36+02.36	14+85.83	85-105	575.7-555.7	110
P-22U	631.98	31+23.23	28+52.24	47.8-62.8	583.9-568.9	62.8
P-23U	626.92	24+09.67	61+78.15	35.6-55.3	591.0-571.3	55.3

¹Feet below ground level.

3-11

2

SYMMAN MARKA PROVING BASSAN NYAWA ANGGOR BASANG

1.2556

1.000000

200200

22.22

40°,40°,50°,54°,54°

ķ

.

RADIAN

Hand augering was performed at FDTA 6. Samples were retrieved from depths ranging from six to 18 inches at six locations within the perimeter of the FDTA. The sampling technique involved precleaning the auger, augering down until the auger was filled with soil and inspecting the contents for visual or aromatic indications of hydrocarbons. At the surface, each sample was placed in a clean one quart glass mason jar with a Teflon-lined lid and cooled to 4°C for shipment to Radian's laboratory for analysis. Soils were analyzed for the parameters listed in Table 1-2. Soil samples were also taken with Shelby tubes and split-spoon samplers during drilling in the upper zone, as described in Section 3.1.3.

3.1.6.2 Surface Water Sampling

During Phase II Stage 1 efforts at AF Plant 4, surface water samples were collected from five creek locations, five outfalls, and two French drains. Surface water grab samples were collected directly in the clean container to minimize handling of the sample.

Field measurements of specific conductance, pH, and temperature were taken on an aliquot of the sample recovered. Specific conductance and pH were taken using a Myron L EP-10/pH meter. Temperature was taken using a mercury thermometer. Before the field measurements the pH meter was calibrated using pH 4, 7, and 10 standard solutions and the conductivity meter was calibrated using a 1413 umhos/cm KCl conductivity standard solution. All instruments used during field measurements were cleaned with deionized water before use.

At each location, observations were made regarding the flow conditions and other pertinent featurs. Table 1-2 shows the analytical schedule for surface water samples. The samples were stored and preserved according to the requirements outlined in Table 3-3.

3.1.6.3 Groundwater Sampling

Groundwater samples were collected from 113 monitor wells. Sampling was conducted once for each well except for one Paluxy Formation well (P-23),



TABLE 3-3. COLLECTION AND PRESERVATION OF WATER SAMPLES, IRP PHASE II STAGE 1, AF PLANT 4, TEXAS

Parameter	Container*	Preservation
Purgeable Halocarbons	Glass, 40 mL VOA vial	Cool to 4°C
Purgeable Aromatics and Xylene	Glass, 40 mL VOA vial	HCl to pH<2; Cool to 4°C
Base/Neutral and Acid Extractable Organics and Pesticides	Amber glass, 1 liter	Cool to 4°C
Metals	Plastic, 500 mL	HNO ₃ to pH<2; cool to 4°C
Hydrocarbon Fuels	Glass, 1 quart	H_2SO_4 to pH<2; cool to 4°C
Oil and Grease	Glass, 1 quart	H_2SO_4 to pH<2; cool to 4°C

*All containers with Teflon[™]-lined lids.



which was sampled twice. Groundwater sampling activities began in January 1986, and were concluded in August 1986. Table 1-2 shows the analytical schedule for groundwater samples. The samples were stored and preserved according to the requirements outlined in Table 3-3. Field sampling methodologies and equipment are detailed in the following sections and in the Technical Operations Plan in Appendix K.

Water Level Measurements

Water level measurements at each monitor well were taken with an Actat Olympic well probe immediately before sample collection. The probe and associated electrical line were washed with laboratory grade detergent, rinsed with tap water, then rinsed with deionized water between measurements to preclude the possibility of cross-contamination. Measurements were taken to the nearest 0.01 foot with respect to the top of the PVC casing. Water level measurements taken before each sampling operation are shown in Section 4.

Purging Operations

Each monitor well was purged either immediately before sample collection or within one day of sample collection (for low yield wells) to ensure that representative formation water was collected. Purging operations were conducted using a Teflon bailer, a hand pump or an installed electric submersible pump. Purging operations were considered complete when three wetted well volumes had been evacuated from a monitor well. All downhole equipment used for purging operations was cleaned with laboratory grade detergent and tap water, rinsed with tap water, then rinsed with deionized water to prevent crosscontamination. In cases where overt evidence of chemical contamination was noted in a well (color, odor, oil, etc.) the bailer was washed with technical grade methyl alcohol, thoroughly rinsed with deionized water, and allowed to dry completely.

RAPIAN

Field Measurements

Specific conductance, pH, and temperature measurements were taken from an aliquot of the sample recovered. Specific conductance and pH were taken with a Myron L EP-10/pH meter. Temperature was taken with a mercury thermometer. Before the field measurements, the pH meter was calibrated using pH 4, 7, and 10 standard solutions and the conductivity meter was calibrated using a 1413 umhos/cm KCl conductivity standard solution. In some instances, field data were not collected due to instrument malfunction, possibility of instrument fouling, or oversight. These are noted in the data tables in Section 4.3.

Sample Capture

Samples were collected after each well was purged of standing water to ensure that groundwater characteristics were representative. Samples were collected for the analyses shown in Table 1-2. After collection, samples were placed in containers and preserved according to the requirements listed in Table 3-3. All samples were chilled to 4°C after collection. Samples were placed in ice chests and shipped by overnight express or hand carried to the analytical laboratory. Field quality assurance measures (e.g., chain of custody forms) were employed to ensure that, once samples were collected, sample integrity was maintained during shipping and handling and before analysis. QA/QC procedures are discussed in the Technical Operations Plan (Appendix K).

3.1.7 Numbering System for Wells and Sampling Locations

The wells and borings drilled at Air Force Plant 4 during the IRP Phase II Stage 1 investigation are identified by an alpha-numeric label. Upper zone monitor wells are labeled with the letters HM or F, followed by a dash and a number. The HM series represents monitor wells installed by Hargis & Montgomery. Hargis & Associates and Radian. Hargis & Montgomery and Hargis & Associates installed monitor wells 1 through 85. Radian installed HM wells 100 through 108. The F series of wells represents those monitor wells installed by Intellus.



Monitor wells drilled into the Paluxy Formation are denoted by the prefix P, followed by a number. In most cases, the number is followed by a U or an M, representing completion in the upper Paluxy or middle Paluxy. Where the P is followed by a number only, the well was completed throughout multiple Paluxy Formation aquifers.

Soil borings are labeled with the letters SB, followed by a number. Surface water sampling points on Meandering Road Creek are designated by the letter C, followed by a number. Outfall samples are labeled OF, followed by the outfall number.

3.1.8 Field Safety

Before field work began, a detailed field Safety Plan was prepared which anticipated likely field hazards and prescribed appropriate personnel protective equipment for the field team. EPA Level C protection (impervious clothing, gloves, boots, and full-face or half-face cartridge respirators) was employed during drilling and/or well installation at Landfills 1 and 3. During drilling operations soil samples and drill cuttings were monitored for signs of contamination with a photoionization detector and Draeger tubes. If no contamination was detected, then EPA level D protection (same as level C, except that respirators were carried, but not worn) was deemed appropriate. Level D protection was used at all other drilling locations. The complete text of the Safety Plan used for this project is contained in the Appendix K.

3.2 Site Activities

Field activities conducted at each site in the AF Plant 4 IRP Phase II Stage 1 investigation are described in the following subsections. Site activities are presented and discussed in the order in which they appear in the delivery order. Wells and sampling activities are discussed with their associated sites in the Statement of Work. Figure 3-1 is an index map showing



EXECUTION DOWN

1 . 12 . 13 . 12 . 14 . 15



the areas covered by the site-specific maps presented in the following sections. Site activities are summarized in Table 3-4.

3.2.1 Site 1, Landfill 1

Investigation activities at Landfill 1 included collection of surface water samples from five creek locations, the creek seep, and two French drains; and collection of groundwater samples from six upper zone monitor wells and two Paluxy monitor wells. The St-5 outfall was identified in the delivery order as a sampling location; however, the outfall was dry during the field program, precluding sample collection. Figure 3-2 shows the location of Landfill 1.

Surface Water Sampling: The purpose of surface water sampling was to determine if contaminants were present in Meandering Road Creek west of AF Plant 4. Sampling at 5 locations along the creek and at one creek seep location was performed by Radian personnel January 29 and 30, 1986. These locations were spaced along the full length of the creek; however, only 2 sampling locations, C-2 and C-3, are located directly opposite (west) of Landfill 1 and Landfill 3. The French drains and the drainpipe were sampled on 12 February 1986. Details of the sampling procedures are outlined in Section 3.1.5.2. The surface water samples were analyzed for VOC, BNA, and heavy metals. Results of the surface water analyses are presented in Section 4.3.1.

<u>Groundwater Sampling</u>: One sample from each of 6 upper zone wells and 3 Paluxy wells was collected by Radian personnel during 29 January through 1 March 1986. The purpose of groundwater sampling and analysis was to confirm earlier investigations and to gain additional data concerning the nature and extent of contamination. Details of the sampling procedures are outlined in Section 3.1.5.3. Groundwater samples were analyzed for VOC, BNA, and heavy metals. Results of the groundwater analyses are presented in Section 4.3.1.
Site	Investigation Activities	Purpose	Location
Landfill 1	1. Sample surface water at 5 loca- tions on Meandering Road Creek and the creek seep.	 Evaluate possible impact of landfill leachate on creek. 	Figure 3-2.
	2. Sampling French drains 1 and 2 and the drain pipe.	 Characterize leachate discharge. Evaluate possible effect of landfill on groundwater. 	
	3. Sample groundwater from 6 existing upper zone and 2 Paluxy wells.		
Landfill 3	 Conduct a terrain conductivity survey. 	 Determine presence and possible configuration of the landfill. Use data to determine if 	Figure 3-2.
	 Install Paluxy monitor well screened in upper sand member. 		
	 Sample groundwater from 9 existing upper zone wells and 1 new Paluxy well. 	 Evaluate Faluxy groundwater directly downgradient of landfill. 	
		 Bvaluate possible effect of landfill on groundwater. 	
Chrome Pit 3	 Sample groundwater from 8 existing upper zone wells and 1 Paluxy well. 	1. Determine post-closure groundwater quality.	Figure 3-4.

 \mathbb{C}^{n}

1.0

2

31	Ð		¥?	7
R				
1 333				
R				
F				
Ŕ				
Ę				
E.				
ř				
Ŕ				
Ŕ				
Ċ,				
F.				
5				
6				
Ľ,				
E				
E.	•	-		
20	<u>i</u>		Ś	6

RA	DIA	N .		
	Map Location	Figure 3-5.	Figure 3-5.	Figure 3-5.
(Continued)	Ригрове	 Provide data on groundwater flow and quality in Radar Range area. Evaluate possible effect of former fuel storage site on groundwater. 	 Determine landfill limits and horizontal extent of contamination, if any. Evaluate Paluxy groundwater directly downgradient of landfill. Evaluate possible effects of landfill on groundwater. 	1.2. Determine possible effects of landfill on groundwater.
TABLE 3-4. (Continued)	Investigation Activities	 Install 1 upper zone well. Sample groundwater from 2 existing and wells 1 new upper zone well. 	 Conduct a terrain conductivity survey. Install Paluxy monitor well in upper sand member. Sample groundwater from 7 upper zone wells and 1 new Paluxy well. 	 Install 1 upper zone well and 1 Paluxy well screened in middle sand unit (upper unit was dry). Sample groundwater from 3 upper zone wells and 1 Paluxy well.
	Site	Former Fuel Storage Site	Landfill 2	Landfill 4

(continued)

Sector of Sectors

122222250

JALLAL

2000000

L'INDERIC'

٠,

3-20

when a lot

N 8 9 W

 Drill borehole at FDTA 5. Drill borehole at FDTA 5. Sample groundwater from 7 existing upper zone wells and 1 Paluxy well. Sample groundwater from 7 existing vell. Drill 3 boreholes along the length of the buried fuel line. Drill 3 boreholes along the length of the buried fuel line. Drill 3 boreholes along the length of the buried fuel line. Sample groundwater: Sample groundwater: Braine groundwater. Install 1 upper zone well. Install 1 upper zone well. Install 1 upper zone monitor well. Braine groundwater. Braine on groundwater. Braine on groundwater. Braine on groundwater. Braine of contamination in subsurface. Determine location and depth of buried sampling and analysis of soil. 		Investigation Activities	Purpose	Map Location
 well. 2. Evaluate possible effects of disposal pits and fire training area on groundwater. Drill 3 boreholes along the length of the buried fuel line. Drill 3 boreholes along the length of the buried fuel line. Sample groundwater and measure fuel lens at 1 upper zone well. Install 1 upper zone monitor well Inted sanitary and industrial waste lines. Evaluate need for subsurface. Determine location and depth of buried sanitary and industrial waste lines. Evaluate need for subsurface. 	7 1	Drill borehole at FDTA 5. Sample groundwater from 7 upper zone wells and 1 Pal	 Determine soil conditions and presence of water underlying FDTA 5. 	Figure 3-4.
Drill 3 boreholes along the length of the buried fuel line. Of the buried fuel line. Install burger and measure fuel lens at 1 upper zone well. Install 1 upper zone monitor well Install 2. Evaluate possible effect of fuel Figure Sample groundwater from 5 upper Sample g		well.	Evaluate disposal area on g	
<pre>fuel lens at 1 upper zone well. 2. Evaluate possible effect of fuel Install 1 upper zone monitor well 1.2. Evaluate possible effects of Figure (located south of site because of basins on groundwater. access restrictions). 3. Evaluate effect of lines, if any. Sample groundwater from 5 upper on movement of contamination in subsurface. Determine location and depth of buried sanitary and industrial waste lines. Evaluate need for additional sampling and analysis of soil.</pre>	7 7	Drill 3 boreholes al of the buried fuel 1 Sample groundwater a	Determine soil conditions ar presence of fuel in the subsurface.	Figure 3-6.
<pre>Install 1 upper zone monitor well (located south of site because of basins on groundwater. (located south of site because of basins on groundwater. access restrictions). 3. Evaluate effect of lines, if any, on movement of contamination in subsurface. Determine location and depth of buried sanitary and industrial waste lines. Evaluate need for additional sampling and analysis of soil.</pre>			Bvaluate possible effect of line on groundwater.	
Sample groundwater from 5 upper on movement of contamination zone wells. Subsurface. Subsurface. Betermine location and depth of buried sanitary and industrial waste lines. Evaluate need for additional sampling and analysis of soil.		. Install 1 upper zone monitor well (located south of site because of access restrictions).	2. Evaluate possible effects obside the constant of the conductor.	Figure 3-4.
Determine location and d buried sanitary and indu waste lines. Evaluate r additional sampling and of soil.	2	Sample groundwater / zone wells.	on movement of contamination subsurface.	
	Ś	Determine location and c buried sanitary and indu waste lines. Evaluate r additional sampling and of soil.		

N.S.S.S.

CALCON REPRESE

>

3

.....

Site	Inv	Investigation Activities	Purpose	nap Location
Fuel Saturation Area 3	1. Sample well.	.e groundwater at 1 upper zone	 Evaluate possible effect of buried fuel line on groundwater. 	Figure 3-7.
Fire Department Training		Hand-auger and collect shallow soil samples for analysis.	 Determine possible effects of fire training exercises on shallow soil. 	Figures 3-7, 8.
Area o	2. Sampl well.	Sample groundwater from 1 Paluxy well.	 Evaluate groundwater quality of Paluxy upgradient of AF Plant 4. 	
Fire Department Training	1. Insta west alrea	Install 1 upper zone monitor well west of FDTA 3 (monitor well already exists at site).	 Determine groundwater flow conditions west of FDTA 3. 	Figure 3-5.
Area 3	2. Sampl zone	Sample groundwater from 2 upper zone wells.	 Evaluate possible effects of FDTA 3 on groundwater. 	
Fire Department Training Area 4	1. Condu	1. Conduct a soil-gas survey.	 Determine location of FDTA 4 and possible effects on shallow soil. 	Figure 3-7.
Solvent Lines	1. Insta 2 Somol	Install 1 upper zone monitor well.	 Improve monitoring coverage of groundwater conditions. 	Figure 3-6.
		bampie groundwater from 4 upper zone monitor wells.	 Evaluate possible effect of solvent lines on groundwater. 	

SAMMAN MALAN DESERTING DESERT DESERTE DESERTE

100000000

25,555,555

22222222

Site		Investigation Activities		Purpose	Location
Chrome Pit 1	1.	Install upper zone monitor well east of site (outside of	1.	Evaluate groundwater conditions east of Chrome Pit 1.	Figure 3-4.
	2.		2.	Evaluate possible effect of Chrome Pit 1 on groundwater.	
Fire Department	1.	Conduct a terrain conductivity survey.	1.	Determine the extent of a hydrocarbon plume, if any.	Figure 3-2.
Training Area 2	2.	Sample groundwater from 7 upper zone wells.	2.	Evaluate possible effect of FDTA 2 on groundwater.	
Fuel Saturation	1.		1.	Determine soil conditions and presence of fuel in subsurface.	Fígure 3-2.
Area 1	2.	Sample groundwater from 2 upper zong wells, 1 upper zone borehole, and 2 Paluxy wells.	2.	Evaluate effects of FSA 1 on groundwater.	
NARF	1.	Drill 4 boreholes in the upper zone.	1.	Determine shallow soil conditions and presence of RAM in soil.	Figure 3-7.
	2.	Sample groundwater from 3 upper zone wells.	2.	Evaluate possible effect of NARF on groundwater.	
	з.	Conduct review of NARF and White Settlement records for radionuclide analyses.	з.	Determine past operations and exact location of NARF.	

TABLE 3-4. (Continued)

Site		Investigation Activities	Purpose	Location
Jet Engine Test Stand	1. đ	1. Install 3 upper zone wells and drill 2 boreholes.	 Collect and analyze soil samples and evaluate groundwater flow and chemistry. 	Figure 3-7.
	й Х. 7		2. Evaluate possible effect of Jet Engine Test Stand on groundwater.	
	ສູ່ແ ເ	Sample surface water at Outfalls 1 and 5.	 Evaluate effect of discharge to surface water near facility. 	
Ambient Monitoring	1. ž	Sample groundwater from 11 upper zone wells and 8 Paluxy wells.	 Evaluate groundwater quality in areas away from waste disposal spill sites. 	Figure 3-9.
East Parking Lot	1. 4 2 2	Sample groundwater from 3 upper zone monitor wells and 2 Paluxy wells.	1. Evaluate groundwater quality at east portion of AF Plant 4.	Figure 3-10.
Fuel Storage Tank	1. § Ø	1. Sample groundwater from upper zone well HM-23.	1. Evaluate possible effect of the Fuel Storage Tank on groundwater.	Figure 3-5.
Lake Worth Monitor Well	1. I. W.	1. Install 1 Paluxy well near Lake Worth.	 Evaluate groundwater quality of Paluxy aquifer near recharge source, Lake Worth. 	Figure 3-9.

DEREN SERVE EXERCE EXERCE

za storica usersal necessar concernantes

a the second and a strategy of

RADIAN



3-25

10000

1.1.



ロビード・トート

3.2.2 Site 3, Landfill 3

Investigation activities at Landfill 3 included: performance of geophysical (EMP) surveying, installation of one monitor well (P-22U) completed in the upper member of the Paluxy Formation, and collection of groundwater samples from nine existing upper zone monitor wells and the newly installed Paluxy well. Figure 3-2 shows the location of Landfill 3.

<u>Terrain Conductivity Survey</u>: The purpose of the terrain conductivity survey conducted in November 1985 was to determine the presence and configuration of waste disposal areas. This was accomplished using electromagnetic profiling to measure electrical conductivities of the subsurface material. To perform the survey at Landfill 3, a grid was established, as described in Section 3.1.1. Figure 3-3 illustrates the area of the survey at Landfill 3, as well as the other survey locations at AF Plant 4. The area of the terrain conductivity survey grid is approximately 2100 by 250 feet. Electromagnetic profile readings were taken every 30 feet.

Monitor Well Installation: One monitor well (P-22U) was installed in the Landfill 3 area as part of this investigation. The well was completed in the upper member of the Paluxy Formation. Drilling and well installation activities took place from 18 to 21 March 1986. Soil samples and samples of the drilling fluids were collected as described in Section 3.1.3. The Paluxy well location was selected in conjunction with existing Paluxy monitor wells to further define groundwater flow direction on the eastern boundary of AF Plant 4 and to note possible contamination migrating onto the AF Plant 4 boundary.

<u>Groundwater Sampling</u>: One sample from each of 9 upper zone wells (HM-21, 26, 27, 34, 35, 36, 37, 38, and 39) was collected by Radian personnel during 31 January through 27 February 1986. The Paluxy well (P-22U) was sampled on 10 April 1986. The purpose of groundwater sampling and analysis was to confirm earlier investigations and to gain additional data on the nature and extent of contamination. Details of the sampling procedures are



3-27



outlined in Section 3.1.5.3. The groundwater samples were analyzed for VOC, BNA, and heavy metals. Results of the groundwater analyses are presented in Section 4.3.2.

3.2.3 Site 12, Chrome Pit 3

Investigation activities at Chrome Pit 3 included: collection of groundwater samples from eight existing upper zone monitor wells and one Paluxy monitor well. Figure 3-4 shows the location of Chrome Pit 3.

<u>Groundwater Sampling</u>: The purpose of groundwater sampling and analysis was to evaluate post-closure groundwater conditions at the site. One groundwater sample was collected and analyzed from each of wells HM- 13, 15, 16, 17, 30, 32, 41, 45, and P-2. Field sampling was performed by Radian personnel from 11 to 25 February 1986. Details of the sampling procedures are outlined in Section 3.1.5.3. The groundwater samples were analyzed for VOC, BNA, and heavy metals. Results of the groundwater analyses are presented in Section 4.3.3.

3.2.4 Site 17, Former Fuel Storage Site

Investigation activities at the Former Fuel Storage Site included: installation of one upper zone monitor well (HM-100), collection of upper zone groundwater samples from two existing monitor wells and the newly installed monitor well, and investigation of the existence of a fuel lens, if any, in these monitor wells. Figure 3-5 shows the location of the Former Fuel Storage Site.

<u>Monitor Well Installation</u>: One monitor well (HM-100) was installed southeast of the former fuel storage site and completed in the upper zone alluvium. Soil samples were collected using split-spoons as described in Section 3.1.3. The purpose of installing monitor well HM-100 was to improve the definition of groundwater flow in the upper zone and to note any contamination that may exist in the vicinity of Site 17. The flow direction of water in the



-55



upper zone is difficult to identify because of the variability of the strata in the upper zone and the resultant scattered occurrence of groundwater. The upper zone material at this site consists of either alluvium or the Goodland Formation. The Goodland Formation, a calcareous shale underlying most of AF Plant 4, acts as a barrier to flow.

<u>Groundwater Sampling</u>: One sample from each of two existing upper zone wells (HM-8 and HM-14) was collected during 13 to 19 February 1986, and from one new well (HM-100) on 11 April 1986. The purpose of groundwater sampling and analysis was to confirm earlier investigations and to gain additional data on the nature and extent of contamination. Details of the sampling procedures are outlined in Section 3.1.5.3. The groundwater samples were analyzed for hydrocarbon fuels. Results of the analyses are presented in Section 4.3.4.

3.2.5 <u>Site 2, Landfill 2</u>

Investigation activities at Landfill 2 included: performance of geophysical surveying, installation of a monitor well (P-21U) completed in the upper member of the Paluxy Formation, and collection of groundwater samples from seven existing monitor wells and the newly installed Paluxy monitor well. Figure 3-5 shows the location of Landfill 2.

Terrain Conductivity Survey: The purpose of the terrain conductivity survey was to determine the presence and configuration of waste disposal areas. This was accomplished by using electromagnetic profiling to measure electrical conductivities of the subsurface material. To perform the survey at Landfill 2 a grid was established, as described in Section 3.1.1. Figure 3-3 illustrates the area of the survey at Landfill 2, as well as the other EMP survey locations at AF Plant 4. The area of the EMP survey grid is approximately 1800 by 800 feet. Electromagnetic profile readings were taken every 30 feet. The survey was performed in November, 1985.



RADIAN

<u>Monitor Well Installation</u>: Cae new monitor well (P-21U) was installed in the Landfill 2 area. The well was completed in the upper member of the Paluxy Formation. Soil samples were taken as described in Section 3.1.3, along with samples of drilling fluids. The location of P-21U was selected to complement the existing Paluxy well network and determine the quality of groundwater downgradient of Landfill 2 and of other waste disposal areas.

<u>Groundwater Sampling</u>: One sample from each of the upper zone wells (HM-2, 22, 40, 42, 43, 44, and 46) was collected during the period 20-25 February 1986. The Paluxy well (P-21U) was sampled on 10 April 1986. The purpose of groundwater sampling and analysis was to confirm earlier investigations and to gain additional data on the nature and extent of contamination in the upper zone and the Paluxy aquifer. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for VOC, BNA, and heavy metals. Results of the analyses are presented in Section 4.3.5.

3.2.6. <u>Site 4, Landfill 4</u>

Investigation activities at Landfill 4 included: installation of two monitor wells (HM-101 and P-20M) and collection of groundwater samples from the two existing monitor wells and the two newly installed monitor wells (Figure 3-5).

Monitor Well Installation: Monitor well HM-101 was completed in the upper zone, and P-20M was completed in the middle member of the Paluxy Formation. Monitor well P-20M was intended to be installed in the upper member of the Paluxy Formation, but the upper member was dry so the well was installed to coincide with uppermost occurrence of water, this being the middle member of the Paluxy. Both monitor wells were installed south of Landfill 4 on the AF Plant 4 Radar Range. Soil samples collected from split spoons, as described in Section 3.1.3, and samples of drilling fluids were also taken. HM-101 is located along the fence line of the Radar Range, an area with few monitor wells. This location was chosen to more accurately define groundwater

RAPIAN

flow and the occurrence of contamination in the upper zone. The P-20M location was selected to note migration of contamination, if any, from east of the AF Plant 4 boundary, and also because it was upgradient of the Paluxy well installed in the Landfill 2 area.

<u>Groundwater Sampling</u>: One sample from each of the existing upper zone wells (HM-5, HM-9) was collected during 12 to 27 February 1986. Monitor well HM-101 was sampled on 18 April 1986 and P-20M was sampled on 9 April 1986. The purpose of groundwater sampling and analysis was to confirm earlier investigations and to gain additional data on the nature and extent of contamination in the upper zone and the Paluxy aquifer. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for VOC, BNA, and heavy metals. Results of the analyses are presented in Section 4.3.6.

3.2.7 Zone 1: Site 13, Die Pits; Site 11, Chrome Pit 2; Site 8, FDTA 5

Since Zone 1 contains three sites in close proximity to each other, the field activities were designed to account for their proximity. Field activities at Zone 1 included: drilling of a borehole (SB-5) in the vicinity of the Site 8 Fire Department Training Area (FDTA) 5 and collection of groundwater samples from seven existing upper zone monitor wells and from one existing Paluxy monitor well. Figure 3-4 shows the location of the borehole wells, and sites.

<u>Borehole</u>: Soil boring SB-5 was drilled in the area of FDTA 5 (Figure 3-4). Based on color, odor, and OVA readings, soil samples were collected from split spoons, as described in Section 3.1.3, and submitted for analysis of hazardous constituents. Originally, the purpose of the borehole was to allow for the installation of an upper zone monitor well; however, no water was noted in the borehole even after it had remained open for two days. Therefore, the borehole was grouted from the total depth to the land surface. <u>Groundwater Sampling</u>: One sample from each of the upper zone wells (HM-11, 12, 24, 25, 28, 60, and 77) and Paluxy well P-1 was collected during 6 to 19 February 1986. The purpose of groundwater sampling and analysis was to confirm earlier investigations and to gain additional data on the nature and extent of contamination in the upper zone and the Paluxy aquifer. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for VOC, BNA, and heavy metals. In addition, groundwater from HM-77 was analyzed for hydrocarbon fuels and oil and grease. Results of the analyses are presented in Section 4.3.7.

3.2.8 Site 15, Fuel Saturation Area 2

Investigation activities in Fuel Saturation Area (FSA) 2 included: drilling of three boreholes along the length of a buried fuel line, collection of a groundwater sample from an existing upper zone monitor well, and observation of the monitor well for the existence of a fuel lens. The location of FSA 2 is shown in Figure 3-6.

<u>Boreholes</u>: Three boreholes (SB-1, 2 and 3) were drilled just west of the fence along a buried fuel line in the upper zone. Locations for drilling were selected after verbal communication with Resource Engineering, Inc., the subcontractor to Intellus for Phase IV field work. Drilling was accomplished between 24 and 26 January 1986. These boreholes were drilled to determine if hydrocarbons existed in the soil and groundwater near the fuel line. Soil samples were collected from split spoons as described in Section 3.1.3. The boreholes were left open overnight to see if groundwater was present and if hydrocarbons existed on the surface of the water table. After the observations were completed, each borehole was grouted to the surface.

<u>Groundwater Sampling</u>: Groundwater samples were collected from HM-80 on 5 February 1986. Samples were collected and analyzed to determine if contamination was present in the groundwater at the buried fuel line. Details of the sampling procedures are given in Section 3.1.5.3. The samples were



1.1.1.1.1.1.1

A. B. W. J. W. D. V. A. C. D.

3-35



analyzed for VOC and hydrocarbon fuel. Results of the analyses are presented in Section 4.3.8.

3.2.9 Site 20, Wastewater Collection Basins

Investigation activities at the Wastewater Collection Basins (Figure 3-4) included: installation of an upper zone monitor well (HM-104), collection of groundwater samples from four existing monitor wells and the newly installed well, determination of the location and depth of the buried sanitary and industrial waste lines, as well as of the need for soil sampling along the waste lines (based on soil and groundwater analysis).

Monitor Well Installation: Monitor well HM-104 was installed between 22 and 23 January 1986. Soil samples were collected from split spoons, as described in Section 3.1.3. The monitor well was located southeast of the basins at the southwest corner of Building 188. This location was selected to gain information on the direction of groundwater flow and on contamination in the area.

<u>Groundwater Sampling</u>: One sample from each of the upper zone monitor wells HM-31, 47, 69, and 70 was taken during 6 and 21 February 1986. Monitor well HM-104 was sampled on 10 April 1986. The purpose of groundwater sampling and analysis was to confirm earlier investigations and to gain additional data on the nature and extent of contamination in the upper zone. The groundwater samples were analyzed for VOC, oil and grease, hydrocarbon fuel, and heavy metals. Results of the analyses are presented in Section 4.3.9.

3.2.10 Site 16, Fuel Saturation Area 3

Investigation activities in Fuel Saturation Area 3 (FSA 3) (Figure 3-7) included the collection of a groundwater sample from one existing well and observation of the existence of a fuel lens, if any, in the monitor well.



<u>Groundwater Sampling</u>: Sampling of upper zone monitor well, HM-78, was performed on 5 February 1986 to confirm earlier investigations and to gain additional data on the nature and extent of contamination in the upper zone. The groundwater sample was analyzed for VOC and hydrocarbon fuel. Results of the analysis are presented in Section 4.3.10.

3.2.11 Site 9, Fire Department Training Area 6

Investigation activities in Fire Department Training Area 6 (FDTA 6) (Figure 3-7) consisted of hand augering, the collection of six soil samples, and the collection of one groundwater sample from an existing Paluxy monitor well.

<u>Hand Augering</u>: The purpose of collecting soil samples by hand augering was to determine whether shallow soil contamination exists at this suspected fire training site. Field work occurred on 12 May 1986. Each hand auger site was within the suspected boundary of FDTA 6, as shown in Figure 3-8. The perimeter of FDTA 6 was shown to Radian personnel by the AF Plant 4 fire department and marked with flags. Four of the auger sites were placed along the perimeter on the northern half of FDTA 6 and the remaining two were placed along the center line of the area. Since the southern half of FDTA 6 had very little topsoil, it was not sampled. Soil samples were taken as described in Section 3.1.5.1. One soil sample was submitted for analysis for VOC, BNA, oil and grease, and hydrocarbon fuel. Results of the analyses are presented in Section 4.3.11.

<u>Groundwater Sampling</u>: Radian personnel collected a groundwater sample from P-3 on 4 February 1986. The purpose of this effort was to gain additional data on the groundwater quality conditions in the Paluxy aquifer at this site. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater sample was analyzed for VOC, BNA, oil and grease, and hydrocarbon fuel. Results of the chemical analysis are presented in Section 4.3.11.



1222.022

NELENCE

- 2222455

\$150050505050505050505050505



A'1..........

RADIAN

3.2.12 Site 6, Fire Department Training Area 3

Investigation activities at Fire Department Area 3 (FDTA 3) (Figure 3-5) included the installation of a groundwater monitor well HM-102 and collection of groundwater samples from monitor wells HM-33 and HM-102.

<u>Monitor Well Installation</u>: One new monitor well (HM-102) completed in the upper zone was installed west of FDTA 3 in the vicinity of Landfill 2. Soil samples were collected from split spoons as described in Section 3.1.3. The location for this monitor well was selected to improve coverage of groundwater flow monitoring points.

<u>Groundwater Sampling</u>: Groundwater samples were collected from HM-33 on 28 February 1986 and from HM-104 on 10 April 1986. The purpose of collecting and analyzing the groundwater was to determine the nature and extent of contamination at the fire training site. Details of the sampling procedures are given in Section 3.1.5.3. Groundwater samples were analyzed for VOC, oil and grease, and hydrocarbon fuel. Results of the analyses are presented in Section 4.3.12.

3.2.13 Site 7, Fire Department Training Area 4

Investigation activities at Fire Department Training Area 4 (FDTA 4) (Figure 3-7) included an attempt to determine the location of the old fire training area by conducting a soil-gas survey.

<u>Soil-Gas Survey</u>: Soil-gas samples were collected and analyzed, as described in Section 3.1.2, from a total of 28 points at FDTA 4. The target depth for soil-gas sampling was 5 to 6 feet below the ground surface at most of the probe locations. At eight of the sampling locations, probes were driven to the 8 to 12 foot depth interval to determine if volatile organic compounds (VOCs) were being biodegraded in the shallow soil gas. Since petroleum hydrocarbons are particularly susceptible to oxidation biodegradation in shallow soil where aerobic conditions prevail, sampling for petroleum



hydrocarbons is usually conducted at greater depths than is sampling for halocarbons.

Soil-gas samples were analyzed for methane, total petroleum hydrocarbons, carbon tetrachloride, 1,1,1-trichloroethane, trichloroethylene, and perchloroethylene. Total petroleum hydrocarbons were defined as C_4 to C_{10} aliphatic and alicyclic compounds and benzene, toluene and xylene isomers. These petroleum compounds are the major constituents of aviation fuels. Methane is produced as a by-product of subsurface biodegradation, and the other four compounds are the most common industrial solvents.

After 28 soil-gas samples were taken and no hydrocarbons were noted, Radian personnel contacted the AF Plant 4 fire department and requested a more precise location for the fire training area described in the Phase I report. Three fire department personnel, each with from 20 to 30 years experience at AF Plant 4, came to the site and stated that no FDTA had ever existed at this location; therefore, activities at this site were terminated.

3.2.14 Site 18, Solvent Lines

R

Investigation activities along the Solvent Lines included the installation of a monitor well (HM-106) and collection of groundwater samples from four existing upper zone monitor wells and from HM-106. Figure 3-6 shows the location of this site.

<u>Monitor Well Installation</u>: Monitor well HM-106 was installed along the Solvent Lines and completed in the upper zone (Figure 3-6). Drilling and well installation took place on 26 January 1986. The purpose of the additional upper zone monitor well was to complement the existing well network at the Solvent Lines by providing additional information on the direction of groundwater flow and groundwater quality in the upper zone. Soil samples were collected from split spoons, as described in Section 3.1.3. The new monitor



well was installed at the northwest bend along the solvent line, approximately 400 feet east of Building 119.

<u>Groundwater Sampling</u>: One groundwater sample was collected from each of the four existing wells (HM-72, 73, 74, and 74) from 5 to 26 February 1986. Monitor well HM-106 was dry and could not be sampled. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for oil and grease, xylene, and methyl ethyl ketone. Results of the chemical analyses are presented in Section 4.3.14.

3.2.15 Site 10, Chrome Pit 1

Investigation activities at Chrome Pit 1 included the installation of a monitor well (HM-103) and collection of groundwater samples from an existing monitor well (HM-48) and HM-103.

Monitor Well Installation: One new monitor well (HM-103) was installed east of Chrome Pit 1 and completed in the upper zone (Figure 3-4). Well installation activities took place on 21 January 1986. Soil samples were taken as described in Section 3.1.3. The monitor well is located east of the chrome pit on the northeast corner of Building 188. This location was selected to gather data on groundwater flow in the upper zone in this area and to determine if contamination exists there. The monitor well was not installed directly next to Chrome Pit 1 because the site is located beneath the floor of Building 181.

<u>Groundwater Sampling</u>: Groundwater sampling was performed by Radian personnel at HM-48 on 14 February 1986 and at HM-103 on 10 April 1986. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for VOC and chromium. Results of the analyses are presented in Section 4.3.15.

RAPIAN

3.2.16 Site 5, Fire Department Training Area 2

Investigation activities at Fire Department Training Area (FDTA) 2 (Figure 3-3) included performance of a terrain conductivity survey and collection of groundwater samples from seven existing upper zone monitor wells.

<u>Terrain Conductivity Survey</u>: The purpose of the terrain conductivity (EMP) survey performed in November, 1985 was to determine the presence and configuration of the fire training area and to gain information on the northern limits of Landfill 1, located just south of FDTA 2. The survey used electromagnetic profiling to measure electrical conductivities of the subsurface material. To perform the survey, a grid was established, as described in Section 3.1.1. Figure 3-3 illustrates the area of the EMP survey at FDTA 2, and the other survey locations at AF Plant 4. The area of the EMP survey grid is approximately 800 feet by 250 feet. Electromagnetic profile readings were taken every 30 feet.

Groundwater Sampling: Radian personnel collected one groundwater sample from each of monitor wells HM-19, 49, 51, 65, 66, and 76 during 11 February to 1 March 1986. The purpose of the groundwater sampling and analysis program was to gain additional information on the nature and extent of contamination at FDTA 2. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for VOC, oil and grease, and hydrocarbon fuel. Results of the analyses are presented in Section 4.3.16.

3.2.17 Site 14, Fuel Saturation Area 1

Investigation activities at Fuel Saturation Area (FSA) 1 (Figure 3-2) included drilling one borehole (SB-4); collecting groundwater samples from the borehole, from two existing upper zone monitor wells, and from two existing Paluxy monitor wells; and observing the monitor wells for a fuel lens, if any, on the surface of the water table.

<u>Borehole</u>: One borehole (SB-4) was drilled into the upper zone just west of FSA 1 on the east side of Building 14 (Figure 3-2). Soil samples were collected from split spoons, as described in Section 3.1.3. The borehole was drilled to the water table and the liquid was sampled. The sample contained only fuel; therefore the complete thickness of the fuel lens on the water table could not be determined. After sampling, the borehole was grouted to the surface and the sample was analyzed for hydrocarbon fuels. Results of the analysis are presented in Section 4.3.17.

<u>Groundwater Sampling</u>: One groundwater sample was collected from each of monitor wells HM-53, HM-55, P-6M, and P-6U during 4 to 10 February 1986. Samples were collected and analyzed to determine the presence of contamination in the upper zone and to gain additional information on the groundwater quality in the Paluxy aquifer. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for hydrocarbon fuel. Results are presented in Section 4.3.17.

3.2.18 Nuclear Aerospace Research Facility (NARF) Area

Investigation activities at the Nuclear Aerospace Research Facility (NARF) (Figure 3-7) included drilling four boreholes (SB-6, 7, 8 and 9), collecting soil samples, collecting groundwater samples from three existing monitor wells, researching past analytical data from wells at Plant 4 and White Settlement for radionuclide results during the NARF operations period, and ascertaining the exact locations of past operations. The purpose of these activities was to ascertain the presence or absence of radioactive material (RAM) on the surface and subsurface of the former reactor locations, as well as at the old aircraft decontamination area.

A representative of White Settlement stated that no information on radionuclides is available for White Settlement. AF Plant 4 personnel stated that the information on the wells at AF Plant 4 during NARF operations was archived and that separate action would be required to retrieve it.

RAPIAN

Boreholes: A total of four boreholes (SB-6, 7, 8 and 11) were drilled within the NARF area in the upper zone (Figure 3-7). Initially, only three boreholes were to be drilled in the NARF area, one as a monitor well. However, no water was noted in any of the three boreholes; therefore, a fourth borehole was attempted. The fourth boring was also dry and no more attempts were made. Soil samples were collected from split spoons, as described in Section 3.1.3, except that samples were collected at two foot intervals. Soil samples were submitted for analysis of alpha, beta, and gamma radionuclides. Each borehole was grouted to the surface after sampling was completed. Results of the analyses are presented in Section 4.3.18.

<u>Groundwater Sampling</u>: Radian personnel performed groundwater sampling at HM-83, 84, and 85 on 21 August 1986. Details of the sampling procedures are given in Section 3.1.5.3. Groundwater samples were analyzed for VOC, BNA, heavy metals, oil and grease, RAM, and hydrocarbon fuels. Results of the analyses are presented in Section 4.3.20.

3.2.19 Jet Engine Test Stand, Bldg. 21

Investigation activities at the Jet Engine Test Stand (Figure 3-7) included installation of three monitor wells (HM-105, 107, and 108), drilling of two boreholes (SB-9 and 10), collection of groundwater samples from the three newly installed monitor wells, and collection of surface water samples from Outfall 5 and Outfall 001.

Monitor Well Installation: Three monitor wells (HM-105, 107 and 108) were installed around the Jet Engine Test Stand, two west and one south of the facility. The monitor wells were all completed in the upper zone. Drilling and well completion were performed on 11 August 1986. Soil samples were collected from split spoons as described in Section 3.1.3. The monitor well locations were chosen because of the lack of monitor wells in the immediate vicinity of Building 21. Their purpose was to gain information on groundwater flow in the upper zone and data on the groundwater chemistry downgradient of the facility. Originally, the monitor wells were to be installed at



⋒[⋬]▙⋶⋻₩⋳[⋎]⋇∜⋻⋵⋻⋵⋭⋎⋺₩⋻⋎⋺⋐⋳⋎⋺⋐⋧⋎⋺⋐⋺⋎⋺⋐⋳⋎⋺⋎∊⋶⋺⋪⋺⋎⋳⋎⋺⋎⋳⋎⋺⋳⋳⋎⋳⋳∊⋎⋺⋳⋼⋎⋵⋳∊⋎⋳⋳⋳⋎⋳⋺⋐⋎⋳⋺⋑⋎⋳⋺⋓⋎⋒

the north, south, and west sides of the facility; however, the boring on the north side of the facility was dry. This is probably because the Goodland Formation, generally a dry or low-yield formation, is on the north side of the facility.

Boreholes: Two boreholes (SB-9 and -10) were drilled at the Jet Engine Test Stand, one north and one south of the facility. Soil samples were collected from split spoons, as described in Section 3.1.3. Drilling and sampling was conducted 22 July 1986. The boring on the north side of the facility was dry while the boring on the south side encountered groundwater and liquid hydrocarbon. The hydrocarbons may be related to abandoned underground fuel storage tanks located approximately 6 feet north of the borehole. Each borehole was grouted to the surface after sampling was complete.

<u>Groundwater Sampling</u>: Radian personnel performed groundwater sampling from 19 to 20 August 1986. Details of the sampling procedures are given in Section 3.1.5.3. Groundwater samples were analyzed for oil and grease and hydrocarbon fuel. Results of the analyses are presented in Section 4.3.19.

Surface Water Sampling: Radian personnel performed field sampling on 20 August 1986. Outfalls 1 and 5 were sampled every hour over an eight hour period. A total of nine samples was collected, consisting of the eight samples and a composite of the eight. Details of the sampling procedures are outlined in Section 3.1.5.2. Surface water samples were analyzed for oil and grease and hydrocarbon fuel. Results of the surface water analyses are presented in Section 4.3.19.

3.3 Additional Study Topics

In addition to the investigations related to the waste disposal sites presented above, four additional study topics were identified for field investigations. Each is discussed below:



3.3.1 <u>Ambient Monitoring</u>

Investigation activities for Ambient Monitoring involved the collection of groundwater samples from eleven existing upper zone monitor wells and eight existing Paluxy monitor wells (Figure 3-9). The purpose of this activity was to gain continuing information on groundwater quality in the upper zone and Paluxy aquifer in areas not directly associated with waste disposal sites.

Radian personnel performed groundwater sampling from 31 January through 26 February 1986. One sample was collected from each of HM-29, 52, 54, 56, 57, 58, 59, 61, 64, 79, and 81; and from P-5U, 5M, 9U, 9M, 10U, 10M, 11U, and 11M. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for VOC, BNA, oil and grease, hydrocarbon fuel, and heavy metals. Chromium was also specified for analysis in the delivery order, but this parameter was already included in the list of heavy metals. Results of the analyses are presented in Section 4.3.20.

3.3.2 East Parking Lot

Investigation activities for the East Parking Lot involved the collection of groundwater samples from three existing upper zone monitor wells and two Paluxy monitor wells (Figure 3-10). The purpose of this activity was to gain continuing information on the quality of groundwater in the upper zone and Paluxy aquifer in areas not directly associated with waste disposal sites.

Radian personnel performed groundwater sampling from 3 to 28 February 1986. One sample was collected from each of HM-68, 71, and 82; and P-8U and 8M. Details of the sampling procedures are given in Section 3.1.5.3. Groundwater samples were analyzed for VOC, BNA, oil and grease, hydrocarbon fuel, and heavy metals (chromium was also specified for analysis in the delivery order, but this parameter was already included in the parameter list of heavy metals). Results of the analyses are presented in Section 4.3.21.



3-48

S-345-5-346



3-49

2009 AUDIO BUDDU ANDARY WSSSSS PREDUCTION

3.3.3 Fuel Storage Tank

Investigation activities for the Fuel Storage Tank consisted of collecting a groundwater sample from upper zone monitor well HM-23 (Figure 3-5). The purpose of this activity was to determine groundwater quality in the upper zone at the Fuel Storage Tank.

Radian personnel performed groundwater sampling at HM-23 on 18 February 1986. Details of the sampling procedures are given in Section 3.1.5.3. The groundwater samples were analyzed for oil and grease and hydrocarbon fuel. Results of the analyses are presented in Section 4.3.22.

3.3.4 Lake Worth Monitor Well

Investigation activities at this area included installing Paluxy monitor well (P-23U) and collecting two groundwater samples, one month apart.

Monitor Well Installation: A Paluxy monitor well (P-23U) was installed west of Meandering Creek Road, and west of Building 21 in the upper member of the Paluxy Formation (Figure 3-9). Soil samples and samples from the drilling fluids were taken as described in Section 3.1.3. This location was selected to determine the groundwater quality in the Paluxy aquifer directly downgradient of a major recharge source, Lake Worth.

<u>Groundwater Sampling</u>: Radian personnel performed groundwater sampling at P-23U on 18 April and 21 August 1986. Details of the sampling procedures are given in Section 3.1.5.3. Groundwater samples were analyzed for VOC, BNA, and heavy metals. Results of the analyses are presented in Section 4.3.22.

3.4 Hazardous Constituents in Soil and Drilling Fluids

Selected samples of soils and drilling fluids from drilling and well installation operations were analyzed for a variety of parameters to determine



if they should be designated hazardous materials. Selection of samples for analysis was based on the presence of discoloration, odor or positive OVA readings. Soils were analyzed for metals (EP Toxicity), chromium, and ignitability. Drilling fluids used during the installation of wells P-20M, P-21U, P-22U and P-23U were also analyzed for volatile organic compounds. Results of these analyses are given in Table 3-5. Complete analytical reports appear in Appendix A.

Table 3-6 shows the maximum concentrations of contaminants for characterization of EP Toxicity. Mercury, cadmium, and selenium were detected at concentrations above these maximum levels in soils from P-20M. Furthermore, analyses for volatile organics showed high levels of trichloroethylene in the drilling fluid used during the installation of P-22U. Therefore, on the basis of these tests for hazardous constituents, materials from P-20M and P-22U should be considered hazardous materials.

In addition, it was previously recommended (based on further analytical work reported in Section 4), that materials from HM-103, SB-2, and SB-4 also be considered hazardous material. Analyses detected high levels of chromium and TCE in soils from HM-103 and elevated hydrocarbons in soils from SB-2 and SB-4. SOIL SAMPLES COLLECTED FOR HAZARDOUS CONSTITUTENTS, AF PLANT 4, TEXAS **TABLF 3-5.**

					4					
		SB-1-B	SB-2-D	SB-3-C	SB-4-C	SB-5	SB-9-B	HM-103-C	HM-104-A	HM-105
	Method	FSA2	FSA2	FSA2	FSAI	Zone 1	Bldg.	Chrome Dit 1	WWCB	Bldg.
Date Sampled	Limit	1/24/86	1/24/86	1/26/86	1/26/86	1/23/86	7/22/86	1/21/86	1/22/86	8/11/86
Analytical Parameters ^a										
RCRA Metals (mg/L)										
	0.002	0.027	0.017	0.017	0.018	0.019	0.018	0.023	0.015	0.019
	0.002-0.005	9	Q	Q	Q	Ð	Q	Ð	Ð	Q
Ва	0.009	0.046	0.15	0.30	0.25	0.23	1.5	0.20	0.12	0.64
Cd	0.002	23		0.005*	0.003*	0.005*	93		0.007*	0.005*
Gr	0.005	0.02*	0.02*	0,02*	0.01*	*600°0	0.027	0,02*	0.01*	0.02*
	0.0002	2	0.0002*	93			0.08	93	0*0004*	0,0003+
Pb Se	0.001-0.002 0.002-0.004	0,002# ND@	600.0	90.0 M	0.010 UN	0,003* ND	0.015 ND(.02)	9000 0	900 0000	0.008 ND(.02)@
Ignitability		No	No	No	No	No	No	No	No	No
Volatile Organics (mg/kg)	~	NA	M	M	M	VN	VN	W	M	M
Method 8240 Concentration Factor										
Methylene Chloride trans-1,2-Dichloro- ethylene	0.0028 0.0016									
Trichlorosthulene	0.0019									
Toluene	0.006									
1.1.1-Trichloroethane	0									
1.2-Dichloroethane										
Ethvlbenzene	0.0072									
Tetrachloroethylene	0.0041									
Method 601 (Halogenated Hydrocarbonm) Concentration Factor	d Hydrocarbone	a)								
Mathulana Chlorida	1/211 26 0									
netuytene ontortue Chloroform	0.05 ue/L									
trans-1.2	0.10 ug/L									
Dichloroethylene	b									
Trichloroethylene	0.12 ug/L									
Method 602 (Aromatic Compounds) Concentration Factor	ompounds)									
Ē										
Toluene	0.2 ug/L									

EXECUTION STATES - EXECUTION -

(Continued)

RADIAN

Continued
9
3-5.
TABLF

				Monito	Ы	Soil Boring ID	20.4		B-73
	M - 44 - M	HM-100-B	HM-10/	801-108 8140	1 54	1.82	5-22 1.F3	1.P3	Lake Worth
	Detection	SOVENT Tines	-3nra	21 21	r 1	4	Water	Mud	Mud
Date Sampled	Limit	1/26/86	8/11/86	8/11/86	2/28/86	2/28/86	3/20/86	3/20/86	3/20/86
Analytical Parameters ^a									
RCRA Metals (mo/l.)									
AR	0.002	0.005*	0.017	0.004*	0.039	0.014	M	*600*0	0.0130
	0.002-0.005	2	9	0.003*	1.6	ND(.06)	N	ND(.06)	0.06+8
	600°0	0.031	0.95	0.16	18	0.72	M	0.099	0.15
Cd	0.002	0,004*	*600*0	Ð	3.2	0.077	VN	0.003#	0.005+0
Cr	0.005	Ð	0,02#	0.005*	0.031	0.02*	M	0.02*	0.02+8
Hg	0.0002	Ð	0.0013	0 ,0006*	3.7	0.10	M	£	0.0048
	0.001-0.002 0.002-0.004	0.007 MD	0.35 ND	0.064 ND	0.18 2.1	ND(.08) ND(.08)	¥ ¥	ND(.08) ND(.08)	ND(.08)0 ND(.08)0
Ignitability		No	No	No	No	No.	NA	9N N	No
Volatile Organics (mg/kg)		NA	M	W					
Method 8240									
Concentration Factor					1	1		1	1
Mathulana (h)aride	0.0028				0.005BL	0.003BL		0.069BL	Ð
trang-1,2-Dichloro-	0.0016				Q	QN		0.0054	
ethylene					!	1		с .	ŝ
Trichloroethylene	0.0019				29	2 9		2.1	2 9
Toluene					2 9			0 0053	5 5
1,1.Trichloroethane					2 5	2 5		0.019	9 5
1.2-Dichloroethane	0.0028				2 5	5		0.006	
Ethylbenzene Tetrachloroethylene	0.0041				22	2		0.0096	2
Method 601 (Halogenated Hydrocarbons) Concentration Factor	Hydrocarbons								
Mothulana Chlorida	0 25 110/1						0.39 ug/L		
Ghloroform Chloroform	0.05 ue/L						3.5 ug/L		
trans-1,2,-	0.10 ug/L						1.2 ug/L		
Dichloroethylene Trichloroethylene	0.12 ug/L						4.1 ug/L		
Method 602 (Aromatic Compounds) Concentration Factor	(spunds)								
E STATE	1/511 0 0						0.69 ue/L		
auantor	0.2 ug/ L								

BL = Detected in reagent blank; background subtraction not performed. BL = Detected in reagent blank; background subtraction not performed. a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4. ND = Not detected, values in parentheses are actual sample detection limits.

1.0.0

۰,



TABLE 3-6. MAXIMUM CONCENTRATION OF CONTAMINANTS FOR CHARACTERIZATION OF EP TOXICITY

Contaminant ¹	Maximum Concentration (mg/L)
 Arsenic	5.0
Barium	100.0
Cadmium	1.0
Chromium	5.0
Lead	5.0
Mercury	0.2
Selenium	1.0
Silver	5.0

¹Reference: 40 CFR 261.24.
RADIAN

4.0 DISCUSSION OF RESULTS AND SIGNIFICANCE OF FINDINGS

In this section, the Phase II Stage 1 hydrogeologic observations and chemical analyses are discussed on a site-by-site basis. Hydrogeologic interpretations are drawn on the basis of observed variations in groundwater flow and quality. Analytical chemistry data are discussed within the context of available regulatory standards and criteria. Two introductory sections deal with general considerations (the plant-wide geologic and hydrogeologic context) and analytical chemistry considerations (quality confirmation of organic chemical data and available federal regulatory standards and guidelines). The discussion of results and significance of findings for each site appear in separate subsections.

4.1 Hydrogeologic Considerations

The IRP Phase II Stage 1 investigation conducted at AF Plant 4 has been one of several efforts to define the nature and extent of contamination at the facility. Since previous hydrogeologic data (e.g., stratigraphy. groundwater occurrence and flow, water quality) have been gathered and interpreted, this investigation has relied heavily on the available data, both to plan the scope of work and to report and interpret the present information. Specifically, the geologic setting and the occurrence and character of the groundwater have been previously described so that the scope of the investigation reported here has been to fill in gaps in the data and orient the investigation to areas not previously studied. The following subsections describe both the use and interpretation of existing geologic data (4.1.1) and hydrogeologic data (4.1.2).

4.1.1 Geologic Data

The field program was planned and conducted to use the extensive geologic information (presented in Section 2.3) previously developed at AF Plant 4. This subsection describes the geologic data gathered in the Phase II

Stage 1 program that confirm or extend the existing geologic data at AF Plant 4.

The knowledge of geologic conditions defined by previous studies are summarized in Section 2.3. Generally, the results of the IRP Phase II Stage 1 investigation confirm previous interpretations. Since much of the field program involved the strategic placement of additional wells and boreholes to complement existing site monitor networks, no effort was needed to define basic geologic conditions at AF Plant 4. The findings and interpretations of site-specific field data gathering programs appear later in the appropriate subsections. Most of the new geologic information from the Phase II Stage 1 program came from the installation of 4 wells drilled into the Paluxy Formation along the western margin of AF Plant 4.

Figure 4.1.1-1 shows a geologic cross section of the western margin of AF Plant 4. The section is constructed mostly from the geologic logs of the four Phase II Stage 1 Paluxy wells. The location of this cross-section is shown in Figure 2-5. As with the previously developed cross section (Figure 2-6), the new section consists of alluvium at the surface and sedimentary bedrock at depth. The section confirms the undulating nature of the alluviumbedrock contact, reflecting the erosion of bedrock and the subsequent deposition of terrace and fluvial deposits. At P-20M, drilling revealed that the upper member of the Paluxy is dry. The dry upper member at this location may reflect the greater distance from Lake Worth, a recharge source to the Paluxy aquifer. Elsewhere, the upper member of the Paluxy is only partially saturated, whereas the middle member is confined. The additional drilling conducted for the Phase II Stage 1 investigation has shown that the existing geologic interpretations are sound, and that only limited and site-specific drilling or geologic exploration is warranted for the future.

4.1.2 Hydrogeologic Data

Hydrogeologic data gathered for this investigation consisted of water-level readings at existing and new upper zone and Paluxy monitor wells.



Second Second

A series of potentiometric surface maps have been prepared as a result of the water-level measurement program. The following paragraphs discuss the observations of water levels in the upper zone for AF Plant 4, the occurrence of groundwater in the Paluxy aquifer, and groundwater levels in each of the areas of AF Plant 4.

AF Plant 4 Potentiometric Surface Map (Upper Zone)

Figure 4.1.2-1 is similar to the earlier interpretations of Hargis & Associates (1985) with regard to basic inferred groundwater flow patterns. The occurrence and flow of groundwater can be seen to be related closely to the distribution and thickness of alluvial material, the bedrock topography, and surface elevations. For example, the effect of a relatively thick section of coarse-grained (i.e., permeable) alluvial material can be seen in the Radar Range area, which shows widely-spaced potentiometric contours. In other areas, such as north of Landfill 1, the effect of a steeply sloping bedrock surface with thin alluvial cover is shown by closely-spaced contour lines. In general, potentiometric surface elevations are highest (greater than 640 feet MSL) under the central part of AF Plant 4, corresponding to bedrock high areas under the Assembly and Parts Building complexes. A potentiometric high also is evident in the extreme southwestern part of the site where the Goodland Limestone and Walnut Formations crop out near Landfill 4. From those high areas, potentiometric contours decrease either toward Meandering Creek west of the site or toward Grants Lane east of the site. The lowest potentiometric contours are at elevations of 620 or 615 MSL. Inferred directions of groundwater flow are indicated on the figure.

Subtle differences between the current map and earlier versions can be found in the identity of areas without upper zone groundwater. These differences may be attributed to the combination of new data available from the Phase II Stage 1 investigation and to re-interpretation of existing geologic logs from older HM-series wells. The hatchured areas of the potentiometric surface maps should be carefully evaluated in light of the modified



4-5

Real Control of Contro

geologic map presented in Figure 2.5. Specifically, the hatchured areas signify that no upper zone groundwater is present, either because the upper zone material is present but dry or because upper zone material is not present (even though limited groundwater may occur in either the Goodland or Walnut Formations). Thus, the uppermost occurrence of groundwater at any given location may not correlate with adjacent occurrences of groundwater. This approach to the presentation of groundwater data is essentially the same as that of Hargis & Associates (1985).

AF Plant 4 Potentiometric Surface Map (Paluxy Aquifer)

Potentiometric surface maps of both the upper and middle members of the Paluxy aquifer have been prepared from the Phase II Stage 1 data. As with the upper zone maps, recent Paluxy data support earlier Hargis & Associates (1985) interpretations of the potentiometric surface and groundwater flow in the Paluxy. The additional Phase II Stage 1 wells have extended the areal coverage of these maps to the southwest portion of AF Plant 4. However, only data from wells screened in either the upper or middle Paluxy have been contoured; data from wells screened in more than one member of the Paluxy were gathered but not contoured.

Figure 4.1.2-2 illustrates the potentiometric contours and inferred direction of groundwater flow in the upper member of the Paluxy. The upper Paluxy is unconfined at AF Plant 4. Results of drilling in the southwestern corner of the AF Plant 4 property have shown that the upper Paluxy is dry. In general, the highest elevations (590 foot contour line) are to the north and, near Lake Worth, decreasing more than 30 feet to the south and east. The estimated hydraulic gradient in the upper Paluxy at the western portion of AF Plant 4 is approximately 0.03. A potentiometric surface high is evident at the East Parking lot, where the maximum elevation of greater than 570 feet, indicating a limited reversal in the regional southeast direction of groundwater flow. Thus at the East Parking Lot the hydraulic gradient is approximately 0.01 to the west and south. The extent or cause of this



AND YOUN

CONT

4-7

RADIAN

potentiometric surface high is not presently understood, although earlier investigations (Hargis & Associates, 1985) suggested that the Goodland/Walnut aquitard may be locally absent, allowing direct hydraulic communication between the upper Paluxy and upper zone materials. This area was the subject of a field investigation by the U.S. Army Corps of Engineers at the time this report was written.

The water levels in P-10U and P-23U in the northern and western part of the study area are nearly equivalent in elevation to that of the spillway in Lake Worth (594 feet MSL). Thus, these water level data reflect the recharge from Lake Worth, which is directly underlain by sediments of the upper Paluxy. This conclusion was noted in previous investigations (Hargis & Associates, 1985); however, data are so sparse in the portion of AF Plant 4 bordering Lake Worth, that statements regarding recharge of the upper Paluxy from Lake Worth must be viewed as preliminary.

Figure 4.1.2-3 is a potentiometric surface map of the middle Paluxy aquifer. Although groundwater level data are limited to the southern and western portions of AF Plant 4, the pattern of the potentiometric surface contour lines is similar to that of the upper Paluxy, revealing a southeasterly direction of groundwater flow. Contour lines in the middle Paluxy are nearly uniformly spaced. Comparison of water level elevations in the upper and middle Paluxy shows a steeper hydraulic gradient in the upper Paluxy. The hydraulic gradient in the middle Paluxy is approximately 0.004 toward the southeast. Water levels in the middle Paluxy are lower than the levels in the upper Paluxy in the western portion of the site, indicating a downward vertical hydraulic gradient from the upper Paluxy to the lower Paluxy in this region.

Upper Zone Potentiometric Surface Maps

Descriptions in the following paragraphs of upper zone potentiometric surface maps prepared for AF Plant 4, are based primarily on



i Norici

Υ.

rapian

the results of Phase II Stage 1 monitoring. Separate maps are presented for each area of the plant identified in Figure 3-1 and for the East Parking Lot Area.

Area 1 Upper Zone Potentiometric Surface Map

Water levels in the upper zone of Area 1, which covers the northern part of the site, were measured in August, 1986. Data are available from two widely separated areas: the Jet Engine Test Stand and Building 197 near the NARF area. Water level data have been contoured, as shown in Figure 4.1.2-4. The potentiometric contours suggest a groundwater divide resulting from a low north-south ridge separating the two areas. Groundwater flows west in the vicinity of the Jet Engine Test Stand and to the east and north at Building 197.

Little information is available to make definitive statements regarding the groundwater flow in this area, but based on limited evidence at from the Jet Engine Test Stand Building, the hydraulic gradient in the upper zone is approximately 0.016. Long-term predictions or statements regarding groundwater flow in this area would be speculative, since the upper zone is thin to absent across the area.

Area 2 Upper Zone Potentiometric Surface Map

The upper zone water level data shown in Figure 4.1.2-5 are based on measurements made in February 1986. These data reflect the complex subsurface conditions in Area 2. Review of available geologic logs shows that a large area of Fuel Saturation Area 2, extending southeast to the Solvent Lines and under the Assembly Building does not have alluvial materials of the upper zone (see hatchured area in Figure 4.1.2-5). Groundwater occurs in several wells completed in the consolidated bedrock, but these data have not been contoured with the upper zone water levels because of probable differences in flow phenomena and hydraulic gradients. At one location, HM-106, groundwater was



No. Contraction

PERSONAL KUNDER

www

100000000





not present at the time of sampling. Where groundwater occurs in the upper zone, the elevation of the potentiometric surface generally decreases to the northwest, toward Lake Worth. At the Solvent Lines, the hydraulic gradient could be as high as 0.04 to the northwest; other areas have sparse control but probably have much lower gradients as well (based on projecting the hydraulic gradient calculated at the Solvent Lines to the northwest to Fuel Saturation Area 3).

Area 3 Upper Zone Potentiometric Surface Map

Figure 4.1.2-6 is a potentiometric surface contour map of the upper zone at Area 3. Most of the area is underlain by alluvium and fill of the upper zone. Groundwater elevations are highest in the southeast and northeast portions of the area, with a large east-central area of relatively constant groundwater elevations loosely defined by the 630 foot contour line. Closely spaced coutours have been drawn in the Landfill 3 area, corresponding to steeply sloping bedrock near Meandering Road Creek and probably buried bedrock channels near HM-38 and HM-19. The upper zone is absent in a few limited areas near FDTA 2, and in a larger area northeast of Building 15. Despite obvious variations in local groundwater flow patterns, most upper zone groundwater under Area 3 flows to the west and probably discharges to Meandering Road Creek. The Creek Seep at the southwest tip of Landfill 3 is an example of the surface discharge of groundwater in the area. Over much of the area, particularly in the vicinity of Landfill 1, hydraulic gradients are low, approximately 0.008. Much steeper hydraulic gradients are evident at HM-38 near the St-5 outfall and near FDTA 2.

Area 4 Upper Zone Potentiometric Surface Map

Water levels measured in February, 1986, at Area 4 illustrate a complex groundwater flow pattern controlled by bedrock topography (Figure 4.1.2-7). Much of the southern portion of the area is bedrock, providing a sharp groundwater flow boundary. For example, bedrock is either dry (SB-5 at FDTA 5 contained no water) or contains groundwater under different



4-14

ł.

የታለጉ እር የርጉ እና እ



留いりというで

. . .

Rapian

potentiometric conditions (comparison of water levels between HM-11 and HM-47). The western portion of the area is part of the Radar Range. Here groundwater elevations are highest at the Parts Plant and along Clifford Avenue. Thus groundwater elevations decrease to the west and north in the direction of groundwater flow. Toward the west, in the Radar Range, alluvial material is thick and potentiometric contour lines are widely spaced.

Groundwater elevations in the vicinity of FDTA 5 and Chrome Pit 1 begin to decrease to the east and south. Hargis & Associates (1985) reported the existence of a bedrock channel extending from this area eastward under the Warehouse Building and toward the East Parking Lot. Groundwater contour lines in the south part of the Assembly Building and Parts Plant appear to outline this channel. particularly the 625-foot contour line. Data for this area are sparse owing to the inaccessibility caused by the buildings. Groundwater flow in the East Parking Lot ultimately flows past Grants Lane to the east.

Area 5 Upper Zone Potentiometric Surface Map

Figure 4.1.2-8 illustrates the potentiometric surface of the upper zone in Area 5 during February 1986. Groundwater occurs at all locations in the area, but the upper zone occurs in the central and eastern portion of the area and along Meandering Road Creek. The Goodland Limestone and Walnut Formations crop out at the surface at HM-43, northward along Bomber Road, and at Landfill 4. East of this area alluvium of the upper zone reaches a thickness of nearly 50 feet.

Potentiometric contour lines at the Radar Range are widely spaced. Over most of the Radar Range the inferred direction of groundwater flow list the northeast and north. However, precise groundwater flow directions are or certain locally in the center of the Radar Range. The hydraulic gradient with the central part of the area is approximately 0.002 - 0.003 toward the retro Near FDTA 3, contours are closely spaced, illustrating a mode story hydraulic gradient of approximately 0.05.







4-17

RADIAN

East Parking Lot Upper Zone Potentiometric Surface Map

Water level data collected in February 1986, for the East Parking Lot are shown in Figure 4.1.2-9. The southern portion of the parking lot and the area around the Warehouse Building were covered in the Area 4 discussion above. The potentiometric surface contour map shows decreasing elevations to the east, ranging from slightly greater than 630 feet MSL to less than 615 feet MSL along Grants Lane. Thus groundwater in the upper zone flows from the west to the east under a hydraulic gradient of approximately 0.002.

As discussed in earlier paragraphs, the upper zone and upper member of the Paluxy aquifer may be in contact where the Goodland/Walnut aquitards are believed to be absent. Water level data in the upper zone do not reveal the location or describe the characteristics of this phenomenon. The Corps of Engineers was engaged in a drilling and sampling program in the East Parking Lot as this report was being prepared.

4.2 Analytical Chemical Considerations

4.2.1 Regulatory and Human Health Criteria

To determine possible water quality effects on the local groundwater systems, concentrations of organic and inorganic compounds detected in groundwater samples were compared to various water criteria. These criteria, from federal drinking water regulations, standards, and guidelines, include Maximum Contaminant Levels (MCLs) and proposed Recommended Maximum Contaminant Levels (RMCLs) above zero (0), established by the EPA as part of national drinking water regulations. The RMCLs are non-enforceable health goals, set with an adequate margin of safety at levels that would result in no known or anticipated adverse health effects. The MCLs are enforceable standards set at levels as close to the RMCLs as feasible.



Ĺ

POSSESS PURSEAL PULSESS PURSESS

2222222 Distances

2000000

4-19



Table 4.2.1-1 lists inorganic parameters detected and the corresponding federal primary drinking water standard (MCL). Table 4.2.1-2 lists EPA proposed MCLs, proposed RMCLs, and other criteria available for many of the organic compounds detected in groundwater analyses.

The body of literature concerning recommended or suggested guidelines is large and often contradictory. An attempt has been made to select criteria useful for the interpretation of IRP data. With the exception of the Safe Drinking Water Act standards (Table 4.2.1-1), the sole purpose of the criteria presented is to highlight contaminants of interest or of possible concern. Although these criteria do not now have the force of standards, they do provide a valid means for assessing the relative degree of contamination. For each site, parameters detected are compared to these standards and guidelines.

The status of the criteria for one parameter of potential concern, methylene chloride, is particularly ambiguous. The 1986 EPA Quality Criteria for Water recommend a value for "halomethanes" of 1.9 ug/L, corresponding to a 10^{-5} cancer risk. For methylene chloride, this level is substantially below what might be otherwise dismissed as inadvertent sample or laboratory contamination of no health or environmental significance. The State of California regulates methylene chloride at the substantially higher level of 40 ug/L. Although methylene chloride is a common industrial solvent, no RMCL exists for it. Methylene chloride exists at levels of potential concern at only one sampling site (well HM-38 at Landfill 3). In view of the ambiguity of the available documentation, methylene chloride is listed on the appropriate table in Section 4.3, but without an accompanying criterion.

Using human health criteria and standards is a stringent way to evaluate groundwater contamination at AF Plant 4. Since the shallow aquifer is not used as a drinking water supply source, contaminants in-situ will have

TABLE 4.2.1-1. REGULATORY STANDARDS FOR INORGANIC GROUNDWATER CONTAMINANTS

Parameter ¹	Maximum Contaminant Level (mg/L, except as noted)
Arsenic	0.05
Barium	1.0
Cadmium	0.01
Chromium	0.05
Lead	0.05
Mercury	0.002
Selenium	0.01
Silver	0.05
Alpha radiation	15 pCi/L
Beta radiation	50 pCi/L
Gamma radiation	no standard ²

¹Primary Drinking Water MCLs; Regulatory Reference: 40 Code of Federal Regulations, Part 141.

²The Primary Drinking Water Regulations do not set a concentration standard for gamma radiation, but require analyses for specific elements and dose calculations if the beta radiation MCL is exceeded.

RAPIAN

.....

40.215.21

TABLE 4.2.1-2. GUIDELINES FOR ORGANIC COMPOUNDS DETECTED IN GROUNDWATER

Compound	Proposed MCLs (ug/L)	Proposed RMCLs (ug/L)	10 ⁻⁵ I Cancer Risk (ug/L)	Toxicity (ug/L)
Vinyl Chloride	1			
Methylene Chloride - NO DAT	A			
Trichlorofluoromethane - NO	DATA, NOT	A PRIORITY P	POLLUTANT	
1,1-Dichloroethylene	7			
1,2-Dichloroethane	5			
trans-1,2-Dichloroethylene	-	70		
Chloroform	*** ^C			
1,1,1-Trichloroethane	200			
Carbon tetrachloride	5 *** ^c			
Bromodichloromethane	*** ^C			
1,2-Dichloropropane		6		
Trichloroethylene	5 *** ^C			
Bromoform	*** ^C		F .	
Tetrachloroethylene			8 ^f .g	
Chlorobenzene		60		
Benzene	5			
Toluene		2000		
Ethyl Benzene		680		
1,3-Dichlorobenzene				400 ^e
1,2-Dichlorobenzene		620		
1,4-Dichlorobenzene	750			
2,4-Dichlorophenol				3.09 mg/L. ^g
2,4-Dimethylphenol				2120 ^{g,1}
Phenol				3.5 mg/L ^g
Acenaphthene			**** ^d •g	U U
Fluoranthene			**** ^d ,g	
Butyl benzyl phthalate				940 ^{g,i}
bis[2-ethylhexyl]phthalate				15 mg/L^8
di-n-butyl phthalate				$35 \text{ mg/L}^{\text{g}}$
Naphthalene				35 mg/L ^g 620 ^{g • h}
Fluorene			**** ^d •8	-
Phenanthrene			**** ^d .g	

(Continued)

DESERVE DAVEND PRESSER TRANSC

12632933



TABLE 4.2.1-2. (Continued)

Compound	Proposed MCLs (ug/L)	Proposed RMCLs (ug/L)	10 ⁻⁵ Cancer Risk (ug/L)	Toxicity (ug/L)
N-nitrosodimethylamine N-nitrosodiphenylamine			0.014 ⁸ 49 ⁸	<u> </u>
Isophorone Pyrene			5.2 mg/L ⁸ ****	

a 50 Federal Register 46904, 13 November 1985. Ъ

50 Federal Register 47022, 13 November 1985. с

*** CFR 141.12 Primary Drinking Water MCL for all trihalomethanes, sum less than 100 ug/L. d

**** 28 ng/L. for all PNAs. e

EPA 440/5-80-039, Ambient Quality Criteria for Dichlorobenzenes. f

EPA 440/5-80-073, Ambient Quality Criteria for Tetrachloroethylene.

g EPA 440/5-86-001, Quality Criteria for Water 1986. h

Lowest Observed Effect Level - Chronic.

i Lowest Observed Effect Level - Acute.

<u> Carlo a Carlo a</u>

no immediate human health consequences. Since these contaminants exit from the shallow groundwater system, however, they may encounter potential receptors. Where waters come to the land surface, either as seeps or as groundwater outflow to streams, the potential exists for exposure. Groundwater within the Paluxy Formation is used directly as a drinking water source. A formal assessment of the environmental and human health risks associated with the occurrence of contaminants is beyond the scope of this program; therefore, the use of human health standards and criteria is both reasonable and prudent.

No guidelines exist concerning the maximum allowable or recommended pollutant concentrations in soils for protection of human health and the environment. Table 4.2.1-3 lists the normal ranges of several heavy metal concentrations in soils. However, natural occurrences of metals can be far above or below the normal range, depending on local geologic conditions.

4.2.2 Quality Confirmation of Organic Chemical Data

As indicated in the Statement of Work (Appendix B), a second gas chromatographic column was needed to qualitatively verify the presence of many organic compounds in AF Plant 4 groundwater samples. The chemical analyses requiring second column confirmations are EPA Methods 601 (8010) and 602 (8020). All applicable organic chemical data were confirmed when quantitative results exceeded values specified by USAFOEHL. Table 4.2.2-1 shows these action limits for EPA 601 and 602 compounds.

The analytical methods noted above require a secondary chromatographic column to distinguish a particular compound from any interferences. The positive presence of a compound is reported when the retention times on both the primary and secondary gas chromatographic columns match standard values for that specific constituent. If the compound is not detected in the second column at the appropriate retention time, then the compound is not present or not detected.



TABLE 4.2.1-3. NORMAL RANGES OF HEAVY METAL CON	CENTRATIONS FOUND IN SOILS
---	----------------------------

Metal	Normal Range (ppm)	
Chromium	10-50	
Lead	2-20	
Arsenic	5-10	
Selenium	0.2-0.6	
Silver	0.04-0.1	

Source: Rose, A.W., H.E. Hawkes, and J.S. Webb, 1979, <u>Geochemistry in Mineral</u> <u>Exploration</u>: Academic Press, New York, 675 p.



TABLE 4.2.2-1. ACTION LIMITS FOR SECOND COLUMN CONFIRMATION ON EPA 601 AND 602 ORGANIC COMPOUNDS

Compound	Concentration (ug/L)	
Benzene	0.7	
Carbon Tetrachloride	4.0	
1,2-Dichloroethane	0.1	
Methylene Chloride	4.0	
Tetrachloroethylene	4.0	
Trichloroethylene	1.0	
Vinyl Chloride	1.0	
Dichlorobenzene isomers	Sum greater than 10	
All other organics	Greater than 10	

4. "ab. val. "al. "ab." ab. "ab. "at. "at. at. at. at. at.



Ċ.,

Table 4.2.2-2 shows the method detection limits for EPA Methods 601 (volatile halocarbons by GC) and 624 (volatile organics by GC-MS). Table 4.2.2-3 shows method detection limits for EPA Method 602 (volatile aromatics by GC). Method detection limits for EPA Method 625 (acid and base/neutral extractables by GC-MS) are shown in Table 4.2.2-4.



TABLE 4.2.2-2.ANALYTES AND DETECTION LIMITS FOR METHODS 601,
8010, 624, AND 8240

	Detection Limits (ug/L)		
Analyte	Method 601 (8010) ^a	Method 624 (8240) ^a	
Chloromethane	0.08	N/A	
Bromomethane	1.2	N/A	
Vinyl Chloride	0.18	N/A	
Chloroethane	0.52	N/A	
Methylene Chloride	0.25	2.8	
Trichlorofluoromethane	0.4	N/A	
1,1-Dichloroethylene	0.13	2.8	
1,1-Dichloroethane	0.07	4.7	
trans-1,2-Dichloroethylene	0.1	1.6	
Chloroform	0.05	1.6	
1,2-Dichloroethane	0.03	2.8	
1,1,1-Trichloroethane	0.03	3.8	
Carbon Tetrachloride	0.12	2.8	
Bromodichloromethane	0.10	2.2	
1,2-Dichloropropane	0.04	6.0	
cis-1,3-Dichloropropylene	0.2	6.0	
Trichloroethylene	0.12	1.9	
Benzene	N/R	4.4	
Dibromochloromethane	0.09	3.1	
1,1,2-Trichloroethane	0.02	5.0	
trans-1,3-Dichloropropylene	0.34	N/A	
2-Chloroethyl vinyl ether	0.13	N/A	
Bromoform	0.2	4.7	
1,1,2,2-Tetrachloroethane	0.03	6.9	
Tetrachloroethylene	0.03	4.1	
Toluene	N/R	6.0	
Chlorobenzene	0.25	6.0	
Ethylbenzene	N/R	7.2	
1,3-Dichlorobenzene	0.32	N/A	
1,2-Dichlorobenzene	0.15	N/A	
1,4-Dichlorobenzene	0.24	N/A	

^aDetection limits for Methods 8010 and 8240 are in ug/kg.

N/A = Not Available.

N/R = Not Required for Method.



C.C.C.

TABLE 4.2.2-3. ANALYTES AND DETECTION LIMITS FOR METHODS 602 AND 8020

TABLE 4.2.2-3. ANALYTES AND DETECTION LIN	MITS FOR METHODS 602 AN
Analyte	Detection Limi
Benzene	0.2
Toluene	0.2
Ethylbenzene	0.2
Chlorobenzene	0.2
1,4-Dichlorobenzene	0.3
1,2-Dichlorobenzene	0.4
1,3-Dichlorobenzene	0.4



<u>՟ֈ֎ֈֈ՟֎ֈֈ՟ֈֈֈ՟ֈֈֈ՟ֈֈֈ՟֎ֈֈ՟֎ֈֈ՟ֈֈֈֈ՟֎ֈֈ՟ֈֈֈ՟ֈֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟ֈֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟֎ֈֈ՟</u>

12.44*.44*.44*.44*.44*.44*.44*.

TABLE 4.2.2-4. ANALYTES AND DETECTION LIMITS FOR METHODS 625 AND 8270

Analyte	Detection Limit (ug/L) ^a
ACIDS	
2,4,6-Trichlorophenol	2.7
4-Chloro-3-methy1pheno1	3.0
2-Chlorophenol	3.3
2,4-Dichlorophenol	2.7
2,4-Dimethylphenol	2.7
2-Nitrophenol	3.6
4-Nitrophenol	2.4
2,4-Dinitrophenol	42
2-Methy1-4,6-dinitrophenol	24
Pentachloropheno1	3.6
Phenol	1.5
BASE/NEUTRALS	
Acenaphthene	1.9
Benzidine	44
1,2,4-Trichlorobenzene	1.9
Hexachlorobenzene	1.9
Hexachloroethane	1.6
Bis(2-chloroethyl)ether	5.7
2-Chloronaphthalene	1.9
1,2-Dichlorobenzene	1.9
1,3-Dichlorobenzene	1.9
1,4-Dichlorobenzene	4.4
3,3'-Dichlorobenzidine	17
2,4-Dinitrotoluene	5.7
2,6-Dinitrotoluene	1.9
1,2-Diphenylhydrazine	N/A
Fluoranthene	2.2
4-Chlorophenyl phenyl ether	4.2
N-Nitrosodimethylamine	10
N-Nitrosodiphenylamine	1.9
N-Nitrosodi-n-propylamine	12
Bis(2-ethylhexyl)phthalate	2.5
Butyl benzyl phthalate	2.5
Di-n-butyl phthalate	2.5
Di-n-octyl phthalate	2.5
Diethyl phthalate	1.9
Dimethyl phthalate	1.6

(Continued)

10.00

BULLES

1.1.1.1.1.1.1

55555



TABLE	4.2.2-4.	(Continued)
-------	----------	-------------

Analyte	Detection Limit (ug/L) ²
BASE/NEUTRALS (Continued)	
Benzo(a)anthracene	7.8
Benzo(a)pyrene	2.5
Benzo(b)fluoranthene	4.8
Benzo(k)fluoranthene	2.5
Chrysene	2.5
Acenaphthylene	3.5
Anthracene	1.9
4-Bromophenyl phenyl ether	1.9
Bis(2-chloroisopropy1)ether	5.7
Bis(2-chloroethoxy)methane	5.3
Hexachlorobutadiene	0.9
Hexachlorocyclopentadiene	6.0
Isophorone	2.2
Naphthalene	1.6
Nitrobenzene	1.9
Benzo(ghi)perylene	4.1
Fluorene	1.9
Phenanthrene	5.4
Dibenzo(a,h)anthracene	2.5
Indeno(1,2,3-cd)pyrene	3.7
Pyrene	1.9

^aDetection limits for Method 8270 are in ug/kg.

N/A = Not Available.

RAPIAN

4.3 Discussion of Results and Significance of Findings

The following subsections discuss the results of the geophysical surveys, drilling, sampling and chemical analyses performed at AF Plant 4. Sites are discussed in the order in which they appear in the Delivery Order. Discussions of additional study topics from the delivery order follow the site discussions. Lithologic and well-completion logs for all wells and borings installed during this program are presented in Appendix E. Raw analytical data appear in Appendix A.

4.3.1 Site 1, Landfill 1

IRP Phase II Stage 1 work at Landfill 1 consisted of collecting and analyzing samples from existing wells and drains where data had been collected in previous investigations. These data revealed contamination in shallow groundwater at the landfill, which was partially excavated in 1983. Reportedly, the landfill had received mostly construction rubble, trash, and fill dirt. However, hazardous wastes such as solvents, thinners, and paints; contaminated oils and fuel; and chromate sludges and cyanide were disposed of in pits at the landfill and burned. The resulting sludges were dredged from the pits and deposited in the landfill. After the landfill was closed, a series of drainage pipes were installed to channel liquids to a stormwater outfall. The drainage system was later modified to route liquids to a sump. This drainage system is called French Drain No. 1. In 1983, the waste oil pits were excavated (from the land surface to bedrock) and 11,000 cubic yards of earth were removed. Following the excavation, a drain field (French Drain No. 2) was installed.

Figure 4.3.1-1 shows the locations of groundwater sampling points and the physical features of the site. The locations of Meandering Road Creek surface water samples are shown in Figure 4.3.1-2. Groundwater samples were collected and analyzed at HM-7, 10, 18, 20, 62, 63, P-4, P-7 Middle (P-7M), and P-7 Upper (P-7U). Monitor wells HM-18, 20, 62, and 63 are located hydraulically upgradient of the landfill. Monitor wells HM-19 and HM-50 are located within Landfill 1 but were sampled in association with FDTA 2, as required in

4-32



4-33

64° A



الديدين
the statement of work. Surface water samples were collected and analyzed from Meandering Road Creek west of the landfill at the following locations: C-1, C-2, C-3, C-4, and C-5. The C-1 and C-4 locations are upstream of the landfill, while the other locations are downstream of the landfill and other disposal sites. Other water samples were collected and analyzed from French Drains 1 and 2, the drain pipe and the creek seep. St-5 outfall was dry at the time of sampling. Groundwater and surface water samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, acid and base/neutral extractable compounds by EPA Method 625, and heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2.

4.3.1.1 <u>Results of Investigation</u>

The investigation conducted at Landfill 1 focused on the collection, analysis, and interpretation of groundwater quality samples. Results of these activities and descriptions of the topographic and geologic features of the site are discussed in the following paragraphs.

Topography

The land surface in the vicinity of Landfill 1 ranges from approximately 660 feet MSL at the southeast to 635 feet MSL on the west side of the landfill. The surface of the landfill is paved and used as an employee parking lot.

Geologic Features

Geologic information for Landfill 1 is available from the logs of borings of several monitor wells. In the Hargis and Associates (1985) report, definitions and logs of the formations were based on the following characteristics:

Alluvium: Unconsolidated poorly-sorted to well-sorted clay, silt, sand and gravel.

Goodland Limestone: Dense, thinly to massively bedded fossiliferous limestone interbedded with stiff clay and shale.

Walnut Formation: Indurated fossiliferous limestone and shell agglomerate with thin-bedded calcareous shale and clay. Black fissile shale encountered immediately above the shell agglomerate in several locations.

Paluxy Formation: Fine- to medium-grained, friable, white quartz sand interbedded with gray to blue-green silty clay and shale, and poorly cemented sandstone.

To make the lithologic logs from the previous investigation by Hargis and Associates comparable to those of this report, formation definitions have been based on the same characteristics.

The logs from Hargis and Associates (1985) and from this report (Appendix E) indicate that alluvial and terrace materials of the upper zone underlie the entire site. Thicknesses of these materials range from 13 feet at monitor well HM-7 to 33 feet at monitor well HM-62.

The Goodland Limestone was encountered only at HM-19, which recorded two feet of limestone. The Walnut Formation was present in all wells and ranged in thickness from two feet at HM-7 and HM-10 to 30 feet at P-4.

Of the four Paluxy wells located in the vicinity of Landfill 1. only P-4 fully penetrates the Paluxy Formation. At this well, the Paluxy Formation is approximately 170 feet thick. This description of geologic features at Landfill 1 is consistent with that of the Margis and Associates (1985) report.

Occurrence of Groundwater

Groundwater at Landfill 1 occurs in both the materials of the upper zone and in the Paluxy Formation. Results of water level measurements are



provided in Table 4.3.1-1. The water level map for the up er zone in this area of AF Plant 4 (Figure 4.1.2-6) was previously described in Section 4.1.2. Water levels in upper zone wells monitored for water quality at this site ranged from 639.96 feet MSL at HM-63 to 627.34 feet MSL at HM-7 during January-February 1986. Groundwater flow at Landfill 1 is generally to the west. toward Meandering Road Creek.

Water levels in the Paluxy (at P-4, P-7U, and P-7M) have been measured in the upper, middle, and lower members of the formation. Comparison of the water levels in the upper and middle members indicates that the direction of the hydraulic gradient is from the middle to the upper member. However, comparison of the higher water level measured in P-4 (which is screened through the entire thickness of the Paluxy, but representative of lower Paluxy water levels) to the P-7 levels suggests that the lower member of the Paluxy has the greatest hydraulic head, indicating that there is an overall upward vertical hydraulic gradient from the lower to the upper member of the Paluxy aquifer at this location. Groundwater flow in the Paluxy aquifer is to the southeast in the vicinity of AF Plant 4 (see Figures 4.1.2-2 and 4.1.2-3).

Results of an aquifer pumping test conducted at the P-7 wells indicate that the upper Paluxy is in hydraulic communication with the middle Paluxy in the vicinity of P-7 (Hargis & Associates, 1985).

Groundwater and Surface Water Quality

Groundwater samples were collected and analyzed from the following locations: HM-7, 10, 18, 20, 62, 63, P-4, P-7M, and P-7U. Surface water samples were collected from French Drains 1 and 2; the drain pipe; creek seep; and creek locations C-1, -2, -3, -4, and -5. Results of the water analyses and field measurements (pH, temperature, and conductivity) appear in Tables 4.3.1-2 and -3. Complete analytical reports for these analyses are shown in Appendix A.



TABLE 4.3.1-1. GROUNDWATER ELEVATIONS AT SITE 1, LANDFILL 1, AF PLANT 4, TEXAS (JANUARY THROUGH MARCH 1986)

Monitor Well		
	Elevation (feet MSL)	Date
HM-7	627.34	2-13-8
HM-10	628.93	2-7-86
HM-18	630.16	2-11-8
HM-20	628.68	1-31-8
HM-62	629.66	3-1-8
HM-63	639.96	2-7-80
P-4	570.15	1-30-6
P-7U	567.67	1-30-8
P-7M	568.27	1-30-8

				Monit	Monitor Well				
Method Method Decention HM-7 HM-10 Decention Limit 2-13-86	n HM-7 2-13-86	HM-10 2-7-86	НМ-18 2-11-86	НМ-20 1-31-86	НМ62 3-1-86	н м- 63 2-7-86	₽-4 1-30-86	Р-7М 1-30-86	53 P-4 P-7M P-7U -86 1-30-86 1-30-86
Field Parameters Temnarature (°C)	15.5	21.5		22.0	21.0	19.2	22.0	22.0	19.0
pH Conductivity (umhos)	6.6	6,8 1900	6.4 450	6.8 1300	6.6 1200	6.4 900	6.9 900	6.2 680	6.7 840
Analytical Parameters ^a									
Metais (mg/L) Armenic 0.002-0.00	5 0.038	<u>n</u>		Q	0.005*	0.075	Q	9	£
	0.11.0 6	0.15	0.14	0.11	0.17	0.52	0.022	0.068	660.0
Cadmium 0.00	2 ND			22				22	8
Chromitum 0,000 Lead 0.001-0.00	0,02# 0,006#	0.042		41.0 M	0.015	0,008*			0.016
Mercury 0.000	2	0.0002*		2	Ð	Ð	9	2	0.0003*
Selenium 0.002-0.00 Silver 0.00	4 ND6	0.007*	ê e	ND6 0,005*	0.013	0.015	UN 0.011	0.014	0.017
Volatile Organics (ug/L)									
Method 601 (Halogenated Hydroca Concentration Factor	rbons) 1	1	1	25	1	250	10	10	10
Vinyl Chloride 0.1	9 56	QN		9	9	9	9	9	Ð
Methylene Chloride 0.2.	5 2.2	Q !		£ !	£ !	£ !	£ !	6.7	
Trichlorofluoromethane 0.	99	99		29			29		0.64
1.1-Dichloroethane 0.0	7 4.2	2 9	2 2	2 9	2	2 2	2 2	2 2	2 2
trans-1,2-Dichloro-0.1	0	**ON		6.4	Ð	**QN	# #QN	3.2	QN
ethylene Chiorofore 0.00		Đ		ĥ	AF 0	Ę.	Ş	Ş	Ş
1,1.1.Trichloroethane 0.0		29		240	11	2 2	2 9	99	2 2
Trichloroethylene 0.1. 1.2-Dichlorobenzene 0.1	2 5 11 13	37 M	<u>9</u> 9	2300 ND	0.0 0.0	4500 ND	2 2	22	<u>ę</u> 9
(
method ou/ (Aromatic Compounds) Concentration Factor	1	1	1	1	1	25	1	1	1
Benzene 0.:	2 61	Ð		2	9	Ð	Ð	2	Ŷ
Toluene 0.3	2 71	Đ		Ð	£	QN	Ð	Ð	Ð
Ethyl Benzene 0. 1.2-Dichlorobenzene 0.	2 590 3 44	99	99	22	2 2	29	99	99	9 9

TABLE 4.3.1-2. (Continued)

					Monit	Monitor Well				
Date Sampled	Method Detection Limit	НМ-7 2-13-86	HM-10 2-7-86	HM-18 2-11-86	нм-20 1-31-86	HM-62 3-1-86	HM-63 2-7-86	P-4 1-30-86		Р-7М Р-7U 1-30-86 1-30-86
Extractable Organics (ug/L)	(ng/L)									
Method 625 Acid Concentration Factor	ctor	1	1	1	1	1	1	1	1	1
2,4-Dimethylphenol Phenol	ol 2.7 1.5	68 2	99	99	Q 1	9 9	9 9 9	99	99	99
Method 625 Base/Neutral Concentration Factor	utral ctor	. 1	1	1	1	1	1	1	1	1
Aconanht hene		2	9	2	2	2	Ð	Ð	Ð	IN
1.2-Dichlorobenzene		23	QN	Q	Q	9	QN	Q	Q	Ĭ
1.4-Dichlorobenzene		2	Q	9	2	2	Ð	Ð	9	Ð
bis(2-ethylheryl)	2.5	3BL	QN	2	25	Ð	Q	Q	Ð	2
phthalate		ļ		ġ	Í	9	Ę	Ģ	ſ	N
di-n-butyl phthalate		Ê	4	R	Ð	R		2 9	יי	2 9
Naphthalene	1.6	16	9	Ð	9	2	9	ð	Ð	N

ND = Not detected.

* = Value less than five times detection limit.
Errors may range up to 100 percent.
* = Spike recovery not within acceptable limits.

BL = Detected in reagent blank; background subtraction not performed. ** = Compound not confirmed by second-column analysis. - = Data not collected. Indicates interferent.

45

= Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

RADIAN

G

RESULTS OF SURFACE WATER SAMPLE ANALYSES, SITE 1, LANDFILL 1, AF PLANT 4, TEXAS

TABLE 4.3.1-3.

					Moni	Monitor Well				
	Method Detection	5	(-)		4-0	J	(Wanay	C FINAUAS	444545	AD LON
Date Sampled	Limit	1-29-86	1-29-86	1-30~86	1-30-86	1-29-86	2-12-86	2-12-86	1-30-86	2-12-86
Field Parameters		-	-							
lemperature (*U) _u		12.0	12.0	0 °	19.0	19.0	1	0.12 4 8	1.01	I
ри Conductivity (umhom)		920	1100	620	790	700	. 1	1400	820	• 1
Analytical Parametera ⁸ Motolo (2011)										
Artenic Artenic	0.002-0.005	Q	2	g	Q	9	9	9	9	0.007*
Berium	0.009	0.053	0.059	0.065	0.035	0.052	0.068	0.46	0.038	0.062
Cadmium	0.002	Ð	0.003*	2	Q	Ð	Ð	Ð	9	Ð
Chromium	0.005	Ð	0.02*	0.01*	9	9	0,006*	9	0.067	0.008*
Lead	0.001-0.002	9	ę	Ð	Q	2	Ð	0QN	2	2
Mercury	0.0002	Ð	2	Ð	9	• 0000 • 0	9	Ð	2	2
Selenium	0.002-0.004	2	9	2	2	2	2		2	2
Silver	0.002	0.012	0.013	0.015	0,004*	0.008*	Ð	Q	0.013	2
Method 601 (Helosenated Hvdrocarbona)	ted Hydrocarbon	(a)								
Concentration Factor	or	10	10	10	10	10	1000	100	10	1000
Vinvl Chloride	0.18	9	Q	9	9	Ð	1200	33	9	970
trans-1,2-Dichloro-		Ð	7.1	5.8	9	2	**QN	**QN	##QN	ND**
einyrene		ę	į	9	9	-	ŝ	.,	Í	9
I, I, I-Trichloroethane m : - + 1 + - 1	ane 0.03	2 9	29	29	2	29		0055	28	
iricnioroetnyiene Tetrachloroethylene		29	2 2	29	29	2 9		B D D	2 9 9	150
Method 602 (Aromatic Compounds)	Compounds)									
Concentration Factor	or .	1	1	1	1	-	1000	50	1	1000
Toluene	0.2	Q	Ð	Q	9	Đ	Ð	Đ	Ð	006
Extractable Organics (ug/L)	ug/L)									
Method 625 Acid					•		•	-	-	-
Concentration Factor	or	1	1	1	-	1	-	-	-	-1
2.4-Dichlorophenol	2.7	9	Ð.	₽ '	9 9	9 9	с с	£.	29	-3 6
Phenol	C.I	7	7	m	â	â	7	7	Ð	7

ł

5

NEXTRA NOTES STATES

125.15.1.1.1 222222

27.2

22222222

22022555

ر مر

کر

(Continued)

RADIAN

4-41

(Continued) TABLE 4.3.1-3.

					Moni	Monitor Well				
	Method Detection		C-2	C-3	C-4		FRDRN#1	FRDRN#2	CRSEEP	DRNPIPE
Date Sampled	Limit	1-29-86	1-29-86	1-30-86	1-30-86	1-2986	2-12-86	2-12-86	1-30-86	2-12-86
Method 625 Base/Neutral	1									
Concentration Factor		1	1	-	1	1		1	-	1
1,2-Dichlorobenzene	1.9	Ð	Ð	Ð	2	2	220	Ð		240
1,4-Dichlorobenzene	4.4	Ð	Q	Q	Ð	Ð	40	Ð	Ð	35
Bis(2-ethylhexyl) phthalate	2.5	ŝ	QN	4	Q	2	17	Ê		4
Di-n-butyl phthalate	2.5	4	3.2	Ð	Ð	Ð	6BL			4
Naphthalene	1.6	Ð	Ð	Q	Q	Q	6			16
Pyrene	1.9	Q	Q	QN	Ð	Ð	ę		Q	7
Fluoranthene	2.2	Ð	Ð	Ð	Q	Q	Ð			7

Not detected. ę

4-42

- Value less than five times detection limit. 0 0 *
 - Errors may range up to 100 percent.
- Spike recovery not within acceptable limits. 41 9
 - Indicates interferent.

- BL = Detected in reagent blank; background subtraction not performed.
 ** = Compound not confirmed by second-column analysis.
 = Data not collected, measurement omitted in field (FRNDRN #1) and deleted to prevent fouling of instrument (DRNFIPE).
 a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor.
 Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

KARANSI NASASIN KANANA KAN

24446666 Y2274468

REEL DISTRICT DISTRICT DISTRICT DISTRICT

Radian

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Tables 4.3.1-4 and -5 summarize of these comparisons.

4.3.1.2 Significance of Findings

This section discusses the significance of the investigations's findings, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) have previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates previous findings and trends identified by the Hargis report.

Significance of Metals in Groundwater and Surface Water

Surface water at sites associated with Landfill 1 and groundwater in the upper zone contained concentrations of heavy metals generally below federal MCLs. The exceptions are chromium, which was detected at levels exceeding the MCL of 0.05 mg/L at three locations, and arsenic, which was detected at levels exceeding the federal MCL of 0.05 mg/L at one location. The discovery of chromium in concentrations greater than the MCL is not surprising because of the known disposal of chromium-bearing sludges and wastes at the landfill. The concentration of chromium was 0.069 mg/L at HM-62, 0.14 mg/L at HM-20 and 0.067 mg/L in the creek seep.

Previous groundwater analyses have identified chromium in excess of MCL levels in HM-20. Similarly, arsenic, cadmium, and lead were detected in well HM-62 in excess of MCL levels. These earlier findings were not duplicated during the January-February 1986 sampling and analysis of groundwater.

4-43

Farameter Criteria* HM-7 HM-10 HM-20 HM-62 METALS (mg/L) METALS (mg/L) MM-20 HM-60 HM-60 HM-60 Marenic 0.05 0.05 0.14 0.069 Arsenic 0.05 0.05 0.14 0.069 ORCANIC COMPOUNDS (ug/L) 37 240 8 11,1-Trichloroethane 200 5 61 11,1,1-Trichloroethane 0.028 2 8 Acemaphthene 0.0228 2 2 * See Tables 4.2.1-1 and 4.2.1-2 for source of criteria. *	Parameter
(mg/L) c 0.05 um 0.05 c COMPOUNDS (ug/L) C COMPOUNDS (ug/L) C COMPOUNDS (ug/L) C ComPOUNDS (ug/L) C ComPOUNDS (ug/L) C ComPOUNDS (ug/L) C C COMPOUNDS (ug/L) C C COMPOUNDS (ug/L) C C C C C C C C C C C C C C C C C C C	
and 0.05 0.05 0.14 and 0.05 0.05 0.14 C COMPOUNDS (ug/L) C COMPOUNDS (ug/L) Chloride 1 56 37 240 Chloride 200 5 61 37 2300 e 5 61 37 2300 e 0.028 2 61 hthene 0.028 c criteria. Tables 4.2.1-1 and 4.2.1-2 for source of criteria.	METALS (Bg/L)
C COMPOUNDS (ug/L) Chloride 1 56 240 Trichloroethane 200 37 2300 oroethylene 5 61 37 2300 e 61 200 238 61 hthene 4.2.1-1 and 4.2.1-2 for source of criteria.	Argenic Chromium
Chloride 1 56 240 Trichloroethane 200 37 2300 oroethylene 5 61 e 0.028 2 Tables 4.2.1-1 and 4.2.1-2 for source of criteria.	ORGANIC COMPOUNDS
See Tables 4.2.1-1 and 4.2.1-2 for source of	Vinyl Chloride 1,1,1-Trichloroeth Trichloroethylene Benzene Acemaphthene
	See Tables
· · · · · · · · · · · · · · · · · · ·	



• Charles

TABLE 4.3.1-5. COMPARISON OF SURFACE WATER QUALITY AT SITE 1, LANDFILL 1, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

╈┙╇╧╋╧╄╝╅╝╅╢╅╝╪╝╪╧╄╪╝╪╧┾╝┾╅╬╪╪╪╝╝╡╝┝╝┝╝╞╝┝╝╞╝╞╝╞╲╞╢╸┧╝╪╝┍╝╝┝╝╝┝╝╝╪╝╌╝╸╪╝╧╧╸╝╝┪╝╕╝╝╝╝╝╝╝╝╝╝╝╝╝╝╝╝╝

			Sampling	Location	
Parameter	Criteria*	FRDRN#1	FRDRN#2	CRSEEP	DRNPIPE
METALS (mg/L)					
Chromium	0.05			0.067	
ORGANIC COMPOUNDS (ug/L	.)				
Vinyl Chloride	1	1200	33		970
Trichloroethylene	5	6700	2300	20	2300
Tetrachloroethylene	8				150
Fluoranthene	0.028				2

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.

Significance of Organic Compounds in Groundwater and Surface Water

Synthetic organic compounds were detected predominantly in upper zone ground water and in water draining from the landfill. Other minor occurrences of these compounds were noted in the Paluxy aquifer and in Meandering Road Creek. These findings are generally consistent with the results of earlier sampling and analysis conducted by Hargis & Associates.

Based on results from the HM-series wells, groundwater in the upper zone contains the highest concentrations of organic contaminants. Monitor well HM-7 displays concentrations of vinyl chloride (56 ug/L), and benzene (61 ug/L) in excess of MCL levels. Trichloroethylene (TCE) was detected in four of six upper zone wells, with the highest readings from HM-20 (2300 ug/L) and HM-63 (4500 ug/L). Water sampled from HM-18 appears to be the least affected by contamination; the only organic compound detected was toluene, at a concentration of 0.6 ug/L.

Water sampled from the French drains and the drainpipe showed the highest concentrations of organic compounds of any sampling point at Landfill 1. In particular, vinyl chloride (33 - 1200 ug/L) and TCE (2300 - 6700 ug/L) were detected at these three locations. Also, an unknown (non-method) compound eluting on both columns at retention times near that of trans-1,2-dichloroethylene was discovered. Several other contaminants, including toluene, tetrachloroethylene, and 1,1,1-trichloroethane were detected in these samples. Water sampled at these locations probably represents a mixture of landfill liquids and groundwater because of the high observed concentrations of organic compounds and the origin of the water from the drainfield under the landfill area.

Groundwater samples collected from three Paluxy aquifer wells revealed several organic compounds in the water. The sample from P-4, which is closest to the landfill and screened over multiple zones in the Paluxy, showed



328 ug/L of an unknown (non-method) compound eluting on both columns at retention times near that of trans-1,2-dichloroethylene. However, this was the only contaminant detected. In the other Paluxy wells, P-7U and P-7M, several contaminants were revealed but in very low concentrations that did not exceed federal criteria. Trans-1,2-dichloroethylene, undetected in the upper member of the Paluxy, was detected at a concentration of 3.2 ug/L in the lower member of the Paluxy.

Surface water samples collected from five locations along Meandering Road Creek had concentrations of metals below federal MCL levels and concentrations of organic compounds generally below the limits of detection. The upstream locations, C-1 and C-4, showed virtually no levels of contamination in the creek water. Samples sites C-2 and C-3, the closest locations to Landfill 1, showed very low levels of contaminants. At these locations, it is possible that a variety of disposal sites, specifically Landfill 3 and FDTA 2, could contribute contaminants to the surface water.

RAPIAN

4.3.2 <u>Site 3. Landfill 3</u>

From about 1942 to 1945, Landfill 3 was used for disposal of miscellaneous wastes, including mixed oils and solvents. One pit (and possibly more) present in this area during the 1940s was used for holding and burning some of the liquid wastes. It is suspected that other wastes were disposed of on the ground and then buried. There was no apparent activity at this site from the late 1940s until 1966, when fill dirt and construction rubble were used to finish the filling and grading of the site.

The IRP Phase II Stage 1 activities at Landfill 3 consisted of performing a terrain conductivity survey, installing a Paluxy aquifer monitor well (P-22U), and collecting and analyzing groundwater samples. Upper zone monitor wells at this site include HM-21, 26, 27, 34, 35, 36, 37, 38, and 39. Paluxy well P-22U was also sampled. Figure 4.3.2-1 shows the locations of these sampling points. Samples were collected and analyzed for volatile organic compounds by EPA Methods 601 and 602, acid and base/neutral extractable compounds by EPA Method 625, and heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2.

4.3.2.1 Results of Investigation

The investigation at Landfill 3 focused on the performance of the terrain conductivity survey, the construction of the Paluxy monitor well, P-22U, and the collection and analysis of groundwater samples. Discussions of these activities and descriptions of the topography and geologic features of the site follow.

Topography

The land surface at Landfill 3 is generally flat, with elevations ranging from approximately 638 feet MSL to 630 feet MSL. The area is bounded on the east by Meandering Road and on the west by the slope break at Meandering



. The take is the take

4-49

-49

Road Creek. Construction rubble is visible in the steep slopes above the creek. The level of Meandering Road Creek is approximately 600 feet MSL in the area of Landfill 3. The landfill area itself is landscaped with grass and is sometimes used for parking overflow from the General Dynamics employee lots east of Meandering Road.

Geologic Features

Lithologic logs of nine upper zone wells (Hargis and Associates, 1985) and one Paluxy monitor well (Appendix E, this report) provide information on the subsurface features at Landfill 3.

The upper zone is relatively thin at this site, ranging in thickness from three feet at P-22U to 20 feet at HM-38. The Goodland Limestone was absent at all wells drilled in Landfill 3.

The Walnut Formation, composed of shell agglomerate and calcareous shale and clay, crops out along Meandering Road Creek just west of Landfill 3 and was encountered at all wells drilled at this site. The formation was fully penetrated only at P-22U, where it is 25 feet thick.

The upper member of the Paluxy Sand was fully penetrated during the construction of monitor well P-22U, where it is 26 feet thick.

Occurrence of Groundwater

At Landfill 3, groundwater occurs in both the materials of the upper zone and the Paluxy Formation. Table 4.3.2-1 gives results of water level measurements. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-6) was described previously in Section 4.1.2. Water levels in the upper zone at this site ranged from 628.91 MSL at HM-34 at the south end of the landfill to 611.35 MSL at HM-38 in the west-central portion of the landfill. As shown in Figure 4.1.2-6, groundwater flow in the upper zone is



1.111.111.1

VCV29292

A. 110. Phy. 81

1.1.1.1.1.1.

0.1.1.

Monitor Well	Elevation (feet MSL)	Date
HM-21	627.71	2-7-86
HM-26	627.39	1-31-86
HM-27	627.83	1-31-86
HM-34	628.91	2-27-86
HM-35	628.29	2-27-86
HM-36	627.74	2-27-86
HM-37	627.56	1-31-86
HM-38	611.35	2-21-86
HM-39	618.58	2-27-86
P-22 U	575.94	4-10-86

TABLE 4.3.2-1. GROUNDWATER ELEVATIONS AT SITE 3, LANDFILL 3, AF PLANT 4, TEXAS (JANUARY THROUGH FEBRUARY 1986)



555 A A A

to the west, toward Meandering Road Creek. In some areas where upper zone deposits overlie the Walnut Formation, groundwater seeps from the upper zone have been observed. Quoting from Hargis & Associates (1985):

Water level elevation contours in the upper zone beneath the West Parking Lot and west of Bomber Road suggest that groundwater from the upper zone discharges in the vicinity of wells HM-34, HM-35, and the St. 5 outfall...Field inspections of the Walnut Formation outcrops between well HM-9 and Lake Worth have confirmed the presence of other groundwater seeps west of HM-36 along Meandering Road Creek. In this area, Meandering Road Creek has cut approximately 15 to 20 feet into the Walnut Formation. The west bank of the creek is a steep cliff with limestone ledges, where underlying shale interbeds have been eroded. A depression in the top of the Walnut Formation occurs along the west bank in the area of the seeps. At the top of the west bank, upper zone deposits overlie the Walnut Formation. Groundwater apparently seeps from the upper zone deposits in the depression and trickles over the limestone ledges...Although groundwater discharge from the upper zone has not been observed elsewhere along the creek adjacent to the Plant, lithologic data from well HM-38 suggests that a buried subsurface tributary channel to the creek may occur in the vicinity of the St. 5 outfall. The tributary is buried by fill material and is a potential pathway for migration of upper zone groundwater.

Water levels in the upper Paluxy at AF Plant 4 are shown in Figure 4.1.2-2.

Terrain Conductivity Survey

A terrain conductivity survey using electromagnetic profiling was performed in November 1985. A grid was established, as described in Section 3.1.1, measuring approximately 2100 feet (north-south) by 250 feet (east-west). Electromagnetic profile readings were taken at 30 foot intervals on the grid.

Figure 4.3.2-2 illustrates the area of the survey. Survey results are presented in Appendix I. Ranges of conductivity in the area were generally 30 millimhos/meter to 45 millimhos/meter. Anomalous readings could be



AND THE PROPERTY NUMBER OF THE PROPERTY PROPERTY IN THE

كتحيث فتعريد

22.22.22

41111111

WWWWWWW

C N

RAPIAN

attributed to a number of sources, such as buried metal, variable thicknesses of fill material, or shallow contamination in water or soil.

The results of the terrain conductivity survey indicate a random pattern of anomalies that may be due to relict disposal features at Landfill 3. The recorded conductivity values for this area reflect no major conductivity trends; however, large anomalous areas may suggest moderate amounts of fill and high contamination, which has been documented in the drilling and groundwater sampling and analysis program. In addition, there are smaller areas interpreted to reflect high amounts of metal, moderate amounts of fill, and moderate amounts of fill with low contamination.

Groundwater Quality

Groundwater samples were collected and analyzed from the following locations: HM-21, 26, 27, 34, 35, 36, 37, 38, 39, and P-22U. All samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, acid and base/neutral extractable compounds by EPA Method 625, and heavy metals by EPA Method 200.7, 206.2, 239.2, 245.1, and 270.2. Results of the groundwater analyses and field measurements (pH, temperature, and conductivity) are provided in Table 4.3.2-2. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 Sampling were compared to federal standards and guidelines. Table 4.3.2-3 provides a summary of this comparison.

4.3.2.2 Significance of Findings

This section discusses the significance of the findings of the investigation, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) have previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report

	κ.	ĥ
	K.	
	Ŀ.	
	C.	
	6	
	F	
	2	
	Ľ.	
	K.	
	D.	
	100025-5-56	
	6	
	DK.	
	R.	
1	K.	
	Ě.	
	Ŋ	
ļ		
ļ	Ē4	
ļ	C.	
ļ		
ļ	4	
ļ		
ļ	7	
	Ê.	
	K.	
	K.	
	C	
	ţ.,	
	[-]	
	-	
l	Ŀ	
	K	
	Π.	
	2	
	Ν.	
	μ.	
	Ľ	
	С.	
	C	
	Ì	
	C.,	
	K-	
	Ν.	
	N	
	5	
	F.	
	Ĉ.	
	F.	
	h	
	5	
	F	
	Ŀ	
	[·	
	وكالمستعد والمتعالم والمستعم والمراجع والمراجع والمعالي والمعالي والمعالي والمستعم وال	
	ALLA DERRECCE	
	5	
	Ľ	
	K	
	N.	
	h	
	1	

TABLE 4.3.2-2.	RESULTS OF	RESULTS OF GROUNDWATER SAMPLE ANALYSES,	SAMPLE ANA	LYSES ,	SITE 3, L	LANDFILL 3.
	AF PLANT 4	TEXAS				

	Method Detection						Monitor Well	11				
Date Sampled	Limit	HM-21 2-7-86	HM-26 1-31-86	нм-27 1-31-86	HM-34 2-27-86	HM-34D 2-27-86	HM-35 2~27-86	HM-36 2-27-86	HM-37 1-31-86	HM-38 2-21-86	нн-39 2-27-86	P-220 4-10-86
Field Parameters												
Temperature (°C)		14.0	17.0	19.0	1	1	18.5	I	18.0	1	1	20.0
pH Conductivity (u∎hos)		5.8 2000	2100	260		1 3	1000	11	6./ 2600	. 1	1 (9.9 086
Analytical Parameters ^a												
ng/L)			!	!					9	:		
•	0.002-0.005	2			0.040	0.062	0,006#	0,005#		¥	0.13	0.008
Barice	600 J	0.10	0.023	0.12	0.45 0	5.0 1	060.0	0.01	0.053	23	0.77	0.093
	700.0									5 3		
Coromatuma Lond	0,00-0-100-0	0.004#		0.009	0.00	0.024	0.030	010-0		5 2	0.250	0.036
174	0.0002	0.0002	9	E S	Ĩ					ž	9	0,0004*
a	0.002-0.004	ę	2	2	9	2	2	OQX	9	ž	Q	Q
	0.002	0.004*	0.012	0.019	0.011	0.011	*600°0	0.005*	0.016	W	0.027	0.011
Volstile Organics (ug/L)	-											
Method 601 (Halogenated Hydrocarbons)	d Hydrocarbon											
Concentration Factor		100	50	10	1	7	1	1	25	10000	1	1
Vinyl Chloride	0.18	9	Ð	Ð	Ð	9	Ð	2	2	N	Q	20
Methylene Chloride	0.25	<u>E</u>	ĝ	Q	QN	QN	0.33	0.33	Q	160000	9	Q
trans-1.2-Dichloroethylene		**0	300	**QN	-	9	9	0.74	##QN	120000	8.9	Ð
1.2-Dichloroethane	0	QN	2	QN	2	Q	9	Ð	Ð	24000	Ð	2
Trichloroethvlene	0.12	Q	2000	390	Q	2	2	0.31	400	450000	0.16	46
Tetrachloroethylene	0.03	Q	9	Q	Q	Q	0.07	0.08	9	19000	Q	2
1.2-Dichlorobenzene	0.15	2	9	Ð	Ð	2	Ð	9	Ð	300000	2	Ð
Method 602 (Aromatic Compounds)	ompounds)											
Concentration Factor		s	S	\$	1	1	1	1	\$	20000	1	1
Toluene	0.2	2	9	2	1.5	130	55	Ð	2	0.6%	13	21

4-55

PITTE ZZZZZZA SKOROW PARAZA SKASKA SKASKA SKASKA SKASKA PARAZA PARAZA PARAZA PARAZA SKASKA PARAZA SKASKA PARA

rete i e i e i e

(Continued)

Method Detection	Method Detection						Monitor Well	11				
Li Date Sampled	Limit	HM-21 2-7-86	HM-26 1-31-86	HH-27 1-31-86	HM-34 2-27-86	HM-34D 2-27-86	HH-35 2-27-86	HM-36 2-27-86	HM-37 1-31-86	HM-38 2-21-86	HM-39 2-27-86	P-22U 4-10-86
Extractable Organics (ug/L)												
Method 625 Acid Concentration Factor		1	1	1	1	1	1	1	1	ŝ	1	1
Phenol	1.5	C	2	3	Đ	Ð	Ð	Q	16	35	Ð	2
Method 625 Base/Neutral Concentration Factor		1	1	1	1	1	I	1	1	Ś	1	1
Acenaphthene	1.9	Ð	Ð	Ð	9	Ð	Ð	9	9	250	Ð	9
1.4-Dichlorobenzene	4.4	9	9	ę	Ð	Ð	9	9	Ŷ	4400	9	9
Big(2-ethylhexyl)phthalate	2.5	Ð	Ð	Ē	9	Ð	Ð	Ð	Ð	1200	9	16
butyl benzyl phthalate	2.5	Q	2	QN	Ŷ	2	Ð	Ð	2	150	2	9
Di-n-butyl phthelate	2.5	e	Ð	9	2	Û	9	Q	9	850BL	Ż	SBL
Naphthalene	1.6	Ð	Ð	g	Ð	Ð	9	2	9	1400	9	9
Fluorene	1.9	Q	9	Ð	9	9	Ð	Ð	9	190	Ð	Đ
Phenant hr ene	5.4	9	9	Q	9	9	Q	9	9	590	Q	Q

(Continued) TABLE 4.3.2-2. Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4. Data not collected, omitted in field (NM34 and 36), insufficient volume (HM 38 and 39). 11 46

11 ı

 Value less than five times detection limit. Errors may range up to 100 percent.
 Spike recovery not within acceptable limits. Indicates interferent.
 Compound not confirmed by second-column analysis.
 Detected in reagent blank; background subtraction not performed. NA = Not snalyzed. ND = Not detected. * = Value less than @ = Spike recovery ** = Compound not co BL = Detected in res D = Duplicate sampl

TANKAR PARAMA PARAMA

2222222

222222222

7

Duplicate sample.

RADIAN

0.05 0.05 0.05 1 1 70 300	0.062 0.13 0.088		Sampling Location HM-35 HM-36 HM-37	HM-36 HM-	-39 P-22U
0.05 0.05 0.05 1 200 200 200					
(ug/L) 1 5 ethylene 70 300		0.065 0.055	55	000	0.13 0.14 0.25
e 1 e 5 oethylene 70 300					
ethylene 70 300				160000	20
0007	390		400	24000 120000 19000	46
				6000000 4400	
nzene 620 0.028				300000 250	
620 0.028 0.028				1400 190 590	



recognizes and incorporates previous findings and trends identified by the Hargis report.

Significance of Metals in Groundwater

Three metals, chromium, arsenic, and lead, were detected in upper zone groundwater in concentrations exceeding the federal MCL criteria. In particular, chromium, noted at these levels in five wells, ranged in concentration from 0.055 mg/L at HM-36 to 0.14 mg/L at HM-39. Hargis & Associates (1985) reported that chromium was observed in water from HM-36 during December 1984. These findings are also consistent with the presence of chromium in the creek seep (reported in Section 4.3.1.2) at the margin of Landfill 3. Metals in the upper member of the Paluxy aquifer at P-22U were reported at concentrations below federal MCL criteria.

The origin of chromium and the other metals in upper zone groundwater could be Landfill 3 or several other upgradient sources. Landfill 1 and 2 are possible nearby upgradient contributors of heavy metals. In addition, Chrome Pit 3 is a more distant upgradient source of chromium and other heavy metals.

Significance of Organic Compounds in Groundwater

A variety of volatile organic compounds exists in upper zone groundwater in all monitor wells at Landfill 3, an observation consistent with the results of previous monitoring (Hargis & Associates, 1985). In addition, acid and base/neutral compounds were detected in HM-38. Away from HM-38, the main contaminant at Landfill 3 is TCE. TCE ranged from "not detected" to 2,000 ug/L. Monitor well HM-38 contained much higher levels of these contaminants, as well as the highest level of phenol (35 ug/L) and a variety of base/neutral organic compounds.

These observations agree with those used by Hargis & Associates (1985) to suggest that groundwater contamination in upper zone wells in the



vicinity of Landfill 3 may be the result of contaminant transport via the shallow groundwater flow system from the former Waste Oil Pits located upgradient at Landfill 1; however, the nature and magnitude of contaminants observed at HM-38 strongly suggest a local source of contamination at Landfill 3 in the vicinity of HM-38.

Two volatile organic compounds were detected in monitor well P-22U. These compounds are vinyl chloride (20 ug/L), and TCE (46 ug/L). Although vinyl chloride was not detected in upper zone groundwater during this sampling and analysis program. Hargis & Associates (1985) noted the previous occurrence of vinyl chloride in several upper zone monitor wells. With regard to the occurrence of contamination in the Paluxy aquifer. Hargis & Associates (1985) noted that contaminants were detected in the P-10 well suite, located west of Meandering Road Creek and upgradient of Landfill 3. The IRP Phase II Stage 1 sampling and analysis program also detected volatile organic compounds in P-10U. The Hargis report hypothesized that the Paluxy Formation west of the creek may be recharged by upper zone groundwater and creek underflow. The Walnut Formation aquitard has been deeply incised by the creek as it flows adjacent to Landfill 3. Quoting from the Hargis report (p. 72):

> The extent of contamination in the Paluxy Formation due to recharge from the upper zone in the vicinity of the P-10 well suite is unknown; however, movement of contaminants from this area toward the southeast is possible. Field inspection indicates that the source of the contaminants which are believed to enter the upper Paluxy Formation in the vicinity of the P-10 well suite may not be on-site. Water level elevation data for the Paluxy Formation indicate that contaminants from known on-site sources would be transported to the south-southeast, not the north-northwest toward the monitor well P-10 suite.

Thus the contamination observed in P-22U possibly may possibly be the result of a combination of the sources attributed to the contamination noted in P-10 and of the local movement of contaminated upper zone groundwater through the Walnut Formation.



4.3.3 Site 12, Chrome Pit 3

Work at Chrome Pit 3 consisted of collecting and analyzing samples from existing wells where data had been collected in previous investigations. These previous data revealed contamination in shallow groundwater at the site, which reportedly contained barium chromate sludge, dilute metal solutions, and drums of unidentified liquids. In 1983 and 1984, the pits were excavated and 8,900 cubic yards of earth were removed. Testing during excavation indicated that most of the contaminated soils were removed.

Groundwater samples were collected and analyzed at HM-30, 16, 15, 45, 17, 32, 41, and P-2. Figure 4.3.3-1 shows the locations of these sampling points and the physical features of the site.

4.3.3.1 Results of Investigation

The investigation conducted at Chrome Pit 3 focused on the collection, analysis, and interpretation of groundwater quality samples. Discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

The ground surface in the vicinity of Chrome Pit 3 is relatively flat, with an elevation of approximately 658 feet MSL.

Geologic Features

Upper zone monitor well HM-1 was located within this site but has since been destroyed by excavation. The lithologic log of this well (Hargis and Associates, 1985) shows an upper zone thickness of 29 feet. Five feet of Goodland Limestone was encountered.





Southern Revenue

Lithologic logs of monitor wells surrounding the site (Hargis and Associates, 1985) give additional information on subsurface features. The thickness of the upper zone in the wells surrounding the site ranges from 18 to 29 feet. The Goodland Limestone formation is absent at HM-13, HM-32, and HM-41, but where present it ranges in thickness from 6 feet at HM-30 to 18 feet at HM-17.

The thickness of the Walnut Formation at this site can be determined from the lithologic log of Paluxy monitor well P-2 (Hargis and Associates, 1985). The thickness of the Walnut Formation is 35 feet at this well.

Occurrence of Groundwater

Groundwater occurs at Chrome Pit 3 in both the materials of the upper zone and the Paluxy Formation. Table 4.3.3-1 gives results of water level measurements. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-7) was described previously in Section 4.1.2. Water levels in the upper zone at this site ranged from 638.42 feet MSL at HM-17 to 630.21 feet MSL at HM-41. The groundwater elevation at P-2 was measured at 561.98 feet MSL.

As shown in Figure 4.1.2-7, groundwater in the upper zone at this location flows predominantly to the west. Groundwater flow in the Paluxy Formation at AF Plant 4, discussed in Section 4.1.2, is predominantly to the southeast.

Groundwater Quality

Groundwater samples were collected and analyzed from the following locations: HM-30, 16, 15, 45, 17, 32, 13, 41 and P-2. All samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, acid and base/neutral extractable compounds by EPA Method 625, and heavy metals by EPA Methods 200.7 206.2, 239.2, 245.1, and 270.2. Results of groundwater analyses and field measurements (pH, temperature, and conductivity) are shown in Table 4.3.3-2. Complete analytical reports for these analyses appear in Appendix A.



, A

<u>ͺ</u>

Monitor Well	Elevation (feet MSL)	Date
HM-13	643.93	2-19-86
HM-15	633.58	2-25-86
HM-16	635.52	2-11-86
HM-17	638.42	2-11-86
HM-30	633.52	2-13-86
HM-32	633.96	2-14-86
HM-41	630.21	2-25-86
HM-45	634.19	2-25-86
P-2	561.98	2-19-86

TABLE 4.3.3-1. GROUNDWATER ELEVATIONS AT SITE 12, CHROME PIT 3, AF PLANT 4, TEXAS (FEBRUARY 1986)

STREET NEWS

222222

G

TABLE 4.1.3-2. RESULTS OF GROUNDWATER SAMPLE AMALYSES, SITE 12, CHROME PIT 3, AF PLANT 4, TEXAS

RESULTS OF GROUNDWATER SAMPLE AMALYSES, SITE 12,

TABLE 4.3.3-2.

CHROME PIT 3, AF PLANT

	Method				-	Monitor Well	_			
	Detection	HH-13	HM-15	HM-16	1	HM-30	HM-32	HH-41	HM-45	P-2
Date Sampled	Limit	2-19-86	2-25-86	2-11-86	2-11-86	2-13-86	2-14-86	2-25-86	2-25-86	2-19-86
Field Parameters										
Temperature (°C)		22.5	,	16.8	21.0	18.0	16.0	21.0	21.0	20.0
μd		7.0	'	6.8	6.6	6.8	1.0	6.8	6.8	1.2
Conductivity (umbom)		890	,	1400	006	910	600	690	1000	710
Analytical Parameters ^a										
Arsenic 0	0.002-0.005	Ð	N	0.038	OQN	Ð	Ð	Q	*600*0	Ð
Barium	600.0	0.11	VN	0.12	0.034	0.087	0.068	0.11	0.095	0.089
Cadmium	0.002	Ð	M	Ð	0.002*	Ð	2	9	● QN	ĝ
Chromium	0.005	0.2*	YN	0.21	2	2	9	9	0.02#	0.01*
	0.001-0.002	0.031	VN	0.10	0.020	Ð	0.009	QN	0.009	Ð
Mercury	0.0002	2	A	9	0.0003*	9	2	9	0.0002*	2
믹	0.002-0.004	9	¥	0.007*	Ð	9 Q.	€QN	Q	Ŷ	NDO
Silver	0.002	0.006*	¥	9	9	Q	0.003#	0.004*	0.006*	0.002*
Volatile Organics (ug/L)										
Method 601 (Halogenated Hvdrocarb	d Hvdrocarbons)									
Concentration Factor		1	250	250	500	10	1	1	10	1
Methvlene Chloride	0.25	Q	2	g	9	9	0.50	9	Q	g
1.2-Dichloroethvlene		2	E E	g	Ş	Ş	ĒN	Ę.	12	E E
trans-1.2-Dichloro-		9	2	2	9	£	2	9	**QN	2
ethylene										
Chloroform		0.46	ę	9	Ð	Q	1.1	Ð	9	Q
1,1,1-Trichloroethane	e 0.03	9	Ð	Q	9	9	9	9	10	9
Trichloroethylene	0.12	Ð	5900	10000	15000	180	Q	2.7	130	Đ
Tetrachloroethylene	0.03	9	Q	9	9	Ð	0.08	Ð	3.0	Ð
Method 602 (Aromatic Hydrocarbons	ydrocsrbons)									
Concentration Factor		1	50	25	250	1	1	1	1	1
Toluene	0.2	2.0	Ð	Q	9	0.78	6.1	0.84	Ð	10
Extractable Organics (ug/L)	/T)									
Method 625 Acid										
Concentration Factor		1		1	1	1	1	1	1.4	1
Phenol	1.5	Q	VN	Ð	9	2	2	9	Ð	2
Method 625 Base/Neutral	П									
Concentration Factor		1		1	1	1	1	1	1.4	1
Bis(2-ethylhexyl)phthalate 2.5	halate	9	W	12BL	Ð	Ð	2	9	Ð	9
Di-n-butyl phthalate	2.5	9	NA	4BL	3BL	QN	Ð	QN	QN	Ð

ND = Not detected. NA = Not analyzed.

* = Value less than five times detection limit. Errors may range up to 100 percent.
• = Spike recovery not within acceptable limits. Indicates interferent.
BL = Detected in reagent blank; background subtraction not performed.
** = Compound not confirmed by second-column analysis.
a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor.
** = Data not collected, insufficient volume.

RADIAN

rapian

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Table 4.3.3-3 summarizes this comparison.

4.3.3.2 Significance of Findings

This section discusses the significance of the findings of the investigation, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) have previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates previous findings and trends identified by the Hargis report.

Significance of Metals in Groundwater

Chromium and lead were detected in upper zone groundwater in concentrations exceeding the federal MCL criteria in HM-16. The concentration of chromium detected at this well was 0.21 mg/L and the concentration of lead was 0.10 mg/L. Concentrations of chromium in excess of MCL levels would be expected at this site because of the known disposal of barium chromate sludge in Chrome Pit 3. Metals in the Paluxy aquifer at P-2 were reported at concentrations below federal MCL criteria. The Hargis and Associates (1985) report contains no data for metals concentrations for this site; therefore, no comparisons can be made.

Significance of Organic Compounds in Groundwater

A number of organic compounds are present in the groundwater in monitor wells at Chrome Pit 3. Two compounds, 1,1-dichloroethylene and trichloroethylene, were detected at levels exceeding the federal MCL criteria. High levels of trichloroethylene were reported in groundwater from monitor wells HM-15, HM-16, HM-17, HM-30, and HM-45. These levels ranged from 130 ug/L at HM-45 to 15000 ug/L at HM-17. COMPARISON OF WATER QUALITY AT SITE 12, CHROME PIT 3, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA TABLE 4.3.3-3.

			Sam	Sampling Location	ion	
Parameter	Criteria*	HM-15	HM-16	HM-17	HM-30	HM-45
METALS (mg/L)						
Chromium . Lead	0.05 0.05		0.21 0.10			
ORGANIC COMPOUNDS (ug/L)						
1,1-Dichloroethylene Trichloroethylene	5	5900	10000	15000	180	32 130

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.



These findings are generally consistent with earlier analyses by Hargis & Associates except for monitor well HM-30, which had much higher levels of TCE reported in 1983-85. Monitor well HM-17 had the highest concentration of TCE among the wells sampled in the Phase II program (15000 ug/L), followed by HM-16 and HM-15 with TCE concentrations of 10000 ug/L and 5,900 ug/L, respectively. The presence of TCE in the groundwater at Site 12 indicates that drums of unidentified liquids reportedly disposed of at Site 12 probably contained solvents.

Water sampled from HM-45 revealed the concentration of 1,1dichloroethene to be 32 mg/L, which is over the federal MCL criteria for this compound.

Water samples collected from the only Paluxy well in Site 12 revealed no organic compounds in excess of the federal MCL criteria.

RADIAN

4.3.4 Site 17, Former Fuel Storage Site

Site 17 was the former location of a 100,000-gallon, above-ground JP-4 storage tank from the early 1940's until it was relocated in 1962. Sampling at this site in 1982 confirmed that soils and upper zone groundwater were contaminated by fuels and other organic compounds.

Site activities during this study consisted of installing an upper zone monitor well (HM-100), and collecting and analyzing soil and groundwater samples. Existing upper zone monitor wells also sampled at this site include HM-8 and -14. Figure 4.3.4-1 shows the locations of these sampling points. Soil obtained during the drilling of monitor well HM-100 and groundwater from the three wells were collected and analyzed for hydrocarbon fuels by EPA Method 418.1.

4.3.4.1 <u>Results of Investigation</u>

This investigation focused on the construction of the upper zone monitor well HM-100 and on the collection and analysis of soil and groundwater samples. A discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

This site is located at the southwest corner of the AF Plant 4 facility, near the center of the Radar Range. The land surface across the site is flat, with an elevation of approximately 670 feet MSL.

Geologic Features

The lithologic logs of HM-8. HM-14 (Hargis and Associates, 1985) indicate that the upper zone in the vicinity of the Former Fuel Storage Site is 49 feet thick at both wells. The lithologic log of HM-100 (Appendix E, this report) also indicates an upper zone thickness of at least 49 feet. At



4-69

- YOU (- ''



HM-100 the upper zone consists of six feet of clay underlain by layers of sand and silt.

The Goodland Limestone was absent at all locations. The Walnut Formation was encountered during the construction of HM-14 but was absent at HM-8 and HM-100.

This description of geologic features at the Former Fuel Storage Site is consistent with that of the Hargis and Associates (1985) report.

Occurrence of Groundwater

Groundwater at the Former Fuel Storage Site occurs in the materials of the upper zone. Table 4.3.4-1 shows the results of water level measurements. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-8) was described previously in Section 4.1.2. Water levels in the upper zone at this site ranged from 635.28 feet MSL at HM-14 to 634.02 feet MSL at HM-100. As shown in Figure 4.1.2-8, the piezometric surface in this area is, for the most part, flat.

Groundwater Quality

Groundwater samples were collected and analyzed from the following locations: HM-8, 14, and 100. All samples were analyzed for hydrocarbon fuels by EPA Method 418.1. Results of the groundwater analyses and field measurements (pH, temperature, and conductivity) are given in Table 4.3.4-2. Complete analytical reports for these analyses appear in Appendix A.

4.3.4.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater and soil chemistry at the site. Hargis & Associates (1985) have previously discussed the significance of


TYX.

TABLE 4.3.4-1. GROUNDWATER ELEVATIONS SITE 17, AT FORMER FUEL STORAGE SITE AF PLANT 4, TEXAS (FEBRUARY THROUGH APRIL 1986)

₩₩₩₩₩₩₩₽₩₽₩₩

Monitor Well	Elevation (feet MSL)	Date
HM-8	634.82	2-13-86
HM-14	635.28	2-19-86
HM-100	634.02	4-11-86

-	
. SITE 17.	
. RESULTS OF SOILS AND GROUNDWATER SAMPLE ANALYSES, S	EXAS
SAMPLE	NT 4. T
NDWATER	FORMER FUEL STORAGE SITE. AF PLANT 4. TEXAS
ND GROU	E SITE.
SOILS A	STORAGI
TS OF	ZR FUEL
RESUI	FORME
TABLE 4.3.4-2.	
TABLE	

6

			Monitor Well			Soil Analysis	sis
	Method	HM-8	HM-14	HM-100	HM-100-A (11-12)	HM-100-B (25-26)	HM-100-C (35-36)
Date Sampled	Limit	2-13-86	2-19-86	4-11-86	1-20-86	1-20-86	1-20-86
Field Parameters							
Temperature (°C)		18.0	21.5	14.5			
DH		6.8	6.7	6.5			
Conductivity (umhos)		006	006	750			
Analytical Parameters							
Hydrocarbon Fuels (mg/L)	ц) 1	QN	Q	QN			
Hydrocarbon Fuels (mg/kg)	kg) 4–6				QN	Q	QN

ND = Not detected.

4-72

A STATES AND A STATES AND A STATES A STATES AND A



33 State 1 - 1 - 1



groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates previous findings and trends identified by the Hargis report.

Significance of Hydrocarbon Fuels in Groundwater and Soil

Hydrocarbon fuels were below instrument detection limits in soil and groundwater samples, indicating a lack of contamination at the site. These findings are consistent with those of Hargis & Associates (1985).

. Na sana na katala na Na sana na katala na k

RADIAN

4.3.5 Site 2, Landfill 2

Landfill 2 originally consisted of some low areas and a livestock watering hole. Most of this site was reportedly filled with construction rubble, plaster, and fill dirt during the early 1940s. Tires and lumber were reportedly disposed of, and later burned there. Although there are no records of hazardous materials being disposed of at the site, it is possible that small quantities of hazardous wastes could be present.

Activities at Landfill 2 consisted of performing a terrain conductivity survey, installing a Paluxy aquifer monitor well (P-21U), and collecting and analyzing groundwater samples. Upper zone monitor wells at this site include HM-2, 22, 40, 42, 43, 44, and 46. Figure 4.3.5-1 shows the locations of these sampling points. Samples were collected and analyzed for volatile organic compounds by EPA Methods 601 and 602, acid and base/neutral extractable compounds by EPA Method 625, and heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2.

4.3.5.1 Results of Investigation

The IRP Phase II Stage 1 investigation at this site consisted of a geophysical survey, the construction of the Paluxy monitor well P-21U, and the collection and analysis of groundwater samples. A discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

The land surface at Landfill 2 is flat to slightly sloping, with an elevation of approximately 665 feet MSL along the southern and western portions of the landfill to 640 feet MSL near Bomber Road. The site is part of the Radar Range at the southwestern portion of AF Plant 4. Most of the site is landscaped with grass.





Geologic Features

Monitor wells located within the boundaries of Landfill 2 are HM-2, HM-42, HM-102, and P-21U. Monitor well HM-102 was constructed at the northwestern margin of Landfill 2. Although this well was initially designated to detect contamination, if any, attributable to FDTA 3, the placement of HM-102 has also served to help define geologic and groundwater quality conditions at Landfill 2.

According to lithologic logs of drilling operations (Hargis and Associates, 1985 and Appendix E, this report) the thickness of the upper zone at this site ranges from 31 feet at HM-102 to 44 feet at HM-42. At all of these locations, the Walnut Formation directly underlies the upper zone.

Paluxy well P-21U, penetrates both the Walnut Formation and the upper member of the Paluxy Sand. Thicknesses of these formations are 40 feet and 25 feet, respectively.

Monitor well HM-102 was constructed at the northwestern margin of Landfill 2. According to the geologic log from drilling and well construction operations, the thickness of the upper zone at HM-102 is 31 feet. The alluvial material coarsens with depth, with coarse sand and gravel encountered at the base of the upper zone. At this location, the Walnut Formation directly underlies the upper zone.

Occurrence of Groundwater

Groundwater occurs in both the materials of the upper zone and in the Paluxy Formation. Table 4.3.5-1 gives the results of water level measurements. Water levels in the upper zone ranged from 668.06 feet MSL at HM-22

TABLE 4.3.5-1. GROUNDWATER ELEVATIONS AT SITE 2, LANDFILL 2, AF PLANT 4, TEXAS (FEBRUARY THROUGH APRIL 1986)

Monitor Well	Elevation (feet MSL)	Date
HM-2	631.67	2-20-86
HM-22	668.06	2-20-86
HM-40	633,39	2-25-86
HM-42	633,80	2-20-86
HM-43	655.12	2-20-86
HM-44	636,17	2-25-86
HM-46	630,97	2-20-86
HM-102	632.15	4-9-86
P-21U	569.32	4-10-86

RAPIAN

south of the landfill (actually closer and more representative of conditions at Landfill 4) to 630.97 feet MSL at HM-46 at the northeastern portion of the landfill. Groundwater flow in the upper zone is to the northeast, generally parallel to Meandering Road Creek. The appearance of the water level contours in the upper zone resembles the pattern of the elevation contours at the base of the upper zone. Hargis & Associates (1985) reported that the direction of groundwater flow in the upper zone is to the north or northeast beneath most of the Radar Range. A small seep ("Radar Range Seep") near the north edge of LandFill 2 (north part of the Radar Range and south of Bomber Road) was also noted by Hargis & Associates (1985).

Geophysical Survey

A geophysical survey consisting of electromagnetic profiling was performed in November 1985. A grid was established, as described in Section 3.1.1. The area of the survey grid was approximately 1800 feet by 800 feet. Electromagnetic profile readings were taken at 30 foot intervals on the grid.

The location of the survey is illustrated in Figure 4.3.5-2. Ranges of conductivity in the area were generally 30 millimhos/meter to 60 millimhos/ meter. Conductivity values generally increased from the northeast to the southwest. Anomalous readings could be attributed to a number of sources, such as buried metal, variable thicknesses of fill material, or shallow contamination in water or soil. A moderately large zone in the northeastern part of the survey area indicated moderately high amounts of fill material. Near HM-43, however, high conductivity values may possibly reflect shallow soil contamination.

Groundwater Quality

Groundwater samples were collected and analyzed from the following locations: HM-2, 22, 40, 42, 43, 44, 46, and P-21U. All samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, acid and



4-79



and the last of the last of the last and the last of t

base/neutral extractable compounds by EPA Method 625, and heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2. Table 4.3.5-2 shows the results of the groundwater analyses and field measurements (pH, temperature, and conductivity). Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Table 4.3.5-3 summarizes this comparison.

4.3.5.2 Significance of Findings

This section discusses of the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates previous findings and trends identified in the Hargis report.

Significance of Metals in Groundwater

Two metals, chromium and lead, were detected in upper zone groundwater at concentrations exceeding the federal MCL criteria. In particular, chromium was noted at these levels in three wells, HM-22, HM-42, and HM-46 at concentrations ranging from 0.054 ug/L at HM-22 to 0.093 ug/L at HM-46. Lead was detected above the MCL criteria at HM-46 at a concentration of 0.070 mg/L. Hargis & Associates (1985) reported that chromium was observed in water from HM-46 and from the Radar Range seep. Metals in the upper member of the Paluxy aquifer at P-21U were reported at concentrations below federal MCL criteria.

The origin of chromium and lead in upper zone groundwater could be Landfill 2 or some unknown upgradient source. Chrome Pit 3 is hydraulically upgradient of portions of Landfill 2 and could represent a source of chromium and other heavy metals. RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 2, LANDFILL 2, AF PLANT 4, TEXAS TABLE 4.3.5-2.

	Method				Monitor Well	le11			
Date Sampled	Detection Limit	HM-2 2-20-86	HM-22 2-20-86	HM-40 2-25-86	HM-42 2-20-86	HM-43 2-20-86	HM-44 2-25-86	HM-46 2-20-86	P-21U 4-10-86
Field Parameters									
"		21.0	21.0	I	21.0	21.0	20.5	19.5	19.0
remperature (-0)		6.9	1.7	I	7.0	7.0	7.0	7.0	7.4
pn Conductivity (umhos)		710	2200	ł	570	550	640	700	520
Analytical Parameters ^a									
					i	1,00 0	ŝ	L10 0	100 / 100
Argenic 0	0.002-0.005	• * * 0 * 0	0.015	0.029	E S	0.004	2 Z	0.047	(70°) (MN
Barium	0.009	0.13	0.32	0.13	0.066	0.06/	0.0/4	87 O	0.00/
Cadaiua	0.002					0.028		0.03	
1 U B	COU.U	550 0	0.014	0.013	0.038	0.031	0,006	0.070	0.023
	200.0 0		E S	ĺ.	<u>E</u>	Ð	ę	9	Ð
setenty Setenting	0.002-0.004	2	•Q	NDe	90N	Ð	Ð	9	(CO.) ON
	0.002	0.010	0.019	0.004*	0,003*	0.011	0.005*	0.017	Q
Volatile Organics (ug/L)									
Method 601 (Halogenated Hydrocarbons)	Hydrocarbons)	~		,			•	•	•
Concentration Factor		1	1	1	1	I	1	I	-
Methylene Chloride	0.25	0.57	0.45	Ð	0.65	0.40	Ð	2	9
Tetrachloroethylene	0.03	Ð	2	60.0	Ð	Q	Ð	Ð	£
Method 602 (Aromatic Compounds)	a pounds)		•	·	•	•	-	-	-
Concentration Factor		1	1	1	1	7	-	-	-
Toluene	0.2	0.56	0.97	0.61	12	Q	0.73	9	12
Extractable Organics (ug/L)	L)								
Method 625 Acid Concentration Factor		1	1	1	1.2	1	1	2	1
Pheno1	1.5	Q	Q	Q	4	2	Ē	Ð	Q
Method 625 Base/Neutral Concentration Factor		1	ų	1	1.2	1	1	1	1
Big(2-ethvlbervl)phthalate		e	Ð	Ð	Ð	Ð	Ð	2	ŝ
Di-n-butyl phthalate	2.5	3BL	5.81.	Ð	Q	3BL	Q	QN	12BL

4-81

Errors may range up to 100 percent.

Spike recovery not within acceptable limits. Indicates interferent.
BL = Detected in reagent blank; background subtraction not performed.
= Data not collected, measurement omitted in field.
= Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.
ND = Not detected, values in parentheses are actual sample detection limits.

RADIAN

TEXAS **4**, LANDFILL 2, AF PLANT COMPARISON OF WATER QUALITY AT SITE 2. WITH APPLICABLE WATER QUALITY CRITERIA TABLE 4.3.5-3.

				Sampling	Location		
Parameter	Criteria*	HM-22	HM-42	HM-43	HM-43 HM-44	HM-46	P21U
METALS (mg/L)							
Chromium Lead Selenium	0.05 0.05 0.01	0.054	0.069			0.093 0.070	0.03
* See Table 4.2.1-1 for sou	for source of criteria.	ia.					

4-82

TRANSFE MARCHINE RANSFER TRANSFER TAGE

EXERCISE SERVICE SECOND

DODAR 200309 NUMBER

RADIAN

فخال والإعراب فالمتحدر فالمعامل فالمتراجع للتناوي للالتحد فالمركز والمتعاد فيليك والمعالية



Significance of Organic Compounds in Groundwater

Monitoring of upper zone wells in the vicinity of Landfill 2 has not detected significant numbers or levels of organic compounds. Results of the sampling and analysis of HM-102, although not reported in this section, confirm that no levels of organic compounds in excess of MCL criteria were found. This agrees with the data presented by Hargis and Associates (1985), which indicated no organic contamination of the upper zone groundwater at Landfill 2.

The IRP Phase II Stage 1 sampling and analysis program detected toluene (12 ug/L) in P-21U. The source of the toluene in the well is not certain, since no identified sources exist directly upgradient of the well. However, toluene was the only volatile organic compound detected in the well.

4.3.6 <u>Site 4, Landfill 4</u>

From about 1956 to the early 1980s Landfill 4 was used to dispose of clean construction rubble. Other types of wastes may have been disposed of at this site from about 1966 to 1973.

Activities at Landfill 4 consisted of installing one upper zone well (HM-101) and one Paluxy monitor well (P-20M) and collecting and analyzing groundwater samples. Upper zone monitor wells at this site include HM-5, 9, and 101. Figure 4.3.6-1 shows the locations of these sampling points. Samples were collected and analyzed for volatile organic compounds by EPA Methods 601 and 602, acid and base/neutral extractable compounds by EPA Method 625, and heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2.

4.3.6.1 Results of Investigation

The investigation at Landfill 4 focused on the construction of the Paluxy monitor well P-20M and the upper zone well HM-101 and the collection and analysis of groundwater samples. A discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

This landfill is located near the southwest boundary of the AF Plant 4 facility, in the area west of Meandering Road. The fill site was formerly part of the low area bordering Meandering Road Creek. The land surface generally slopes toward the creek (northwest) and elevations vary from about 660 feet to 640 feet MSL across the site.



 $\mathbf{X} \leftarrow \mathbf{X}$



Geologic Features

Monitor well HM-5 is located within the boundaries of Landfill 4. The lithologic log of this well (Hargis and Associates, 1985) indicates an upper zone thickness of 26 feet.

Additional information is provided by the lithologic logs of monitor wells HM-9, HM-101, and P-20M, which surround the site. These logs (Hargis and Associates, 1985 and Appendix E, this report) indicate that the upper zone thins to the southwest while the Walnut Formation thins to the north.

The Goodland Limestone was encountered only at monitor well P-20M, approximately 400 feet south of Landfill 4, where it was 38 feet thick. The Walnut Formation and the upper member of the Paluxy Sand were also fully penetrated at P-20M. The thicknesses of the two units here are 33 feet and 39 feet, respectively.

This description of geologic features at Landfill 4, while generally consistent with that of the Hargis and Associates (1985) report, adds information regarding the presence of the Goodland Limestone and the Paluxy Sand gained from the installation of two new monitor wells at this site.

Occurrence of Groundwater

and the second stands and the second states and the second states and the second states and the second states a

Groundwater at Landfill 4 occurs in both the materials of the upper zone and in the Paluxy Formation. Table 4.3.6-1 presents results of water level measurements. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-8) was discussed previously in Section 4.1.2. Water levels in the upper zone monitor wells at this site ranged from 650.67 feet MSL at HM-5 to 633.79 feet MSL at HM-9 during January and February 1986. As shown in Figure 4.1.2-8, groundwater flow is to the north-northeast, toward Meandering Road Creek. The water level in the middle member of the Paluxy



TABLE 4.3.6-1. GROUNDWATER ELEVATIONS AT SITE 4, LANDFILL 4 AF PLANT 4, TEXAS (FEBRUARY THROUGH APRIL 1986)

HM-5650.672-12-HM-9633.792-27-HM-101646.354-18-			
HM-5650.672-12-HM-9633.792-27-HM-101646.354-18-	TABLE 4.3.6-1.		
HM-9633.792-27-HM-101646.354-18-	Monitor Well	Elevation (feet MSL)	Date
HM-101 646.35 4-18-			2-12-
P-20 563.96 4-19-	HM-101	040.30	4-18-
	P-20	563.96	4-19-8
	······································		· <u> </u>



Formation, measured at P-20M, is 563.96 feet MSL. Groundwater levels in the middle Paluxy at AF Plant 4 are shown in Figure 4.1.2-3.

Groundwater Quality

Groundwater samples from HM-5, HM-9, HM-101, and P-20M were collected and analyzed for volatile organic compounds by EPA Methods 601 and 602, for acid and base/neutral extractable compounds by EPA Method 625, and for heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2. Table 4.3.6-2 gives results of the groundwater analysis and field measurements. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Table 4.3.6-3 summarizes the results of this comparison.

4.3.6.2 Significance of Findings

This section discusses the significance of the findings of the investigation, emphasizing the groundwater chemistry at the site.

Significance of Metals in Groundwater

Analyses of groundwater from HM-101 revealed concentrations of chromium and silver slightly in excess of federal MCL criteria. However, because of poor spike recovery (indicating the presence of an interferent) the analytical results for silver may be lower than the actual concentration. A similar situation exists for cadmium in HM-101. Although the observed concentration is below the MCL, the actual concentration may approach or exceed the MCL. A potential error is associated with the detected level of chromium (0.24 mg/L), since this value is less than five times the detection limit. Metals in HM-5, HM-9, and P-20M were reported at concentrations below federal MCL criteria.

	F GROUNDWA				
LANDEILL	4, AF PLAN		E ANALYSES .S	3, SITE 4,	•
					
Method Detection	HM-5	<u> </u>	Monitor Well HM-101	P-20M	P-20MD
Limit	2-12-86	2-27-86	4-18-86	4-19-86	4-19-86
	14.0	14.5	_	21.0	21.0
	6.4	6.8	7	7.6	7.6
	1600	3200	1200	440	440
0.002005	0.011	0.01*	ND(.02)	ND	ND
0.009	0.032	0.039	0.26	0.057	0.041
0.002	ND	ND 🕈	ND(.02)@	ND	0.003
0.005	0.02*	0.01*	0.2*	0.013	0.005
					0.013 ND
					ND
0.002	0.008*	0.020	0.130	ND	0.007
ydrocarbons)	1	1	1	1	1
0.12	0.16	ND	ND	ND	ND
ounds)	1	1	1	1	1
0.2	ND	ND	ND	7.4	11
	1	1	1	1	1
			1		
					1
ate 2.5 2.5	ND 3BL	ND	4 4BL	5	6 3
	Detection Limit 0.002005 0.009 0.002 0.005 0.001-0.002 0.002-0.004 0.002 0.002-0.004 0.002 0.002-0.004 0.002 0.002-0.004 0.002	Detection HM-5 Limit 2-12-86 14.0 6.4 1600 0.002005 0.011 0.009 0.032 0.002 ND 0.005 0.02* 0.0012 ND 0.002 0.048 0.0002 ND 0.002 ND 0.002 0.048 0.002 ND 0.002 0.008* ND 0.002 0.008* ND 0.12 0.16 pounds) 1 0.2 ND 1 0.2 ND 1 1 1 1 1 1 1 1 1 1 1 1 1	Detection Limit HM-5 2-12-86 HM-9 2-27-86 14.0 14.5 6.4 6.8 1600 0.002005 0.011 0.01* 0.009 0.032 0.039 0.002 ND ND 0.005 0.02* 0.01* 0.001-0.002 0.048 0.004* 0.002-0.004 ND ND 0.002 0.008* 0.020 0.002 0.008* 0.020 0.002 0.008* 0.020 1 1 1 0.12 0.16 ND ND ND ND 1 1 1 0.2 ND ND 1 1 1 0.2 ND ND 1 1 1 1 1 1 1 1 1 0.12 ND ND 1 1 1 .ste 2.5 3BL	Detection Limit HM-5 2-12-86 HM-9 2-27-86 HM-101 4-18-86 14.0 14.5 - 6.4 6.8 7 1600 3200 1200 0.002005 0.011 0.01* ND(.02) 0.009 0.032 0.039 0.26 0.002 ND ND ND ND(.02) 0.005 0.02* 0.01* 0.2* 0.001-0.002 0.048 0.004* 0.044 0.002 ND ND ND 0.002-0.004 ND ND ND 0.002 0.008* 0.020 0.13@ 0.002 0.008* 0.020 0.13@ 1 1 1 1 0.12 0.16 ND ND nounds) 1 1 1 1 1 1 1 0.2 ND ND ND 1 1 1 1 1 1	Detection Limit HH-5 2-12-86 HH-9 2-27-86 HH-101 4-18-86 P-20M 4-19-86 14.0 14.5 - 21.0 6.4 6.8 7 7.6 1600 3200 1200 440 1200 440 0.002005 0.011 0.01* ND(.02) ND 0.002 ND ND@ ND(.02) ND 0.002 ND ND@ ND(.02) ND 0.005 0.02* 0.01* ND(.02) ND 0.005 0.02* 0.01* 0.2* 0.013 0.001-0.002 0.04* 0.04* 0.04* 0.04* 0.002-0.004 ND ND ND ND 0.002 0.008* 0.020 0.13@ ND 1 1 1 1 1 1 0.12 0.16 ND ND ND ND iydrocarbons) 1 1 1 1 1 1 1



122 C

333333

TABLE 4.3.6-3. COMPARISON OF WATER QUALITY AT SITE 4, LANDFILL 4, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

RADIAN		
	OMPARISON OF WATER QUALITY EXAS WITH APPLICABLE WATER	AT SITE 4, LANDFILL 4, AF PLA QUALITY CRITERIA
Parameter	Criteria*	Sampling Location HM-101
METALS (mg/L)		
Chromium Silver	0.05 0.05	0.2 0.13
* See Table 4.2.2	-1 for source of criteria.	



Significance of Organic Compounds in Groundwater

Monitoring of wells in the vicinity of Landfill 4 has not detected significant numbers or levels of organic compounds. These findings agree with the data presented by Hargis and Associates (1985).

4.3.7 Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; Site 8, FDTA 5)

These three sites have been combined into one zone. Descriptions are given below.

Site 13, Die Pits

These pits were used for disposal of chromate sludges, metal solutions, and other chemical wastes until 1962, when the site was graded and the entire die yard was paved. The site of the original pits was excavated in 1983-84. Soils from other parts of the die yard were not analyzed at that time.

Site 11, Chrome Pit 2

Miscellaneous liquid and solid wastes, in addition to chromate solutions, were reportedly disposed of at this site. The actual location of this site could not be accurately confirmed during the Phase I (Records Search) investigation.

Site 8, FDTA 5

Previous groundwater analyses indicated elevated levels of organic compounds at this site, which consisted of a shallow pit into which waste fuels, oils, or chemicals were deposited and ignited for training exercises. It is located in the die yard area south of Warehouse 1 and has been graded and paved.

The IRP Phase II Stage I activities at Zone 1 consisted of the collection and analysis of groundwater samples. Monitor wells at this site include HM-11, 12, 24, 25, 28, 60, 77, and P-1. Figure 4.3.7-1 shows the locations of these sampling points and the physical features of the sites.



É





Additionally, soil boring SB-5 was drilled in the area of FDTA 5. Based on color, odor, and OVA readings, soil samples were collected from split spoons, as described in Section 3.1.3, and submitted for analysis of hazardous constituents, as reported in Section 3.4. Originally, the purpose of the borehole was to allow for the installation of an upper zone monitor well; however, no water was noted in the borehole even after it had remained open for two days. Therefore, the borehole was grouted from the total depth to the land surface.

4.3.7.1 Results of Investigation

The investigation conducted at Zone 1 focused on the collection, analysis, and interpretation of groundwater quality samples. Results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

The Die Pits are located just southwest of Warehouse No. 1. The ground surface is flat and the surface elevation is about 652 feet MSL.

Chrome Pit 2 is located about 600 feet east of the Die Pits. The ground surface is flat at this site also and the surface elevation is 652 feet MSL.

FDTA 5 is located just south of Chrome Pit 2. The ground surface is flat and the surface elevation is 652 feet MSL.

Geologic Features

Monitor wells HM-3a, HM-3b, and HM-4a were installed within the Die Pits site but have since been destroyed by excavation. Drill logs of these wells indicate an upper zone composed of about ten feet of fill material overlying the Goodland Limestone.

Monitor wells HM-12, HM-24, HM-25, and HM-28 are located near the Die Pits. Examination of the lithologic logs found in the Hargis and Associates report (1985) indicates an upper zone thickness ranging from five to 28 feet and a range of Goodland Limestone thickness from eight feet at HM-28 to 28 feet at HM-12.

This description of the geologic features at the Die Pits site differs somewhat from that of the Hargis and Associates report (1985). The Hargis report gave the thickness of the upper zone as 18 to 40 feet and that of the Goodland Limestone as 6 to 18 feet. The Walnut Formation was encountered during the construction of all monitor wells at this site.

Monitor well HM-77 is located at Chrome Pit 2. The lithologic log (Hargis and Associates, 1985) shows an upper zone thickness of 30 feet. The upper zone is directly underlain by the Walnut Formation.

Upper zone monitor well HM-25 is located due west of FDTA 5. The lithologic log of this well (Hargis and Associates, 1985) indicates an upper zone thickness of 29 feet. No Goodland Limestone was encountered.

Occurrence of Groundwater

Groundwater in Zone 1 occurs in both the materials of the upper zone and in the Paluxy Formation. Table 4.3.7-1 gives results of water level measurements. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-7) was discussed previously in Section 4.1.2. Water levels in the upper zone in this area ranged from 644.42 feet MSL at HM-11 to 629.94 feet MSL at HM-77. As shown in Figure 4.1.2-7, groundwater flow in the upper zone is to the northeast in Zone 1. The groundwater elevation at P-1 was measured at 558.44 feet MSL. Groundwater flow in the Paluxy is discussed in Section 4.1.2.



TABLE 4.3.7-1. GROUNDWATER ELEVATIONS AT ZONE 1, (SITES 13, 11 and 8) AF PLANT 4 TEXAS, (FEBRUARY 1986)

112012 A 2.612 A

Monitor Well	Elevation (feet MSL)	Date
HM-11	644,42	2-10-86
HM-12	641,82	2-14-86
HM-24	642,52	2-19-86
HM-25	634.89	2-11-86
HM-28	632,52	2-6-86
HM-60	634.75	2-6-86
HM-77	629.94	2-6-86
P-1	558,44	2-6-86

4-96

RAZIAN

Groundwater Quality

Groundwater samples from HM-11, 12, 24, 25, 28, 60, 77, and P-1 were collected and analyzed for volatile organic compounds by EPA Methods 601 and 602, for acid and base/neutral extractable organic compounds by EPA Method 625, and for heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1 and 270.2.

Table 4.3.7-2 shows the results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling events were compared to federal standards and guidelines. Table 4.3.7-3 summarizes the results of this comparison.

4.3.7.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified in the Hargis report.

Significance of Metals in Groundwater

Two metals, arsenic and chromium, were reported at concentrations in excess of federal MCL criteria. Arsenic was detected in water sampled from HM-25 at 0.140 mg/L. Chromium was detected in water from HM-28 at 0.120 mg/L.

The Hargis report detected chromium at approximately the same concentration but arsenic was below the instrument detection limit. TABLE 4.3.7-2. RESULTS OF GROUNDWATER SAMPLE ANALYSES, ZONE 1 (SITES 13, 11 AND 8), AF PLANT 4, TEXAS

Dete Li 0.002-0 0 0.001-0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	HM-11 2-10-86 7.3 660 7.3 660 0.040 0.006 0.006 0.006 0.006 0.006	HM-12 2-14-86 19.5 6.8 6.8 21.00 21.00 0.017 0.017 ND ND	HM-24 2-19-86		HM-28 2-6-86	HM-28D 2-6-86	HM-60	HM-77	P-1
ield Parameters Temperature (°C) pH Conductivity (umhos) Conductivity (umhos) alytical Parameters ^a rela (mg/L) Arsenic Arsenic Cadmium Concolo 0.002 Cadmium Concolo 0.002 Cadmium Concolo 0.002 Head Selenium Selenium Concentration Factor Vinyl Chloride Concentration Factor Wethylene Chloride Concolo 0.25 Methylene Chloride Concolo 0.25 Methylene Chloride Concolo 0.25 Methylene Chloride Concolo 0.25 Methylene Chloride Concolo 0.25	18.0 7.3 660 660 0.040 0.006 0.028 0.028 0.028 0.028 0.028	19.5 6.8 21.00 0.007* 0.017 0.017 0.018		7-11-80			7-0-80	00-0-7	2-6-86
Temperature ('U) pH Conductivity (umhos) Analytical Parameters ^a Metala (mg/L) Arsenic Barium Concentration Factor Cadmium 0.002-0.005 Barium 0.002-0.002 Chromium 0.002-0.004 0.002-0.004 Silver Volatile Organics (ug/L) Method 601 (Halogenated Hydrocarbo) Concentration Factor Vinyl Chloride 0.25 Methylene Chloride 0.25	7.3 660 0.040 ND 0.0064 0.0064 0.0064 0.0064	21.00 0.007* 0.017 0.010 0.010 ND	22.0	16.5	17.0	17.0	19.5	19.0	21.0
PH Conductivity (umhos) malytical Parameters ^a etals (mg/L) Arsenic 0.002-0.005 Barium 0.002 Chromium 0.001-0.002 Chromium 0.001-0.002 Chromium 0.002-0.004 Silver 0.002-0.004 Silver 0.002-0.004 Silver 0.002 Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Winyl Chloride 0.25 Methylene Chloride 0.25	0.040 0.040 0.006 0.006 0.006 0.006	21.00 21.00 0.007* 0.017 0.019 ND		5 . 9 9	7.0	7.0	7.0	7.2	7.4
nalytical Parameters ^a Atsenic (mg/L) 0.002-0.005 Barium 0.002 Chromium 0.002 Chromium 0.002 Chromium 0.002 Lead 0.001-0.002 Selenium 0.002-0.004 Silver 0.002-0.004 Silver 0.002 Olatile Organics (ug/L) Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	0.040 0.040 ND 0.006* 0.028 ND 0.006*	0.007* 0.017 ND 0.018* 0.010 ND	3100	950	1000	1000	920	850	720
etals (mg/L) Arsenic 0.005 Barium 0.009 Barium 0.002 Catanum 0.002 Chromium 0.002 Chromium 0.002 Fead 0.002 Mercury 0.002 Selenium 0.002-0.004 Silver 0.002 Selenium 0.002-0.004 Silver 0.002 Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Winyl Chloride 0.25 Methylene Chloride 0.25	ND 0.040 ND 0.006* 0.028 ND ND 0.006*	0.007* 0.017 ND 0.018* 0.010 ND							
Arsenic v.v.v. Barium 0.009 Barium 0.002 Chromium 0.002 Chromium 0.001-0.002 Lead 0.0002 Mercury 0.002-0.004 Silver 0.002-0.004 Silver 0.002 Outile Organics (ug/L) Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	0.040 ND 0.006* 0.028 ND ND 0.006*	0.017 ND 0.008* ND ND	Ę	0.14	QN	NA	0.01*	Q	Q
Concentration (0.002 Chromatum 0.002 Chromatum 0.002 Lead 0.0002 Mercury 0.002 Selentum 0.002 Silver 0.004 Silver 0.002 Outoloo Datile Organics (ug/L) Method 601 (Helogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	0,006* 0,028 ND ND 0,006*	ND 0.008* 0.010 ND ND	060.0	0.47	0.054	NA	0.17	0.099	0.076
Chromium 0.005 Lead 0.001-0.002 Hercury 0.0002 Selenium 0.002-0.004 Silver 0.002-0.004 Silver 0.002 Olatile Organics (ug/L) Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	0, 006* 0, 028 ND ND 0, 006*	0,008* 0,010 ND ND	Ð	Ð	Q	NA	Q	QN	Q
Lead 0.001-0.002 Mercury 0.0002 Seleníum 0.002-0.004 Silver 0.002-0.004 Diatile Organics (ug/L) 0.002 Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	0, 028 ND ND 0, 006#	0.010 MD MD	0.02*	Q	0.12	NA	0.048	Ð	0.007*
Mercury 0.0002 Selenium 0.002 Silver 0.004 Silver 0.002-0.004 Outile Organics (ug/L) Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	ND UN 0.006*	99	0.027	0.035	0.003*	N	0.005*	0.003*	9 CIN
Selenium 0.002-0.004 Silver 0.002 Datile Organics (ug/L) Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	ND 0.0064	Đ :	Ð	Ð	Ð	NA	Ð	Ð	0.0002*
Silver 0.002 Silver 0.002 Method 601 (Halogenated Hydrocarbo Concentration Factor 0.18 Vinyl Chloride 0.25 Methylene Chloride 0.25	0.006*		Ð	Ð	Ð	NA	Q	Ð	• 00.03
olatile Organics (ug/L) Method 601 (Halogenated Hydrocarbo Concentration Factor Vinyl Chloride 0.18 Methylene Chloride 0.25		0.00/*	0.013	Q	0.003*	NA	Ð	Ð	2
Concentration Factor Vinyl Chloride 0.18 Methylene Chloride 0.25	ns)								
	1	1	1	25	50		25	1	1
	QN	Ð	Ð	16	QN	NA	QN	Ð	Ð
	9	0.92	2.5	Q	Ð	NA	Q	Q	QN
trans-1.2-Dichloro- 0.10	**UN	Ð	Ð	**QN	**QN	NA	Đ	**ON	Q
ethylene				I	!		!	Í	ţ
	Ð	ę	37	2	2	N :		2 (
1,2-Dichloropropane 0.04	Ð	ę	9.6	13		NA :		Ð	2 9
Trichloroethylene 0.12	26	0.23	7.0	2	600	N	400	7. 5	
Tetrachloroethylene 0.03	1.3	g	0.98	Ð	Ð	NA	Ð	ÎN	(IN
Chlorobenzene 0.25	Ð	Ø	Q	4700	Q	NA	2	2	
1.2-Dichlorobenzene 0.15	Ð	£	Ð	190	g	Ð	AN	Q	Û
	Q	Ð	Ð	870	Ð	M	Ð	2	9
Method 602 (Aromatic Hydrocarbons)							:		
Concentration Factor	1	1	1	25	10		10	1	1
Benzene 0.2	Ð	Ð	Ð	470	Ð	NA	QN	Ð	Ŷ
	4.1	1.2	23	Ð	9	M	9	9	Ŷ
enzene	Ø	Q	Q	0006	ę	NA	Q	Ð	ę
enzene	Q	Q	Ð	370	9	M	Ð	ę	Ð
1.4-Dichlorobenzene 0.3	Ð	Q	Q	730	Ð	M	Ð	9	£

(Continued)

and and a second property property and the property and a second property of the

S

γ.

Y. Y. K.

RAPIAN

4-98

(Continued) TABLE 4.3.7-2.

	Mathod					Monitor Well				
	Detection	NM-11	HM-12	HM-24	1	HM-28	HM-28D	09-MH	HM-77	P-1
Date Sampled	Limit	2-10-86	2-14-86	2-19-86	2-11-86	2-6-86	2-6-86	2-6-86	2-6-86	2-6-86
Extractable Organics (ug/L)	6									
Method 625 Acid Concentration Factor		1	1	1	1	1	1	1	1	1
Phenol 2-Chlorophenol	1.5 3.3	<u>9</u> 9	99	99	4 10	£ £	9 9 9	89	M 2	ND 7
Method 625 Base/Neutral Concentration Factor		1	1	1	1	1	1	1	1	1
1 2-Dichlorohenzene	1.0	Qr	Ð	Q	210		Ð	9	Q	Ð
1,2-Vichiorovensene 1	4.4	9	2	9			Ð	Ð	Ð	N
N-nitrosodiohenvlamine	e 1.9	Ð	9	e			9	9	Ð	ÛN
Rig(2-athulhavul) phth	alate 2.5	6BL	Ð	80			m	Ð	Ð	IJ
Di-n-hutvl phthalate 2.5	2.5	2	Q	3BL		Ð	Ð	9	e	Ð
Naphthalene	1.6	Ð	Q	Q			Ð	Q	Ð	IN
Hydrocarbon Fuels (mg/L)	1	NA	NA	W	NA	NA	NA	NA	Ð	NA
Oil and Grease (mg/L)	1	NA	NA	NA	NA	NA	M	NA	Q	N

£19,618,618,676,676,656,656,656,656,756,756,796,796,796,796,796,796

4-99

ND = Not detected. NA = Not analyzed. * = Value less than five times detection limit.

Errors may range up to 100 percent.

e = Spike recovery not within acceptable limits.

Indicates interferent.

BL = Detected in reserve blank; background subtraction not performed. ** = Compound not rengirmed by second-column analysis. * = Compounds not] sted were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection: limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

Duplicate sample. " Q

RADIAN

and wat shall be a state of the state of the

			Came S	Samuline Location	rion	
Parameter	Criteria*	HM-11	HM-24	HM-25	HM-28	09-MH
METALS (mg/L)						
Arsenic Chromium	0.05 0.05			0.14	0.12	
ORGANIC COMPOUNDS (ug/L)						
Vinyl Chloride	9		9*6	16 13		
Trichloroethylene Chlorobenzene	60 5	26	7	4700	600	400
Benzene 1.4-Dichlorobenzene	5 750			470 870		

COMPARISON OF WATER QUALITY AT ZONE 1, (SITES 13, 11, AND 8), AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA TABLE 4.3.7-3.

A LAND

DG

See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.

NASONAL NAVANA PARANA PARANA NAVANA NAVANA NAVANA NAVANA NAVANA NAVANA

1.1.1.2.2.2.5.1.5.1

4-100

-}*

RADIAN

A STATE AND A STATE AND A STATE AND A STATE



Significance of Organic Compounds in Groundwater

A number of organic compounds exist in upper zone groundwater at levels in excess of federal MCL criteria. Trichloroethylene was detected at HM-11, HM-24, HM-28, and HM-60. These levels ranged from just over the MCL criteria at HM-24, where the concentration of TCE was 7 ug/L, to 600 ug/L at HM-28. Groundwater sampled from HM-25 contained levels of vinyl chloride, 1,2-dichloropropane, chlorobenzene, benzene, and 1,4-dichlorobenzene in excess of federal MCL criteria. In addition, 1,2-dichlorobenzene was detected at 190 ug/L at HM-25. An unknown compound with a retention time near that of trans-1,2-dichloroethylene was detected at HM-25 and HM-28.

These findings are generally consistent with the results of earlier analyses conducted by Hargis and Associates (1985) with the exception of monitor well HM-25. The Hargis report identified trans-1,2-dichloroethylene in concentrations ranging from 40 ug/L (August, 1984) to 260 ug/L (June, 1983). In addition, instead of the high concentrations of vinyl chloride Radian found at HM-25, the Hargis study reported this substance as not detected. Since the Hargis study did not analyze for 1,4-dichlorobenzene, no comparisons could be made.

RADIAN

4.3.8 <u>Site 15, Fuel Saturation Area 2</u>

Between the 1970s and early 1980s, the soils at Fuel Saturation Area 2 became saturated by fuels from leaking buried fuel lines. IRP Phase II Stage I activities at Fuel Saturation Area 2 consisted of drilling three boreholes along the length of a buried fuel line, collecting a groundwater sample from an existing upper zone monitor well (HM-80), and observing the monitor well for the existence of a fuel lens. Monitor well HM-80 was sampled for volatile organics by EPA Methods 601 and 602 and for hydrocarbon fuels by EPA Method 418.1. Soil samples were collected and analyzed for halogenated volatiles and aromatics by EPA Method 8010 and 8020 and for hydrocarbon fuels by EPA Method 418.1. Figure 4.3.8-1 shows the locations of groundwater and soil sampling points along with the physical features of the site.

4.3.8.1 Results of Investigation

The investigation focused on the collection and analysis of samples and the interpretation of data on soil and groundwater quality. Results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

Fuel Saturation Area 2 is located just northwest of the Paint Shop (Building 176). The ground surface, essentially flat, has an elevation of approximately 648 feet MSL.

Geologic Features

Monitor well HM-80 and soil borings SB-1, -2, and -3 are located in the vicinity of Fuel Saturation Area 2. The lithologic log of HM-80 (Hargis and Associates, 1985) reveals the presence of approximately 18 feet of Goodland Limestone overlain by 6 feet of concrete and fill. SB-1, -2, and -3 are located just west of the site. Lithologic logs of these soil borings



44° 44° 44° 44° 44° 44° 44°

K.

500

4-103



(Appendix E, this report) reveal an upper zone thickness of four to five feet at all three soil boring locations. Goodland Limestone was encountered at all three soil boring locations, ranging in thickness from 14 feet at SB-2 to 10 feet at SB-3. About three feet of Walnut Formation was encountered during the drilling of SB-1, -2, and -3.

Occurrence of Groundwater

Because the upper zone is absent at this location, the uppermost occurrence of groundwater is found in the Goodland Limestone at this site. The water level elevation in monitor well HM-80 was 638.99 feet MSL. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-5) was discussed previously in Section 4.1.2. However, because upper zone sediments are not present at well HM-80, water level contours for the upper zone are not drawn through the area of FSA 2. Groundwater flow in this area is to the north-northwest, toward Meandering Road Creek.

Groundwater Quality

Groundwater samples were collected from monitor well HM-80 and analyzed for volatile organic compounds by EPA Methods 601 and 602 and for hydrocarbon fuels by EPA Method 418.1. Table 4.3.8-1 gives results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

4.3.8.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, the present report recognizes and incorporates those findings and trends identified by the Hargis report.

and all all a last a



Date Sampled	Method Detection Limit	<u>Monitor Well HM-80</u> 2-5-86
Field Parameters		
Temperature (°C)		19.5
рН		6.8
Conductivity (umhos)		2000
Analytical Parameters ^a		
Volatile Organics (ug/L)		
Method 601 (Halogenated Hydrocan Concentration Factor = 1	rbons)	
1,2-Dichloropropane	0.04	1.7
Method 602 (Aromatic Compounds) ¹ Concentration Factor = 1	0	
Hydrocarbon Fuels (mg/L)	1	ND

TABLE 4.3.8-1. RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 15, FUEL SATURATION AREA 2, AF PLANT 4, TEXAS

(4%) 18% (4%) 18% 18% 18% 18% 18%

ND = Not detected.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4. b = No compounds were detected by this analysis.

Significance of Organic Compounds in Groundwater

IRP Phase II Stage 1 sampling and analysis of the single monitor well at this site (HM-80) detected only 1.2-dichloropropane, which was present at a level below the MCL criteria. No fuel lens was observed in the well. Hydrocarbon fuels were below the instrument detection limit in the groundwater sample. These results are consistent with those of the previous sampling and analysis conducted by Hargis and Associates.

Significance of Organic Compounds in Soils

Table 4.3.8-2 presents the results of soil sample analyses for this site. Organic compounds were detected at low levels except for soil sampled at location SB-2 at 5 to 6 feet. This soil sample had a concentration of hydrocarbon fuels of 4600 mg/kg. However, the concentration decreased at increased depths.
TABLE 4.3.8-2. RESULTS OF SOIL SAMPLE ANALYSES, SITE 15, FUEL SATURATION AREA 2, AF PLANT 4, TEXAS

			So	Soil Boring (depth - feet)	depth - fee	(t)		
	Method	SB-1-A	SB-1-C	SB-2-A	SB-2-B		SB-3-A	10
De	Detection	(4-2)	(14-14.5)	(2–6)	(10-11)	(10-11)	(01-6)	
Date Sampled	Limit	1-24-86	1-24-86	1-24-86	1-24-86	1-24-86	1-26-86	
Analutical Parameters								
Hydrocarbon fuels (mg/kg)	4-6	QN	QN	4600	QN	QN	Q	
Volatile Organics (mg/kg)								
Mathod 8010 (Halosenated	4 Hvdrocarbons)	rhons)						
Concentration Factor		12.5	12.5	12.5	12.5	12.5	12.5	
1 1-Dichlorosthylene	0.013	ÛZ	Q	0.012	Q	Q	Q	
	0.005	ND(2.8)	ND(2.8)	Q.	Q	Q	Ð	
Tetrachloroethvlene	0,003	ND (1.6)	ND (1.6)	Ð	ND(2.1)	ND (2.1)	ND(2.1)	
Trichloroethylene	0.019	<u>N</u>	QN	Ð	ND(1.6)	ND(1.6)	ND(1.6)	
Method 8020 (Aromatic Compounds	ompounds)							
Concentration Factor	• • •	125	125	125	125	125	125	
Toluene	0.0002	ND (0,68)	ND (0.68)	ND (0.68)	ND (0.68)	ND (0.83)	ND (0.83)	
Benzene	0.0002	QN	Ð	QN	Q	ND (0.11)	ND (0.11)	
Et hyl benzene	0.0002	Ð	Ð	Ð	Ð	ND (0.14)	ND(0.14)	
and and and and and and and		thodog are	actual gamo	la detection	limita			
ND = Not detected, values in parentheses are actual sample detection limits	in paren	theses are	actual samp]	le detection	limits.			

ND = Not detected, values in parentneses are actual sample verection timite

Sample detection limits are the method detection limit Method detection limits times the concentration factor except when species of interest were found in the extractant. The resultant detection limits are reported in parentheses in mg/kg. Compounds not listed were not detected. are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4. 41 65

RADIAN

4-107

RADIAN

4.3.9 Site 20, Wastewater Collection Basins

This site consists of two concrete-lined waste basins, each with an approximate capacity of 85,000 gallons. The basins are used to collect and settle suspended solids from plant wastewaters and have been in use since approximately 1966. The Phase I investigation determined that several spills of vapor degreaser from tanks in the Process Building occurred here. Spilled liquids may have flowed to the basin via floor drains in the Process Building. Phase II Stage I studies were conducted to determine if underlying soil and upper zone groundwater have been affected by the spills.

Activities at Site 20 consisted of installing an upper zone monitor well (HM-104) southeast of the basins, collecting groundwater samples from four existing monitor wells and the newly installed well, and determining the location and depth of the buried sanitary and industrial waste lines. Soil samples from HM-104 were not submitted for chemical analysis. The need for soil sampling along the waste lines, based on groundwater analysis, was also evaluated. Upper zone monitor wells at this site include HM-31, 47, 69, 70, and 104. Figure 4.3.9-1 shows the locations of the sampling points. Samples were collected and analyzed for volatile organic compounds by EPA Methods 601 and 602, for acid and base/neutral extractable compounds by EPA Method 625, for heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2, for oil and grease by EPA Method 413.2, and for hydrocarbon fuels by EPA Method 418.1.

The sanitary sewer line runs on an east-west line through the site. The storm drain is on a northwest-southeast line located approximately 75 feet south of the Wastewater Collection Basins. The depths of these lines are not specified on the underground utilities maps provided by General Dynamics. There is no reason to suspect that the sanitary sewer or the storm drain would carry high levels of hazardous contaminants. Although contaminants were present in the existing site monitor wells, the levels do not suggest massive releases of contaminants; therefore, no soil sampling along these waste lines is deemed necessary.



and a stand with the stand of the



4.3.9.1 Results of Investigation

The investigation focused on the installation of the upper zone monitor well, HM-104, and on the collection and analysis of groundwater samples. Discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

This site is located just south of Building 181. The ground surface is almost flat, but slopes very slightly to the south. The surface elevation on the south side of the concrete-lined basins is about 652 feet MSL.

Geologic Features

Monitor well HM-47 is located about 50 feet east of the Wastewater Collection Basins. The lithologic log of this well (Hargis and Associates, 1985) indicates an upper zone thickness of 28 feet. The upper zone consists primarily of clayey silt and sand and gravel. The Goodland Limestone is two feet thick. Six feet of Walnut Formation were encountered.

The log of well HM-104, installed during this study, indicates an upper zone thickness of ten feet underlain by 26 feet of Goodland Limestone. About three feet of Walnut Formation were encountered.

This description of the subsurface features at the Wastewater Collection Basins is consistent with that of the Hargis and Associates report (1985).

Occurrence of Groundwater

Water levels in upper zone wells are reported in Table 4.3.9-1. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-7)



Checkler Corners

29.0.9.029729

7 • F

TABLE 4.3.9-1. GROUNDWATER ELEVATIONS AT SITE 20, WASTEWATER COLLECTION BASINS, AF PLANT 4, TEXAS (FEBRUARY THROUGH APRIL 1986)

Monitor Well	Elevation (feet MSL)	Date
HM-31	_	(not measured)
HM-47	620,95	2-6-86
HM69	630,97	2-21-86
HM-70	624.72	2-19-86
HM-104	622,59	4-10-86

የሚሰሩ ግድ ማስት በትርስ የሚሰር በባለ የግድ እንዲሆን እን እንዲሆን እንዲሆን እንደ አልሆን እንዲሆን እ

2020

was discussed previously in Section 4.1.2. During January to August 1986, water levels in monitor wells ranged from 630.97 feet MSL at HM-69 to 622.59 feet MSL at HM-104. As shown in Figure 4.1.2-7, groundwater flow is to the southeast in this area.

Groundwater Quality

Groundwater samples were collected from HM-31, 47, 69, 70 and 104 and analyzed for heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270, for volatile organic compounds by EPA Methods 601 and 602, for acid and base/neutral extractable compounds by EPA Method 625, for oil and grease by EPA Method 413.2, and for hydrocarbon fuels by EPA Method 418.1. Table 4.3.9-2 gives results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Table 4.3.9-3 summarizes the results this comparison.

4.3.9.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

Significance of Metals in Groundwater

Three metals (arsenic, chromium, and lead) were detected at this site in concentrations exceeding federal MCL criteria. All were detected in samples taken either from HM-70 or from a duplicate sample from that well. Arsenic was detected at HM-70 at a concentration of 0.062 mg/L. Chromium was







RADIAN									
RADIAN TABLE Date Sampled Field Parameters Temperature (°C) pH Conductivity (unho	4.3.9-2.	RESUL:	IS OF GROUN	DWATER S	AMPLE ANAL ANT 4. TEX	YSES, SITE As	20, WASTE	WATER	
	Me	thod							
	Dete Li	ction mit	HM-31	HM-47	HM-47D	Monitor W HM-69	e11 HM-70	HM-70D	HM-104
Date Sampled			2-14-86	2-6-86	2-6-86	2-21-86	2-19-86	2-19-86	4-10-86
Field Parameters									
Temperature (°C)			22.0	19.0	19.0	19.5	23.0	23.0	22.0
pH Conductivity (umbo	B)		950	950	950	710	840	840	1600
Analytical Paramete	rs ^a								
Metals (mg/L)									
Arsenic Barium	0.002-	0.005	ND 0.031	ND 0.51	ND 0.53	ND 0.054	ND 0.092	0.062 0.082	ND 0.045
Cadmium		0.002	0.031 ND	ND	ND	ND	ND	ND	ND
Chromium Lead	0.001-	0.005	ND 0.004*@	0.02* ND	0.02* ND	0.032 0.045	0.075 0.040	0.064 0.068	0.009* ND
Mercury	0	.0002	ND	ND	0.0003*	ND	ND	ND	0.002
Selenium Silver	0.002-	0.004	ND CONTRACTOR	ND 0.003*	0.003* ND	ND 0.011	ND 0.01≭	ND 0.011	0.003* 0.005*
Volatile Organics (
	-	• • • •	>						
Method 601 (Haloger Concentration Fac		ocarbor	18) 100	1000	1000	1	1000	1000	1
Methylene Chlorid	2	0.25	ND	ND	ND	0.63	ND	ND	ND
Trichlorofluorome		0.40	ND	ND	ND	19	ND	ND	ND
trans-1,2-Dichlor Trichloroethylene	pethylene	0.10 0.12	ND** 2400	ND** 11000	ND** 12000	ND** 3.2	ND 11000	ND 11000	ND ND
Chlorobenzene		0.25	ND	310	770	ND	ND	ND	ND
Method 602 (Aromat: Concentration Fac		ds)	100	1000	1000	·1	1	1	1
		<u> </u>							
Toluene 1,1,2,2-Tetrachlo:	roethane	0.2 0.03	ND ND	ND ND	ND ND	ND 0.18	ND ND	ND ND	0.74 ND
Chlorobenzene	· · · · · · · · ·	0.2	ND	740	710	ND	ND	ND	ND
Method 624							-		
Concentration Fact							1		
Tetrachloroethyle	he	1.9	NA	NA	NA	NA	6500	NA	NA
Extractable Organic:	s (ug/L)								
Method 625 Acid Concentration Fac			1						
Phenol		1.5	2	NA	NA	NA	NA	NA	NA
Method 625 Base/Net Concentration Fact			1						
bis(2-ethylhexyl)		2.5	2	NA	NA	NA	NA	NA	NA
Hydrocarbon Fuels (1	ND	ND	ND	ND	ND	ND	ND
	-								
Oil and Grease (mg/)	ر ما	1	4	1*	1*	ND	1*	ND	1*
 a = Compounds not times the conce and 4.2.2-4. NA = Not analyzed. ND = Not detected. * = Value less than # = Spike recovery * = Compound not compound to the compound of the compound to the compound tothe compound to the compound to the compound to the compound tot	entration : n five time not within	factor. es dete n accep	. Method o ection lim: ptable lim:	detection it. Erro its. Ind	limits ar rs may ran	e found in ge up to 1	Tables 4.	2.2-2. 4.2	
D = Duplicate.									



RADIAN					
TABLE 4.3.9-3. COMPARIS BASINS,	SON OF WATER QUAI AF PLANT 4, WITH				
Parameter	Criteria*	Samp HM-31	oling Loca HM-47	tion HM-70	
METALS (mg/L)					
Arsenic Chromium Lead	0.05 0.05 0.05			0.062 0.075 0.068	
ORGANIC COMPOUNDS (ug/L))				
Trichloroethylene Chlorobenzene Tetrachloroethylene	5 60 8	2400	12000 770	11000 6500	
				·	

found in groundwater from HM-70 and in a duplicate sample from that monitor well at concentrations of 0.075 mg/L and 0.064 mg/L, respectively. Lead was detected at a concentration of 0.068 mg/L in a duplicate sample from HM-70. Hargis & Associates (1985) detected similar levels of arsenic and chromium from HM-70 in 1985, but detected lead at levels below federal MCL criteria. Chrome Pit 2 is a possible upgradient source of chromium and other heavy metals.

Significance of Organic Compounds in Groundwater

An unknown compound with the same retention time as trans-1,2dichloroethylene was detected at high levels on the first column from groundwater samples from HM-31 and 47. Hargis & Associates (1985) reported high levels of trans-1,2-dichloroethylene in these wells. Groundwater samples from monitor wells HM-31, 47, and 70 revealed high concentrations of trichloroethylene, ranging from 2400 ug/L in HM-31 to 12000 ug/L in HM-47. Chlorobenzene was also detected in HM-47 at a concentration of 770 ug/L. Tetrachlorethylene was detected in HM-70 at a concentration of 6500 ug/L. These results are generally consistent with those of Hargis & Associates (1985), except for monitor well HM-47, which was found to have TCE concentrations ranging from 27,000 ug/L in 1984 to 3,900 ug/L in 1985. The probable source of the TCE found in upper zone groundwater at this site is the vapor degreaser spilled from tanks in the Process Building located nearby and upgradient of the well.

RAPIAN

4.3.10 Site No. 16, Fuel Saturation Area 3

From the mid-1970s until the early 1980s (CH2M Hill, 1984), this site reportedly became saturated by fuels from leaking buried fuel lines.

IRP Phase II Stage 1 activities at this site included the collection of groundwater samples from monitor well HM-78 to determine the presence of a fuel lens. Figure 4.3.10-1 shows the location of well HM-78.

4.3.10.1 Results of Investigation

The IRP Phase II Stage 1 investigation focused on the collection and analysis of groundwater samples from HM-78. Discussions of the results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

This site is located just southwest of Building 142. The land surface elevation varies from about 642 feet to 644 feet MSL, north to south across the site.

Geologic Features

Upper zone monitor well HM-78 is located within the boundaries of Fuel Saturation Area 3. The lithologic log (Hargis and Associates, 1985) indicates an upper zone thickness of about 20 feet. At the base of the upper zone, four feet of clayey limestone gravel were encountered. About four feet of Walnut Formation were encountered at this location, and Goodland Limestone was absent.



4-117

PLACEDCING

TOWNS HILLS DIGTON DIMIN NAVAS

No. of the second second

1535555554

2222222222

Occurrence of Groundwater

Limited water level data from upper zone wells near this site suggest a groundwater elevation between 620 and 624 feet MSL in the vicinity of the site and a generally northwestward flow direction (see Figure 4.1.2-5).

Groundwater Quality

Groundwater was sampled at well HM-78 on 5 February 1986. The sample was analyzed for volatile organic compounds (EPA Methods 601 and 602) and hydrocarbon fuels (EPA Method 418.1). To prevent fouling of the instrument and loss of data at subsequent sampling events, field measurements of temperature, pH and conductivity were not obtained because of the visual evidence of significant contamination of the well. Table 4.3.10-1 gives results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

4.3.10.2 <u>Significance of Findings</u>

This section discusses the significance of the findings of this investigation, emphasizing the groundwater and soil chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified in the Hargis report.

Significance of Organic Compounds in Groundwater

To evaluate the degree of groundwater contamination, detected organic parameters were compared to their proposed federal MCLs and RMCLs, as available. The results of this comparison show that benzene, at 8400 ug/L, is the only compound exceeding its proposed MCL. The reported concentration of hydrocarbon fuels was 26 mg/L. No fuel lens was observed at the time of the sampling event, even though one was anticipated.



	Method Detection	Monitor Well HM-7
Date Sampled	Limit	2-5-86
Field Parameters		
Temperature (°C)		-
pH Conductivity (umhos)		
Analytical Parameters ^a		
Volatile Organics (ug/L)		
Method 601 (Halogenated Hydrocar	bons)	
Concentration Factor		1
Vinyl Chloride	0.18	0.33
Methylene Chloride	0.25	0.32
Trichlorofluoromethane	0.40	2.5
trans-1,2-Dichloroethylene	0.10	ND**
Trichloroethylene	0.12	0.96
Method 602 (Aromatic Compounds)		
Concentration Factor		1000
Benzene	0.2	8400

TABLE 4.3.10-1. RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 16, FUEL SATURATION AREA 3. PLANT 4. TEXAS

ND = Not detected.

****** = Compound not confirmed by second-column analysis.

- = Data not collected, omitted to prevent fouling instrument.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.



Concentrations of hydrocarbon fuels at HM-78 were not reported in the Hargis report. In addition to the high levels of benzene reported here, the Hargis report identified high concentrations of chlorodibromomethane, ethylbenzene, and toluene in groundwater from HM-78. The absence of a fuel lens at the time of sampling disagrees with the previously reported fuel lens.



4.3.11 <u>Site No. 9, FDTA 6</u>

FDTA 6 was the designated FDTA from the late 1950s until 1980, when it was closed. This site consisted of a 50-foot diameter gravel-lined ring surrounded by a low, earthen berm. Before 1970, training exercises were conducted twice a year at this site. After 1970, exercises were conducted at monthly intervals. Approximately 250 gallons of waste fuels and oils were reportedly used for each exercise. In addition, it is suspected that larger quantities of waste fuels and oils were deposited in the FDTA between exercises (CH2M Hill, 1984).

In 1983 FDTA 6 was excavated and removed as part of the hazardous waste remedial action program conducted at the plant. The excavated material was analyzed and disposed of at approved hazardous waste disposal sites (CH2M Hill, 1984).

IRP Phase II Stage 1 investigative activities entailed hand augering, collection of soil samples, and the collection of one groundwater sample from Paluxy monitor well P-3. Soil samples were analyzed for acid and base/neutral extractable compounds (EPA Method 8240 and 8270), for halogenated and aromatic volatiles (EPA Methods 8010 and 8020), for hydrocarbon fuels (EPA Method 418.1), and for oil and grease (EPA Method 413.2). The groundwater sample was analyzed for volatile organic compounds, acid and base/neutral extractable compounds, hydrocarbon fuels, and oil and grease. Figure 4.3.11-1 shows the locations of sampling points.

4.3.11.1 <u>Results of Investigation</u>

The investigation at this site included hand augering and the collection and analysis of soil samples, and the collection and analysis of a groundwater sample from well P-3. A discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow.



*_Da*_Ja*_Ja*_Ja*_Ja*_Da*_da*_da*_d

4-122

RAPIAN

Topography

This former Fire Department Training Area is located west of Meandering Road and north of the North Parking Lot. The ground slopes westward toward Meandering Road Creek and surface elevations are generally between 620 feet and 630 feet MSL.

Geologic Features

No monitor wells have been installed at FDTA 6. Information on test hole TH-26 at this site is provided by Hargis and Associates (1985). At TH-26 the upper zone consists of two feet of fill. The Walnut Formation was encountered at two feet below land surface. The Goodland Limestone was not encountered.

Occurrence of Groundwater

On 4 February 1986, the static water level was measured in Paluxy well P-3, the only well associated with FDTA 6. The groundwater elevation was 591.03 feet MSL. However, since this well is screened through the upper, middle, and lower sections of the aquifer, this water level cannot be correlated with groundwater elevations in other Paluxy wells screened across individual zones. The projection of contours on the potentiometric surface of the upper Paluxy (Figure 4.1.2-2), however, suggest a generally southeastward flow direction in this area.

Groundwater Quality

The groundwater sample from well P-3 was collected on 4 February 1986 and analyzed for acid and base/neutral extractables (EPA Method 625); for volatile organics (EPA Methods 601 and 602); for hydrocarbon fuels (EPA Method 418.1); and for oil and grease (EPA Method 413.2). In addition, the water sample was field-checked for temperature, pH and conductivity. Table 4.3.11-1 summarizes the results of these analyses. Complete lab reports for all analyses appear in Appendix A. All positive analytical results were compared to



コンフィン・シン・シン

うしんしんしん

Ì

TABLE 4.3.11-1. RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 9, FIRE DEPARTMENT TRAINING AREA 6, AF PLANT 4, TEXAS

KUX

STATES BARRED REALED FRANKER

SCORESSE, DODDODS

1222222

226226

	Method Detection	Monitor Well P-3
Date Sampled	Limit	2-4-86
Field Parameters		
Temperature (°C)		20.0
pH Conductivity (umhos)		6.8 720
Analytical Parameters ^a		
Volatile Organics (ug/L)		
Method 601 (Halogenated Hydroca Concentration Factor = 1	arbons) ^b	
Method 602 (Aromatic Compounds) Concentration Factor = 1	b	
Extractable Organics (ug/L) Method 625 Acid Concentration Factor = 1		
Phenol	1.5	2
Method 625 Base/Neutral Concentration Factor = 1		
Di-n-butyl phthalate	2.5	3BL
Hydrocarbon Fuels (mg/L)	1	ND
Oil and Grease (mg/L)	1	ND

ND = Not detected.

BL = Detected in reagent blank; background subtraction not performed.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

b = No compounds were detected by this analysis.

RADIAN

federal MCLs and guidelines to evaluate the degree of groundwater contamination. Results of this comparison are discussed in Section 4.3.11.2.

4.3.11.2 Significance of Findings

A discussion of the significance of the analytical results obtained for soil and groundwater samples collected in the Phase II Stage 1 effort follows. These data are compared to data from earlier efforts, where available, and any apparent trends are noted.

Significance of Organic Contaminants in Groundwater

A comparison of organic analyses from well P-3 groundwater samples with federal proposed MCLs and RMCLs revealed no exceedances. None of the regulated compounds were present in detectable quantities. Low concentrations of phenol (2 ug/L) and di-n-butyl phthalate (3 ug/L) were identified. However, di-n-butyl phthalate was detected in the blank. Therefore, its presence in the groundwater sample may be spurious. Results of the groundwater analyses indicate that local groundwater quality has not been degraded by past fire training activities in this area. These observations are generally consistent with those of the Hargis and Associates (1985) report.

Significance of Organic Compounds in Soil

On 12 May 1986, six shallow (6 to 18 inches) soil samples were collected from hand-augered borings in the vicinity of FDTA No. 6. Figure 4.3.11-2 illustrates the locations of the borings. Each sample was analyzed for acid and base/neutral extractable organics; hydrocarbon fuels; and for oil and grease. Table 4.3.11-2 summarizes the results of all positive analyses.



11146 (A. 2004)

SSSSS SSSS

1.11

4-120

M.A. .

and the second and



TABLE 4.3.11-2. RESULTS OF SOIL SAMPLE ANALYSES, SITE 9, FIRE DEPARTMENT TRAINING AREA 6, AF PLANT 4, TEXAS

				Soil Boring	(depth-inch	nes)	
	Method	HA-1	HA-2	HA-3	HA-4	HA-5	HA-6
Date Sampled	Detection Limit	(8) 5-12-86	(8) 5-12-86	(6) 5-12-86	(12-18) 5-12-86	(10-15) 5-12-86	(11-15) 5-12-8
Analytical Parameters [®]							
Hydrocarbon Fuels (mg/kg) 4-6	2600	4000	3800	230	14000	N
Oil and Grease (mg/kg)	4-6	4700	5600	5600	830	13000	N
Volatile Organics (mg/kg)						
Method 8010 (Halogenat	ed Hydrocar	bons)					
Concentration Factor		12.5	12.5	12.5	12.5	12.5	12.
1,1,1-Trichloroethan	e 0.00003	.020	.021	.022	ND	ND	N
Method 8020 (Aromatic Concentration Factor	Compounds) ^b	2500	2500	2500	2500	250000	250
Extractable Organics (mg	/kg)						
Method 8270 Acids ^b							
Concentration Factor		37	36	37	39	120	4:
Method 8270 Base/Neutr	als						
Concentration Factor		37	36	37	39	120	4:
Bis (2-ethylhexyl) phthalate	0.0025	2.5	3.1	ND	0.15	ND	N
Di-n-butyl phthalate		ND	0.95	1.5	0.10	0.71	N
Di-ethyl phthalate	0.0019	ND	0.22	ND	ND	ND	N
Naphthalene Phenanthrene	0.0016 0.0054	ND	ND	ND ND	ND	2.3 0.83	N

ND = Not detected.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

b = No compounds were detected by this analysis.



Five of the six shallow soil samples collected at this site show evidence of residual contamination associated with past fire training activities. Sample HA-6 most closely reflects background soil conditions; all analytical parameters were below detection. The remaining samples contained hydrocarbon fuels in concentrations ranging from 230 to 14,000 mg/kg and a similar range of oil and grease concentrations (830 to 13,000 mg/kg). Several phthalate species were detected in concentrations from 0.15 to 3.1 mg/kg, and naphthalene and phenanthrene were detected in sample HA-5 at concentrations of 2.3 and 0.83 mg/kg, respectively. The only volatile compound detected was 1,1,1-trichloroethane. It was identified at 0.020 to 0.022 mg/kg in the three shallowest soil samples (HA-1, HA-2, and HA-3).

The moderate levels of organic contaminants associated with the soils reflect the environmental effects of past waste disposal activities. However, these contaminants are apparently being retained within the soils.



4.3.12 Site 6, FDTA 3

During the mid-1960s, FDTA 3 was reportedly the site of routine fire department training exercises which used about 250 gallons per exercise of waste fuels and oils.

Activities at this site included the installation of upper zone monitor well HM-102 and the collection of groundwater samples from monitor wells HM-33 and HM-102. Figure 4.3.12-1 shows the locations of these wells. Groundwater samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, for hydrocarbon fuels by EPA Method 418.1, and for oil and grease by EPA Method 413.2.

4.3.12.1 Results of Investigation

A discussion of the results of IRP Phase II Stage 1 activities at FDTA 3 and descriptions of the topographic and geologic features of the site follow.

Topography

FDTA 3 is located northeast of Landfill 4 between Bomber Road and Meandering Road Creek. The land surface slopes slightly toward the creek. Surface elevations in the general vicinity vary from about 650 feet to 640 feet MSL.

Geologic Features

Monitor well HM-33 is located at FDTA 3. The lithologic log of this well (Hargis and Associates, 1985) reveals an upper zone thickness of 20 feet, consisting of about 18 feet of fill and two feet of clay. Approximately 2.5 feet of Goodland Limestone and six feet of Walnut Formation were encountered.





Monitor wells HM-101 and HM-102, which were installed south and east of the site, yielded additional information on the geologic features at this site. The lithologic logs of these wells (Appendix E) show that the Goodland Limestone is absent south of FDTA 3 at HM-101; however, at HM-102, 28 feet of Goodland Limestone (consisting of shale and sand and limestone gravel) were encountered.

Occurrence of Groundwater

Static water levels in the upper zone were measured in HM-33 (located within the suspected boundaries of FTDA 3) and in HM-102, installed about 850 feet east of the site as part of this study. The groundwater elevation in HM-33 was 635.37 feet MSL on 8 February 1986. The elevation in HM-102 was 632.15 feet MSL on 9 April 1986. These data are insufficient to define groundwater flow direction. Figure 4.1.2-8 is the water level map for the upper zone in this area of AF Plant 4 which suggests a northerly groundwater flow direction through the site.

Groundwater Quality

Before this investigation, toluene was detected at a concentration of 8 ug/L in a sample from HM-33 collected in August 1984. For this study, groundwater samples were collected from monitor well HM-33 and from newly installed HM-102 on 2 February 1986 and 9 April 1986, respectively. The samples, including a duplicate split from HM-33 (HM-33D) were analyzed for volatile organic compounds (EPA Methods 601 and 602), for hydrocarbon fuels (EPA Method 418.1) and for oil and grease (EPA Method 413.2). Temperature, pH and conductivity were also measured in the field. Table 4.3.12-1 summarizes all positive analyses. Complete lab reports appear in Appendix A.

RADIAN

	Method		Monitor Well	
Date Sampled	Detection Limit	HM-33 2-28-86	HM-33D 2-28-86	HM-102 4-9-86
Field Parameters				
Temperature (°C)		20.0	20.0	20.0
рН		6.9	6.9	6.5
Conductivity (umhos)		900	900	630
Analytical Parameters ^a				
Volatile Organics (ug/L)				
Method 601 (Halogenated H Concentration Factor	ydrocarbons)	1	1	1
Methylene Chloride	0.25	ND	0.34	ND
Method 602 (Aromatic Comp	ounds)			
Concentration Factor		1	1	1
Toluene	0.2	ND	0.64	0.95
Hydrocarbon Fuels (mg/L)	1	ND	ND	ND
Oil and Grease (mg/L)	1	ND	ND	1*

TABLE 4.3.12-1.RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 6, FIRE
DEPARTMENT TRAINING AREA 3, AF PLANT 4, TEXAS

ND = Not detected.

* = Value less than five times detection limit.

Errors may range up to 100 percent.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

D = Duplicate sample.



4.3.12.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater and soil chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

Significance of Organic Compounds in Groundwater

None of the organic parameters analyzed exceeded federal MCL criteria. Oil and grease, and hydrocarbon fuels concentrations were at or below the detection limit (1 mg/L) in all samples analyzed. Methylene chloride and toluene were the only volatile organic compounds detected. Both compounds were identified at low levels (<1 ug/L) in one split of the duplicate samples from HM-33. Toluene was also detected at a concentration of 0.95 ug/L in the sample from well HM-102.

As previously mentioned, toluene was initially detected in a sample from well HM-33 in 1984. However, the concentration was almost an order-ofmagnitude higher than determined in this study. Also, considering the generally northward flow direction through the site, it is unlikely that the presence of toluene in well HM-102 is related to past fire training activities. Well HM-102 is located just within the boundary of Landfill 2, which represents a more likely source of the observed low-level organic contamination. While the trace amounts of volatile organics identified at HM-33 may be related to former site activities, their extent and thus the potential for adverse environmental effects are considered minimal. These findings are generally consistent with those of the Hargis and Associates report (1985).



4.3.13 Site 7, Fire Department Training Area 4

During the late 1960s, FDTA 4 was reportedly the site of fire department training exercises which used about 250 gallons per exercise of waste fuels and oils. Figure 4.3.13-1 shows the reported location of the site.

Activities at FDTA 4 consisted of a soil-gas survey conducted to determine the location of the old fire training area. Results of the soil gas survey appear in Appendix I. After 28 soil-gas samples had been taken and no definite hydrocarbons noted, AF Plant 4 fire department personnel were requested to pinpoint the fire department training area noted in the Phase I report. The fire department personnel contacted stated that no FDTA had ever existed at this location; therefore, no further work was conducted at this site.



NALAS-NA



4.3.14 Site 18, Solvent Lines

Formerly, these solvent lines reportedly carried xylene, methyl ethyl ketone, and kerosene. Leaks occurred during the 1940's before the lines were drained, capped, and abandoned in place in 1944. The actual locations of the leaks are not known (CH2M Hill, 1984).

IRP Phase II Stage 1 activities at Site 18 consisted of the installation of a downgradient monitor well in the upper zone, HM-106, collection of soil samples from that well, and collection and analysis of groundwater from HM-72, 73, 74, 75, and 106. Figure 4.3.14-1 shows the locations of these sampling points. Soil samples were analyzed for oil and grease by EPA Method 413.2 and for xylene. Groundwater samples were analyzed for oil and grease by EPA Method 413.2, for xylene by EPA Method 602, and for methyl ethyl ketone by EPA Method 8015.

4.3.14.1 Results of Investigation

A discussion of the results of IRP Phase II Stage 1 activities at Site 18 and descriptions of the topographic and geologic features of the site follow.

Topography

This site is located along the route of the former buried solvent lines from the northeast corner of Facility No. 15 to the Test Lab, Facility No. 80. The topography along this corridor has a maximum relief of about four feet in the area north of the Assembly Building, but it is nearly flat, with a surface elevation of approximately 650 MSL, along the north-south trending section (east of Building 176).





Geologic Features

Information on shallow subsurface features derived from lithologic logs of HM-72, 73, 74, and 75 is taken from Hargis and Associates (1985). The lithologic log of well HM-106 appears in Appendix E of this report.

Upper zone monitor wells HM-72, HM-73, HM-74, HM-75, and HM-106 were drilled adjacent to the buried solvent lines. The upper zone at HM-73, 74, and 75 ranges from 4 to 32 feet in thickness and is dominated by clay. No Goodland Limestone was encountered at well HM-74, but 20 feet were encountered at well HM-73 and 18 feet were encountered at well HM-75. The upper zone at HM-106 is 29 feet thick and composed of sand, clay, and shale. A 0.5-foot layer of Goodland Limestone was found at HM-106 and the Walnut Formation was encountered at a depth of 32 feet.

Occurrence of Groundwater

Groundwater occurs in the upper zone of this site. Table 4.3.14-1 gives the results of water level measurements. The water level map of the upper zone in this area of AF Plant 4 (Figure 4.1.2-5) was discussed previously in Section 4.1.2. Water levels measured in the upper zone ranged from 640.31 MSL at HM-73 to 629.71 MSL at HM-74. HM-106 was dry when measured in April, 1986, although it contained water when installed in January. As shown in Figure 4.1.2-5, groundwater flow in the upper zone is to the northwest.

Groundwater Quality

Groundwater samples were collected from monitor wells HM-72, 73, 74, and 75 and analyzed for oil and grease, xylene and methyl ethyl ketone. Table 4.3.14-2 gives results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.



a construction of the

TABLE 4.3.14-1. GROUNDWATER ELEVATIONS AT SITE 18, SOLVENT LINES AF PLANT 4, TEXAS (FEBRUARY 1986)

Monitor Well	Elevation (feet MSL)	Date
HM-72	642.26	2-14-86
HM-73	640.31	2-26-86
HM-74	629.71	2-5-86
HM-75	639.20	2-18-86
HM-106	_	- (dry)

. TEXAS
4•
PLANT
АТ
ES. SITE 18. SOLVENT LINES. AT PLANT 4. TI
SOLVENT
æ
SITE 18.
ANALYSES.
SAMPLE
RESULTS OF GROUNDWATER SAMPLE ANALYSES.
OF
RESULTS
TABLE 4.3.14-2.

5

	Method		2	Monitor Well		
Date Sampled	Detection Limit	HM-72 2-14-86	HM-73 2-26-86	HM-74 2-5-86	HM-75 2-18-86	HM-106 4-18-86
Field Parameters Temnerature (oC)		23.0	24.5	20.5	23.0	SN
		6.8	6.5	6.1	6.1	NS
Conductivity (umhos)		1500	1700-3600	5400	2300	NS
Analytical Parameters						
Oil and Grease (mg/L)	1	UD	QN	QN	QN	NS
Xylenes (ug/L)						
m-Xvlene	0.2	Ð	2	ᢓ	Ð	NS
p-Xvlene	0.2	Q	Q	Q	Q	NS
o-Xylene	0.2	Q	ÛN	Ð	QN	NS
Methyl Ethyl Ketone (ug/L)	1000	Ð	Ð	Q	QN	NS

= Not detected.
= Not sampled. Well was dry.

NS NG

EXAMPLE DEVICED A LEVER

LUDING STRUNT BURGES PERSON

20101111

222222

11 555523

4-140

RADIAN


Ċ

4.3.14.2 Significance of Findings

This section discusses of the significance of the findings of this investigation, emphasizing the groundwater and soil chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

Significance of Organic Compounds in Soils

Soil samples collected at 9-9.5 feet and at 24-24.5 feet during the installation of HM-106 were analyzed for oil and grease, xylene, and methyl ethyl ketone (MEK). Table 4.3.14-3 shows the results of soil sample analyses. Xylene, oil and grease and MEK were not detected in either sample.

Significance of Organic Compounds in Groundwater

IRP Phase II Stage 1 sampling and analysis of the monitor wells at this site detected no organic compounds in concentrations in excess of federal MCL criteria. Methyl ethyl ketone, oil and grease, and xylene were undetected in all four of the samples analyzed. Hargis & Associates (1985) did not analyze for these parameters; therefore, no comparison can be made.

4.3.15 Site 10, Chrome Pit 1

Chrome Pit 1 was used during the early 1940s. It is suspected that miscellaneous liquid and solid chemical wastes, in addition to chrome wastes, were disposed of at this site. Although the location of this site is believed to be somewhere beneath the Process Building, its exact location could not be determined from interviews or aerial photographs.

RADIAN

		<u>Soil Boring (</u>	depth - feet)
Date Sampled	Method Detection Limit	HM-106-A (9-9.8) 1-26-86	HM-106-C (24-24.5) 1-26-86
Analytical Parameters			
Oil and Grease (mg/kg)	4-6	ND	ND
Xylenes (mg/kg)			
p-Xylene	0.11	ND	ND
m-Xylene	0.16	ND	ND
o-Xylene	0.12	ND	ND
Methyl Ethyl Ketone (mg/kg)	1	ND	ND

TABLE 4.3.14-3. RESULTS OF SOIL SAMPLE ANALYSES, SITE 18, SOLVENT LINES, AF PLANT 4, TEXAS

1121

N-0.0.0

ND = Not detected.

Activities at Chrome Pit 1 consisted of installing upper zone monitor well HM-103 east of the chrome pit, collecting and analyzing soil samples, and collecting and analyzing groundwater samples from wells HM-48 and HM-103. Figure 4.3.15-1 shows the locations of these sampling points. The location of monitor well HM-103 was selected because of the need for further information on groundwater flow in the upper zone in this area and to note contamination, if any, present in this area. Soil samples were analyzed for halogenated and aromatic hydrocarbons by EPA Methods 8010 and 8020, and for chromium by ICPES. Groundwater samples were also analyzed for halogenated and aromatic hydrocarbons (EPA Methods 601 and 602) and for chromium by ICPES.

4.3.15.1 Results of Investigation

The IRP Phase II Stage 1 investigation focused on the installation of upper zone monitor well HM-103 and on the collection, analysis and interpretation of soil and groundwater samples. A discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow.

Topography

This site is located beneath Facility No. 181, in the south part of the AF Plant 4 complex. The surface elevation under the building in the vicinity of the site is between 554 feet and 556 feet MSL.

Geologic Features

Upper zone monitor well HM-48 is located approximately 500 feet northwest of Chrome Pit 1. Monitor wells HM-60 and HM-77 are located slightly further away to the southwest of the site. Information on shallow subsurface features is derived from the lithologic logs of these wells (Hargis and Associates, 1985).

The upper zone in this area ranges in thickness from 17 feet at HM-48 to 30 feet at HM-60. The Goodland Limestone was fully penetrated by



4-144



HM-48, where it is only two feet thick. The Goodland Limestone was not encountered at HM-77, where 30 feet of silty clay and gravel overlie limestone of the Walnut Formation.

Upper zone well HM-103 is located approximately 800 feet east of Chrome Pit 3. The upper zone at this well is about 36 feet thick and consists primarily of clayey silt. Approximately three feet of Goodland Limestone were encountered.

This descripton of geologic features at Chrome Pit 1 is generally consistent with that in the Hargis and Associates report (1985).

Occurrence of Groundwater

Groundwater occurs in the upper zone of this site. Table 4.3.15-1 gives results of water level measurements. Water levels in the upper zone range from 642.42 feet MSL at HM-48 to 623.57 feet MSL at HM-103. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-7) was discussed previously in Section 4.1.2. As shown on that map, groundwater flow in this area is to the south-southwest.

Groundwater Quality

Groundwater samples were collected from monitor wells HM-48 and HM-103 and analyzed for volatile organic compounds by EPA Methods 601 and 602, and for chromium by ICPES. Table 4.3.15-2 gives results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling events were compared to federal standards and guidelines. Table 4.3.15-3 summarizes this comparison.



AF PLANT 4, TEXAS (FEBRUARY THROUGH APRIL 19	Monitor Well Elevation (feet MSL) Date HM-48 642.42 2-14-	AF PLANT 4, TEXAS (FEBRUARY THROUGH APRIL 1 Monitor Well Elevation (feet MSL) D. HM-48 642.42 2-
Monitor Well Elevation (feet MSL) Da	HM-48 642.42 2-14-8	HM-48 642.42 2-

n 19 19 - Maria Andrea ann an Anna ann an An



	Method		or Well
	Detection	HM-48	HM-103
Date Sampled	Limit	2-14-86	4-10-86
Field Parameters			
Temperature (°C)		18.0	22.0
рН		7.6	6.3
Conductivity (umhos)		300	620
Analytical Parameters ^a Metals (mg/L)			
Chromium	0.005	ND	0.01*
Volatile Organics (ug/L)			
Method 601 (Halogenated Hy	drocarbons)		
Concentration Factor		5	500
Chloroform	0.05	3.0	ND
Trichloroethylene	0.12	20	5500
Chlorobenzene	0.25	ND	ND
Method 602 (Aromatic Compo	unds)		
Concentration Factor		1	1
Toluene	0.2	0.83	ND
Chlorobenzene	0.2	ND	18

TABLE 4.3.15-2. RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 10, CHROME PIT 1, PLANT 4, TEXAS 2443422

12222222

ALL REPART NUMBER NEWS

2121252

222255222

JALEN 22222

ND = Not detected.

* = Value less than five times detection limit. Errors may range up to 100 percent.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.



E

TABLE 4.3.15-3. COMPARISON OF WATER QUALITY AT SITE 10, CHROME PIT 1, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

Parameter	Criteria*	Sampling HM-48	Location HM-103
ORGANIC COMPOUNDS (ug/L)			
Trichloroethylene	5	20	5500

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.



4.3.15.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater and soil chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

Significance of Organic Compounds and Metals in Soils

Soil samples were collected at depths of 29-30 feet BGL and 34-35 feet BGL during the installation of HM-103 and analyzed for halogenated and aromatic volatiles by EPA Methods 8010 and 8020 and for chromium by EPA Method 6010. Table 4.3.15-4 gives results of these analyses. Two compounds, trans-1.2-dichloroethylene and trichloroethylene, were detected in both samples. Concentrations of trans-1.2-dichloroethylene ranged from 0.039 mg/kg at 29-30 feet to 0.096 mg/kg at 34-35 feet. Levels of trichloroethylene also increased with depth. TCE was present at 0.065 mg/kg in soil from 29-30 feet and 0.17 mg/kg in soil from 34-35 feet.

Chromium was also detected in both soil samples. However, concentrations detected were below the normal range of chromium concentrations found in soils (Table 4.2.1-3).

Significance of Metals in Groundwater

Groundwater samples from HM-48 and HM-103 were analyzed for chromium. Concentrations were less than instrument detection limits in both cases. These findings agree with the data presented by Hargis & Associates (1985).

Significance of Organic Compounds in Groundwater

Of the volatile organic compounds analyzed for, trichloroethylene was the only compound detected at levels in excess of federal MCL criteria.

D

Date Sampled	Method Detection Limit	HM-103-A (29-30) 1-21-86	HM-103-B (34-35) 1-21-86
Analytical Parameters ^a			
Chromium (mg/kg)	0.5	7.4	7.1
Volatile Organics (mg/kg)			
Method 8010 (Halogenated Hydro Concentration Factor	carbons)	12.5	12.5
trans-1,2-Dichloroethylene Trichloroethylene	0.00010 0.00013	0.039 0.065	0.096 0.17
Method 8020 (Aromatic Compound Concentration Factor	s) ^b	125	125

KANNA PANAN KAREN KARAN KARANA DININA

Sec. Sec.

22.22.22

22222222

Figura States

TABLE 4.3.15-4. RESULTS OF SOIL SAMPLE ANALYSES, SITE 10, CHROME PIT 1 AF PLANT 4 TEXAS

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.
b = No compounds were detected by this analysis.

RAPIAN

TCE ranged from 20 ug/L at HM-48 to 5500 ug/L at HM-103. The Hargis & Associates (1985) report also noted the presence of TCE in monitor well HM-48, but at lower levels. TCE was reported in this study at concentrations ranging from 55 to 93 ug/L.

4.3.16 Site 5. Fire Department Training Area 2

FDTA 2 was used for approximately two years during 1955-1956. This site consisted of a 50-foot diameter earthen ring located just north of Land-fill 1. Exercises were held twice a year. However, disposal of waste oils and fuels, as well as uncontrolled burns, may have occurred more frequently.

Activities at FDTA 2 included a geophysical survey and collection of groundwater samples from seven existing upper zone monitor wells. The geophysical survey was conducted to determine the extent of any hydrocarbon plume. Upper zone wells sampled at this site included HM-19, 49, 50, 51, 65, 66, and 76. Figure 4.3.16-1 shows the locations of these wells. Groundwater samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, for hydrocarbon fuels by EPA Method 418.1, and for oil and grease by EPA Method 413.2.

4.3.16.1 Results of Investigation

A discussion of the results of IRP Phase II Stage 1 activities at FDTA 2 and descriptions of the topographic and geologic features of the site follow.

Topograp iy

This site is located under the pavement in the west employee parking lot. The land surface across the site slopes gently to the northwest, toward Meandering Road Creek from elevations of about 640 feet to 636 feet MSL.



4-152

+-152



Geologic Features

Information on the shallow subsurface features was provided by the lithologic logs of monitor wells HM-49, HM-51, HM-65, HM-66, and HM-76 (Hargis and Associates, 1985). The thickness of the upper zone at this site ranges from four feet at HM-66 to 16 feet at HM-76. The Goodland Limestone, encountered at HM-65, HM-66, and HM-76, consists of shale and clay and ranges in thickness from four feet at HM-65 to 13 feet at HM-66. The Goodland Limestone is absent at HM-49 and HM-51.

The thickness of the Walnut Formation is unknown in the immediate vicinity of FDTA 2; however, on the basis of lithologic data obtained from Paluxy monitor wells P-10U, P-10M, and P-4, it is believed to be at least 25 feet thick.

Occurrence of Groundwater

Elevations of water levels of groundwater occurring in the upper zone at this site range from 641.34 feet MSL to 622.09 feet MSL. Table 4.3.16-1 presents results of water level measurements. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-6) was discussed previously in Section 4.1.2. As shown on that map, groundwater flow in this area is to the west-northwest, toward Meandering Road Creek.

Geophysical Survey

A terrain conductivity survey consisting of electromagnetic profiling was performed in November 1985. A grid was established as described in Section 3.1.1.

Figure 4.3.16-2 illustrates the location of the survey. Complete results appear in Appendix I. The conductivity values recorded indicate two large anomalous zones and one small anomalous zone. The anomalous zone in the center of the site is interpreted to be shallow soil contamination and/or a chemical reaction between contamination and subsurface materials. The second

4-153



Τ,

17

A10.610.010.010.01

እስለ እና እና የሚያስት እና ትርጉ የሚያስት እና ትርጉ የሚያስት እና ትርጉ እና ትር እና ትርጉ እና ትርጉ

TABLE 4.3.16-1.GROUNDWATER ELEVATIONS AT SITE 5, FIRE DEPARTMENT TRAINING
AREA 2, AF PLANT 4, TEXAS (FEBRUARY THROUGH MARCH 1986)

Monitor Well	Elevation (feet MSL)	Date
HM-19	622.09	3-1-86
HM-49	627.19	2-12-86
HM-50	629.95	2-12-86
HM-51	630.79	2-11-86
HM-65	627.19	2-12-86
HM-66	-	(not measured)
HM-76	641.34	2-13-86



4-155

- A CARACANA

RADIAN

zone, may indicate moderate amounts of fill. The third zone located in the southwestern corner of the area, is interpreted to be large amounts of metal.

Groundwater Quality

Groundwater samples were collected from wells HM-19, 49, 50, 51, 65, 66, and 76 and analyzed for volatile organic compounds by EPA Methods 601 and 602, for hydrocarbon fuels (EPA Method 418.1) and for oil and grease (EPA Method 413.2). Table 4.3.16-2 presents results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Table 4.3.16-3 summarizes the results of this comparison.

4.3.16.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis & Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those previous findings and trends identified by the Hargis report.

Significance of Organic Compounds in Groundwater

Three volatile organic compounds, trichloroethylene, toluene, and 1,2-dichloroethane, were detected in upper zone groundwater in concentrations exceeding the federal MCL criteria. An unknown compound with the same retention time as trans-1,2-dichloroethylene was detected on the first column in groundwater samples from HM-50 and HM-51. Methylene Chloride was detected in HM-51 at a concentration of 200,000 ug/L. Levels of TCE ranged from 9.0 ug/L in HM-19 to 420,000 ug/L in HM-51. Toluene was present only in groundwater sampled from HM-51, at a concentration of 96,000 ug/L. 1,2-Dichloroethane was



TABLE 4.3.16-2. RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 5, FIRE DEPARTMENT TRAINING AREA 2, AF PLANT 4, TEXAS

	Method			Moi	nitor Well	L		
Date Sampled	Detection Limit	HM-19 3-1-86	HM-49 2-12-86	HM-50 2-12-86	HM-51 2-11-86	HM-65 2-12-86	HM-66 2-13-86	HM-76 2-13-8
ield Parameters								
Temperature (°C)		21.0	15.5	-	19.0	17.5	19.0	19.
рН		6.3	7.4	-	5.8	6.5	6.6	6.
Conductivity (umhos)	l i	340	1200	-	9700	3900	1200	210
nalytical Parameters ⁴	Ł							
olatile Organics (ug/	'L)							
Method 601 (Halogena	ated Hydrocar	bons)						
Concentration Fact	or	1	1	500	10000	1	1	
Methylene Chloride	0.25	ND	ND	ND	200000	ND	ND	N
trans-1,2-Dichlord ethylene		ND**	ND**	ND**	ND**	n ND	ND**	' N
1,2-Dichloroethane	0.03	ND	ND	ND	27000	ND	ND	N
Trichloroethylene	0.12	9.0	1.9	ND	420000	ND	29	N
Tetrachloroethyler	ne 0.03	0.41	1.1	ND	ND	ND	ND	N
Method 602 (Aromatic	Compounds)							
Concentration Fact	or	5	1	100	1000	1	1	
Toluene	0.2	ND	1.4	ND	96000	1.5	1.8	0.9
ydrocarbon Fuels (mg/	'L) 1.	ND	ND	ND	11	ND	ND	N

ND = Not detected.

** = Compound not confirmed by second-column analysis. - = Data not collected, insufficient volume.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

2.1



and a start of the s

TABLE 4.3.16-3. COMPARISON OF WATER QUALITY AT SITE 5, FDTA 2, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

		Sam	pling Loca	tion
Parameter	Criteria*	HM-19	HM-51	HM-66
RGANIC COMPOUNDS (ug/L	.)			
l,2-Dichloroethane	5		27000	
richloroethylene	5	9	420000	29
Toluene	2000		96000	
lethyl Chloride			200000	

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.



التكنينينين

present in HM-51 at a concentration of 27,000 ug/L. Hydrocarbon fuels and oil and grease were present in levels above instrument detection limit only in groundwater from HM-51. At this well, hydrocarbon fuels were detected at 11 mg/L and oil and grease was detected at 17 mg/L.

Hargis & Associates (1985) noted a hydrocarbon fuel concentration of 63 mg/L in groundwater from HM-51 in comparison to the 11 mg/L detected during the present study. Otherwise, Hargis & Associates' observations generally agree with present observations of hydrocarbon fuel levels and oil and grease levels in upper zone groundwater. Hargis & Associates (1985) also report levels of volatile organic compounds generally consistent with those reported here. However, they noted high levels of 1,2-trans-dichloroethylene in groundwater from HM-50 and HM-51.

The highest concentrations of contaminants are found in groundwater samples taken from HM-51, directly in the center of FDTA 2. Downgradient from this well (HM-49 and HM-65), concentrations of all organic compounds drop off to much lower levels, indicating that the contamination is localized.



4.3.17 Site 14, Fuel Saturation Area 1

Fuel Saturation Area 1 (FSA 1) is located just north of a fuel tank truck unloading and pumping station. From the 1970s until the early 1980s (CH2M Hill, 1984), the ground at this site reportedly became saturated by fuels from leaking buried fuel lines.

Activities at FSA 1 included drilling one borehole, collecting groundwater samples from the borehole, two upper zone monitor wells and two Paluxy wells, and observing the monitor wells for the existence of any fuel lens on the water table. Monitor wells sampled at this site included HM-53, 55, P-6M and P-6U. Groundwater samples were analyzed for hydrocarbon fuels (EPA Method 418.1) and for oil and grease (EPA Method 413.2). Soil samples collected during the drilling of soil boring SB-4 were also analyzed for hydrocarbon fuels. Figure 4.3.17-1 shows the locations of all sampling points.

4.3.17.1 Results of Investigation

A discussion of the results of IRP Phase II Stage 1 activities at FSA 1 and descriptions of the topographic and geologic features of the site follow.

Topography

This area is located west of, and adjacent to, Facility No. 5 (Parts Plant) across from Building 14. The surface elevation in this area is about 654 feet MSL.

Geologic Features

Paluxy monitor wells P-6M and P-6U are located in the immediate vicinity of Fuel Saturation Area 1. The lithologic logs of these wells (Hargis and Associates, 1985) show an upper zone thickness of about 27 feet.



4-161

SUSSESS TRANSPORT SUSSESS SUSSESS SUSSESS SUSSESS DEPENDENT DEPENDENT DEPENDENT DEPENDENT DEPENDENT 1111



The Goodland Limestone is approximately four feet thick and the Walnut Formation is about 33 feet thick.

Soil boring SB-4 was drilled at this site. The lithologic log of this soil boring (Appendix E, this report) reveals an upper zone thickness of at least 28 feet. The upper zone is composed of sand and sand and gravel overlain by 3.5 feet of fill.

Occurrence of Groundwater

Groundwater occurs in both the upper zone and in the Paluxy Formation. Table 4.3.17-1 provides results of water level measurements. Water levels in the upper zone ranged from 638.91 feet MSL at HM-55 to 629.50 feet MSL at HM-53. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-6) was discussed previously in Section 4.1.2. Upper zone groundwater flow in this area is to the north. Groundwater elevations in the Paluxy Formation ranged from 571.69 feet MSL in the upper Paluxy Formation to 566.69 feet MSL in the middle Paluxy Formation, indicating that the direction of hydraulic gradient is from the upper to the middle member. Figures 4.1.2-2 and 4.1.2-3 show groundwater flow in the upper and middle Paluxy.

Groundwater Quality

Groundwater samples were collected from monitor wells HM-53, 55, P-6M, and P-6U and analyzed for hydrocarbon fuels and oil and grease. Table 4.3.17-2 gives the results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

4.3.17.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the soil and groundwater chemistry at the site.



 Ч.,

Y. Y. Y.

TABLE 4.3.17-1. GROUNDWATER ELEVATIONS AT SITE 14, FUEL SATURATION AREA 1, AF PLANT 4, TEXAS (FEBRUARY 1986)

Monitor Well	Elevation (feet MSL)	Date
HM-53	629.50	2-10-86
HM-55	638.91	2-4-86
P-6M	566.69	2-4-86
P-6U	571.69	2-5-86

RADIAN

		Method	HM-53	М НИ-55	SATURATION AREA 1, AF PLANT 4, TEXAS A HM-53 HM-55 P-6M	RESULTS OF GROUNDWATER SAMPLE ANALYSES, SITE 14, FUEL SATURATION AREA 1, AF PLANT 4, TEXAS A TEXAS Monitor Well HM-53 HM-55 P-6M P-6U	SB-4-D
	Date Sampled	Detection Limit	2-10-86	2-4-86	2-4-86	2-5-86	Water Grab 1-27-86
	Field Parameters Temperature (°C)		20.5	21	20	17	~
	pH Conductivity (umhos)		6.8 950	7.0 730	7.2 760	11.2	NA
	Analytical Parameters						
	Hydrocarbon Fuels (mg/L)	1	1.0	QN	QN	QN	212
-164	Oil and Grease (mg/L)	ц.	NA	Ð	QN	Q	NA

= Not detected. Not analyzed. П QN AN ALLER DILLER TILLER PLEASE DILLER PLEASE PLACE DILLER DILLER DILLER DILLER DILLER DILLER DILLER

Received.

11111111

المحدقة

Hargis and Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

Significance of Organic Compounds in Soil

Soil samples were collected during the drilling of Soil Boring 4 (SB-4) at two depths: 9 to 10 feet BGL, and 25 to 25.5 feet BGL. Samples were analyzed for hydrocarbon fuels. Table 4.3.17-3 gives the results of these analyses. Soil from the 9- to 10-foot depth contained the lowest concentrations of hydrocarbon fuels, lower than the instrument detection limit. At greater depths, levels of hydrocarbon fuels increased to 59,000 mg/kg at 25 to 25.5 feet.

Significance of Organic Compounds in Groundwater

Hydrocarbon fuels and oil and grease were not detected except in groundwater from well HM-53, where hydrocarbon fuels were detected at a concentration of 1.0 mg/L. These observations generally agree with those of the Hargis and Associates (1985) report with one exception. In the Hargis report, oil and grease was detected in water from HM-53 at 240 mg/L, and hydrocarbon fuel concentrations ranged from 12 to 18 mg/L. This suggests a decrease in contaminant levels over time.

The borehole for SB-4 was allowed to stand open over night and a grab sample of accumulated liquid was collected the next day. Analysis of this sample revealed 97% hydrocarbon fuels.



SA	TURATION AREA 1, AF P	LANT 4, TEXAS	
Date Sampled	Method Detection Limit	SB-4-A (9-10) 1-26-86	SB-4-B (25-25.5) 1-26-86
Analytical Parameters			
Hydrocarbon Fuels (mg/kg)	4–6	ND	59000

TABLE 4.3.17-3. RESULTS OF SOIL SAMPLE ANALYSES, SITE 14, FUEL SATURATION AREA 1, AF PLANT 4, TEXAS

ND = Not detected.



4.3.18 Site 19, NARF Area

The Nuclear Aerospace Research Facility (NARF) formerly located at this site reportedly housed several experimental atomic reactors between 1953 and 1974. Large quantities of nuclear activation material were produced at this site as an undesirable side effect of neutron bombardment. Those activation products were reportedly contained on site, and the entire facility was decommissioned and disposed of by contractor in 1974. A total of over two million pounds of miscellaneous parts and 15 million pounds of concrete rubble were hauled off site to Barnwell, South Carolina. Post-closure inspection of this site reportedly revealed no remaining contamination (CH2M Hill, 1984).

Investigation activities at this site included drilling four boreholes, collecting and analyzing soil samples, collecting groundwater samples from three upper zone monitor wells (HM-83, 84 and 85), researching past analytical data from wells at AF Plant 4 and the City of White Settlement for radionuclide results during the NARF operations period, and determining the exact locations of past operations.

According to a representative of the City of White Settlement, no information on radionuclides is available for White Settlement municipal wells. In addition, AF Plant 4 personnel stated that the information from wells at Plant 4 during NARF operations was archived and that separate action would be required to retrieve this information.

Soil samples were collected during the drilling of four soil borings, SB-6, SB-7, SB-8, and SB-11. These samples were analyzed for alpha, beta, and gamma radiation. Figure 4.3.18-1 shows the locations of soil boring locations.



4-168



4.3.18.1 Results of Investigation

The investigation focused on the collection and analysis of soil samples. A discussion of the results of these activities and descriptions of the topographic and geologic features of the site follow. Wells HM-83, -84 and -85 are relatively far removed from the NARF. Because of this, monitoring results for these wells are reported in the discussion of ambient monitoring in Section 4.3.20.

Topography

The former NARF area is located at the north end of AF Plant 4, just south of Lake Worth. The topography of the site is relatively flat. The surface elevation is about 622 feet MSL, but the slope steepens northward towards the shoreline of Lake Worth.

Geologic Features

e Berren manaren - berren antikari - Barrenan - Barrenan - Darrenan - Sannan - Sannan - Sannan - Barrenan - Barrenan

Although two monitor wells installed in the early 1960s are located near this area, lithologic logs are not available. Therefore, the thickness and nature of the upper zone at these locations are unknown.

Occurrence of Groundwater

Water level data from upper zone wells HM-83, HM-84, and HM-85 suggest a groundwater elevation between 628 and 630 feet MSL and a westerly flow direction. The water level map of the upper zone in this are of AF Plant 4 (Figure 4.1.2-4) was discussed previously in Section 4.1.2.

4.3.18.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the soil chemistry at the site. No previous soil chemistry data exist for this site; therefore, no comparisons can be made.

Significance of Radioactive Materials in Soil

Soil samples were collected during the drilling of Soil Boring 6, 7, 8, and 11. Table 4.3.18-1 shows the depths of sample collection at each borehole. Samples were analyzed for alpha, beta, and gamma radiation. Table 4.3.18-1 shows the results of these analyses. Complete analytical reports for these analyses appear in Appendix A.

Levels of alpha radiation ranged from 6.7 pCi/g at SB-6-A to 12.4 pCi/g at SB-6-C. Beta radiation was detected at levels ranging from 16.1 pCi/g at SB-11-A to 23.1 pCi/g at SB-8-A. Instead of directly measuring gamma radiation, which is a fission by-product, analyses were conducted for the detection of Cesium 137, which is also a fission by-product. If Cesium 137 was detected at high levels, then an actual test of gamma radiation levels could have been run. However, levels of Cesium 137 were below instrument detection limit; therefore it was not necessary to analyze directly for gamma radiation.

While no background radiological data are available, as discussed above, all other data suggest that there is no residual radiation above acceptable levels at this site.

RESULTS OF SOIL SAMPLE ANALYSES, SITE 19, NARF AREA, AF PLANT 4, TEXAS TABLE 4.3.18-1.

					Soil	Soil Boring (depth - feet)	:h - feet)			
	Method	SB-6-A	SB-6-B	SB-6-C	SB-6-D	SB-7-A	SB-8-A	SB-11-A	SB-11-B	SB-11-C
Date	Detection	(2-3)	(4-5)	(6-7)	(8-6)	(2-3)	(2-3)	(2-3)	(2-3)	(4-4.5)
Sampled	Limit	7-21-86	7-21-86	7-21-86	7-21-86	7-21-86	7-21-86	7-24-86	7-24-86	7-24-86
Analytical Parameters										
Alpha radiation (pCi/g)	0.5	6.7(<u>+</u> 6.6)	6.7(<u>+</u> 6.6) 11.3(<u>+</u> 6.8)	7.1(<u>+</u> 6.5)	12.4(<u>+</u> 6.9)	9.2(<u>+</u> 5.9)	9.7 (<u>+</u> 6.5)	7.1(<u>+</u> 6.5) 12.4(<u>+</u> 6.9) 9.2(<u>+</u> 5.9) 9.7(<u>+</u> 6.5) 7.1(<u>+</u> 6.5) 8.7(<u>+</u> 6.4)	8.7(<u>+</u> 6.4)	7.6(<u>+</u> 5.8)
Beta radiation (pCi/g)	0.8	19.6(<u>+</u> 4.2)	15.2(<u>+</u> 4.2)	15.8(<u>+</u> 4.2)	19.5(<u>+</u> 4.2)	22.9(<u>+</u> 4.4)	23.1(<u>+</u> 4.4)	19.6(<u>+</u> 4.2) 15.2(<u>+</u> 4.2) 15.8(<u>+</u> 4.2) 19.5(<u>+</u> 4.2) 22.9(<u>+</u> 4.4) 23.1(<u>+</u> 4.4) 16.1(<u>+</u> 4.2) 18.3(<u>+</u> 4.2)	18.3(<u>+</u> 4.2)	10.0(<u>+</u> 3.9)
Gamma radiation (pCi/g) (Cesium 137)	40-60	Ð	9	₽	2	Ð	9	2	Ð	9

ND = Not detected.

4-171

AND TOWN OF THE REAL PROPERTY AND THE PROPE



4.3.19 Jet Engine Test Stand, Building 21

The purpose of the investigation in the vicinity of Building 21 was to determine if soil and groundwater have been affected by activities at the site. Reportedly, jet fuel and gasoline had collected downslope of the facility in a sump.

Activities at the Jet Engine Test Stand included installation of three upper zone monitor wells, HM-105, 107, and 108, drilling of soil borings SB-9 and SB-10, collection of soil samples during drilling operations, collection of groundwater samples from the three upper zone monitor wells, and collection of surface water samples from two outfalls, Outfall 1, and Outfall 5. Soil samples were analyzed for hydrocarbon fuels by EPA Method 418.1 and for oil and grease. Samples of groundwater, surface water, and Outfall 1 were analyzed for volatile organic compounds by EPA Methods 601 and 602, for hydrocarbon fuels by EPA Method 418.1, and for oil and grease by EPA Method 413.2. Outfall 5 surface water samples were analyzed for hydrocarbon fuels and oil and grease. Figure 4.3.19-1 shows the locations of the monitor wells. Figure 4.3.19-2 shows the two outfalls.

4.3.19.1 Results of Investigation

A discussion of the results of collecting and analyzing soil, surface water, and groundwater samples at the Jet Engine Test Stand follows. Other aspects of the site, such as topography and geology are also discussed.

Topography

The Jet Engine Test Stand is located in the northwestern corner of AF Plant 4, just northeast of Fuel Saturation Area 3. The topography of the site is relatively flat. The surface elevation is about 640 feet MSL.



22222222

4-173



Γ.

23355

4-174

25552



Geologic Features

Upper zone monitor wells HM-105, HM-107, and HM-108 are located in the vicinity of the Jet Engine Test Stand. Soil borings SB-9 and SB-10 were also drilled in the vicinity of this site. The upper zone is 15 to 19 feet thick at these wells and consists of clay, sand, and silt gravel. About 2 feet of Walnut Formation were encountered at HM-105 and HM-108; about 8 feet of Walnut Formation were encountered at HM-107.

Occurrence of Groundwater

Table 4.3.19-1 shows groundwater elevations in the upper zone at this site. Groundwater elevations range from 626.74 feet MSL at HM-105 to 624.45 feet MSL at HM-108. The map of water levels in the upper zone in this area of AF Plant 4 (Figure 4.1.2-4) was discussed previously in Section 4.1.2. As shown on Figure 4.1.2-4, groundwater flow at this site is to the southwest.

Groundwater and Surface Water Quality

Groundwater samples were collected from monitor wells HM-105, 107, and 108 and surface water samples were collected from Outfall 1. All samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, for hydrocarbon fuels, and for oil and grease. Surface water samples collected from Outfall 5 were analyzed for hydrocarbon fuels and oil and grease. Tables 4.3.19-2 (Outfall 1) and 4.3.19-3 (Outfall 5) present the results of surface water analyses. Although Outfall 3 (not 5) is associated with the Jet Engine Test Stand facility, Outfall 5 was sampled as stated in the statement of work. Table 4.3.19-4 gives results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.



.....

6 T. T

7 X I X

TABLE 4.3.19-1. GROUNDWATER ELEVATIONS AT JET ENGINE TEST STAND, BUILDING 21, AF PLANT 4, TEXAS (AUGUST 1986)

Monitor Well	Elevation (feet MSL)		Date
HM-105	641.92	626.74	8-19-86
HM-107	642.33	625.26	8-19-86
HM-108	639.76	624.45	8-20-86

STALL ALLER ?
RESULTS OF SURFACE WATER SAMPLE ANALYSES, OUTFALL 1, JET ENGINE TEST STAND, BUILDING 21, AF PLANT 4, TEXAS TABLE 4.3.19-2.

Á	Method Detection					Sa	Sample Site				
Date Sampled	Limit	041-1030 8-20-86	0F1-1130 8-20-86	0£1-1230 8-20-86	0£1-1330 8-20-86	0F1-1430 8-20-86	0fr1-1330 8-20-86	0£1-1030 8-20-86	0f1-1/30 8-20-86	0f 1-1/300 8-20-86	0f 1 - CUMPUST 15 8-20-86
Field Parameters Temperature (°C)		33.0	I	33.0	33.0	34.0	33.0	31.0	32.0	32.0	¥2
pH		6.9	1	7.4	7.4	8.0	8.1	8.1	8.1	8.1	N
Conductivity (umhos)		069	1	450	410	410	410	410	400	400	¥
Analytical Parameters ^a											
Volatile Organics (ug/L)											
Method 601 (Halogenated Hydrocarbons) Concentration Factor	ydrocarbons)	1	1	1	1		1	1	1	1	1
Mathulana (hlorida	75	Ę	Ş	Ę	Ş	N	E	Ę	Ĩ	ΕV	Y
Trichlorofluoromethane	0.4	2 9	2	9	2	NA.	2 2	2	2	2.3	2
Chloroform	0.05	3.0	4.3	2.7	2.6	VN	2.4	2.2	2.8	2.6	2.9
1,2-Dichloroethane	0.03	ę	Q	Ð	ę	NA	Ð	Ð	Q	9	E
1.1.1.Trichloroethane	0.03	ę	Ð	2	Ð	NA	Ð	Ð	Ð	9	Ż
Carbon Tetra chloride	0.12	ę	Ð	9	Ð	VN	Ð	Ð	£	Ð	Z
Bromodichloromethane	0.1	4.0	4.8	4.1	3.9	VN	3.7	3.5	4.7	3.7	4.1
Trichloroethylene	0.12	Q	0.13	9	0.19	VN	0.29	0.23	0.25	9	W
Dibromochloromethane	60.0	4.4	4.8	4.3	4.4	VN	4.6	4.2	5.6	4.5	5.2
Bromoform	0.2	1.7	2.0	1.8	1.9	NA	1.8	1.8	2.0	1.8	1.9
Method 602 (Aromatic Compounds)	(spuno									•	
Concentration Factor		-	1	1	-		1	-	1	-	1
Toluene	0.2	Ð	Ð	Ð	Q	NA	Ð	Ð	Q	1.5	Đ
Oil and Grease (mg/L)	1	Q	QN	QN	QN	NA	Q	Ŕ	QN	1*	<u>R</u>
Hydrocarbon Fuels (mg/L)	1	Ð	Q	Ð	<u>CIN</u>	NA	Q	Ð	Q	Q	Ð

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.
ND = Not detected.
NA = Not analyzed.
NA = Outfall.
e Value less than five times detection limit. Errors may range up to 100 percent.

Method

Data not collected, instrument not available.

	TABLE 4.3.19-3.		RESULTS OF SURFACE WATER ANALYSES, OUTFALL 5, JET ENGINE TEST STAND. BUILDING 21, AF PLANT 4, TEXAS	E WATER ANAL LANT 4. TEXU	LYSES, OUTFI AS	ALL 5, JET E	INGINE TEST	STAND,		
					San	Sample Site				
Date Sampled	Method Detection Limit	0F5-930 8-20-86	OF5-1130 8-20-86	OF5-1230 8-20-86	0F5-1330 8-2 J- 86	0F5-1430 8-20-86	0F5-1530 8-20-86	0F5-1630 8-20-86	0F5-1730 8-20-86	0F5-Comp. 8-20-86
Field Parameters										
Temperature (°C)		I	ı	34.5	33.0	33.0	33.0	33.5	32.0	Ņ
PH		I	١	7.3	7.6	7.8	7.9	7.2	7.5	VN
Conductivity (umhog)		I	I	590	370	510	390	520	610	¥
Analytical Parameters										
Volatile Organics (ug/L)		NA	VN	NA	NA	NA	NA	NA	NA	N
Oil and Grease (mg/L)	1	Ð	Q	Ð	₽	Q	Q	Q	Q	Q
Hydrocarbon Fuels (mg/L)) 1	Ð	QN	Q	Q	QN	QN	Ð	Ð	Ð

4-178

ND = Not detected. NA = Not analyzed. - = Data not collected, instrument not available.

<u>ssen</u> denen kære fræks

<u> 22.000 - 20.000 - 20.000 - 20</u>

RADIAN

	Method		Monitor Well	
Date Sampled	Detection Limit	HM-105 8/19/86	HM-107 8/19/86	HM-108 8/20/86
Field Parameters				
Temperature (°C) pH Conductivity (umhos)		26.6 7.2 670	24.0 6.8 620	27.5 7.1 870
Analytical Parameters				
Volatile Organics (ug/L)		NA	NA	NA
Oil and Grease (mg/L)	1	2*	2*	ND
Hydrocarbon Fuels (mg/L)	1	ND	ND	ND

TABLE 4.3.19-4. RESULTS OF GROUNDWATER ANALYSES, JET ENGINE TEST

ND = Not detected.

* = Value less than five times detection limit. Errors may range up to 100 percent.

4.3.19.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the soil and groundwater chemistry at the site.

Significance of Organic Compounds in Soil

Soil samples were collected during the drilling of HM-105, 107, and 108. Samples were analyzed for hydrocarbon fuels and oil and grease. Table 4.3.19-5 gives the results of these analyses. The highest concentration of hydrocarbon fuels was 1700 mg/kg found in soil from HM-107 at a depth of 14 to 15 feet. At SB-9, 1300 mg/kg of hydrocarbon fuels were found in soil at a depth of 9 to 10 feet. At HM-105 and HM-108, hydrocarbon fuels were less than the instrument detection limit. The highest concentration of oil and grease was also found at HM-107, at 2000 mg/kg. At SB-9, 1800 mg/kg of oil and grease was found at 9 to 10 feet. The concentrations of oil and grease at HM-105 and HM-108 were 170 mg/kg and 310 mg/kg, respectively.

Significance of Organic Compounds in Groundwater

Concentrations of hydrocarbon fuels and oil and grease were less than the instrument detection limit in samples from all three monitor wells at the Jet Engine Test Stand.

Significance of Organic Compounds in Surface Water

Concentrations of organic compounds in water sampled from Outfalls 1 and 5 were all below federal MCL criteria. Outfall 1 was sampled every hour for eight consecutive hours. One eight-hour hand-composited sample was also collected. Samples were analyzed for volatile organic compounds, hydrocarbon fuels, and oil and grease. For one time period, the volatile organics samples were not analyzed; however, no trends in contaminant concentration were observed. Eight samples plus one eight-hour composited sample were also collected from Outfall 5 and analyzed for hydrocarbons and oil and grease. These

555555 J

JET ENGINE TEST STAND.	
RESULTS OF SOIL SAMPLE ANALYSES.	4. TEXAS
SAMPLE	PLANT 4
SOIL	. AF
OF	21
RESULTS (BUILDING 21, AF PLANT 4, 7
TABLE 4.3.19-5.	

	De	Method etection	HM-105	So HM-107 (1 4-15)	il Sample (HM-108 (9-10)	Soil Sample (depth - feet) HM-108 SB-9-A (9-10) (9-10)	t) SB-9-C (19-20)	SB-10-A (9-10)
	Analytical Parameters							
ytical Parameters	Hydrocarbon Fuels (mg/kg)	4–6	QN	1700	Q	1300	Q	QN
4–6 ND 1700 ND 1300 ND	Oil and Grease (mg/kg)	4–6	170	2000	310	1800	Q	QN
4-6 ND 1700 ND 1300 ND 4-6 170 2000 310 1800 ND								

NT = Not detected.

RADIAN

1.5.8.4"5.5"5



.

samples were collected at the times shown in Table 4.3.19-3. All concentrations were below the instrument detection limit and, again, no trends in contaminant concentration were observed. These outfalls were not sampled during the Hargis and Associates investigation (1985); therefore, no comparisons can be made.

ዀዀዀዀዀዀዀ



4.3.20 Ambient Monitoring

Several wells installed at AF Plant 4 are not associated with a particular waste disposal site or suspected spill area. Groundwater from these and other wells was sampled and analyzed to gain a complete view of groundwater quality at AF Plant 4.

IRP Phase II Stage 1 ambient monitoring activiteis entailed the collection of groundwater samples from eleven upper zone monitor wells and eight Paluxy monitor wells. Upper zone monitor wells included in this group are the following: HM-29, 52, 54, 56, 57, 58, 59, 61, 64, 79, 81, 83, 84, and 85. Paluxy wells included are: P-5M, P-5U, P-9M, P-9U, P-10M, P-10U, P-11M, and P-11U. Figure 4.3.20-1 shows the locations of these wells. Groundwater samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, for acid and base/neutral extractable compounds by EPA Method 625, for heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2, for hydrocarbon fuels by EPA Method 418.1, and for oil and grease by EPA Method 413.2. In addition, samples from HM-83, HM-84, and HM-85 (sampled as part of the NARF area activities, Section 4.3.18) were analyzed for radioactive materials.

4.3.20.1 Results of Investigation

A discussion of results of the IRP Phase II Stage 1 ambient monitoring follows.

Occurrence of Groundwater

Groundwater occurs in both the materials of the upper zone and in the Paluxy Formation in the ambient monitoring wells. Table 4.3.20-1 presents results of water level measurements. Water levels in the upper zone ranged from 649.26 feet MSL at HM-57 to 623.78 feet MSL at HM-79. Groundwater elevations in the upper member of the Paluxy Formation ranged from 596.05 feet MSL at P-10U to 555.66 feet MSL at P-11U. Groundwater elevations in the



4-184



RAPIAN	DUNDWATER ELEVATIONS AT AMBIENT MO	
FABLE 4.3.20-1. GRO AF	DUNDWATER ELEVATIONS AT AMBIENT MO PLANT 4, TEXAS (JANUARY THROUGH A	NITORING LOCATIONS, UGUST 1986)
Monitor Well	Elevation (feet MSL)	Date
HM-29	629.04	2-21-86
HM-52	643.95	2-10-86
HM-54	638.52	2-10-86
HM-56	632.01	2-21-86
HM-57	649.26	2-26-86
HM-58	643.10	2-26-86
HM-59	639.45	2-30-86
HM-61	640.65	2-26-86
HM-64	628.89	2-26-86
HM-79	623.78	2-3-86
HM-81	634.32	2-7-86
HM-83	629.41	8-21-86
HM-84	628.22	8-21-86
HM-85	630.01	8-21-86
D 5W	570.91	1 - 21 . 06
P-5M P-5U	570.81 564.78	1-31-86 1-31-86
P-9M P-9U	564.25	2-5-86 2-5-86
	565.15	
P-10M	575.83	2-17-86
P-10U	596.05	2-17-86
P-11M	555.46	2-28-86
P-11U	555.66 557.53	2-28-86 2-28-86
P-12M	557.74	
P-12U	557.74	2-28-86
	4-185	



middle member of the Paluxy Formation ranged from 575.83 feet MSL at P-10M to 555.46 feet MSL at P-11M. Maps of water levels in these zones are presented and discussed in Section 4.1.2.

Groundwater Quality

Groundwater samples were collected from the following upper zone monitor wells: HM-29, 52, 54, 56, 57, 58, 59, 61, 64, 79, 81, 83, 84 and 85. The following Paluxy wells were sampled: P-5M, P-5U, P-9M, P-9U, P-10M, P-10U, P-11M and P-11U. Water levels were also measured at P-12M and P-12U. Samples were analyzed for volatile organic compounds (EPA Methods 601 and 602), for acid and base/neutrals (EPA Method 625), for heavy metals (EPA Methods, 200.7, 206.2, 239.2, 245.1, and 270.2), for hydrocarbon fuels (EPA Method 418.1), and for oil and grease (EPA Method 413.2). In addition, groundwater sampled from HM-83, HM-84, and HM-85 was analyzed for radioactive materials. Table 4.3.20-2 presents results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Table 4.3.20-3 summarizes this comparison.

4.3.20.2 Significance of Findings

This section discusses of the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis and Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

YORK KNOT IN BARAKANAN KANANAN KANAN

Paradora Presserves

Personal processors

Hethod Detection Lisit Hethod Het-28 He-52 He-52 He-54 He-56 He-57 He-58 <	RAPI/	CABLE 4.3.	20-2.	RESU AMBI	LTS OF ENT MO	GROUN	DWATER NG, AF	SAMPL PLANT	E ANAL 4, TI	LY SES, EXAS		
Date Supple Period Pe		Hethod										
Temporeture (°C) 21,8 18,0 18,0 20,3 - - 18,0 18,0 20,3 - - 18,0 18,0 20,3 - - 18,0 18,0 20,3 100 1000 - - 18,0 18,0 20,3 100 1000 - - 18,0 18,0 20,3 1000 1000 - - 18,0 18,	Date Sempled	List					+ .=					1 84 2-3-
pH 7.0 7.2 7.2 7.0	Field Persmeters											
Dendeutiting (unkes) 1000 550 550 1100 1100 - - 1200-1800 120	Temperature (°C)											2
Nature (mg/L) Argentic 0.002-0.005 ND 0.002* 0.007* 0.002* NA 0.01* ND 0.045 0.047 0.005 Chramma 0.005 0.14 0.005* 0.02* 0.007* 0.00** 0.01** 0.00** 0.01** 0.00** 0.01** 0.00** 0.01** 0.01** 0.01** 0.01** 0.01** 0.01** 0.01** 0.01** 0.01** 0.01** 0.01** 0.01*** 0.01*** 0.01***<		1										1:
Matsis (mg/l) Arrento 0.000-0.000 ND 0.000 0.007 0.0007 ND 0.007 ND 0.017 0.017 0.014 0.007 ND 0.017 0.014 0.007 ND 0.017 0.014 0.007 ND 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017 0.014 0.017	Analytical Parameter											
Arrento 0.005 0.00 0.007 0.007 N.0.007 N.0.017 N.0.048 0.005 0.017 0.005 Gate tum 0.002 N.0 0.003 0.003 0.003 N.0	-	•										
Bertum 0.008 0.10 0.008 0.11 0.008 0.12 0.003 NM 0.04 0.06 0.07 0.07 0.07 0.07 0.07 0.07 0.07 0.07 0.07 0.07 0.07 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.022 0.023 0.029 0.021 0.020 0.029 0.024 0.022 0.011 0.021 0.029 0.021 0.029 0.011 0.021 0.029 0.011 0.021 0.029 0.011 0.011 0.011 0.021 0.029 0.011 0.011 0.021 0.029 0.011 0.011 0.021 0.029 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011 0.011		0,002-0 00=	-	0.0090	0 0079	0_0050	-	0_010		0.044	0 047	
Cheastum 0.005 0.14 0.0025 0.270 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.070 0.00 NO NA NO												0.0
Lead 0.001-0.002 NO 0.023 0.23 0.27 0.040 NA 0.007 NO 00 0.022 0.02 0.02 0.03 NO										-	ND	
Herecury Solarity 0.002 ND ND <td></td> <td>0.0</td>												0.0
Bitser 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.002 0.001 0.011 0. Sitver 0.002 0.012 ND 0.002 ND												0.00
Volatile Grgenice (ug/L) Method B01 (Malogeneted Mydrocentons) Generatizion Fector 250 1 <td>Selentum</td> <td>0.002-0.004</td> <td>ND</td> <td>0,003*</td> <td>ND</td> <td>ND</td> <td>NA</td> <td>ND</td> <td>ND</td> <td>ND</td> <td>ND</td> <td></td>	Selentum	0.002-0.004	ND	0,003*	ND	ND	NA	ND	ND	ND	ND	
Mathed 501 (Mailogenetic Hydrocarbone) Concentration Factor 200 1 <th1< th=""> 1</th1<>	Silver	0.002	0.012	ND	0.004*	ND	NA	0.012	0,008*	0.010	0.011	0.0
Method B01 (Mailogenerate Hydrocerbone) Generatizion Factor 250 1 <th1< th=""> 1</th1<>	Volatila Organica (u	9/L]										
Concentration Fector 250 1	•	-	1									
Vingi Chlaride 0,18 NO				1	1	1		1	1	1	1	
Mathylane Chieride 0.25 NO NO 0.74 NO NO<		••		•	•	•		•	•	•	•	
treme-f_2-Dickicrosthylene 0.1 ND ND ND ND ND ND ND ND ND NA 4.2 ND 1.3 2.0 Ditordam 0.03 ND ND ND NA ND												
Chickorotheme 0.05 ND 4.3 3.8 ND MA ND												1
Browd folio-resettene 0.10 NO 1.8 NO NA NO 1.9 NO NO Trichio-resettiene 0.09 0.03 0.035 6.4 NA NO 0.44 0.75 0.14 0.74 0.54 0.45 NO NO 0.45 NO NO 0.45 NO N											-	
Trichisrosthylene 0.12 5600 0.26 8.4 NA 0.75 0.14 0.74 0.24 0.25 Disroschisrosthylene 0.05 NO 0.42 0.72 NO NA NO 0.45 NO NO Tetrechisrosthylene 0.45 NO				. –								
Dibramechicrimestame 0.08 ND 0.42 0.72 ND NA ND 0.45 ND ND Testenchicrimestheme 0.4 ND ND <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></td<>												
Tetrachlorosthylene 0,03 NO NO </td <td></td> <td></td> <td></td> <td></td> <td></td> <td>-</td> <td></td> <td>_</td> <td>-</td> <td></td> <td></td> <td>6</td>						-		_	-			6
Method 602 (Arcmetic Compounds) Concentration Factor 10 1 1 1 10 1 1 1 Benzene Toluene 0.2 ND ND ND ND NA 110 ND ND <td>TetrachLoroethyLen</td> <td>0.03</td> <td>ND</td> <td>ND</td> <td>ND</td> <td>0.17</td> <td>NA</td> <td>ND</td> <td>0,1</td> <td>ND</td> <td>ND</td> <td></td>	TetrachLoroethyLen	0.03	ND	ND	ND	0.17	NA	ND	0,1	ND	ND	
Concentration Factor 10 1 1 1 1 10 1	Trichlorofluoromet	hene 0,4	ND	ND	ND	ND	NA	ND	ND	ND	ND	
Benzene D.2 ND ND ND ND NA 110 ND	Hethod 602 (Arometi	c Compounds)										
Toluene 0.2 ND 3.0 0.90 10 NA ND 0.92 ND ND Mathod 624 Concentration Fector 1 1 1 1 1 1 NA	Concentration Fact	or	10	1	1	1		10	1	1	1	
Toluene 0,2 ND 3,0 0,90 10 NA ND 0,92 ND ND Method 624 Concentration Factor 1 1 1 1 1 1 1 NA	Baasaaa		-		5	5	MA	440	1 0			
Method 824 Concentration Factor 1 trans-1,2-Dichloroethyleme 1.8 NA			-	. –		-				-		5
Concentration Factor 1 trans-1,2-Dichloroethylene 1.8 NA		-		-	-				-			-
trans-1,2-Dichlorosthylane 1.8 NA		~~						•				
Benzeme 4,4 NA <		••						•				
Extractable Organica (ug/L) Mathod 825 Acid Concentration Fector 1 1 1 1 1 1 1.06 1 Phenol 1.5 2 2 NA 3 2 9 ND 2 2 Mathod 825 Base/Neutral Concentration Fector 1 1 1 1 1 1 1.06 1 M-nitrosediphenylamine 1.8 11BL ND NA ND												
Mathod 825 Acid Concentration Factor 1	benzene	4,4		-	NA	NA	NA.	35	NA	MA	NA	
Concentration Factor 1	Extractable Organica	(ug/L)										
Concentration Factor 1	Mashad 878 Andd											
Phenol 1.5 2 2 NA 3 2 9 ND 2 2 Hethod 825 Base/Neutral Concentration Fector 1 1 1 1 1 1 1 1.06 1 M-nitrosodiphenyLemine 1.9 11BL ND NA ND		or	1	1		1	1	1	1	1_06	1	
Hethod 825 Base/Neutral Concentration Fector 1 <th1< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th1<>												
Concentration Fector 1	rm en o l	1.5	5	2	NA	3	2	9	ND	2	5	
H-nitrosodiphenylmsine 1,8 118L ND NA ND	Hethod 625 Base/Neu	trel										
Bis[2-sthylhexyl]phthslate 2.5 ND ND NA ND ND <t< td=""><td>Concentration Fect</td><td>or</td><td>1</td><td>1</td><td></td><td>1</td><td>1</td><td>1</td><td>1</td><td>1,08</td><td>1</td><td></td></t<>	Concentration Fect	or	1	1		1	1	1	1	1,08	1	
Bis(2-ethylhexyljphthelate 2,5 ND ND NA ND			448)					-		<u>.</u>	-	
D1-m-butyl phthelata 2.5 ND ND NA												
Alphe redistion (pCi/L) 2 Bets redistion (pCi/L) 4-8 Gemme redistion*** (pCi/L) 9-13 Hydrocarbon Fuels (mg/L) 1 ND ND ND ND ND NA NA ND ND ND ND Oiland Greess [mg/L] 1 ND ND ND ND NA NA NA ND ND ND	• • •											
Alpha radiation (pCi/L) 2 Bata radiation (pCi/L) 4-0 Gemme radiation*** (pCi/L) 9-13 Hydrocarbon Fuels (mg/L) 1 ND ND ND ND NA ND ND ND ND ND Oil and Greese (mg/L) 1 ND ND ND ND NA NA ND ND ND	Badicantius Material	•	-	MA	-	MA	-	-	-	M A	NA	
Bete redistion (pCi/L) 4-6 Gemme redistion ^{ese} (pCi/L) 9-13 Hydrocerbon Fuels (mg/L) 1 ND ND 01Land Grasse (mg/L) 1 ND ND ND ND ND NA NA ND ND ND			ruA.							rs#		
Hydrocarbon Fuels (mg/L) 1 ND ND ND ND ND NA ND	Bets redistion (pCi.	/L] +0										
01 Land Greeses [mg/L] 1 ND ND ND ND NA NA ND ND ND	Genne redistion***	(pC1/L) 9-13										
01 Land Grease [mg/L] 1 ND ND ND ND NA NA ND ND ND	Hydrocerbon Fueis (m	a/L) 1	ND	NO	ND	ND	NA	ND	ND	ND	ND	
		-										
[Cantinued]	Oil and Greess [mg/L] 1	ND	ND	ND	ND	NA	NA	ND	ND	NC	
									(Continu	ied)		
4~187						4-197						



TY.

TABLE 4.3.20-2. (Continued)

WHOM I

1. N. M.

M.

~	Nethod staction					Manda	a Matt				
Ű	Limit	191-61	101-64	H#4-79	HH-81	P-6H	<u>pr Well</u> P~6U	HH-63	101-84	194-85	H#1-850
Date Sampled			2-26-86	2-3-96	2-7-88	1-3186	1-31-86	8-21-96	8-21-86	8-21-86	8-21-96
ield Parameters											
Temperature (^O C)		22.0	21.5	22.0	18.0	21.0	21.0	21.5	24.0	25.0	25.0
pH		7.1	7.1	6.8	6.4	7.1	5.6	8,7	6.7	6.5	6.5
Conductivity [umhos]		650	1500	1000	1500	910	1000	800	1300	760	760
malytical Parameters ⁴											
·											
Hetals (mg/L)											
	02-0.005	0.11	0.072	ND 0,082	0.007**	MD 0.095	ND 0,05	0,26	0,004*	0,095	0.004*
Berium Cedeium	0.009	0,18 0,003*	0.072 ND	0,082 ND	0.077	0.003*		0,16	0,081 ND	U_12 ND	0.003*
Chromium	0,002	0.003*	0,006*	0.02*	ND 0.050	0.007*	0.003* 0.01*	0,006* 0,009*	0.02*	0,02*	0.007*
+ · · · • • • • • • • • • • • • • • • •	0,005	0.08	0.007	ND ND	0.20	0,00/* NO®	0.003*	0,008*	0.020	0.020	0,007
	2000,0-10 2000,0	U,UB	0.007 ND	ND		ND		0.0002*	U.UUS ND	U,UZZ	0,006 NC
Hercury			0.0040	0.0030	ND ND #	NO	ND		-	NO	NC
Selenium 0.0 Silver	0.004	ND 0.020	0.003*	0.008*	0,019	0.011	ND 0.008*	0,006* ND	ND NG	NO	ND
	0,002	0,020	0.000	01004	0.018	0.011	0,000				
olatile Organics (ug/L)											N
Method 601 (Helogenated H Concentration Fector	ydrocerboi	10 (m	10	1	1	1	1	1	1	1	
		-									
Vinyl Chioride	0.18	ND	ND	ND	ND	ND	NO	ND	ND	ND	
Hethylene Chloride	0.25	ND	ND	ND	ND	ND	NO	ND	NO	ND	
trans-1,2-Dichlorosthyis		ND	ND	ND	ND	ND++	ND++	ND	0.58	ND	
Chloroform	0.05	ND	ND	ND	ND	ND	ND	ND	ND	ND	
1,1,1-Trichloroethene	0.03	ND	ND	ND	ND	ND	ND	ND	ND	ND	
Sramodich Lo romethane	0.10	ND	ND	ND	ND	ND	ND	ND	ND	ND	
Trichloroethylene	0.12	ND	75	170	ND	ND	ND	7.4	0.30	NO	
Dibromochicromethane	0.09	ND	ND	ND	ND	ND	ND	ND	ND	ND	
Tetrechloroethylane	0.03	ND	ND	13	ND	ND	ND	0,10	ND	ND	
Trichlorofluoromethene	0,4	ND	ND	ND	NC	ND	ND	1.5	1.2	1.3	
Method 602 (Arometic Comp Concentration Fector	ounds)	1	10	. 1	1	5	5	1	1	1	
Benzene Toluene	0.2 0,2	ND 2,7	ND ND	ND 1.4	ND 3.5	ND ND	ND ND	ND 4,4	ND 1.2	ND 1.4	
Method 824											
Concentration Fector											
trens-1,2-Dichloroethyle	ne 1.5	NA	NA	NA	NA	NA	NA	NA	NA	NA	
Benzene	4,4	NA	NA	NA	NA	NA	NA	NA	NA	NA	
ixtractable Organica (ug/L	}										
Nethod 825 Acid											
Concentration Factor			1	1	1	1	1	1	1	1	٩
Phenol	1.5	NA	ND	2	3	ND	3	ND	ND	ND	NC
Method 525 Been/Neutral											
Concentration Fector			1	1	1	1	1	1	1	1	٩
N-nitrosodiphenyLamine	1,9	NA	ND	ND	ND	ND	ND	ND	ND	ND	NC
Bis(2-sthythexyt)phthele		NA	ND	ND	ND	ND	ND	ND	ND	ND	NC
Di-n-butyl phthelate	2,5	NA	ND	ND	4	ND	ND	ND	ND	ND	NC
adioactive Materials	_	NA	NA	NA	NA	NA	NA				
Aiphe rediction [pCi/L]	2							5,1(<u>+</u> 4,4)		3.6(<u>+</u> 4.0)	
Beta radiation (pCi/L) Gemme radiation ^{ees} (pCi/L	4-6) 9-13							4.7(<u>+</u> 3.8) 10.7	ND ND	ND ND	7.2(<u>+</u> 2.2) NC
ydrocerbon Fuels (mg/L)	1	ND	ND	ND	ND	ND	ND	ND	ND	ND	N
					-	-	-				
il and Grease [mg/L]	1	ND	ND	ND	ND	ND	NO	ND	ND	ND	N

(Continued)

ي من

12322322

たんかいのため、たんななの

RADIAN	
TABLE 4, 3. 20-2. (Continued)	
Mathod	
Detection <u>Monitor Well</u> Limits P-ON P-10M P-10MD P-10U P-10UD P-11M P-11M	
Liait P-500 P-100 P-1000 P-1000 P-1000 P-1000 P-1000 P-1000 P-110 P-110 Dete Sempled 2-6-86 2-17-86 2-17-86 2-17-86 2-17-86 2-28-86 2-28-66 2-	
Field Permaneters	
Temperature (°C) 22.0 22.0 19.5 21.0	4 9.6
Conductivity (umhom) 720 740 825 825 800 800 700 70	
Analytical Parameters ⁶	
Motals (mg/L)	
	D ND
	50,024 D NO
Chromitum 0,005 0,007* NO NO NO NO NO NO 0,031 N	0.05.
	D 0.038
·····	D ND D ND
Seten fail 0,002-0,004 fabre fab fabre fab fabre fab f Stiver 0,002 0,007* 0,007* 0,008 0,002* 0,007* 0,004* 0,003* 0,002	
Volatile Organics (ug/L)	
Method 601 [Helogenated Hydrocarbone]	
Concentration Factor 1 1 1 1 1 1 1 1	1 1
•	D NO
• • • • • • • • • • • • • • • • • • • •	D 14 D NO
	U NU D NG
1,1,1-Trichlarosthene 0,03 ND ND ND ND ND ND ND ND ND	D ND
	D NO
	D NO D NO
•- •	D NO D NO
	D ND
	-
Method 602 (Arometic Compounde) Concentration Factor 1 1 1 1 1 1 1	1 1
Concentration region	
Benzene 0,2 ND	
Taluene 0,2 ND ND 16 2,8 ND 5,9 ND 1.	8 7.0
Nothed 824 Concentration Fector	
trana-1,2-Dichlorosthylene 1,5 NA NA NA NA NA NA NA NA NA	a na
	A NA
Extrectable Organica (ug/L)	
Nethod 625 Acid	
	1 1
	D ND
Method 825 Basa/Neutral Concentration Fector 1 1 1 1 1 1 1 1	1 1
	1 1
	D ND _
Bis(2sthylhexyl)phthelete 2,5 ND	
Di-m-butylphthalata 2,5 ND	D ND
Redioective Materiels NA NA NA NA NA NA NA NA	a na
Alphe redistion (pCi/L) 2	
Bete rediston (pC//L) 4-6	
Genue redistion*** [pCi/L] 9-13	
Hydrocarbon Fuels (mg/L) 1 ND ND ND ND ND ND ND ND ND	D ND
	_
01LendiGrease (mg/L) 1 ND NO	D NO

a = Compounds not listed were not detected. Semple detection limits are the method detection limit times the concentration

factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

NA = Not analyzed.

ŗ

ND = Not detected.

BL = Detected in reagent blank; background subtraction not performed.

BL = Detected in reagent blank; deckground section to performe. ** = Compound not confirmed by second-column enalysis. * = Value Less than five times detection limit. Errors may range up to 100 percent. * = Spike recovery not within acceptable limits. Indicates interferent. - = Date net collected, insufficient values (NR-60), emitted in field (NR-67).

D = Duplicate semple. *** = Cestum 137.

4-189

Martin S.

a de la compañía de l

4.	
PLANT	
AF	
MBLENT MONITORING,	CRITERIA
COMPARISON OF WATER QUALITY, AMBIENT MONITORING, AF PLANT	WITH APPLICABLE WATER OUALITY CRITERIA
TABLE 4.3.20-3. COMP.	TIN
TA	

			WITH APPLICABLE WATER QUALITY CRITERIA	LE WAIER U	NALITY CKI	IEKIA			
Parameter	Criteria*	HM-29	S HM-52	Sampling Location HM-54 HM-	cation HM-56	HM-58	HM-59		
METALS (mg/L)									
Arsenic Chromium Lead	0.05 0.05 0.05	0.14	0.27			0.075			
ORGANIC COMPOUNDS (ug/L)									
Trichloroethylene Tetrachloroethylene Benzene	νøν	5600		6.4	110		6.6 43		
				TABLE	TABLE 4.3.20-3.	(Continued)	0		
			Se	Sampling Location	cation				
Parameter	Criteria*	HM-61	HM-64	HM-79	HM-81	HM-83	HM-85	P-10U	P-1 1U
METALS (mg/L)									
Arsenic Chromitim	0.11	0.11			20.0	0.28	0.086		
uitom tum Lead	0.08	0.08			0.2	0.05			
ORGANIC COMPOUNDS (ug/L)									
Trichloroethylene Tetrachloroethylene Benzene Methylene Chloride	νœν		75	170 13		7.4		91	14

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.

SIMM THE PARTY PR

5



TEXAS

RADIAN

Significance of Metals in Groundwater

Four metals, arsenic, cadmium, chromium, and lead, were reported in samples from upper zone wells at concentrations that exceeded federal MCL criteria. Arsenic was detected in water sampled from HM-61 at 0.11 mg/L and in water sampled from HM-83 and HM-85 at 0.26 mg/L and 0.086 mg/L, respectively. Chromium was present in groundwater sampled from HM-29, 58, 61, and 81 at levels exceeding federal standards. The highest recorded concentration of chromium was 0.17 mg/L at HM-61. Concentrations of lead that exceeded MCLs ranged from 0.08 mg/L at HM-61 to 0.27 mg/L at HM-52. Lead was detected at HM-83 at 0.05 mg/L, which is the MCL for that metal. Concentrations of metals were below federal MCL criteria in all Paluxy wells sampled as part of ambient monitoring activities.

These observations differ slightly from the Hargis and Associates (1985) report, which reported concentrations of metals in these wells at levels below federal MCL criteria. Low concentrations of heavy metals were noted in the Paluxy wells in the Hargis study.

Significance of Organic Compounds in Groundwater

Compounds detected at levels above federal standards included trichloroethylene, tetrachloroethylene, and benzene. Groundwater samples from HM-54, 59, 64, 79 and 83 contained trichloroethylene in excess of the MCL at levels ranging from 6.4 ug/L at HM-54 to 170 ug/L at HM-79. Tetrachloroethylene was detected above the MCL of 8 ug/L at HM-59 and 79. Concentrations ranged from 13 ug/L at HM-79 to 43 ug/L at HM-59. Benzene was detected at levels exceeding federal criteria at one well, HM-56, at a reported concentration of 110 ug/L.

The Hargis and Associates (1985) report noted similar concentrations of trichloroethylene in the above-mentioned wells. However, concentrations of tetrachloroethylene and benzene were noted at different levels. Wells HM-59 and 79 reportedly contained lower levels of tetrachloroethylene, while HM-64 had a higher concentration, 400 ug/L. Furthermore, the concentration of

4-191



benzene in HM-56, noted in this report at 110 ug/L, was detected at concentrations ranging from 700 to 850 ug/L in 1985.

Hydrocarbon fuels and oil and grease were not detected in groundwater samples from all ambient monitoring wells.

These observations generally agree with those of the Hargis and Associates (1985) report, with a few exceptions. In 1985, Hargis reported levels of oil and grease of 21 mg/L, 16 mg/L and 12 mg/L in HM-56, 57, and 61, respectively. Otherwise, hydrocarbon fuels and oil and grease were reported at levels below the instrument detection limit.

Significance of Radioactive Materials in Groundwater

Levels of alpha radiation present in groundwater at HM-83, HM-84, and HM-85 ranged from <2.1 pCi/L at HM-85 to 5.1 pCi/L at HM-83. Beta radiation was detected at levels ranging from <4.2 pCi/L at HM-84 to 7.2 pCi/L at HM-85. Gamma radiation was not measured directly, but a test for Cesium 137, also a fission byproduct, was conducted. Since radiation was detected at very low levels, it was not necessary to run a test for gamma radiation. Levels of Cesium 137 were below the instrument detection limit for groundwater samples from all three wells. All values are below the corresponding federal criteria.



4.3.21 East Parking Lot Wells

This site has five wells, HM-68, HM-71, HM-82, P-8M, and P-8U, located in the vicinity of the East Parking Lot.

Investigation activities at East Parking Lot Wells entailed the collection and analysis of groundwater from three upper zone monitor wells and two Paluxy monitor wells. Figure 4.3.21-1 shows the locations of these wells.

4.3.21.1 Results of Investigation

A discussion of the results of IRP Phase II Stage 1 activities at the East Parking Lot and the topographic and geologic features of the site follows.

Topography

This area is located at the southeastern border of AF Plant 4. The surface elevation in this area is about 640 to 648 feet MSL.

Geologic Features

Lithologic logs of monitor wells P-8M, P-8U, HM-68, HM-71, and HM-82 (Hargis and Associates, 1985) provide information on the subsurface at this site.

The upper zone, from 30 to over 50 feet thick in this area, consists of silty and sandy clay, clayey sand, and sand and gravel. The Walnut Formation is 26 to 28 feet thick. The Goodland Limestone is absent.

Previous investigators (Hargis and Associates, 1985) have described a buried bedrock valley, trending east-northeast across the East Parking Lot. It has been suggested that contaminants from the plant area are moving in the



4-194



upper zone down this valley along the contact between the upper zone and the Walnut. It has been further suggested that the Walnut Formation is thin or absent in the floor of this valley, providing a "window" through which contaminants could enter the Paluxy. The Corps of Engineers was actively studying this topic at the time this report was being prepared.

Occurrence of Groundwater

Groundwater occurs in both the materials of the upper zone and in the Paluxy Formation. Table 4.3.21-1 presents results of water level measurements. The water level map for the upper zone in this area of AF Plant 4 (Figure 4.1.2-9) was discussed previously in Section 4.1.2. Water levels in the upper zone in the area of the East Parking Lot ranged from 647.09 feet MSL at HM-82 to 615.38 feet MSL at HM-68. As shown in Figure 4.1.2-9, upper zone groundwater flow in this area is to the east and northeast. The effects of the buried bedrock valley are not well displayed by the contours shown. Groundwater elevations in the Paluxy Formation ranged from 573.84 feet MSL in the upper Paluxy Formation to 559.58 feet MSL in the middle Paluxy Formation. Figures 4.1.2-2 and 4.1.2-3 shows groundwater flow in the Paluxy.

Groundwater Quality

Groundwater samples were collected from monitor wells HM-68, 71, 82, P-8M, and P-8U and analyzed for volatile organic compounds by EPA Methods 601 and 602, for acid and base/neutral extractable compounds by EPA Method 625, and for heavy metals by EPA Methods 200.7, 206.2, 239.7, 245.1, and 270.2. Table 4.3.21-2 shows the results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

To evaluate the degree of groundwater contamination, analytical results from IRP Phase II Stage 1 sampling were compared to federal standards and guidelines. Table 4.3.21-3 summarizes the results of this comparison.



TABLE 4.3.21-1. GROUNDWATER ELEVATIONS AT EAST PARKING LOT WELLS, AF PLANT 4, TEXAS (FEBRUARY 1986)

Monitor Well	Elevation (feet MSL)	Date
HM-68	615.38	2-28-86
HM-71	615.76	2-3-86
HM-82	647.09	2-4-86
P-8M	559.58	2-4-86
P-8U	573.84	2-7-86

1.2233



TABLE 4.3,21-2. RESULTS OF GROUNDWATER SAMPLE ANALYSES, EAST PARKING LOT, AF PLANT 4, TEXAS

TABLE 4.3.21-					
TABLE 4.3.21-					
TABLE 4.3.21-					
IMDLE 4.5.21-				NALYSES, E	ACT DADUTI
	LOT, AF PLA			WALISES, E	ADI FARRI
	Method				
	Detection		HM-71	Monitor We	11 P-8M
Date Sampled	Limit	HM-68 2-28-86	2-3-86	HM-82 2-4-86	2-4-86
Field Parameters					
Temperature (°C)		19.5	22.0	22.0	22.0
рН		6.8	6.8	6.8	7.3
Conductivity (umhos)	820	1100	940	810
Analytical Parameter	sa				
Metals (mg/L)					
Arsenic	0.002-0.005	ND	ND	ND	ND
Barium	0.009	0.11	0.091	0.13	0.082
Cadmium	0.002	ND	ND	ND	ND
Chromium	0.005	ND	0.14	0.33	0.006*
Lead	0.001-0.002	0.021	ND	0.004*	ND@
Mercury	0.0002	ND	ND	ND@	ND
Selenium	0.002-0.004	ND	ND	ND@	0.003*
Silver	0.002	0.004*	0.012	0.012	0.007*
Volatile Organics (u	g/L)				
Method 601 (Halogen					
Concentration Fact	or	1	100	1000	1
Methylene Chloride		0.37	ND	ND	ND
trans-1,2-Dichloro		ND**	ND**	ND**	ND
1,1,1-Trichloroeth		0.07	ND	ND	ND
Trichloroethylene	0.12	32	1300	19000	ND
Tetrachloroethylen		0.17	23	ND	ND
1,2-Dichlorobenzen	e 0.15	5.0	ND	ND	ND
Method 602 (Aromati					
Concentration Fact		1	1	500	1
Toluene	0.2	11	7.6	ND	0.64

(Continued)



	Method Detection	Monitor Well					
Date Sampled	Limit	HM-68 2-28-86	HM-71 2-3-86	HM-82 2-4-86	P-8M 2-4-86	P-80 2-7-86	
Extractable Organics (ug	g/L)						
Method 625 Acid							
Concentration Factor		1	1	1	1	1	
Phenol	1.5	ND	2	4	3	ND	
Method 625 Base/Neutral	L						
Concentration Factor		1	1	1	1	1	
1,2-Dichlorobenzene	1.9	2	ND	ND	ND	ND	
Di-n-butyl phthalate	2.5	ND	ND	4BL	ND	ND	
Hydrocarbon Fuels (mg/L)) 1	ND	ND	ND	ND	ND	
Oil and Grease (mg/L)	1	ND	ND	ND	ND	ND	

TABLE 4.3.21-2. (Continued)

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables 4.2.2-2, 4.2.2-3, and 4.2.2-4.

* = Value less than five times detection limit. Errors may range up to 100 percent.

** = Compound not confirmed by second-column analysis.

@ = Spike recovery not within acceptable limits. Indicates interferent.

ND = Not detected.

BL = Detected in reagent blank; background subtraction not performed.



Later and a state of the state of the

Ń

H-1000000

h

TABLE 4.3.21-3. COMPARISON OF WATER QUALITY, EAST PARKING LOT, AF PLANT 4, TEXAS WITH APPLICABLE WATER QUALITY CRITERIA

Parameter		Sampling Location			
	Criteria*	HM-68	HM-71	HM-82	P-8U
METALS (mg/L)					
Chromium	0.05		0.14	0.33	
ORGANIC COMPOUNDS (ug/L))				
Trichloroethylene	5	32	1300	19000	7200
Tetrachloroethylene	8		23		

* See Tables 4.2.1-1 and 4.2.1-2 for source of criteria.



4.3.21.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis and Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

Significance of Metals in Groundwater

Groundwater in the upper zone and in the Paluxy Formation at the East Parking Lot contained concentrations of heavy metals generally below federal MCLs. The one exception is chromium, which was detected at levels exceeding the MCL of 0.05 mg/L in HM-71 and HM-82. The detected value at HM-71 was 0.14 mg/L. At HM-82 the detected level of chromium was 0.33 mg/L. The results of previous efforts by Hargis and Associates (1985) agree with present observations.

Significance of Organic Compounds in Groundwater

Trichloroethylene and tetrachloroethylene were detected in upper zone water at this site in levels in excess of federal MCL criteria. An unknown compound, with a retention time near that of trans-1,2-dichloroethane, was also present in high concentrations in upper zone groundwater. The highest concentration of TCE observed was 19,000 ug/L in HM-82.

Groundwater samples collected from two Paluxy aquifer wells revealed the presence of organic compounds at levels exceeding federal MCL criteria only in the upper zone of the Paluxy. The middle member had no organic compounds in excess of federal criteria. Trichloroethylene was detected at 7200 ug/L in P-8U. An unknown compound with a retention time near that of trans-1,2-dichloroethane was also detected in P-8U.



These observations differ somewhat from those of the Hargis and Associates (1985) report. In the previous report, TCE was detected at HM-68 at lower levels and other purgeable halocarbons and aromatics were reported as not detected. Similarly, at HM-71 only TCE was reported at a detectable level (1200 ug/L and 2500 ug/L for separate sampling episodes in 1985). In addition, in the 1985 report, levels of TCE in groundwater from HM-82 were noted at much lower levels than in this report. The reported value in 1985 was 9500 ug/L, while this study shows the concentration of TCE at HM-82 to be 19,000 ug/L.

The results of this study and those of previous work support the suggestion that contaminants are entering the upper Paluxy through a window in the Walnut. The detailed investigation being conducted by the Corps of Engineers will establish the flow relationships with more certainty.



4.3.22 Fuel Storage Tank

This site has been the location of a 100,000-gallon above-ground JP-4 storage tank since 1962.

Investigation activities for the Fuel Storage Tank site entailed the collection of a groundwater sample from the one upper zone well in the area, HM-23. The sample was analyzed for hydrocarbon fuels by EPA Method 418.1 and oil and grease by EPA Method 413.2. Figure 4.3.22-1 shows the location of HM-23.

4.3.22.1 Results of Investigation

A discussion of the results of IRP Phase II Stage 1 activities at the Fuel Storage Tank and the topographic and geologic features of the site follows.

Topography

The Fuel Storage Tank is located just north of Clifford Avenue in the southwest corner of the Plant 4 facility. The ground surface is relatively flat and has an elevation of approximately 680 feet MSL.

Geologic Features

The lithologic log of monitor well HM-23 (Hargis and Associates, 1985) gives shallow subsurface information for the Fuel Storage Tank site.

The upper zone is about 42 feet thick and consists of silty to sandy clay grading down into sand and gravel and clayey sandy gravel. About 8 feet of Walnut Formation were encountered. The Goodland Limestone is absent at this location.



4-203



Occurrence of Groundwater

The water level elevation at HM-23 was 638.19 during February, 1986. The water level map of the upper zone in this area of AF Plant 4 (Figure 4.1.2-8) was discussed previously in Section 4.1.2. As shown on this map, upper zone groundwater flow in this area is to the north.

Groundwater Quality

Groundwater sampled from HM-23 was analyzed for hydrocarbon fuels and oil and grease. Table 4.3.22-1 presents results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

4.3.22.2 Significance of Findings

This section discusses the significance of the findings of this investigation, emphasizing the groundwater chemistry at the site. Hargis and Associates (1985) previously discussed the significance of groundwater quality data collected in 1985 and earlier. Where appropriate, this report recognizes and incorporates those findings and trends identified by the Hargis report.

Significance of Organic Compounds in Groundwater

Hydrocarbon fuels and oil and grease were not detected in the upper zone groundwater at this site.

The Hargis and Associates (1985) report also notes low levels of these compounds; however, in 1984, hydrocarbons were detected at 5.7 mg/L and oil and grease was detected at 7.1 mg/L.

TABLE 4.3.22-1.	RESULTS OF GROUNDWATER SAMPL STORAGE TANK, AF PLANT 4, TE	
Date Sampled	Method Detection Limit	Monitor We 2-17
Field Paraméters		
Temperature (°C) pH Conductivity (umhos)		22 6 5
Analytical Parameters		
Hydrocarbon Fuels (mg/L)	1	
Oil and Grease (mg/L)	1	
ND = Not detected.		

B



4.3.23 Lake Worth Monitor Well

This site comprises one monitor well installed along the northern area of AF Plant 4 that borders Lake Worth. A monitor well was installed at this location to determine the occurrence and character of groundwater in the vicinity of Lake Worth.

Investigation activities at this site included the installation of a Paluxy monitor well completed in the upper member of the formation, and the collection of two sets of water samples, approximately four months apart. Figure 4.3.23-1 shows the location of this Paluxy well, P-23U. Groundwater samples were analyzed for volatile organic compounds by EPA Methods 601 and 602, for acid and base/neutral extractable compounds by EPA Method 625, and for heavy metals by EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2.

4.3.23.1 Results of Investigation

A discussion of the results of the IRP Phase II Stage 1 activities for the Lake Worth monitor well and descriptions of the topographic and geologic features of the site follow.

Topography

The Lake Worth well is located in the northwest corner of the AF Plant 4 facility. The land surface near well P-23U slopes down rather steeply toward Lake Worth. Elevations range from 620 feet MSL at P-23U to 590 feet MSL at Lake Worth.

Geologic Features

The upper zone is absent at this site. The Walnut Formation was encountered at the surface when drilling and is about 25 feet thick. It is



4-207



composed of layers of fossiliferous limestone and calcareous shale. Monitor Well P-23U penetrates approximately 20 feet of sand in the Upper Paluxy Formation.

Occurrence of Groundwater

Groundwater at this site occurs in the upper member of the Paluxy Formation. The water level elevation at P-23U was 587.71 feet MSL in April 1986 and 588.35 feet MSL in August, 1986. Groundwater in the upper member of the Paluxy Formation flows to the east-southeast in this area.

Groundwater Quality

Groundwater samples were collected and analyzed from monitor well P-23U at two separate sampling periods; April, 1986, and August, 1986. Samples were analyzed for volatile organic compounds (EPA Methods 601 and 602), for acid and base/neutral extractable compounds (EPA Method 625), for heavy metals (EPA Methods 200.7, 206.2, 239.2, 245.1, and 270.2), for hydrocarbon fuels (EPA Method 418.1), and for oil and grease (EPA Method 413.2). Table 4.3.23-1 presents results of the groundwater analyses. Complete analytical reports for these analyses appear in Appendix A.

4.3.23.2 Significance of Findings

This section discusses the significance of the findings : ... investigation, emphasizing the groundwater chemistry at the site.

Significance of Metals in Groundwater

All metals detected in groundwater at F-230 were trations below federal MCL criteria.

Significance of Organic Compounds it is a final state of Analysis for organic compounds terms of number of compounds, all at concentrations of the state of the s





	Method	Monitor Well		
Date Sampled	Detection Limit	P-23U 4-18-86	P-23U 8-21-86	
Field Parameters				
Temperature (°C)		-	-	
pH		6.3	-	
Conductivity		1000	-	
Analytical Parameters ^a				
Metals (mg/L)				
Arsenic	0.002-0.005	0.015	0.017	
Barium	0.009	0.06	0.036	
Cadmium	0.002	ND	0.002*	
Chromium	0.005	0.02*	0.006*	
Lead	0.001-0.002	0.023	0.007	
Mercury	0.0002	ND	0.0004*	
Selenium	0.002-0.004	ND(0.03)	ND	
Silver	0.002	0.01*	ND	
Volatile Organics (ug/L)				
Method 601 (Halogenated Hydro	carbons)			
Concentration Factor		1	1	
Trichlorofluoromethane	0.40	ND	1.9	
Method 602 (Aromatic Compounds	s)			
Concentration Factor		1	1	
Toluene	0.2	7.0	3.6	
Ethyl Benzene	0.2	1.2	ND	
Extractable Organics (ug/L)				
Method 625 Acid ^b				
Concentration Factor		1	1	
Method 625 Base/Neutral				
Concentration Factor		1	1	
Bis(2-ethylhexyl)phthalate	2.5	3	4	
Di-n-buty1 phthalate	2.5	3BL	ND	

TABLE 4.3.23-1. RESULTS OF GROUNDWATER SAMPLE ANALYSES, LAKE WORTH WELL (P-23), AF PLANT 4, TEXAS 2

ND = Not detected, values in parentheses are actual sample detection limits.

BL = Detected in reagent blank; background subtraction not performed.

- = Data not collected, omitted in field.

a = Compounds not listed were not detected. Sample detection limits are the method detection limit times the concentration factor. Method detection limits are found in Tables X-1, X-2, and X-3.

b = No compounds were detected by this analysis.

* = Value less than five times detection limit. Errors may range up to 100 percent.

RADIAN

4.3.24 White Settlement Groundwater Pumping Effects

An investigation of of the pumpage effects of White Settlement's municipal wells was performed on the groundwater to determine the zone of influence, long-term drawdowns (if any) and the influence on the direction of groundwater flow in the Paluxy aquifer beneath the Plant 4 area. Groundwater flows and drawdowns were determined by reviewing existing geologic and hydrologic information and current pumpage records for the City of White Settlement. These data were supported by comparing actual water level data to simulated results from a simple finite-difference steady-state groundwater model (Koch, 1984). Background data used in executing the simulated pump test of the Paluxy aquifer were gathered from a variety of sources, including the City of White Settlement Engineers Office, historical records of the AF Plant 4 Paluxy wells, previous groundwater investigations conducted by General Dynamics, Phase II results, and existing published literature.

Model Application

The application of a groundwater model to the Paluxy aquifer involves data collection, data input for the model, historical matching and predictive simulation.

As a first step, initial estimates of Paluxy aquifer parameters were compared to known regional aquifer parameters by running the model and matching the results with existing conditions in the AF Plant 4 area. By trial and error, the estimated values for aquifer properties were calibrated and refined. A predictive simulation using the estimated hydraulic conductivity, aquifer thickness, specific yield and discharge was then run to compute drawdown in the AF Plant 4 area.

Results of the predictive simulation were compared with observed drawdown. It was not necessary to modify the input data to fine tune the results of the simulated pump test.


The nearest municipal wells to AF Plant 4 are wells WS-6 (re-designated as WS-2 in 1982) and WS-12. Well WS-2 is located about 850 feet west of the plant, and well WS-12 is located about 1500 feet south of the plant (Hargis and Montgomery, 1983). Well records of the Paluxy and Twin Mountain wells are presented in Table 4.3.24-1 below.

The total daily consumption of groundwater from these wells is approximately 750,000 gallons per day. The average yield per well is about 83,000 gallons per day. Currently, the City of White Settlement withdraws approximately 552 acre-feet of water from the Paluxy and about 27 acre-feet from the Twin Mountains aquifers (Otrosky, 1986).

Because of the projected increase in the demand for groundwater by the community and by Fort Worth to the east, the city is planning to install an additional three (3) wells in the next five years. All three wells will be screened in the Twin Mountain aquifer: two wells at existing Paluxy well sites (WS-2 and WS-H3), and the other in Rockwall County.

Previous Pump Test Data

Results of pumping tests conducted on three Paluxy wells within two miles of AF Plant 4 indicate that the transmissivity of the Paluxy ranges between 3100 and 7500 gallons per day per foot (Nordstrom, 1982). Specific yield was estimated to be on the order of 0.15 to 0.20 for the Paluxy. The average rate of water movement in the Paluxy was calculated to be less than 2 feet per year in a south-easterly direction. The present hydraulic gradient is east-southeast, approximately 27 feet per mile (Nordstrom, 1982).

Historical water level records show that groundwater levels have not declined appreciably in outcrop areas of the Paluxy. The aquifer is under water-table conditions in the outcrop area, and observation wells show only minor fluctuations from year to year (Hargis and Montgomery, 1983). This is not unusual since Lake Worth provides a source of constant recharge to the



Well Number	Date Drilled	Total depth (ft)	Aquifer	Discharge (gpd)
WS-1	1945	254	Paluxy	96,000
WS-3	1950	201	Paluxy	70,000
WS-5	1951	720	Twin Mtn.	123,000
WS-5A	1966	305	Paluxy	65,000
WS-6*	1947	170	Paluxy	70,000
WS-8	1966	286	Paluxy	100,000
WS-10	1952	950	Twin Mtn.	120,000
WS-12	1965	195	Paluxy	55,000
WS-H3	1943	282	Paluxy	51,000

TABLE 4.3.24-1. CITY OF WHITE SETTLEMENT WELL DATA

1.179.22

122234

12.22.23

STATES STATES

1.02032

22222

122222

Source: City Engineer's Office, White Settlement *Designation changed to WS-2 (1982)



Paluxy Sand. However, further downgradient in the area of the municipal well field (south of AF Plant 4) groundwater declines of up to 2 feet per year have been observed (Nordstrom, 1982). Because of the growing demand for groundwater by White Settlement and Fort Worth, additional drawdowns of the Paluxy can be expected.

Past Groundwater Flow Investigations

Several investigations have documented the influence of groundwater withdrawals on flow conditions in the Paluxy aquifer. The average rate of water movement in the Paluxy is less than 2 feet per year in an east-southeast direction, except in areas of heavy pumpage (Leggat, 1957). Pump test results indicate that the normal flow patterns within the Paluxy are affected by the withdrawal of groundwater by White Settlement, causing the flow to move toward the pumped wells (Nordstrom, 1982). In the immediate area of AF Plant 4, recent investigations have noted that the City of White Settlement's nine municipal wells create a cone of depression, which in turn induces groundwater to flow in a southerly direction toward the well field and from the plant area (Hargis and Montgomery, 1983).

Aquifer Characteristics

Investigations of the Paluxy aquifer indicate that the average hydraulic conductivity of the Paluxy aquifer is about 44 gallons per day/foot² (Nordstrom, 1982). The specific yield in the outcrop area is about 15 percent. Using this information, combined with geological data on the Paluxy Sand and daily discharge estimates provided by the City of White Settlement (Otrosky, 1986), a simulated pump test was run on six Paluxy wells in the White Settlement-AF Plant 4 area.

4.3.24.1 Background Information

The following sections present the background information used to determine groundwater flow and drawdown in the White Settlement-AF Plant 4 areas.

Geologic Data

The Paluxy Sand (Cretaceous) is part of the Trinity Group, which includes the Travis Peak Formation, Glen Rose Limestone and Paluxy Sand. The Paluxy Sand ranges in thickness from 140-190 feet, with an average thickness of 160 feet in Tarrant County. The Paluxy consists of fine-grained, wellsorted unconsolidated sand with minor amounts of pyrite, lignite and sandy clay and shale. Previous studies have divided the Paluxy into three sand members (upper, middle, and lower) in the local area (Hargis and Montgomery, 1983). Most wells in the White Settlement area obtain groundwater from the middle and lower sand members; however, for the purpose of modeling the drawdown and flow near these wells, the Paluxy Sand was assumed to be a homogeneous aquifer 160 feet thick.

The Paluxy Sand outcrops north of White Settlement in west-central and northwest Tarrant County. The Paluxy dips uniformly at 7 degrees to the southeast at a rate of approximately 35 to 40 feet per mile (Leggat, 1957). Locally, the Paluxy Sand outcrops at the shoreline of Lake Worth west of AF Plant 4 and along the Live Creek drainage basin. No faults are known to exist (based on drill hole information) in the Paluxy underlying AF Plant 4 (Hargis and Associates, 1985).

Groundwater

The Paluxy aquifer, an important source of groundwater for the City of White Settlement, supplies much of the domestic and industrial water



required by the city. Groundwater within the Paluxy aquifer exists in both water table and artesian conditions. Along outcrop and shallow subcrop areas (near White Settlement), the Paluxy aquifer exists as a water table. Downgradient from these outcrop areas, increased hydrostatic pressures caused by overlying and less permeable strata, including the Walnut and Goodland Formations, create artesian conditions within the Paluxy aquifer. Pump test records from the City of White Settlement indicate that the Paluxy aquifer underlying the city exists as a water table.

<u>Groundwater Recharge</u> The major source of recharge to the Paluxy is precipitation and the subsequent infiltration of surface waters along the outcrop areas. Other sources of recharge are stream and reservoir leakage into the Paluxy. The Paluxy forms the lake bottom for Lake Worth, which, it is estimated, contributes about 600,000 gallons/day to the Paluxy aquifer (Leggat, 1957).

<u>Groundwater Discharge</u> Groundwater discharge from the Paluxy occurs near outcrop areas through evapotranspiration and uptake by vegetation. Other natural sources of discharge occur in valleys or road cuts in the form of springs and seeps in outcrop areas. Withdrawal by municipal and private wells in the White Settlement area effects artificial discharge of groundwater from the Paluxy aquifer.

Municipal Water Use

The City of White Settlement obtains groundwater from the Paluxy aquifer and deeper Twin Mountains aquifer (Trinity Group). Currently, the City of White Settlement owns seven municipal wells in the Paluxy aquifer and two deeper wells in the Twin Mountains aquifer. Water from the Paluxy and Twin Mountains aquifers is suitable for domestic, public, stock and some industrial supplies (CH2M Hill, 1984).



4.3.24.2 Simulated Pump Test

After reviewing the information available for estimating groundwater flow and drawdown within the aquifer, it was decided that modeling techniques would be most useful in solving the problem of simultaneously pumping wells near a constant head lake (Lake Worth). The purpose of the simulated pump test was to model the withdrawal of groundwater by White Settlement and assess the direction of groundwater flow and drawdown in the Paluxy aquifer beneath AF Plant 4.

Model Selection

The model selected for this study is based on the steady-state solution of the problem of discharging finite length line sinks (wells) in a homogeneous aquifer of infinite extent. The model, developed by Koch and Associates (Koch, 1984), solves simple drawdown problems involving both constant head (Lake Worth) and constant flow (municipal wells). The resultant drawdowns at selected observation points are also determined. For this study, 30 observation points were selected in the White Settlement-AF Plant 4 area to assess the areal extent of drawdown over time.

Model Operation

The model may be used to predict drawdowns around continuously pumped wells. The well dimensions are represented by a finite line sink one foot in diameter. Given the current (1986) well discharges of the six Paluxy wells, drawdown can be predicted at any point in time. The algorithm is more accurate as time progresses because of the steady state approximation inherent in the program.



Model Assumptions and Limitations

Following are the assumptions and limitations inherent in the hydrologic model selected for the simulated pump test of the Paluxy aquifer:

- o The finite line sinks (wells)/sources (Lake Worth) fully penetrate the aquifer
- o The potentiometric surface is assumed to have a steady state curvature.
- o The aquifer is isotropic and homogeneous. The aquifer is unconfined and the Darcy assumptions are assumed to be valid.
- Lake Worth is assumed to be a constant head, zero drawdown line source.
- o The three sand units within the Paluxy are treated as one aquifer 160 feet thick.

Raw Input Data

Geologic and hydrologic data selected for the model were derived mainly from the existing sources of information discussed earlier, which were carefully checked for accuracy and correctness before incorporation into the model.

To systematize the model, a grid coordinate system was developed whereby White Settlement well, WS-H3 was arbitrarily chosen as the origin. Using existing plant maps, observation points, AF Plant 4 boundary lines, and Lake Worth were located by X-Y coordinates in feet from the origin. Table 4.3.24-2 summarizes of the input parameters used in the finite difference

Sink No.	Well No.	X/Y location (feet from origin)	Flow (gpd) ^a
1	WS-H3	1,1	51,000
2	WS-12	2050,570	55,000
3	WS-8	950,3050	100,000
4	WS-3	900,4100	70,000
5	WS-1,	2600,5100	96,000
6	WS-6 ^D	6900,3900	70,000
7	Lake Worth	8200,1-10,000	
Aquifer Char	acteristics		
l. Water tab 2. Hydraulic 3. Specific y	le conductivity, 44 gpd/ yield, 15 percent uifer thickness, 160 f		

TABLE 4.3.24-2. MODEL INPUT PARAMETERS

^aSource: City Engineer's Office, White Settlement.

^bDesignation changed to WS-2 (1982).

States to take to

والمحاربة والمحاربة والمحاوية والمحاوية والمحاوية والمحاوية والمحاوية والمحاوية والمحاوية والمحاوية والمحاوية



model. The following discussions are based on the simulated pump test results.

Model Results

The pumping test was conducted by observing simulated drawdowns from five Paluxy wells in the AF Plant 4 area. Since one well, WS-8 (old WS-16), is now abandoned and is not expected to return to operating conditions in the future, it was deleted from the pump test. The well is located approximately 2750 feet south of Plant 4. During operation, it supplied about 100,000 gallons a day to the City of White Settlement.

Another Paluxy well, WS-8 (Park), located approximately 6700 feet southwest of AF Plant 4, was deleted from the study because its radius of influence would not sufficiently alter the groundwater flow gradient or drawdowns in the AF Plant 4 area. This well produces about 100,000 gallons a day.

To predict the drawdown over time within the Paluxy aquifer, all wells were assumed to begin pumping at the same time and to be simultaneously pumping at the current capacity for a period of 20 years. Drawdowns were predicted after years 1, 3, 5, 10, 15 and 20 and plotted on maps using CPS-1[®], a Radian plotting system (Figures 4.3.24-1 through 6). Results of these predictions are described in the following paragraphs.

Year 1 (Figure 4.3.24-1) During the first year of the pump test drawdowns are seen in the vicinity of the pumping wells south of AF Plant 4, but little effect is seen within Plant 4. As expected, the largest drawdowns occur near the pumping wells. Little or no drawdown occurs near the Lake Worth shoreline. In the raw output of the drawdown data, the model at some points predicts negative numbers (in the recharge area near Lake Worth). This reflects a net gain of groundwater, indicating that recharge is occurring from Lake Worth at a rate greater than the loss created by the pumping wells.



AND A REPEACE REPORTED REPORTED REPORTED REPORT

L'ELECTION ELECTION

Saw www

AND STREET

4-220

RAPIAN

Year 3 (Figure 4.3.24-2) After three years, drawdowns of almost 5 and 6 feet are seen in the vicinity of wells WS-12 and WS-3. Drawdowns ranging from 1.9 to 4.7 feet occur along the southern boundary of AF Plant 4.

The shoreline area of Lake Worth again shows no drawdown effects from well pumpage except in the area adjacent to well WS-6, where withdrawal of water by the well exceeds recharge from Lake Worth to the north.

Year 5 (Figure 4.3.24-3) Trends similar to those observed in Year 3 are also seen in Year 5. The most significant drawdowns occur near wells WS-H3, WS-3 and WS-1. Within AF Plant 4, the drawdown after 5 years ranged from less than 1 to 6 feet, in the northeast and southwestern areas of the plant, respectively.

Year 10 (Figure 4.3.24-4) After 10 years, the drawdown near well WS-3 is approximately 10 feet. Drawdowns along the southeastern boundary of AF Plant 4 range from 5.5 feet in the southeastern corner to 8 feet in the southwestern corner, indicating flow is being directed to the southwest in the southern portion of AF Plant 4. In the central AF Plant 4 area, drawdown ranges from about 2.5 feet to 4.5 feet, increasing to the west in the direction of well WS-6.

Year 15 (Figure 4.3.24-5) After 15 years, drawdowns are still most pronounced near wells WS-H3, WS-3, and WS-1. Along the southern AF Plant 4 boundary, drawdowns are between 6 feet in the southeast corner and 8.5 feet in the southwest corner of the property. Consistent with previous years, the groundwater flows to the south toward wells WS-3 and WS-12 and toward well WS-6 west of the plant area.

Year 20 (Figure 4.3.24-6) After 20 years the drawdowns begin to achieve steady state conditions. The predicted drawdowns near wells WS-H3, WS-3, and WS-1 show very little change from year 15. Drawdown along the southern perimeter of Plant 4 ranges between about 6.5 feet in the





LC: NY SI

REALESS RELEVE

1999 Sec. 19

22222231 3222203

2222227 5525-151

11.91.4



ATTA PERSON PLATED IN THE SAME SAME A VILLES PANDER PERSON PERSON PANDER PERSON PANDER PANDER

 $X_{1}^{\prime} > 2$



- 22.22.24 - 52.65.55 - 55.55.55

Sec. 265, 647, 667, 687,

4-225

I Simulated Drawdowns in the Paluxy Aquifer from White Settlement Municipal Wells Year 20 10000 DRAMDØWN. FEET LAKE WØRTH LINE SOURCE 9000 8000 0 ٦. 7000 n 'S 6000 5000 FEET 4 P**L**ANT 4000 3000 0 ŝ S M ₿ :0, 2000 NS-1000 ε Γ S `0. S WS-HI30 07 n .8 C 1000 Figure 4.3.24-6. 0009 ooos oooz 0008 COOL 0005 3000 T333

ኯዾኯፚኯፚኯፚኯፚኯፚኯፚኯፚኯፚኯፚ



southeastern corner to 9 feet in the southwestern corner of the plant. Drawdown remains fairly steady in the area adjacent to well WS-6; however, Lake Worth continues to recharge this area at a rate approximately equal to the rate of discharge from the well.

4.3.24.4 Conclusions

Preliminary results of the simulated pump test, as well as existing information related to groundwater flow in the Paluxy aquifer, indicate that groundwater in the Middle Paluxy aquifer beneath the Plant 4 area is influenced to some degree by the withdrawal of water by nearby municipal wells.

Figures 4.3.24-1 through 4.3.24-6 show that fairly significant drawdowns can be expected near pumping wells, which in turn could direct groundwater to flow from Lake Worth and AF Plant 4 toward these wells. Considering water quality, there appears to be a logical concern about the potential for contaminants originating at AF Plant 4 to migrate toward these wells.

Since the pump test assumes constant flow rates on a year-to-year basis, we conclude that the model indicates the worst case (i.e., maximum likely water flow rate without regard for contamination content, if any) rather that actual conditions. For instance, some wells expected to shut down for repairs may be completely abandoned. According to the City 4 Engineer's office at White Settlement, no additional Paluxy wells will be drilled by the city and no increase in pumping rate is expected. Therefore the model depicts the maximum expected municipal usage of the Paluxy aquifer in the area of AF Plant 4.

RAPIAN

4.3.25 Well Abandonment

The Statement of Work requested that available well abandonment techniques be evaluated and a candidate method recommended. The basic concept governing the proper sealing of abandoned wells is to restorate, as far as feasible, the hydrogeologic conditions existing before the well was drilled. Any well to be permanently abandoned should be completely filled in such a way that vertical movement of water within the well bore, including vertical movement of water within the annular space surrounding the well casing, is effectively and permanently prevented and the water is permanently confined to the specific zone in which it originated.

The first step in such a process is to comply with applicable regulations. The relevant portion of the Texas Administrative Code is quoted in Table 4.3.25-1. Given the nature of Plant 4 groundwater conditions, Radian strongly recommends filling the well with concrete.

Grouting should placed from the bottom of the well upward, using a tremie pipe or dump bailer, not by pouring from the land surface. A volume of material sufficient to fill the casing and the voids in the sand pack should be calculated and that volume placed in the well. For upper zone wells, emplacement followed by topping off (if required) should be sufficient.

For the Paluxy wells, the sealing material should be placed within the casing and then forced out into the sand pack under pressure, which should be maintained long enough for the cementing mixture to set. This step is particularly important for those Paluxy wells penetrating more than one zone of the aquifer. Some pre-grouting testing will probably be required to determine the proper water-cement mixture that would ensure effective permeation of the sand pack.

Surface features (guard pipes, well pads, etc.) should be removed and a cap of concrete poured at the site.



TABLE 4.3.25-1. EXCERPT FROM TEXAS WATER WELL DRILLERS ACT RELATED RULES, TEXAS ADMINISTRATIVE CODE SECTIONS 319.49 AND 50

319.49. Standards For Plugging Wells That Penetrate Undesirable Water zones.

- (a) If the use of a water well that penetrates undesirable water is to be permanently discontinued, all of the removable casing shall be removed from the well and the entire well filled with cement to the land surface.
- (b) In lieu of the procedure in subsection (a) of this section, either the zone(s) contributing undesirable water, or the fresh water zone(s), shall be isolated with cement plugs and the remainder of the wellbore filled with sand, clay, or heavy mud to form a base for a cement plug extending from land surface to a depth of not less than ten (10) feet.
- 319.50. Standards for Plugging Existing Wells.
- (a) If the use of a well that does not contain any undesirable-water zones is permanently discontinued, all of the removable casing shall be removed from the well and the entire well filled with cement to the land surface.
- (b) In lieu of the procedure in subsection (a) of this section, the well may be filled with fine sand, clay, or heavy mud followed by a cement plug extending from land surface to a depth of not less than ten (10) feet.



4.3.26 Well Inventory

The Statement of Work required that an inventory of monitor wells be made on a site-by-site basis, associating monitor wells with each site, as appropriate. The complete inventory, shown in Table 4.3.26-1, was accomplished by inspecting the most recent well location map supplied by General Dynamics. Table 4.3.26-2 lists the monitor wells by site. 1.0.110

Well	Well	Well	Well	Well	Well
P-1	HM-1	HM-25	HM-51	HM-77	F-208
P-2	HM-2	HM-26	HM-52	HM-78	F-209
P-3	HM-3A	HM-27	HM-53	HM-79	F-210
P-4	HM-3B	HM-28	HM-54	HM-80	F-211
P-5U	HM-4A	HM-29	HM-55	HM-81	
P-5M	HM-4B	HM-30	H M -56	HM-82	
P-6U	HM-5	HM-31	HM-57	HM-83	
P-6M	HM-6	HM-32	HM-58	HM-84	
P-7U	HM-7	HM-33	HM-59	HM-85	
P-7M	HM-8	HM-34	HM-60	HM-100	
P-8U	HM-9	HM-35	н м –61	HM-101	
P-8M	HM-10	HM-36	HM-62	HM-102	
P-9U	HM-11	HM-37	HM-63	HM-103	
P-9M	HM-12	HM-38	HM-64	HM-104	
P-10U	HM-13	HM-39	HM-65	HM-105	
P-10M	HM-14	HM-40	HM-66	HM-106	
P-11U	HM-15	HM-41	HM-67	HM-107	
P-11M	HM-16	HM-42	HM-68	HM-108	
P-12U	HM-17	HM-43	HM-69	F-200	
P-12M	HM-18	HM-44	HM-70	F-201	
P-13U	HM-19	HM-45	HM-71	F-202	
P-13M	HM-20	HM-46	HM-72	F-203	
P-20M	HM-21	HM-47	HM-73	F-204	
P-21U	HM-22	HM-48	HM-74	F-205	
P-22U	HM-23	HM-49	HM-75	F-206	
P-23U	HM-24	HM-50	HM-76	F-207	

TABLE 4.3.26-1. COMPLETE WELL INVENTORY, AF PLANT 4

URXER

TANKA TANAN MANAN KANAN KANAN KANAN KANAN KANAN KANAN KANAN KANAN KANAN

10120-001

N. W. W. W.

Well	Status	Aquifer	Date Installed
Site 1, Landfil	1 1		
HM-6	Destroyed	Upper Zone	-
HM-7	Active	Upper Zone	12-4-82
HM-10	Active	Upper Zone	12-6-82
HM-18	Active	Upper Zone	3-31-83
HM-20	Active	Upper Zone	4-1-83
HM-62	Active	Upper Zone	4-18-84
HM-63	Active	Upper Zone	4-18-84
P-4	To be abandoned	Paluxy	6-15-83
P-7U	Active	Upper Paluxy	1-6-85
P-7M	Active	Middle Paluxy	2-13-85
Site 3, Landfil	<u>1 3</u>		
HM-21	Active	Upper Zone	4-1-83
HM-26	Active	Upper Zone	5-25-83
HM-27	Active	Upper Zone	5-25-83
HM-34	Active	Upper Zone	4-11-84
HM-35	Active	Upper Zone	4-11-84
HM-36	Active	Upper Zone	4-11-84
HM-37	Active	Upper Zone	4-11-84
HM-38	Active	Upper Zone	4-11-84
HM-39	Active	Upper Zone	4-11-84
P-22U	Active	Upper Paluxy	3-20-86
Site 12, Chrome	Waste Pit 3		
HM-1	Destroyed	Upper Zone	12-1-82
HM-13	Active	Upper Zone	3-29-83
HM-15	Active	Upper Zone	4-1-83
HM-16	Active	Upper Zone	4-1-83
HM-17	Active	Upper Zone	3-31-83
HM-30	Active	Upper Zone	5-26-83
HM-32	Active	Upper Zone	4-13-84
HM-41	Active	Upper Zone	4-12-84
HM-45	Active	Upper Zone	4-13-84
P-2	To be abandoned	Paluxy	6-17-83

TABLE 4.3.26-2. WELL INVENTORY OF AF PLANT 4, ARRANGED BY SITE

(Continued)

58

Well	Status	Aquifer	Date Installe
ite 17, Former	Fuel Storage Area		
HM-8	Active	Upper Zone	12-5-82
HM-14	Active	Upper Zone	3-30-83
HM-100	Active	Upper Zone	1-20-86
ite 2, Landfill	2		
HM-2	Active	Upper Zone	12-2-82
HM-22	Active	Upper Zone	5-23-83
HM-40	Active	Upper Zone	4-12-84
HM-42	Active	Upper Zone	4-12-84
HM-43	Active	Upper Zone	4-12-84
HM-44	Active	Upper Zone	4-13-84
HM-46	Active	Upper Zone	4-14-84
P-21U	Active	Upper Paluxy	3-10-86
ite 4, Landfill	4		
HM-5	Active	Upper Zone	12-2-82
HM-9	Active	Upper Zone	12-5-82
HM-101	Active	Upper Zone	1-21-86
P-20M	Active	Middle Paluxy	3-4-86
one 1: Site 13,	Die Pits; Site 11, Chrom	e Pit 2; Site 8, FDT.	<u>A 5</u>
HM-3A	Destroyed	Upper Zone	12-2-82
HM-3B	Destroyed	Upper Zone	12-3-82
HM-4A	Destroyed	Upper Zone	12-3-82
HM-4B	Destroyed	Upper Zone	12 5 02
HM - 11	Active	Upper Zone	3-28-83
HM-12	Active	Upper Zone	3-29-83
HM = 12 HM = 24	Active		
		Upper Zone	5-24-83
HM-25	Active	Upper Zone	5-25-83
HM-28	Active	Upper Zone	5-26-83
HM-60	Active	Upper Zone	4-17-84
HM-77	Active	Upper Zone	1-10-85
P-1	To be abandoned	Paluxy	3-9-83
ite 15, Fuel Sa	turation Area 2		

TABLE 4.3.26-2. (Continued)

4.14

and and an and an and an and a second and an and an

LUNI PRIMI

4.5

0.0

07.27.27.27.27.27.27.27.2

Well	Status	Aquifer	Date installed
Site 20, Wastewa	ater Collection Basins		
HM-31	Active	Upper Zone	5-26-83
HM-47	Active	Upper Zone	4-14-84
HM-69	Active	Upper Zone	1-7-85
HM-70	Active	Upper Zone	1-8-85
HM-104	Active	Upper Zone	1-23-86
Site 16, Fuel Sa	aturation 3		
HM-78	Active	Upper Zone	1-11-85
F-200	Active	Upper Zone	2-24-86
F-201	Active	Upper Zone	2-24-86
F-202	Active	Upper Zone	2-25-86
F-208	Active	Upper Zone	3-6-86
F-209	Active	Upper Zone	3-7-86
F-210	Active	Upper Zone	3-7-86
F-211	Active	Upper Zone	3-7-86
<u>Site 9, FDTA 6</u>			
P-3	Active	Paluxy	6-18-83
Site 6, FDTA 3			
HM-33	Active	Upper Zone	4-11-84
HM-102	Active	Upper Zone	1-21-86
Site 18, Solven	<u>Lines</u>		
HM-72	Active	Upper Zone	1-9-85
HM-73	Active	Upper Zone	1-9-85
HM-74	Active	Upper Zone	1-9-85
HM-75	Active	Upper Zone	1-9-85
HM-106	Active	Upper Zone	1-26-86

TABLE 4.3.26-2. (Continued)

(Continued)

1

فسكست

NAMES OF STREET

Well	Status	Aquifer	Date Installe
Site 10, Chrome	Pit 1		
HM-48	Active	Upper Zone	4-14-84
HM-103	Active	Upper Zone	1-21-86
Site 5, FDTA 2			
HM-19	Active	Upper Zone	4-1-83
HM-49	Active	Upper Zone	415-84
HM-50	Active	Upper Zone	4-15-84
HM-51	Active	Upper Zone	4-15-84
HM-65	Active	Upper Zone	1-4-85
HM-66	Active	Upper Zone	1-5-85
HM-76	Active	Upper Zone	1-10-85
Site 14, Fuel Sa	aturation Area 1		
HM-53	Active	Upper Zone	4-16-84
HM-55	Active	Upper Zone	4-16-84
P-6U	Active	Upper Paluxy	5-20-84
P-6M	Active	Middle Paluxy	5-22-84
F-203	Active	Upper Zone	2-25-86
F-204	Active	Upper Zone	2-25-86
F-205	Active	Upper Zone	3-5-86
F-206	Active	Upper Zone	3-6-86
F-207	Active	Upper Zone	3-6-86
mbient Monitor:	ing Sites		
HM-29	Active	Upper Zone	5-26-83
HM-52	Active	Upper Zone	4-15-84
HM-54	Active	Upper Zone	4-16-84
HM-56	Active	Upper Zone	4-16-84
HM-57	Active	Upper Zone	4-17-84
HM-58	Active	Upper Zone	4-17-84
HM-59	Active	Upper Zone	4-17-84
HM-61	Active	Upper Zone	4-17-84
HM-64	Active	Upper Zone	4-18-84
HM-79	Active	Upper Zone	1-15-85
HM-81	Active	Upper Zone	1-6-85
HM-83	Active	Upper Zone	10-24-85
HM-84	Active	Upper Zone	10-24-85
HM-85	Active	Upper Zone	10-24-85

TABLE 4.3.26-2. (Continued)

(Continued)

Section of the sector of the sector

ZE KOSON DEREKE SANDAL DEREKE NADARE KANAN DEREKE DEREKE DEREKE DEREKE DEREKE DEREKE DEREKE DEREKE DEREKE DERE

N > N > N



<u>Ů,;/0,9),4,9),5,9),6,6,6,6,6,6,6</u>,6

UNITARE

4.2222

Well	Status	Aquifer	Date Installed
Ambient Monitor	ing Sites (Continued)		<u></u>
P-5U	Active	Upper Paluxy	6-3-84
P-5M	Active	Middle Paluxy	6-5 - 84
P-9U	Active	Upper Paluxy	1-15-85
P-9M	Active	Middle Paluxy	2-12-85
P-10U	Active	Upper Paluxy	1-23-85
P-10M	Active	Middle Paluxy	2-10-85
P-11U	Active	Upper Paluxy	8-13-85
P-11M	Active	Middle Paluxy	8-26-85
East Parking Lo	<u>t</u>		
HM-67	Abandoned	Upper Zone	1-5-85
HM-68	Active	Upper Zone	1-6-85
HM-71	Active	Upper Zone	1-8-85
HM-82	Active	Upper Zone	1-16-85
P-8U	Active	Upper Paluxy	1-19-85
P-8M	Active	Middle Paluxy	1-29-85
Fuel Storage Ta	nk		
HM-23	Active	Upper Zone	5-24-83
Lake Worth			
P-23U	Active	Upper Paluxy	3-17-86
Fuel Test Area,	Building 21		
HM-105	Active	Upper Zone	8-11-86
HM-107	Active	Upper Zone	8-11-86
HM-108	Active	Upper Zone	8-11-86

TABLE 4.3.26-2. (Continued)

-Oʻ

RAPIAN

4.3.27 Well Network Selection for Future Monitoring Efforts

The total system of wells at AF Plant 4 has been evaluated with respect to designing an optimal well network for future monitoring efforts. The wells evaluated consisted of the 87 upper zone monitor wells installed by Hargis and Montgomery (Hargis and Associates, 1985) and the nine upper zone wells installed by Radian. The F-series monitor wells recently installed by Intellus Corporation have not been included in this evaluation, since no analytical chemistry data were available for them. This evaluation considers the plant site as a whole, rather than individual sites. Site-specific concerns are best addressed as exceptions after the basic network design has been completed.

In addition to the upper zone monitor wells, the Paluxy Formation monitor wells have been evaluated with respect to future monitoring efforts at AF Plant 4. Since there are so few of them, all of the currently active Paluxy Formation monitor wells should be incorporated into the well network.

The objective of designing an optional monitor well network is to decrease the number of samples needed to represent groundwater quality and the hydrogeologic conditions in the aquifer. Taking too many samples is excessively costly and may add little to an understanding of groundwater quality. Taking too few samples, however, results in insufficient data for observing or modeling the hydrogeologic conditions and changes in contaminant distribution within the system.

One approach to designing an optimal well network for future monitoring efforts at AF Plant 4 involved a geostatistical method, the variogram function. The minerals industry has used variograms as a tool to determine the optimum sampling distance in ore bodies (Blais and Carlier, 1968; Peters, 1978). The variogram function goes beyond classical statistics because it deals with regionalized variables that have specific directional and distance characteristics. Trichloroethylene (TCE) values from upper zone monitor wells

RAPIAN

were used in the same way as ore grade in a mining application of the variogram. TCE was selected as the modeled constituent since it is nearly ubiquitous and commonly occurs in the highest concentration among contaminants of concern. A series of experimental semi-variograms were computer-generated and visually inspected to see if a "range" value could be determined. The "range" value is the optimum sampling distance. The computer-generated variograms did not reveal a "range" value that could be used to determine an optimal sampling (or well spacing distance). Instead, almost all of the variograms exhibited random, inconsistent behavior that could have been due to the lack of a Gaussian or continuous population distribution. An initial logarithmic transformation of the attributes still resulted in a skewed population and random variogram patterns. The distribution of TCE in the upper zone is not sufficiently continuously distributed (mathematically continuous) to be amenable to statistical evaluations of this kind. At this point, the variogram function method was eliminated as a viable method for designing an optimal well network at Plant 4.

A second method used to design the monitor well network on AF Plant 4, involved generating a TCE concentration contour map using the most recent available data, data from all of the IRP Phase II Stage 1 sampling (Figure 4.3.27-1). The data set is not synoptic, since sampling occurred throughout the first eight months of 1986, but constitutes the best data set available at the time of the study. All wells containing 5000 ppb TCE or more were set equal to 5000 ppb TCE to reduce the contrast and obtain additional detail in the low concentration areas. The 5000 ppb level was selected as a threshold of convenience to avoid the numerical difficulties of contouring over many orders of magnitude. Results are not disturbed by selecting an arbitrary, high value. This map was generated using CPS-1®, Radian's Contour Plotting System), which uses an appropriate search algorithm to define contour values. Tables 4.3.27-1 and 4.3.27-2 show the initial and reduced data sets of TCE concentrations used in well network selection.



Ì

4-239

MONITOR WELL	X-COORD INATE	Y-COORD INATE	ELEVATION	TCE CONCENTRATION (ugl)	WELL IDENTIFIER NOTE*	DATE SAMPLED
HM-23	0.017	42.721	625.000	0.0		2/17/86
HM-22	6.828	48.338	631.600	0.0		2/20/86
	14.323	47.035	630 . 900	0.0		2/27/86
HM-24 HM-44	12.85/	40.196 41 769	632.600 633 700	0.0		2/12/80 2/25/86
HM-43	10.822	42.707	647.800	0.0		2/20/86
HM-101	13.291	41.382	621.300	1		4/18/86
HM-33	15.555	42.059	625.600	0.0		2/28/86
HM-14 VM_0	-0.204	3/.803 36 613	624.600 623 100	0.0		2/19/86
HM-42	4./32	38.544	617.400	0.0		2/20/86
	16.399	35.040	620.100	0.0		2/20/86
HM-40	10.168	34.244	622.200	0.0		2/25/86
HM-100	5.846	33.326	621.900	1		4/11/86
HM-13 44-27	/60°0-	31.310	626.200 610 700	0.0		2/19/86
HM-30	9,066	30.452	615.200	177.0		2/13/86
HM-102	16.054	37.559	621.000			4/9/86
HM-41	15.549	30.412	619.000	2.7		2/25/86
HM-46	17.784	32.144	613.900	0.0		2/20/86
HM-45		27.026	618.300	133.0		2/25/86
HM-16 HM-15	10.454 6.975	27.886 27.826	597.200 632.500	5000.0 5000.0	<u>ل</u> ت ليا	2/11/86 2/25/86
						(Continued)

TABLE 4.3.27-1. (Continued)

2/19/86 2/28/86 2/21/86 4/10/86 2/21/86 2/10/86 SAMPLED 2/14/86 2/11/86 2/11/86 2/14/86 4/10/86 2/14/86 2/26/86 2/10/86 2/19/86 2/10/86 2/6/86 2/6/86 2/6/86 2/4/86 2/4/86 2/6/86 2/7/86 DATE IDENTIFIER NOTE* WELL CONCENTRATION (ug1) 0.1 0.8 0.9 14.0 15.0 4485.0 TCE 7.0 0.2 602.0 5000.0 0.0 26.4 3.9 401.0 5000.0 5000.0 32.3 5000.0 2367.0 3.2 ¦ ELEVATION 609.900 609.600 619.400 515.500 531.200 520.000 515.500 520.400 514.400 516.800 619.600 617.800 524.700 626.700 611.400 503.200 506.400 609.300 596.300 516.400 514.900 521.800 519.300 Y-COORDINATE 25.848 23.271 23.691 20.268 10.280 10.974 13.565 5.640 15.651 26.295 26.065 20.958 22.708 18.295 15.274 15.661 22.586 22.628 22.595 26.717 5.705 27.430 24.536 X-COORDINATE 4.075 5.739 3.076 9.907 -0.097 1.977 -0.139 8.844 2.056 3.211 7.035 7.410 10.707 19.789 4.255 -0.105 0.049 23.479 14.691 19.578 15.157 25.102 19.486 MONITOR HM-48 HM-104 **HM-103** WELL HM-60 HM-31 69-MH HM-70 HM-68 HM-25 HM-82 HM-56 HM-24 HM-12 HM-28 HM-17 HM-11 HM-77 HM-47 HM-57 HM-52 HM-55 HM-53 HM-63

(Continued)

TABLE 4.3.27-1. (Continued)

2/21/86 2/7/86 2/12/86 2/13/86 2/13/86 2/26/86 2/10/86 2/13/86 2/11/86 SAMPLED 2/27/86 2/27/86 2/27/86 1/31/86 1/31/86 2/27/86 2/12/86 1/31/86 2/11/86 2/12/86 1/31/86 2/7/86 3/1/86 3/1/86 DATE IDENTIFIER NOTE* WELL Ç۳, ſz. CONCENTRATION (ug1) 2015.0 390.0 TCE 0.0 5000.0 5000.0 0.0 0.3 402.0 0.0 1.929.2 0.0 9.0 0.0 36.6 0.0 8.0 6.4 0.1 2314.0 ELEVATION 615.800 617.500 617.800 618.700 508.700 518.700 617.400 522.300 524.500 622.900 520.100 524.700 521.900 520.100 517.900 515.100 521.500 518.200 518.500 518.400 523.800 616.600 621.500 Y-COORDINATE 34.512 32.450 31.925 31.521 30.707 28.750 27.575 29.328 27.852 28.923 29.106 26.536 26.072 27.020 27.261 32.951 32.372 30.587 30.061 25.586 24.975 22.961 31.664 X-COORDINATE 35.785 33.494 22.288 23.761 25.305 26.210 27.793 29.720 29.494 32.360 32.552 36.334 39.288 37.207 21.279 31.933 30.583 27.068 22.295 24.349 29.480 35.604 34.578 DNITOR WELL HM-36 HM-26 HM-38 HM-39 64-MH HM-66 HM-50 HM-19 HM-10 HM-18 HM-20 HM-65 HM-76 HM-54 HM-62 HM-51 HM-34 HM-35 HM-27 HM-37 HM-21 HM-61 HM-7

RADIAN

4-242

DERE REPORT PROPERTY PROPERTY PROPERTY REPORT REPORT DURING DURING DURING DURING DURING DURING

(Continued)

TABLE 4.3.27-1. (Continued)

8/19/86 8/20/86 8/19/86 SAMPLED 2/21/86 2/26/86 2/26/86 2/14/86 2/26/86 4/18/86 2/18/86 2/5/86 8/21/86 8/21/86 3/21/86 2/5/86 2/5/86 2/3/86 2/3/86 2/3/86 2/7/86 DATE IDENTIFIER NOTE* WELL CONCENTRATION (ng1) 6.6 174.0 1274.0 TCE 0.0 0.0 0.0 7.4 0.3 0.0 1200.0 75.1 0.7 ---ELEVATION 617.200 614.400 614.200 525.600 610.800 504.100 511.400 510.200 514.000 520.000 519.700 619.500 516.200 621.900 621.900 616.600 514.400 513.000 616.000 522.800 Y-COORDINATE 5.716 22.538 20.559 12.890 13.93 19.160 11.618 15.664 22.536 23.230 22.549 22.565 21.343 23.193 15.657 25.141 10.580 20.027 9.601 20.224 X-COORDINATE 29.085 33.010 40.830 37.006 35.650 45.359 43.525 47.745 59.594 56.543 60.217 61.747 60.897 33.739 49.615 40.474 49.626 61.390 63.290 67.85 ONITOR HM-106 HM-107 HM-108 HM-105 HM-75 HM-80 HM-78 HM-81 HM-83 HM-79 HM-73 HM-74 HM-84 WELL **IM-29** 49-MH HM-58 HM-59 HM-71 HM-72 HM-85

* F = Concentration fixed at 5000 ug/L.

Protocory Provi

Concentration data not used (not available at time of TCE concentration contour map generation). п

RAPIAN

4-243



$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$.27-2. TCE CON REDUCED	CENTRATION DA	ATA USED IN WELL NETWO	ORK SELECTION:
$ \begin{array}{c} 6.928 \\ 6.928 \\ 48.338 \\ 47.035 \\ 430.900 \\ 0.0 \\ 14.323 \\ 47.035 \\ 430.900 \\ 0.0 \\ 14.323 \\ 47.035 \\ 430.900 \\ 0.0 \\ 14.323 \\ 47.035 \\ 41.751 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 42.059 \\ 425.600 \\ 0.0 \\ 14.753 \\ 41.752 \\ 42.059 \\ 42.059 \\ 42.050 \\ 1.0630 \\ 14.000 \\ 1.0630 \\ 14.101 \\ 1.0630 \\ 14.101 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.100 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.10$	X-COORDINATE	Y-COORDINATE	ELEVATION		IDENTIFIER
$ \begin{array}{c} 6.928 \\ 6.928 \\ 48.338 \\ 47.035 \\ 430.900 \\ 0.0 \\ 14.323 \\ 47.035 \\ 430.900 \\ 0.0 \\ 14.323 \\ 47.035 \\ 430.900 \\ 0.0 \\ 14.323 \\ 47.035 \\ 41.751 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 41.752 \\ 42.059 \\ 425.600 \\ 0.0 \\ 14.753 \\ 41.752 \\ 42.059 \\ 42.059 \\ 42.050 \\ 1.0630 \\ 14.000 \\ 1.0630 \\ 14.101 \\ 1.0630 \\ 14.101 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.100 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.101 \\ 14.100 \\ 1.0630 \\ 14.10$			(DE 000	0 0 HM 27	
14.32347.035630.9000.0 HM -912.85745.196632.4000.2 HM -57.75141.762633.7000.0 HM -4413.29141.382621.3001.0830 HM -10115.55542.059625.6000.0 HM -425.84633.326621.9001.0830 HM -100-0.07731.310626.2000.0 HM -139.06630.452615.200177.0 HM -3016.05437.559621.0001.0830 HM -10215.54930.412619.0002.7 HM -4114.14027.026618.300133.0 HM -1715.54930.412617.5007.00 HM -244.25526.065615.500602.0 HM -1777.430615.5007.00 HM -244.25526.065615.500602.0 HM -1777.430617.800401.0 HM -1045.73923.691617.800400.0 HM -1777.05609.30032.3 HM -48-0.07715.274603.2002367.0 HM -1040.04915.274603.2002367.0 HM -1040.04915.274603.2002367.0 HM -8319.7895.640596.300500.0.0 HM -8419.7975.705609.30032.3 HM -6319.7985.640596.300 </td <td></td> <td></td> <td></td> <td></td> <td></td>					
12.85745.196 632.400 0.2 HM-5 7.75141.762 633.700 0.0 HM-44 13.29141.382 621.300 $1.0E30 \text{ HM-101}$ 15.555 42.059 625.600 0.0 HM-33 13.235 38.544 617.400 0.0 HM-42 5.846 33.326 621.900 $1.0E30 \text{ HM-100}$ -0.097 31.310 626.200 0.0 HM-133 9.066 30.452 415.200 177.0 HM-30 16.054 37.559 621.000 $1.0E30 \text{ HM-102}$ 15.549 30.412 619.000 2.7 HM-41 14.140 27.026 618.300 133.0 HM-45 6.975 27.826 632.500 500.0 HM-175 7 7.430 615.500 602.0 HM-28 8 844 25.848 620.400 5000.0 HM-27 4.255 26.065 615.500 602.0 HM-28 -0.105 20.958 616.800 26.4 HM-11 5.739 23.691 617.800 401.0 HM-60 3.076 20.268 624.700 5000.0 HM-177 9.907 22.708 616.400 5000.0 HM-103 7.410 13.565 606.400 5000.0 HM-103 7.410 13.565 606.400 5000.0 HM-103 19.789 5.640 596.300 5000.0 HM-36 19.799 5.640 596.300 5000.0 HM-36 21.279 34.512 616.600 0.0 HM-37 <td< td=""><td></td><td></td><td></td><td></td><td></td></td<>					
7.75141.762 633.700 0.0HM-4413.27141.382 621.300 1.0E30HM-10115.55542.059 625.600 0.0HM-3313.23538.544 617.400 0.0HM-345.84633.326 621.900 1.0E30HM-100-0.09731.310 626.200 0.0HM-139.06630.452 615.200 177.0HM-3016.05437.559 621.000 1.0E30HM-10215.54930.412 619.000 2.7HM-4114.14027.026 618.300 133.0HM-456.97527.826 632.500 5000.0HM-244.25526.065 615.500 602.0 HM-28-0.10520.958 614.400 0.0HM-272.05624.536 617.800 401.0 HM-48-0.09718.295 611.400 1.0E30HM-1040.09718.295 611.400 1.0E30HM-1037.41013.565 604.400 5000.0HM-317.03510.974 609.600 1.0E30HM-1037.41013.565 604.400 5000.0HM-3119.7895.640596.3005000.0HM-6321.27934.512 618.400 0.0 HM-4819.7895.641 614.900 0.3 HM-6819.7895.640596.3005000.0HM-5519.48626.717 619.400 4485.0 HM-63 <t< td=""><td></td><td></td><td></td><td></td><td></td></t<>					
13. 29141.382 621.300 1.0E30 HM-10115. 55542.059 625.400 0.0 HM-3313. 23538.544 617.400 0.0 HM-425. 84633.326 621.900 1.0E30 HM-100-0.09731.310 626.200 0.0 HM-139.06630.452 615.200 177.0 HM-3016.05437.559 621.000 $1.0E30$ HM-10215.54930.412 619.000 2.7 HM-4114.14027.026 632.500 5000.0 HM-1244.25526.065 615.500 602.0 HM-288.84425.848 620.400 5000.0 HM-17F2.05624.536 614.400 0.0 HM-27-0.10520.958 616.800 26.4 HM-115.73923.691 617.800 401.0 HM-48-0.09718.295 611.400 $1.0E30$ HM-1040.04915.274 603.200 2367.0 HM-137.03510.974 609.600 $1.0E30$ HM-1037.41013.565 606.400 5000.0 HM-6819.7895.640 596.300 5000.0 HM-2823.47915.661 614.900 0.0 HM-5519.48626.717 617.400 4485.0 HM-6321.27934.512 616.600 2015.0 HM-5519.48626.717 617.400 402.0 HM-3727.77331.925 621.500 2015.0 HM-5423.47915.661 614.900 0.0 HM-3827.77331.925 621.500 402.0				0.0 HM-44	
13. 23538. 544617. 4000.0HM-425. 84633. 326621. 9001.0E30HM-100-0.09731. 310626. 2000.0HM-13016. 05437. 559621. 0001.0E30HM-10215. 54930. 412619. 0002.7HM-4114. 14027. 026618. 300133.0HM-45F1.97727. 430615. 5005000.0HM-15F1.97727. 430615. 500602.0HM-28F8.84425. 848620. 4005000.0HM-17F2.05624. 536614. 4000.0HM-27F-0.10520. 958616. 80026.4HM-115.73923. 691617. 800401.0HM-60F9.90722. 708626. 70019.8HM-48-0.09718. 295611. 4001.0 E30HM-1040.04915. 274603. 2002367.0HM-317.03510. 974609. 30032.3HM-6819.7895.6405000.0HM-55119.7895.64050000.8HM-5419.7895.640618.7000.8HM-6321.27934.512616.6000.0HM-3322.47915.661614.7000.8HM-6319.78922.628621.80014.0HM-7323.47915.641614.4000.0HM-7419.7895.6405000.0HM-73	13.291				
5.84633.326621.9001.0E30HM-100-0.09731.310626.2000.0HM-139.06630.452615.200177.0HM-3016.05437.559621.0001.0E30HM-10215.54930.412619.0002.7HM-4114.14027.026618.300133.0HM-456.97527.826632.5005000.0HM-1777.430615.5007.0HM-288.84425.848620.4005000.0HM-1777.430615.50020.0HM-115.75923.691617.800401.0HM-405.75923.691617.800401.0HM-405.75923.691617.800401.0HM-405.75923.691617.8001.0E30HM-1037.055611.4001.0E30HM-1040.04915.274603.2002367.0HM-317.03510.974609.30032.3HM-6819.7695.640596.3005000.0HM-5519.48626.717619.4004485.0HM-5419.57822.628621.80014.0HM-5519.48626.717619.4000.0HM-3423.76132.450618.7000.0HM-3423.75132.450618.7000.0HM-3625.30532.372615.8002015.0HM-2627.79331.925621.80014.0HM-36 <tr< td=""><td></td><td></td><td></td><td></td><td></td></tr<>					
$\begin{array}{cccccccccccccccccccccccccccccccccccc$					
9.06630.452615.200 177.0 HM-3016.05437.559621.0001.0E30HM-10215.54930.412619.0002.7HM-4114.14027.026618.300133.0HM-456.97527.826632.5005000.0HM-15F1.97727.430615.5007.0HM-244.25526.065615.500602.0HM-288.84425.848620.4005000.0HM-17F2.05624.536614.4000.0HM-4603.07620.268624.7005000.0HM-47F9.90722.708626.70019.8HM-48-0.09718.295611.4001.0E30HM-1040.04915.274603.2002367.0HM-317.03510.974609.4001.0E30HM-1037.41013.565606.4005000.0HM-70F19.7895.640576.30032.3HM-6819.7895.641614.9000.8HM-5519.48626.717619.4004485.0HM-4321.27934.512616.6000.0HM-3425.30532.372615.8002015.0HM-2627.79331.925621.80014.0HM-3729.70031.521608.7005000.0HM-3720.77331.925622.3001.9HM-3421.27934.512616.4000.0HM-3425.705 <td></td> <td></td> <td></td> <td></td> <td></td>					
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$					
15.54930.412 619.000 2.7 HM-4114.14027.026 618.300 133.0HM-45 6.975 27.826 632.500 5000.0 HM-45 6.975 27.826 632.500 5000.0 HM-24 4.255 26.065 615.500 602.0 HM-28 8.844 25.848 620.400 5000.0 HM-17F 2.056 24.536 614.400 0.0 HM-17F 2.056 24.536 614.400 0.0 HM-47F 9.056 22.588 622.700 401.0 HM-60F 5.739 23.691 617.800 401.0 HM-47F 9.907 22.708 622.700 19.8 HM-48 -0.097 18.295 611.400 1.0650 HM-104 0.049 15.274 603.200 2367.0 HM-31 7.035 10.974 609.600 1.0650 HM-104 0.049 15.274 603.200 2367.0 HM-31 7.035 10.974 609.300 32.3 HM-68 19.789 5.640 596.300 5000.0 HM-82F 23.479 15.661 614.900 0.8 HM-55 19.486 26.717 619.400 4485.0 HM-54 21.279 34.512 616.600 0.0 HM-37 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-37 29.7					
14.140 27.026 618.300 133.0 HM-45 F 6.975 27.826 632.500 5000.0 HM-15 F 1.977 27.430 615.500 7.0 HM-24 F 4.255 26.065 615.500 602.0 HM-28 F 8.844 25.848 620.400 5000.0 HM-17 F 2.056 24.536 614.400 0.0 HM-25 F -0.105 20.958 616.800 26.4 HM-11 F 5.739 23.691 617.800 401.0 HM-60 F 9.907 22.708 626.700 19.8 HM-48 F -0.097 18.295 611.400 1.0E30 HM-104 F 0.049 15.274 603.200 2367.0 HM-31 F 10.707 5.705 609.300 32.3 HM-68 F 19.789 5.640 596.300 5000.0 HM-70 F 10.707 5.705 609.300 32.3 HM-68 F 19.588 22.628 621.800 14.0 HM-55 F 19.789 22.628 621.800 0.3 HM-34 <td></td> <td></td> <td></td> <td></td> <td></td>					
6.975 27.825 632.300 3000.0 $MH-24$ 1.977 27.430 615.500 602.0 $HM-24$ 4.255 26.065 615.500 602.0 $HM-28$ 8.844 25.848 620.400 5000.0 $HM-17$ F 2.056 24.536 614.400 0.0 $HM-25$ -0.105 20.958 616.800 26.4 $HM-11$ 5.739 23.691 617.800 401.0 $HM-60$ 3.076 20.268 624.700 5000.0 $HM-47$ F 9.907 22.708 626.700 19.8 $HM-48$ -0.097 18.295 611.400 1.0630 $HM-104$ 0.049 15.274 603.200 2367.0 $HM-31$ 7.035 10.974 609.400 1.0630 $HM-103$ 7.410 13.565 606.400 5000.0 $HM-32$ 7.410 13.565 606.400 5000.0 $HM-82$ F 23.479 5.640 596.300 5000.0 $HM-82$ F 23.479 15.641 614.900 0.8 $HM-56$ 19.578 22.628 621.800 14.0 $HM-34$ 21.279 34.512 616.400 0.0 $HM-34$ 23.505 32.372 615.800 2015.0 $HM-26$ 27.793 31.925 621.500 402.0 $HM-37$ F 29.494 30.707 618.700 0.0 $HM-37$ F 29.494 <td< td=""><td>14.140</td><td>27.026</td><td>618.300</td><td></td><td>c</td></td<>	14.140	27.026	618.300		c
4.255 26.065 615.500 602.0 HM-28F 8.844 25.848 620.400 5000.0 HM-17F 2.056 24.536 614.400 0.0 HM-15 -0.105 20.958 616.800 26.4 HM-11 5.739 23.691 617.800 401.0 HM-60 3.076 20.268 624.700 5000.0 HM-47F 9.907 22.708 626.700 19.8 HM-48 -0.097 18.295 611.400 $1.0E30$ HM-104 0.049 15.274 603.200 2367.0 HM-31 7.035 10.974 609.400 $1.0E30$ HM-103 7.410 13.565 606.400 5000.0 HM-82F 10.707 5.705 607.300 32.3 HM-68F 23.479 15.661 614.900 0.8 HM-55F 23.479 15.661 614.900 0.8 HM-54F 23.479 15.661 614.900 0.8 HM-54F 23.479 35.640 596.300 2005.0 HM-63F 23.479 35.640 541.800 2015.0 HM-24 27.793 31.925 621.500 402.0 HM-37F 27.793 31.925 621.500 402.0 HM-37F 29.494 30.707 618.700 5000.0 HM-38F 29.494 30.707 618.700 0.0 HM-39F					Г
8.844 25.848 620.400 5000.0 HM-17 F 2.056 24.536 614.400 0.0 HM-25 -0.105 20.958 616.800 26.4 HM-11 5.739 23.691 617.800 401.0 HM-60 3.076 20.268 624.700 5000.0 HM-47 F 9.907 22.708 626.700 19.8 HM-48 F -0.097 18.295 611.400 1.0E30 HM-104 F 0.049 15.274 603.200 2367.0 HM-31 F 7.035 10.974 609.600 1.0E30 HM-103 F 10.707 5.705 609.300 32.3 HM-68 F 19.789 5.640 596.300 5000.0 HM-70 F 19.789 5.641 614.900 0.8 HM-56 F 19.789 5.641 614.900 0.8 HM-54 F 19.789 5.641 614.900 0.8 HM-54 F 19.789 5.641 614.900 0.8 HM-54 F 19.797 34.512 616.600 0.0 HM-33 F 27.793					
8.84423.848620.4005000.0 HM-25 2.056 24.536614.400 0.0 HM -11 5.739 23.691617.800401.0 HM-60 3.076 20.268624.7005000.0 HM-47 9.907 22.708626.70019.8 HM-48 -0.097 18.295611.4001.0E30 HM-104 0.049 15.274603.2002367.0 HM-31 7.035 10.974609.4001.0E30 HM-103 7.410 13.565606.4005000.0 HM-82 7.410 13.565609.30032.3 HM-68 19.789 5.640596.3005000.0 HM-82 19.578 22.628621.80014.0 HM-55 19.486 26.717618.7000.3 HM-36 21.279 34.512616.6000.0 HM-34 23.761 32.450618.7000.3 HM-36 27.793 31.925621.500402.0 HM-37 29.720 31.521608.7005000.0 HM-38 29.494 30.707618.7000.0 HM-39 35.785 28.750622.3001.9 HM-49 31.933 27.852620.1009.0 HM-19 30.583 28.923624.7000.0 HM-7					F
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$					
5.739 23.691 617.800 401.0 HM-60 F 3.076 20.268 624.700 5000.0 HM-47 F 9.907 22.708 626.700 19.8 HM-48 F -0.097 18.295 611.400 1.0E30 HM-104 F 0.049 15.274 603.200 2367.0 HM-31 F 7.035 10.974 609.600 1.0E30 HM-103 F 10.707 5.705 609.300 32.3 HM-68 F 19.789 5.640 596.300 5000.0 HM-70 F 23.479 15.661 614.900 0.8 HM-56 F 19.578 22.628 621.800 14.0 HM-55 F 19.486 26.717 619.400 4485.0 HM-63 F 23.741 32.450 618.700 0.3 HM-36 F 27.793 31.925 621.500 402.0 HM-37 F 29.720 31.521 608.700 500.0 HM-37 F 29.720 31.521 608.700 500.0 HM-37 F 29.720 31.521 608.700 5000					
3.076 20.268 624.700 5000.0 HM-47 F 9.907 22.708 626.700 19.8 HM-48 -0.097 18.295 611.400 1.0E30 HM-104 0.049 15.274 603.200 2367.0 HM-31 7.035 10.974 609.600 1.0E30 HM-103 7.410 13.565 606.400 5000.0 HM-68 19.789 5.640 596.300 32.3 HM-68 19.789 5.640 596.300 500.0 HM-82 F 23.479 15.661 614.900 0.8 HM-56 F 19.578 22.628 621.800 14.0 HM-63 F 23.479 15.661 618.700 0.3 HM-43 23.761 32.450 618.700 0.3 HM-26 27.793 31.925 621.500 402.0 HM-37 F 29.720 31.521 608.700 5000.0 HM-38 F 29.7494 30.707 618.700 0.0 HM-39 F	-				
9.90722.708 626.700 19.8 HM-48-0.09718.295 611.400 1.0E30 HM-1040.04915.274 603.200 2367.0 HM-317.03510.974 609.600 1.0E30 HM-1037.41013.565 606.400 5000.0 HM-707.41013.565 606.400 5000.0 HM-8210.7075.705 609.300 32.3 HM-6819.7895.640 594.300 5000.0 HM-8219.57822.628 621.800 14.0 HM-5519.486 26.717 619.400 4485.0 HM-6321.279 34.512 616.600 0.0 HM-3423.761 32.450 618.700 0.3 HM-3625.305 32.372 615.800 2015.0 HM-2627.793 31.925 621.500 402.0 HM-3729.720 31.521 608.700 5000.0 HM-3829.494 30.707 618.700 0.2 HM-3935.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					F
0.049 15.274 603.200 2367.0 HM-31 7.035 10.974 609.600 1.0E30 HM-103 7.410 13.565 606.400 5000.0 HM-70 F 10.707 5.705 609.300 32.3 HM-68 F 10.707 5.705 609.300 5000.0 HM-82 F 23.479 15.661 614.900 0.8 HM-56 19.578 22.628 621.800 14.0 HM-55 19.486 26.717 619.400 4485.0 HM-63 21.279 34.512 616.600 0.0 HM-34 23.761 32.450 618.700 0.3 HM-36 25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 F 29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.0 HM-39 5 35.785 28.750 622.300 1.9 HM-49 1				19.8 HM-48	
7.035 10.974 609.600 1.0E30 HM-103 F 7.410 13.565 606.400 5000.0 HM-70 F 10.707 5.705 609.300 32.3 HM-68 F 19.789 5.640 596.300 5000.0 HM-82 F 23.479 15.661 614.900 0.8 HM-56 F 19.578 22.628 621.800 14.0 HM-55 F 19.486 26.717 619.400 4485.0 HM-63 F 21.279 34.512 616.600 0.0 HM-34 F 23.761 32.450 618.700 0.3 HM-36 25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.0 HM-39 F 32.360 30.587 617.400 0.2 HM-39 F 31.933 27.852 6	-0.097				
7.410 13.565 606.400 5000.0 HM-70 F 10.707 5.705 609.300 32.3 HM-68 F 19.789 5.640 596.300 5000.0 HM-82 F 23.479 15.661 614.900 0.8 HM-56 F 19.578 22.628 621.800 14.0 HM-55 F 19.486 26.717 619.400 4485.0 HM-63 F 21.279 34.512 616.600 0.0 HM-34 F 23.761 32.450 618.700 0.3 HM-36 F 25.305 32.372 615.800 2015.0 HM-26 F 27.793 31.925 621.500 402.0 HM-37 F 29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.0 HM-39 F 32.360 30.587 617.400 0.2 HM-39 F 31.933 27.852 620.100 9.0 HM-19 F 30.583 28.923 624.700 0.0 HM-7 F					
10.707 5.705 609.300 32.3 HM-68 19.789 5.640 596.300 5000.0 HM-82 F 23.479 15.661 614.900 0.8 HM-56 19.578 22.628 621.800 14.0 HM-55 19.486 26.717 619.400 4485.0 HM-63 21.279 34.512 616.600 0.0 HM-34 23.761 32.450 618.700 0.3 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.0 HM-39 F 32.360 30.587 617.400 0.2 HM-39 F 31.933 27.852 620.100 9.0 HM-49 5 31.933 27.852 620.100 9.0 HM-19 5 30.583 28.923 624.700 0.0 HM-7 5					F
19.789 5.640 596.300 5000.0 HM-82 F 23.479 15.661 614.900 0.8 HM-56 19.578 22.628 621.800 14.0 HM-55 19.486 26.717 619.400 4485.0 HM-63 21.279 34.512 616.600 0.0 HM-34 23.761 32.450 618.700 0.3 HM-36 25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.2 HM-39 5 35.785 28.750 622.300 1.9 HM-49 3 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					-
23.479 15.661 614.900 0.8 HM-56 19.578 22.628 621.800 14.0 HM-55 19.486 26.717 619.400 4485.0 HM-63 21.279 34.512 616.600 0.0 HM-34 23.761 32.450 618.700 0.3 HM-36 25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 29.494 30.707 618.700 0.0 HM-21 32.360 30.587 617.400 0.2 HM-39 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					F
19.578 22.628 621.800 14.0 HM-55 19.486 26.717 619.400 4485.0 HM-63 21.279 34.512 616.600 0.0 HM-34 23.761 32.450 618.700 0.3 HM-36 25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 29.494 30.707 618.700 0.0 HM-21 32.360 30.587 617.400 0.2 HM-39 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					
19.486 26.717 619.400 4485.0 HM-63 21.279 34.512 616.600 0.0 HM-34 23.761 32.450 618.700 0.3 HM-36 25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 29.494 30.707 618.700 0.0 HM-21 32.360 30.587 617.400 0.2 HM-39 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					
21.279 34.512 616.600 0.0 HM-34 23.761 32.450 618.700 0.3 HM-36 25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.0 HM-21 F 32.360 30.587 617.400 0.2 HM-39 F 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7				4485.0 HM-63	
25.305 32.372 615.800 2015.0 HM-26 27.793 31.925 621.500 402.0 HM-37 29.720 31.521 608.700 5000.0 HM-38 29.494 30.707 618.700 0.0 HM-21 32.360 30.587 617.400 0.2 HM-39 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7	21.279	34.512			
27.793 31.925 621.500 402.0 HM-37 F 29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.0 HM-21 32.360 30.587 617.400 0.2 HM-39 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					
29.720 31.521 608.700 5000.0 HM-38 F 29.494 30.707 618.700 0.0 HM-21 32.360 30.587 617.400 0.2 HM-39 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					
29.720 31.521 808.700 800.0 HM-38 29.494 30.707 618.700 0.0 HM-21 32.360 30.587 617.400 0.2 HM-39 35.785 28.750 622.300 1.9 HM-49 31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					F
32.36030.587617.4000.2 HM-3935.78528.750622.3001.9 HM-4931.93327.852620.1009.0 HM-1930.58328.923624.7000.0 HM-7					
35.78528.750622.3001.9 HM-4931.93327.852620.1009.0 HM-1930.58328.923624.7000.0 HM-7					
31.933 27.852 620.100 9.0 HM-19 30.583 28.923 624.700 0.0 HM-7					
30.583 28.923 624.700 0.0 HM-7					
				0.0 HM-7	
				(co	ntinued)



TABLE 4.3.27-2. (Continued)

X-COORDINATE	Y-COORDINATE	ELEVATION	TCE CONCENTRATION (ug/L)	WELL IDENTIFIER NOTE *
27.068	29.106	621.900	36.6 HM-10	
22.295	30.061	620.100	0.0 HM-18	
29.480	26.072	615.100	2314.0 HM-20	
35.604	27.020	621.500	5000.0 HM-51	F
37.207	27.261	618.200	0.0 HM-65	
29.085	23.193	614.200	1200.0 HM-29	
33.010	19.160	622.800	75.1 HM-64	
40.830	20.027	625.600	0.7 HM-58	
37.006	15.657	617.200	6.6 HM-59	
35.650	11.618	610.800	174.0 HM-79	
33.739	5.716	604.100	1274.0 HM-71	
49.615	15.664	610.200	0.0 HM-73	
49.626	22.536	620.000	1.0E30 HM-106	
47.745	25.141	619.500	0.0 HM-80	
56.543	23.230	614.400	1.0 HM-78	
60.217	22.549	616.200	1.0E30 HM-107	
60.897	22.538	621.900	1.0E30 HM-108	
59.594	20.559	621.900	1.0E30 HM-105	
61.747	21.343	616.600	0.0 HM-81	
61.390	10.580	614.400	7.4 HM-83	
63.290	12.890	613.000	0.3 HM-84	
67.85	13.93	616.000	0.0 HM-85	

RIGHT PERSON

F = Concentration fixed at 5000 ug/L.

The next step was visual inspection of the TCE concentration map to see how many of the upper zone monitor wells could be considered duplicates (spatially equivalent and displaying similar concentrations of TCE) of nearby wells. The selection was accomplished by inspection and examination of the data, utilizing best engineering judgement. Thirty-two HM-series monitor wells were identified for deletion from the total well system at AF Plant 4. Six of these had previously been destroyed (HM-1, 3A, 3B, 4A, 4B, and 6). The effect of the prior destruction of these wells was accommodated by deleting them from the data base for all computations. The 25 other HM-series wells selected were deemed to be near duplicates of nearby wells that would add no significant data in a future monitoring program. These wells were then eliminated from the data base and a new TCE concentration contour map was generated (Figure 4.3.27-2) based on the smaller data base. Visual inspection of the two contour maps shows that the elimination of the 32 monitor wells did not significantly alter the shape of the contours. This comparison validates the subjective inspection and selection process.

A difference map (contour map of the difference between the two previous contour sets) was also generated to quantify the difference between the two data sets (Figure 4.3.27-3). This map revealed a general pattern of small variations signifying that the smaller data base does not systematically differ from the larger data base. Therefore, plant-wide groundwater conditions are adequately monitored by the reduced well network. An important further step in this process is to examine the effects of well deletion on a site-bysite basis. The purpose of this examination would be to determine whether a given deletion will harm the plant contractor's ongoing monitoring program.

The wells eliminated from the total well system and the wells to be retained in the upper zone monitoring network are listed in Table 4.3.27-3. The selected well network for Plant 4 thus includes 64 upper zone monitor wells (HM-series) and all the currently active Paluxy Formation monitor wells.




arrestaria arresta arresta arresta arresta arrestaria artestaria decensi arrette arresta arresta arresta decend N



GM MANANAN

TABLE 4.3.27-3. WELL NETWORK SELECTION FOR FUTURE MONITORING AT PLANT 4

HM-1	HM-4B	HM-16	HM-43	HM-54	HM-67	HM-76
HM-2	HM-6	HM-27	HM-46	HM-57	HM-69	HM-77
HM-3A	HM-8	HM-32	HM-50	HM-61	HM-72	
HM-3B	HM-12	HM-35	HM-52	HM-62	HM-74	
HM-4A	HM-14	HM-40	HM-53	HM-66	HM-75	
Upper 2	Zone Mon:	itor Well	s To Be R	etained i	n Well Ne	twork
HM-5	HM-20	HM-31	HM-45	HM-63	HM-81	HM-105
HM-5 HM-7	HM-20 HM-21	HM-31 HM-33				
Upper 2 HM-5 HM-7 HM-9	HM-20	HM-31	HM-45	HM-63	HM-81	HM-105
HM-5 HM-7	HM-20 HM-21	HM-31 HM-33	нм-45 нм-47	HM-63 HM-64	нм-81 нм-82	НМ-105 НМ-106
HM-5 HM-7 HM-9	HM-20 HM-21 HM-22	HM-31 HM-33 HM-34	HM-45 HM-47 HM-48	HM-63 HM-64 HM-65	HM-81 HM-82 HM-83	нм-105 нм-106 нм-107
HM-5 HM-7 HM-9 HM-10 HM-11	HM-20 HM-21 HM-22 HM-23	HM-31 HM-33 HM-34 HM-36	HM-45 HM-47 HM-48 HM-49	HM-63 HM-64 HM-65 HM-68	HM-81 HM-82 HM-83 HM-84	нм-105 нм-106 нм-107
HM-5 HM-7 HM-9 HM-10 HM-11 HM-13	HM-20 HM-21 HM-22 HM-23 HM-24	HM-31 HM-33 HM-34 HM-36 HM-37	HM-45 HM-47 HM-48 HM-49 HM-51	HM-63 HM-64 HM-65 HM-68 HM-70	HM-81 HM-82 HM-83 HM-84 HM-85	нм-105 нм-106 нм-107
HM-5 HM-7 HM-9 HM-10 HM-11 HM-13 HM-15	HM-20 HM-21 HM-22 HM-23 HM-24 HM-25	HM-31 HM-33 HM-34 HM-36 HM-37 HM-38	HM-45 HM-47 HM-48 HM-49 HM-51 HM-55	HM-63 HM-64 HM-65 HM-68 HM-70 HM-71	HM-81 HM-82 HM-83 HM-84 HM-85 HM-100	нм-105 нм-106 нм-107
HM-5 HM-7 HM-9 HM-10	HM-20 HM-21 HM-22 HM-23 HM-24 HM-25 HM-26	HM-31 HM-33 HM-34 HM-36 HM-37 HM-38 HM-39	HM-45 HM-47 HM-48 HM-49 HM-51 HM-55 HM-56	HM-63 HM-64 HM-65 HM-68 HM-70 HM-71 HM-73	HM-81 HM-82 HM-83 HM-84 HM-85 HM-100 HM-101	нм-105 нм-106 нм-107

RAPIAN

4.4 Quality Assurance and Quality Control

The objective of the quality assurance and quality control (QA/QC) efforts associated with Plant 4 activities was to ensure that data collected were of known and sufficient calibre to qualitatively and quantitatively characterize the sites of interest. Two primary aspects of the QA/QC effort are related to achieving this objective. First, the QA/QC program forms a framework for controlling data quality within preestablished limits during execution of the sampling and analytical efforts. Second, the QA/QC program estimates uncertainty in the measurement data by identifying and defining qualitative and quantitative limitations associated with these data. Section 4.4.1 briefly describes some of the key elements of the QA/QC effort and assesses QA/QC measurement data quality.

4.4.1 Summary and Approach

QA/QC data associated with the Plant 4 site investigation indicate that the measurement data are acceptable, defensible, and reliable within the expected limits of sampling and analytical error. In some cases, interferences within the sample matrix significantly affected analyte recovery. These are discussed below.

The QA/QC program for the Plant 4 study was designed to fulfill two related purposes. First, by providing an organized framework for the sampling and analytical efforts, the program was designed to control data quality within preestablished limits to ensure that the objectives of the site characterization program were achieved. This involved establishing data quality objectives for the parameters of interest. Next, protocols were defined for critical aspects of the measurement effort, including:

o Sample collection, preservation, and storage;o Sample analysis;

- o Calibration of instrumentation and apparatus;
- o Data reduction, validation, and reporting;
- o Documentation and sample custody; and
- o Internal quality control.

The protocols were tailored to both site characterization and data quality objectives.

The second purpose of a QA/QC program is to assess data quality. Quality control data help identify and define the qualitative and quantitative limitations associated with the measurement data. Following are the key types of QA/QC procedures for quantitatively evaluating the data:

- o Field and laboratory blank samples;
- o Spiked samples;
- o Duplicate samples;
- o Duplicate analyses; and
- o Quality control check samples;

Blank samples play an especially important role in remedial investigation programs. They qualitatively ensure that the analytes detected in the samples are characteristic of the media sampled and not artifacts of the sampling and/or analytical process. Results of the analysis of laboratory blanks, field blanks and trip blanks are presented in the following sections.

Laboratory (reagent) blanks address only the analytical measurement process. Typically included with each batch of samples analyzed, they provide an ongoing check of the analytical system for systematic sample contamination



by contaminated reagents or preservatives. When evidence of contamination is indicated by blank values above preestablished levels, corrective action is taken to identify and eliminate the source of contamination. If possible, the affected samples may be reanalyzed.

Field blanks reflect the combined effects of sample collection, handling, transportation, storage, and analysis. Since often it is not feasible to resample when field blanks indicate possible sample contamination, field blank data are used to define the qualitative limitations of the associated measurement data. The presence of analytes of interest in either the field or laboratory blanks suggests that corresponding field samples may have been similarly contaminated and that results for these analytes should be considered suspect. If the blank data show a given analyte at widely varying concentrations, or at concentrations comparable to those for field samples, then the field sample results should be viewed as possible false positives for that analyte. If, on the other hand, blank data indicate a given analyte at concentrations much lower than typical sample concentrations, then the sample data may usually be viewed as qualitatively reliable, but as having greater than normal quantitative uncertainty.

Trip blanks are sealed VOA vials containing organic-free water which are prepared in the laboratory then shipped to the field with the empty sample containers. The trip blanks are not opened in the field but return to the laboratory with the field samples. Trip blank results are useful for determining whether volatile organic compounds are diffusing across the VOA septum into the sample during storage (e.g., refrigerants) or during the collection/transportation part of sample handling.

Two special types of spiked samples used as part of the protocol for the analysis of organic compounds by gas chromatography are matrix-spiked samples and surrogate-spiked samples. Matrix-spiked samples are field samples to which known amounts of the analytes of interest have been added. Typically, both a spiked and an unspiked aliquot of the sample are analyzed. The difference in results for the two analyses are calculated and compared to the



amount of spike added. Since actual samples are used for the determination, any matrix effects are taken into consideration. Usually expressed as a percentage of the spike amount, spike recovery measures the accuracy of the analysis. For a single sample, this includes the combined effects of bias, or systematic error, and variability due to imprecision. Averaging spike recoveries for multiple samples tends to "average out" the random error caused by imprecision, thus providing an estimate of analytical bias.

Surrogate-spiked samples are similar to normal spiked samples except that an unspiked aliquot is not analyzed. Samples are spiked with a mixture of surrogate compounds chemically similar to the species of interest but not expected to be present in actual field samples. Recovery of these surrogate compounds gives an estimate of the effectiveness of the method during that single analysis.

Duplicate samples and duplicate analyses are used to indicate measurement data precision. Precision indicates the mutual agreement among individual measurements of the same constituents under prescribed similar conditions. Variability among the measurements is attributable to random error caused by imprecision. In the case of duplicate analyses, the analytical process is repeated for separate aliquots of a single sample while prescribed elements of the process are kept constant. For example, duplicate analyses are usually performed on the same day, by the same analyst, using the same instrument and the same calibration. Differences in the results for duplicate analyses, attributable to random variability in the analytical process, reflect analytical precision.

Duplicate samples are another way to measure precision. The analysis of duplicate samples involves replicating sample collection (and the associated sample handling activities) and sample analysis. Precision estimates based on duplicate sample results incorporate imprecision caused by sampling and analytical variability.



Both duplicate sample and duplicate analysis results yield data for estimating precision. Generally, however, duplicate sample data are used in a different way than are results for duplicate analyses. Since analytical precision is primarily a function of the analytical procedures employed. results for duplicate analyses may be used as an ongoing quality control check. Corrective action can be taken when results indicate that analytical precision is not within acceptable limits. Results for duplicate samples, on the other hand, are more often used to assess data quality. Because of the lag between sample collection and the availability of analytical results, it is usually not possible to initiate corrective action based on duplicate sample data. Also, variability in duplicate sample results typically includes a component attributable to inherent nonhomogeneity of the sample or sample matrix. Duplicate data of both types can indicate the degree to which chance may cause results to vary. This information is important whenever measured values are compared. Without it, it is difficult to know whether to attribute observed differences to random measurement error or to actual differences.

Quality control check samples (QCCS) are used to assess analytical performance under a given set of standard conditions. These are synthetic samples (prepared independently of calibration standards) containing some or all of the parameters of interest at known concentrations. Typically analyzed with each set of analyses, QCCS results, by comparing measured values to preestablished acceptability limits, are often used as a real-time check of analytical system performance. By comparing measured results to theoretical concentrations, quality control check sample results may also be used to estimate analytical bias and accuracy Although they do not address matrix effects, as do results for spiked samples, they allow day-to-day consideration of variability and are useful in identifying trends.

Quality control results are summarized and discussed in the following sections. Data for these summaries were obtained from the analytical data reports presented in Appendix A-4. The appendix contains tables of all the quality control information considered.

4.4.2 QA/QC Results For Water Sample Analyses

The QA/QC efforts associated with collecting and analyzing groundwater samples used in the characterization of IRP Phase II Stage I Plant 4 sites included:

- o Analysis of field and laboratory blank samples;
- Analysis of spiked samples;
- Collection and analysis of duplicate samples;
- o Duplicate analyses; and
- o Analysis of QCCS.

Results of these QA/QC analyses are discussed in the following sections.

4.4.2.1 Blank Results

Laboratory, field, and trip blanks were analyzed as part of the QC efforts associated with Plant 4 water sample analyses. Results of the blank analyses indicate the background levels of methylene chloride, trichloro-fluoromethane, trichloroethylene, toluene and phthalates that should be noted if detectable levels of these compounds are reported in actual field samples.

4.4.2.1.1 Laboratory Blanks

Laboratory reagent blanks for groundwater and surface water sample analyses were reagent water samples containing the same proportions of preservatives and reagents as used for field samples. Laboratory blank data for groundwater and surface water samples are contained in Appendix A and summarized in Tables 4.4-1 through 4.4-4. Methylene chloride, trichlorofluoromethane and trichloroethane were detected in the laboratory blanks at concentrations greater than 1.5 times the method detection limit (MDL) but less than 5 times the MDL for Method 601 analyses. These results are presented in Table 4.4-1. LABORATORY BLANK DATA REPORTED ABOVE THE METHOD DETECTION LIMIT FOR WATER SAMPLES ANALYZED FOR ORGANIC PARAMETERS TABLE 4.4-1.

Parameter	Date Analyzed	Concentration (ug/L)	Method Detection Limit (ug/L)	QC Limit ^a (ug/L)
Method 601				
Methylene Chloride	2/12/86	0.44	0.25	0.38
Methylene Chloride	3/5/86	0.51	0.25	0.38
Trichlorofluoromethane	3/4/86	0.81	0.40	0.50
Trichloroethylene	2/6/86	0.18	0.03	0.04
Trichloroethylene	2/6/86	0.18	0.03	0.04
Trichloroethylene	2/26/86	0.20	0.03	0.04
Method 625	Work Order			
di-butyl phthalate	86-02-030	ę	2.5	3.8
di-butyl phthalate	86-02-058	ς	2.5	3.8
di-butyl phthalate	86-02-070	£	2.5	3.8
di-butyl phthalate	86-02-116	ς	2.5	3.8
di-butyl phthalate	86-02-132	с	2.5	3.8
di-butyl phthalate	86-03-021	£	2.5	3.8
di-butyl phthalate	86-03-018	с	2.5	3.8
di-butyl phthalate	86-04-085	9	2.5	3.8
di-butyl phthalate	86-04-135	4	2.5	3.8
bis(2-ethylhexyl)phthlate	86-02-058	с	2.5	3.8
bis(2-ethylhexyl)phthlate	86-02-070	с	2.5	3.8
bis(2-ethylhexyl)phthlate	86-02-116	8	2.5	3.8
N-nit rosodiphenylamine	86-02-078	9	1.9	2.8

^a Estimated as 1.5 times reported detection limit.

PERSON PRESS SECTOR ACCOUNTER DEPENDED DEPENDED DECOMENTER DEPENDENCE

1222243

mu

Q.

TABLE 4.4-2.	LABORATORY	BLANK D	DATA RI	EPORTED	ABOVE	DETECTION	LIMITS	FOR
	METALS ANAL	YSES IN	WATE	R SAMPLE	ES			

Parameter	Number of Samples ^a	Mean Concentration ^b (mg/L)	Number Above Detection Limit	Detection Limit (mg/L)
Silver	10	0.01	1	0.002
Arsenic	13	-	0	0.005
Barium	10	0.003	8	0.001
Cadmium	10	-	0	0.002
Chromium	10	0.021	3	0.005
Mercury	10	0.0003	2	0.0002
Lead	10	0.003	4	0.002
Selenium	10	-	0	0.003

a Total number of reagent blank analyses.

^bMean concentration only for values greater than detection limit.

TABLE 4.4-3. FIELD BLANK GROUNDWATER	SAMPLES	FIELD BLANK DATA REPORTED ABOVE THE METHOD DETECTION LIMIT FOR GROUNDWATER SAMPLES ANALYZED FOR ORGANIC PARAMETERS	DETECTION LIMI	r for	RA
Parameter	Sample ID	Concentration (ug/L)	Method Detection Limit (ug/L)	QC Limit ^a (ug/L)	<u>Pian</u>
Method 601					
Methylene Chloride	86-03-003-06	0.93	0.25	0.38	
Methylene Chloride	86-04-084-06	0.38	0.25	0.38	
Methylene Chloride	86-08-093-05	20.6	0.25	0.38	
Trichlorofluoromethane	86-04-084-06	0.38	0.40	0.60	
Trichlorofluoromethane	86-08-093-05	3.07	0.40	0.60	
Trichloroethylene	86-03-003-06	0.22	0.12	0.18	
Method 602					
Toluene	86-02-031-07	1.73	0.20	0.30	
Toluene	86-02-041-07	2.19	0.20	0.30	
Toluene	86-02-075-08	17.4	0.20	0.30	
Toluene	86-02-087-06	3.33	0.20	0.30	
Toluene	86-04-084-06	0.99	0.20	0.30	
Toluene	86-08-093-05	0.22	0.20	0.30	
Toluene	86-03-003-02	3.56	0.20	0.30	
Toluene	86-02-159-06	0.54	0.20	0.30	
Toluene	86-02-047-06	4.06	0.20	0.30	

B Estimated as 1.5 times reported detection limit. b Territy and the second secon

Instrument detection limit for this analysis allowed quantitation below method detection limit. TRIP BLANK DATA REPORTED ABOVE THE METHOD DETECTION LIMIT FOR WATER SAMPLES ANALYZED FOR ORGANIC PARAMETERS TABLE 4.4-4.

Parameter	Sample ID	Concentration (ug/L)	Method Detection Limit (ug/L)	QC Limit ^a (ug/L)
Method 601				
Methylene Chloride	86-02-075-09	0.34	0.25	0.38
Methylene Chloride	86-02-100-07	0.47	0.25	0.38
Methylene Chloride	86-08-093-06	0.77	0.25	0.38
Trichlorofluoromethane	86-03-031-08	2.52	0.40	0.60
Trichlorofluoromethane	86-02-041-08	3.00	0.40	0.60
Trichlorofluoromethane	86-02-067-06	2.03	0.40	0.60
Trichlorofluoromethane	86-02-075-09	2.11	0.40	0.60
Trichlorofluoromethane	86-02-100-07	2.34	0.40	0.60
Trichlorofluoromethane	86-08-093-06	3.57	0*0	0.60
Trichlorofluoromethane	86-02-047-07	2.75	0.40	0.60
Trichloroethylene	86-02-067-06	0.30	0.12	0.18
Method 602				
Toluene	86-02-067-06	2.50	0.20	0.30
Toluene	86-04-084-07	0.53	0.20	0.30
Toluene	86-08-093-06	3.05	0.20	0.30

As specified in the EPA CLP protocol for organic species; 1.5 times reported detection limit for inorganics and other parameters.

g

RADIAN

Di-butyl phthalate. and bis(2-ethylhexyl) phthalate were detected in several Method 625 blank samples, but at concentrations less than 1.5 times the detection limit. Phthalates are common laboratory contaminants at levels below 5 times MDL. Field samples with phthalate results at this level should be regarded with caution. N-nitrosodiphenylamine was detected in only one reagent blank, at 6 ug/L. This level is three times greater than the laboratory MDL. No major analytical contamination for Method 602 analyses (purgeable aromatics) was indicated by the analytical results for laboratory blanks. Low levels of the halogenated aliphatics reported in laboratory blanks should be considered significant only if any of those compounds were also detected in field samples analyzed by Method 601. With these two exceptions, there is no indication that the levels of organics detected in laboratory blank samples are significant in relation to the analytical results for field samples taken as part of this site characterization.

Results of laboratory blank sample analyses for metals revealed no significant contamination problems. The mean blank concentration reported for barium was 0.003 mg/L, with values for eight out of 10 blank samples greater than the reported detection limit of 0.001 mg/L. This suggests that the actual detection limit during this period might be slightly higher than estimated. Table 4.4-2 summarizes laboratory blank data reported above detection limits for metals analyses of water samples.

4.4.2.1.2 Field Blanks

Samples of organic-free water collected in the field were submitted with the actual field samples for volatile organic analyses by Methods 601 and 602. Table 4.4-3 summarizes the results for field blanks reported at concentrations greater than 1.5 times the MDL. Methylene chloride, trichlorofluoromethane, and trichloroethylene were detected at concentrations greater than 1.5 times the detection limit. Toluene was detected above 1.5 times the MDL (0.2 ug/L) in 8 field blank samples analyzed for volatile aromatic hydrocarbons by Method 602. Reported analytical values for these compounds in actual field samples should be reviewed in light of the field blank background levels. Based on the field blank results, positive values at concentrations less than 20 ug/L for methylene chloride, 3 ug/L for trichloroflouromethane and 0.2 ug/L for trichloroethane should be considered with caution. Toluene values reported at levels less than 17 ug/L should also be regarded with caution in light of possible sample contamination. The effect of samples with higher concentrations will be minimal by comparison. No field blank samples were collected for metals analyses.

4.4.2.1.3 Trip Blanks

Trip blanks were samples of organic-free water in sealed VOA vials that accompanied field samples through transportation and storage, but that were not exposed to sampling procedures. These samples were analyzed for volatile organic compounds by Methods 601 and 602. Table 4.4-4 presents these results, which support the evidence of background levels of methylene chloride and trichloromethane in Method 601 analyses of the field blank samples. Toluene was reported in three of the Method 602 analyses at concentrations greater than 1.5 times MDL. The reported concentrations of these compounds in trip blanks ranged from 0.5 to 3.6 ug/L. Positive values for methylene chloride, trichlorfluoromethane and toluene at concentrations less than 3 ug/L should be considered in the light of possible sample contamination.

4.4.2.2 Spiked Samples

Spiked samples are those to which known amounts of the analytes of interest have been added. These include both matrix spike and surrogate spike analyses. Matrix spikes are made by adding target analytes to an aliquot of the sample. The recovery of matrix spikes indicates matrix effects on sample analyses. Surrogate spikes are readily identifiable species not normally found in the sample matrix that are added to assess method control. Surrogate spikes are added to each sample before analysis for the target compound.



Surrogate and spiked sample results for organic analyses indicated that overall bias and precision for Methods 601, 602, and 625 are within expected limits. Matrix spike results for metals analyses indicate significant matrix interferences for certain metals. These results are discussed below.

4.4.2.2.1 Spiked Sample Results for Organics Analyses

Matrix spike and surrogate spike analyses were performed for the Methods 601, 602 and 625 analyses of water samples. Surrogate spike results for purgeable halocarbons and aromatics (Methods 601 and 602) are summarized in Table 4.4-5. Table 4.4-6 summarizes surrogate spike recoveries for GC/MS BNA extractable organic analyses. Surrogate recoveries indicate small measurement bias and good analytical precision for volatile halocarbons and aromatics analyzed by GC. GC/MS Method 625 (acid and base/neutral extractable organics) surrogate recoveries varied more widely. The mean recoveries for six surrogate compounds ranged from 59 to 105 percent. Overall, Method 625 surrogate recoveries were within the expected ranges of recovery. Recovery of d_5 -phenol and 2-fluorobiphenyl was outside the acceptance criteria for approximately 10 percent of the samples analyzed. This is not excessive, considering that the acceptance criteria represent a distribution of approximately 95 percent. The EPA Contract Laboratory (CLP) expected recovery for each surrogate, using method 625, is provided for comparison in Table 4.4-6.

Table 4.4-7 summarizes matrix spike results (by Methods 601 and 602) for purgeable halocarbons and aromatics. Table 4.4-8 presents a summary of matrix recovery results for Method 625 (BNA extractable organics). Matrix spike recoveries were within acceptable limits for almost every target analyte spiked into the 14 water samples analyzed for volatile organics and the 8 samples analyzed for extractable organics.

Matrix spike results from Method 608 analyses for pesticides and PCBs by GC/MS showed that no target analytes were recovered. The laboratory routinely calibrates for these analytes as part of the instrument tuning TABLE 4.4-5. SUMMARY OF SURROGATE SPIKED SAMPLE RECOVERIES FOR WATER SAMPLES ANALYZED FOR PURCEABLE HALOCARBONS AND AROMATICS

PURGEABLE HALOCARBONS AND AROMATICS

RADIAN

Parameter	Number of Samples ^a	Mean % Recovery	z Relative Standard _b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria (% Recovery)
Method 601						
Bromochloromethane 2-Bromo-1-chloropropane	134 134	105 110	10 13	4 1	n v	84-126 81-139
Method 602						
a,a.a-Trifluorotoluene	140	104 ^d	8d	4	9	87-121
a The number of spiked sample analyses performed on routine samples.	nple analyses	s performed	on routine s	amples.		

States and the states of the

b Relative Standard Deviation of mean x recovery. ^c Taken as x + 2 standard deviations (N-1). ^d One outlier, 298% recovery, was not used in this statistic.

1

Acid Fraction d ₅ -Phenol 106 59 43 2 ⁵ Fluorophenol 106 88 35 2.4.6-Tribromophenol 106 78 48			(% Recovery)	(% Recovery)
1 106 59 1 106 88 ophenol 106 78				
ophenol 106 78	1 2	89 67	8-110 26-150	10-94 21-100
e Fraction	- 1	1 7	3-153	10-123
d-Nitrobenzene 106 81 36	00	5 0	3-139	35-114 42-116
106 105	5 4	л -1	23-187	33-141
es and field duplicates consid atrix spike analyses were not of mean % recovery.	in eveluati ided.) Me range of	ing surrogate	recovery. ithin which	(NOTE: Laboratory
lts lie.		•		
EPA-CLP range of expected surrogate spike recovery results f	for low/medium	i organic level	el water samples	es by Method 625.

SUMMARY OF MATRIX SPIKED SAMPLE RECOVERIES FOR WATER SAMPLES ANALYZED FOR PURGEABLE HALOCARBONS AND AROMATICS TABLE 4.4-7.

Parameter	Number of Samples ^a	Mean % Recovery	% Relative Standard _b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria (% Recovery)
Method 601						
Methylene Chloride	14	94	14	0	0	25-162
1,1-Ďichloroethane	14	75	13	0	0	28-167
trans-1,2-Dichloroethene	14	96	10	0	0	38-155
Chloroform	14	138	13	0	ω	49-133
1,2-Dichloroethane	14	06	21	0	1	51-147
.1.1-Trichloroethane	14	108	6	0	0	41-138
Carbon tetrachloride	14	106	11	0	0	43-143
Bromodichloromethane	14	114	80	0	0	42-172
1,2-Dichloropropane	14	105	6	0	0	44-156
Trichloroethylene	14	122	25	0	ო	35-146
Dibromochloromethane	14	78	34	0	0	24-191
Bronoform	14	108	11	0	0	13-159
Chlorobenzene	14	114	12	0	0	38-150
Tetrachloroethylene	14	85	13	0	0	26-162
Method 602						
Benzene	11	119	10	0	0	39-150
Toluene	11	122	26	0	2	46-148
Ethylbenzene	11	110	13	0	0	32-160
o-Xylene	11	101	17	i	ł	NS
m-Xylene	11	131	23	1	ł	NS
p-Xylene	11	117	12.	ł	ł	SN

The number of spiked sample analyses performed on routine samples and field duplicates.

Relative Standard Deviation of mean % recovery.

c As specified for QC check samples in Method 601 and 602.

TREESE TREESE TREESE TREESE TREETE TREESE TREETE

227 355353

NS - Not specified.

RAPIAN

SAMPLES
WATER
FOR
RESULTS
RECOVERY RESULTS FOR WATER SAMPLE
METHOD 625 MATRIX-SPIKED 1
MATRIX
625 1
OF EPA
SUMMARY
4.4-8.
ABLE

Parameter	Number of Samples	Mean % Recovery	% Relative Standard _b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria ^C (% Recovery)	EPA-CLP Expected Recovery (% Recovery)
Acid Fraction							
•	c		ţ	c	c	D134	
2,4,6-Trichlorophenol	α (00 1	51 25	- c	5 0	U−134 27122	
4-Chloro-3-methylphenol	x c	8 0 6		-	5 0	24-132 24-115	73-07
2-Chlorophenol	οœ	70	2 C 8 C			16-115	27-123
2,4-Dicniorophenol	6 0	44	0 C C		o c	21-72	
4-Dimernyipnenoi Nittonohomol	οα	5 T	67			10-124	
NIT COprenda	οα	, y y	9 L) c		D-165	10-80
4-NIT TOPNENOL 2 4 Distriction	0 00	C 4	104	20	o –	D-129)) •
2.4-DINILLOPMENOL 2-Mathwl-6 6-dinitronhenol	οα	42 87	58	J	• 0	D-189	
z netnyi 4,0 uinittopuon Pentachlorophenol	တ	06	94		0	D-260	9-103
Pheno1	ω	65	18	0	0	42-89	12-89
Base Fraction							
Acenaphthene	ω	83	26	0	0	40-126	
Renzidine	000	24	203	5	1	D-121	
J. 7.4-Trichlorobenzene	0 00	 86	21	0	0	49-123	39-98
Hexachlorobenzene	0 00	106	20	0	0	64-147	
Hexachloroethane	8	82	18	0	0	51-112	
Bis(2-chloroethyl)ether	8	75	25	0	0	38-112	
2-Chloronaphthalene	ω	84	28	0	0	37-130	
1,2-Dichlorobenzene	8	80	20	0	0	48-111	
1,3-Dichlorobenzene	80	77	16	0	0	52-102	1
1,4-Dichlorobenzene	80	73	15	0	0	52-94	36-97
3,3-Dichlorobenzidine	ω	374	178	1	1	D-1714	
4-Dinitrotoluene	ω	93	30	0	0	38-148	24-96
2,6-Dinitrotoluene	ω	93	30	0	0	38-149	
Fluor an thene	8	92	15	0	0	65-119	
4-Chlorophenyl phenyl ether		92	28	0	0	40-143	
N-Nitrosodimethylamine	ω,	40	118	-1 1	0,	D-134	
N-Nit rosodiphenylamine	œ	345	160	0	-1 1	1241-U	
N-Nitrosodi-n-propylamine	ω	72	30	0	0	28-110	41-110

			6	Number	Number		EPA-CLP
Parameter S	Number of Samples ^a	Mean % Recove <i>ry</i>	Relative Standard _b Deviation	Below Acceptance Limits	Above Acceptance Limits	Acceptance Criteria (% Recovery)	Expected Recovery (% Recovery)
Bis(2-ethv1hexv1)phthalate	ω	88	18	0	0	56-120	
Butyl benzyl phthalate	00	47	38	0	0	11-83	
Di-butyl phthalate	œ	88	19	0	0	54-121	
Di-n-octyl phthalate	œ	74	26	0	0	35-113	
Dietyl phthalate	8	81	27	0	0	38-125	
Dimethyl phthalate	8	59	31	0	0	23-95	
Benzo(a) anthracene	œ	16	13	0	0	67-115	
Benzo(a) pyrene	80	87	8	0	0	72-101	
Benzo(b)fluoranthene	8	95	47	0	1	5-185	
Benzo(k)fluoranthene	œ	06	34	0	1	28-152	
Chrysene	80	92	16	0	0	63-121	
Acenaphth1 yene	œ	82	22	0	0	46-117	46-118
Anthracene	80	98	13	0	0	72-124	
4-Bromophenyl phenyl ether	8	105	22	0	0	60-150	
Bis(2-chloroisopropyl)ether	ø	76	36	0	0	22-130	
Bis(2-chloroethoxy)methane	ω	76	18	0	0	48-104	
Hexachlorobut adiene	8	77	47	1	0	4-150	
Hexachlorocyclopentadiene	8	7	141	1	1	D-27	
Isosphorone	8	79	18	0	0	51-107	
Naphthalene	ø	77	20	0	0	47-107	
Nit robenzene	80	75	23	0	0	39-110	
Benzo(ghi)perylene	8	85	32	0	0	30-139	
Fluorene	8	85	22	0	0	48-123	
Phenan threne	ω	06	13	0	0	67-113	
Dibenzo(a,h)anthracene	8	80	37	1	0	21-138	
Indeno(1,2,3-cd)pyrene	80	77	95	0	0	6-149	
Pyrene	8	89	18	0	0	56-121	26-127

4-267

£.

d matrix spike recovery results lie. d EPA-CLP range of expected range of recovery for low/medium organic level water samples by Method 625.

- SAMAN SAMAN MANAN BEERE ENGINE PERSON PARAN PERSON PERSON BEERE

(A. 77. 2.)

1997



procedures for conducting Method 625 analyses. They can also report results for the pesticide and PCB analytes from the same sample injections made for determination of Method 625 analytes. However, the laboratory did not routinely spike pesticide or PCB analytes into the matrix-spiked samples during the period that these samples were analyzed. The reported values of "ND" are correct for these analytes. This is not felt to be a problem since the determination of pesticides and PCBs was not required in the scope of work and the results have been provided at no charge to the client. The Method 608 by GC/MS results for analytes should be considered valid, based on the calibration data, but unqualified, since no surrogate or matrix spike data exist to confirm that the analytes of interest could have been recovered had they been added to each sample.

Of the 14 Method 601 (purgeable halocarbon) target analytes, only the matrix spike results for chloroform indicated unexpected bias. Results for the recovery of chloroform spiked into the matrix indicate a mean percent recovery of 138 percent with eight of 14 spiked sample results outside the acceptance criteria of 49 to 133 percent. This high bias for chloroform should not present a problem unless chloroform was detected in field samples at levels of environmental concern. The possibility of positive bias (relative to the Method 601 published expectations for bias) should be considered when interpreting results. With the exception of the 37 ug/L of chloroform reported at well HM-24, all other positive values for chloroform were less than 10 ug/L, so the bias effect will be negligible.

4.4.2.2.2 Spiked Sample Results for Metals Analyses

Approximately 10 percent of water samples analyzed for metals were spiked with target analytes to assess analytical accuracy in the sample matrices. Spiked sample recovery results for metals analyses in water samples are summarized in Table 4.4-9. These results indicate significant matrix interferences for selenium, lead and cadmium. Sample parameters for which spike recovery results were outside the 75-125 percent recovery range were SUMMARY OF MATRIX-SPIKED SAMPLE RECOVERIES FOR METALS ANALYSES IN WATER SAMPLES TABLE 4.4-9.

RADIAN

Parameter	Number of Samples ^a	Mean % Recovery	Relative Standard _b Deviation	Below Acceptance Limits	Above Acceptance Limits	Acceptance Criteria
Silver	22	88	11.3	2	0	75-125
Arsenic	28	64	6*6	1	0	75-125
Barium	22	91	6.3	0	0	75-125
Cadmium	22	81	14.8	6	0	75-125
Chromium	22	87	8.7	1	0	75-125
Mercury	19	101	17.3	1	1	75-125
Lead	32	62	80.8	13	1	75-125
Selenium	37	64	51.4	17	0	75-125

b Relative Standard Deviation of mean % recovery.

^cAcceptable range of % recovery for indicated analyte.

and seems seemed stated, success second stated bestern balled balled balled broken broken broken and



flagged as suspect. Not every sample was spiked. For this reason, Se. Cd, and Pb results for samples from similar matrices should also be considered suspect because of matrix interferences.

Mean spiked sample recoveries were 64 percent for selenium and 101 percent for mercury. Spiked sample results (mean +1 standard deviation) for cadmium were biased low to a small degree, at 81±12 percent mean recovery. with nine of the 22 spiked sample results less than 75 percent recovery. Results for lead and selenium were also biased low, at 79±64 and 64±32 mean percent recovery, respectively. Thirteen of 32 lead and 17 of 37 selenium spiked sample results showed less than 75 percent recovery, indicating significant matrix interferences. Furthermore, six spiked sample results for selenium showed zero percent recovery. In light of these spiked sample recoveries, reported results for cadmium and lead should be considered to be biased low by about 20 percent. Positive results for selenium should also be considered biased low. Not detected (ND) results for selenium that are flagged or that are from matrices similar to those flagged should be considered with caution.

In several cases, spiked samples that failed the acceptance criteria of 75-110 percent recovery were diluted by a factor of 10, reanalyzed, spiked, and reanalyzed. This procedure usually resulted in acceptable spike recoveries (in the diluted sample) and "not detected" for the endogenous level but a 10-fold increase in the limit of detection. A detailed listing of spike recovery data for metals analyses appears in Appendix A-4.

Quality control check samples (QCCS) containing metals at known concentrations were analyzed to assess analytical performance in the absence of matrix effects. Table 4.4-10 summarizes QCCS recoveries for metals analyses in water samples. Mean percent recoveries were within the laboratory bias objective of 90-110 percent for all eight metals analyzed, and repeatability, expressed as the relative standard deviation (RSD) of QCCS results, was acceptable, ranging from 2.3 to 8.7 percent RSD.

RAPIAN

No. Contraction

Parameter	Number of Samples ^a	Mean % Recovery	% Relative Standard _b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria ^C
Silver	14	97	8.7	2	0	90-110
Arsenic	38	66	7.1	2	1	90-110
Barium	14	102	2.3	0	0	90-110
Cadmium	14	102	2.4	0	0	90-110
Chromium	14	101	2.2	0	0	90-110
Mercury	36	66	7.2	0	0	80-120
Lead	35	102	5.1	0	4	90-110
Selenium	28	96	7.6	Q	0	90-110

4-271

The number of QCCS analyses performed with routine samples.

b_Relative Standard Deviation of mean % recovery.

^CAcceptable range of % recovery for indicated analyte.

51

<u> 1888 - 1988 - 1988</u>

ACCENTED TO

RAPIAN

4.4.2.3 Duplicate Samples and Duplicate Analyses

The QC protocol for Plant 4 groundwater and surface water sampling included collection of duplicate samples as well as the performance of duplicate analyses in the laboratory. Duplicate sample results measure overall variability in sample collection and analysis. Results for the duplicate analysis of individual samples are a measure of analytical precision.

Table 4.4-11 summarizes results of the analyses of duplicate samples. Sampling plus analytical variability averaged 20 percent RSD for Method 601 analyses of duplicate samples, 70 percent for duplicate samples analyzed by Method 602, and 15 percent for analyses of duplicate samples by Method 625 (acid and base/neutral extractable organics).

Table 4.4-12 presents precision estimates for laboratory duplicates for the analyses of volatile organics by Methods 601 and 602. Table 4.4-13 presents the same information for acid and base/neutral extractable organics by EPA Method 625. Analytical variability was as expected for most results, averaging 15 percent RSD for volatile halocarbons, 35 percent for volatile aromatics, and 39 percent for acid and base/neutral extractable organics. The results indicate that results for low levels of phthalates and toluene may be highly variable.

Table 4.4-14 summarizes the metals analyses results for duplicate analysis and duplicate sample variability. Sampling plus analytical variability, expressed as the RSD of duplicate sample results, was 22 percent for arsenic, 23 percent for barium and 72 percent for chromium. Concentrations of the other five metals in the duplicate sample pairs were reported at concentrations less than the limit of reliable quantitation so no variability estimates were possible. Analytical variability, expressed as the RSD for duplicate analyses, ranged from 5 percent for barium to 20 percent for silver.

į
i
ļ
ł
I
į
i

			33. 83.558		8.22.22 8.22.22 8.25				
	TABLE 4.4-11.	ANALYSES OF		LICATE WATER S	FIELD DUPLICATE WATER SAMPLES FOR ORGANIC PARAMETERS	C PARAMETERS		RA	
	Parameter	Monitor Well	Routine Value (ug/L)	Duplicate Value (ug/L)	Mean Concentration (ug/L)	Standard Deviation ^a	% Relative Standard b Deviation	PIAN	9040-9040-909 9040-9040-908
	Method 601								ererer
	Vinvl Chloride	P-10U	0.20	0.74	0.47	0.38	81		
	Methvlene Chloride	P-10U	48.7	91.2	70.0	30.1	43		с Р С
	trans-1.2-Dichloroethene	P-10M	0.30	0.25	0.28	0.035	12		20
	trans-1.2-Dichloroethene	HM-58	1.94	2.02	1.98	0.057	2.9		2.5
	Chloroform	HM-52	4.25	3.79	4.02	0.325	8.1		7.74
	Chloroform	0E1-1730	2.79	2.57	2.68	0.156	5.8		177
	Trichloroethylene	P-10U	1.01	0.82	0.92	0.13	14		
4		HM-47	10,979	12,066	11,522	768.63	6.7		
i-2		НМ-70	10,983	10,866	10,924	82.731	0.8		
273		HM-5 2	0.93	0.95	0.94	0.014	1.5		22
5	Trichloroethylene	HM-58	0.74	0.84	0.79	0.071	0.6		90
	Dibromochloromethane	HM-52	0.42	0.72	0.57	0.21	37		
	Dibromochloromethane	0F1-1730	5.58	4.45	5.02	0.799	16 2		777
	Bromoform	0F1-1730	2.00	1.78	1.89	0.156	8.2		. <u>.</u>
	Chlorobenzene	HM-47	308	171	540	32/	10		- C1
	Bromodichloromethane	0F1-1730	4.69	3.68	4.18	0.714	17		

(Continued)

Method 602 Method 602 Chlorobenzene HM-47 738 709 723 20. Toluene P-10M 15.8 2.79 9.30 9. Toluene HM-52 2.96 0.90 1.9 1.		(ng/L)	Mean Concentration (ug/L)	Standard a Deviation	% Relative Standard Deviation beviation
robenzene HM~47 ene P-10M ene HM~52					
ene HM~34 ene P-20	738 15.8 2.96 1.46 7.39	709 2.79 0.90 125 ^c 11.1	723 9.30 1.9 63 9.24	20.5 9.20 1.5 87 2.62	2.8 99 79 28
<u>od 625</u>					
utyl phthalate 2-ethylhexyl)phthalate P-20	4 0	Q 3	3.5 5.5	0.71 0.71	20 10
butyl phthalate P-20 (2-ethylhexyl)phthalate P-20	4 0	e a	3.5 5.5	0.71 0.71	20 10

PARAMETER
ORGANIC
ND 602
601 A
METHOD
5 FOR
SAMPLES
WATER
NALYSES OF
LABORATORY DUPLICATE ANALYSES OF WATER SAMPLES FOR METHOD 601 AND 602 ORGANIC PARAMETER
LABORATOR
E 4.4-12.
TABL

Routine Value Parameter Sample ID (ug/L)	Routine Value (ug/L)	Duplicate Value (ug/L)	Mean Concentration (ug/L)	Standørd ^a Deviation	% Relative Standard _b Deviation
860153	968	1,114	1,041	103.2	6 . 9
860216	3.9	5.3 10 700	4.6 77 057	0,99 0,77	0 8 77
800103	20,200 8 48	24,04	8.14	0.474	. 8. 0
860216 860216	0.36	0.56	0.46	0.14	30
860216	16.7	21.5	19.1	3.39	18
860153	2.274	3,351	2,812	761.6	27
860146	26.4	22.3	24.4	2.90	12
860216	8.0	10.5	9.2	1.8	20
860179	10,983	12,492	11,738	1,067	9.1
860166	2,367	2,201	2,284	117.4	5.1
860163	177	176	176	0.707	0.40
860153	150	237	194	61.5	0.32
860146	1.31	0.74	1.0	0.40	40
860194	0.09	0.08	0.08	0.007	б
				(Co	(Continued)

57
Þ
5
ي ي ي
5
ζ.
5
مند

(Continued)	
4.4-12.	
TABLE	

Parameter	Routine Sample ID	Routine Value (ug/L)	Duplicate Value (ug/L)	Mean Concentration (ug/L)	Standard Deviation ^a	% Relative Standard b Deviation
Method 602						
Toluene	860156	1.43	1.07	1.25	0.255	20.4
rotuced Tolijane	860159	1.79	1.76	1.78	0.021	1,2
Tolucito Tolucito	860167	0.83	0.76	0.80	0.049	0.062
Toluene	860175 ^C	10.0	0.87	5.44	6.46 _d	119 ⁴
Toluene	860192 ^C	0.84	QN	3	,	I
Toluene	Field Blank	2.19	2.14	2.17	0.35	16
Toluene	Trip Blank	2.50	1.17	1.84	0.940	51.1
d number of Accelor of moon	concentration					

^d Standard deviation of mean concentration. ^b Relative Standard Deviation of % replicate analyses. ^c Data sheet was annotated: "There is a top layer that was in the first run but not in the second run". d Cannot be calculated or result not meaningful.

ND - Not detected.

LABORATORY DUPLICATE ANALYSES OF WATER SAMPLES FOR METHOD 625 ORGANIC PARAMETERS TABLE 4.4-13.

Parameter	Routine Sample ID	Routine Value (ug/L)	Duplicate Value (ug/L)	Mean Concentration (ug/L)	Standard Deviation ^a	% Relative Standard b Deviation
Method 625						
	96 10 90	ç	۳۰	2.5	0.71	28
Phonol	86-01-20 86-01-242	4 C	ب ه ر	2.5	0.71	28
Filenoi Bhonoi	86-01-116	1 1		2.0	1.4	70
fiction di-butul nhthalate	86-02-058	- 2 BL	4 BL	3.0	1.4	47
di-bucyi pucuatate di-butvl nhthalate	860221	12 BL	5 BL	8.5	5.0	59
di-bucyi purunataco di-butul nhthalato	86013	2 BL	8 BL	5.0	4.2	84
di-buty1 phthalate	860152	3 BL	4 BL	3.5	0.71	20
bis(2-ethylhexv1) phthalate	860221	Ś	4	4.5	0.71	16
bis(2-ethylhexyl)phthalate	86013	e	en	3.0	0	0

^a Standard Deviation of mean concentration. ^b polating standard Daviation of % renlicat

4-277

Relative Standard Deviation of % replicate analyses.

^c Work Order number.

BL - Data qualifier used by the analytical laboratory indicating that the compound was also detected in the laboratory blank.



Parameter	Number of Field Duplicate Pairs ^a	Sampling Plus Analytical Variability ^b	Number of Analytical Duplicate Pairs ^a	Analytical Variability ^C
Silver	0	NC	5	20
Arsenic	2	22	7	13
Barium	6	23	17	5.4
Cadmium	0	NC	0	NC
Chromium	5	72	8	5.5
Mercury	0	NC	1	7.5
Lead	0	NC	11	8.8
Selenium	0	NC	0	NC

TABLE 4.4-14. SUMMARY OF PRECISION ESTIMATES BASED ON DUPLICATE WATER SAMPLES AND ANALYSES FOR METALS

<u>}`</u>\$\${`\$}{`\$}_\$\$

NC - not calculated. All values less than limit of quantitation.

^aNumber of duplicate pairs where at least one concentration is greater than the limit of quantitation.

^b% Pooled Relative Standard Deviation for duplicate samples reported above quantitation limit; represents total variability of the measurement process.

^CPooled Relative Standard Deviation for replicate analyses of individual samples reported above quantitation limit; represents analytical variability independent of sampling variability. The concentrations of cadmium and selenium were below the limit of quantitation in all duplicate analysis pairs, so no precision estimates for these two parameters could be made. 2112112

11522233 7222222

4.4.3 QA/QC Results for Soil Sample Analyses

Quality control efforts associated with the collection and analysis of soil samples used in characterization of Plant 4 sites included:

- Analysis of laboratory blank samples;
- Analysis of spiked samples;
- o Collection and analysis of duplicate samples; and
- o Duplicate analyses.

Results for these QC analyses are discussed in the following sections.

4.4.3.1 Blank Sample Results

Only laboratory blanks were analyzed as part of the QC efforts associated with Plant 4 soil sample analyses. Blank sample results of the analysis for purgeable halocarbons and aromatics, acid and base/neutral extractable organic compounds, and trace metals are discussed below. The laboratory blank analyses indicate that no significant sample contamination was due to handling during analysis. Background levels of certain organic compounds must be considered in the interpretation of results.

The laboratory (reagent) blank samples for the soil sample analyses were prepared using the same reagents and preservatives as for the preparation and analyses of field samples. Table 4.4-15 summarizes the results for laboratory blanks reported at concentrations exceeding the QC limits for soil samples analyzed for halogenated volatile organics (Method 8010) and aromatic volatile organics (Method 8020). The results show high blank sample levels for the volatile halocarbons, 1,1,1-trichloroethane, trichloroethylene, and tetrachloro- ethylene, and for the aromatic hydrocarbons, benzene, toluene, and ortho-, meta-, and para-xylene. Any contribution compounds found in

REAGENT BLANK DATA REPORTED ABOVE THE METHOD DETECTION LIMIT FOR SOIL SAMPLES ANALYZED FOR ORGANIC PARAMETERS TABLE 4.4-15.

RADIAN

Parameter	Date Analyzed	Concentration (ug/L)	Method Detection Limit (ug/L)	QC Limit ^a (ug/L
Method 8010				
1,1,1-Trichloroethane	5/14/86	1.26	0.03	0.04
1,1,1-Trichloroethane	5/15/86	1.78	0.03	0.04
Trichloroethylene	5/14/85	0.59	0.12	0.18
Trichloroethylene	5/15/86	1.03	0.12	0.18
Trichloroethylene _k	1/29/86	0.13	0.12	0.18
Tetrachloroethylene ^U	1/29/86	0.17 ^D	0.03	0.04
Tetrachloroethylene	5/14/86	0.34	0.03	0.04
Tetrachloroethylene	5/15/86	0.32	0.03	0.04
Tetrachloroethylene	1/28/86	0.13	0.03	0.04
Method 8020				
Benzene	1/31/86	0.91	0.2	0.3
Benzene	5/24/86	3.72	0.2	0.3
Toluene	1/30/86	5.45	0.2	0.3
Toluene	1/31/86	6.64	0.2	0.3
Toluene	5/24/86	3.77	0.2	0.3
p-Xylene	1/31/86	0.89	0.2	0.3
p-Xylene	5/24/86	0.77	0.2	0.3
m-Xylene	1/31/86	1.24	0.2	0.3
m-Xylene	5/24/86	1.43	0.2	0.3
o-Xylene	1/31/86	0.99	0.2	0.3
o-Xylene	5/24/86	1.03	0.2	0.3

Estimated as 1.5 times reported detection limit.

Data sheet did not distinguish between 1,1,2,2-tetrachloroethylene and tetrachloroethylene. م

APARANA DIMINIA DUNINA MANANA MINILA DIMANA MINILA DIMANA MANIMU ANANIMI DUNINA MANANA MANANA MANANA DI

122.22



laboratory blanks can be related to field soil samples by using the mass of sample extracted, extract volume and final dilution factor if these compounds were reported above detection limits in any field samples. Analyses of reagent blanks by Method 8270 acid and (base/neutral extractable organics) showed no compounds were reported at concentrations above the detection limits.

Laboratory reagent blanks were analyzed as part of the metals analyses of soil samples. Table 4.4-16 summarizes the results for laboratory blanks reported at concentrations above the laboratory method detection limits. These results indicate no serious contamination. Half of the blank results for barium were above the detection limit of 0.001 mg/L. Those results above the detection limit averaged 0.004 mg/L. Blank samples analyzed by ICP for arsenic, lead and selenium in certain soil extracts reflected the higher detection limit for analysis of these metals by ICPES rather than by AA.

Since no soil field blanks or trip blanks were collected during this study, no estimate of the contribution of sample handling to overall measurement variability can be made.

4.4.3.2 Spiked Sample Analysis Results

Surrogate-spiked samples were used to monitor method control for EPA Method 8010 (volatile halocarbons), 8020 (volatile aromatics), 8240 (volatile organics) and 8270 (BNA extractable organics) analyses of soil samples. Surrogate spike results for these analyses are summarized in Table 4.4-17 for Methods 8010 and 8020 and in Table 4.4-18 for GC/MS Methods 8270 and 8240. The results of these analyses indicate acceptable measurement bias and good repeatability. The mean percent recovery for surrogate spike recovery by Method 601 ranged from 101 to 112 percent for the two surrogate compounds. Relative standard deviations for the surrogate recoveries were 8 and 11 percent. The mean recovery for the Method 602 surrogate compounds was 101 percent, with a relative standard deviation of 5 percent for the 19 measurements. Mean surrogate recoveries for GC/MS acid and base/neutral extractable

Å

TABLE 4.4-16.	LABORATORY BLAN	C DATA	REPORTED	ABOVE	DETECTION	LIMITS	FOR
	METALS ANALYSES	OF SO	IL SAMPLES	5			

Number of Samples	Mean Concentration (mg/L)	Number Above Detection Limit (mg/L)	Detection Limit (mg/L)
6	0.003	1	0.002
3	-	0	0.006
6	0.004	3	0.001
6	-	0	0.002
7	0.006	1	0.005
7	-	0	0.0002
3	-	0	0.002 ^c
3	-	0	0.001 ^c
	Samples ^a 6 3 6 6 7 7 7 3	Samples ^a Concentration ^D (mg/L) 6 0.003 3 - 6 0.004 6 - 7 0.006 7 - 3 -	Samples ^a Concentration (mg/L) Detection Limit (mg/L) 6 0.003 1 3 - 0 6 0.004 3 6 - 0 7 0.006 1 7 - 0 3 - 0

^aTotal number of reagent blank analyses.

^bMean concentration only for values greater than detection limit.

 $^{\rm C}$ ICP detection limit is 0.06 mg/L for As and 0.08 mg/L for Pb and Se.
ALYZED	
SAMPLES AN	
SOIL	
FOR	
RECOVERIES	••
SAMPLE	ROMATICS
4-17. SUMMARY OF SURROGATE-SPIKED SAMPLE RECOVERIES FOR SOIL SAMPLES ANALYZED	PURCEARLE HALOCARBONS AND AROMATICS
TABLE 4.4-17. S	4

FOR

Parameter	Number of Samples ^a	Mean % Recovery	% Relative Standard Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria (% Recovery)
Method 8010						
Bromochloromethane 2-Bromo-1-chloropropane	18 10	111 112	8 11	00	0 1	92-130 87-137
Method 8020						
a,a.a-Trifluorotoluene	19	101	2	1	0	90-112
^a The number of spiked sample analyses performed on routine samples. ^b Relative Standard Deviation of mean % recovery. ^c Taken as x ± 2 standard deviations (N-1).	ple analyses ion of mean deviations (<pre>% performed % recovery (N-1).</pre>	on routine s	amples.		

RADIAN

TARGER REPORT TRADES DERIVER DES

XX-22-22

SUMMARY OF EPA METHOD 8270 SURROGATE-SPIKED RECOVERY RESULTS FOR SOIL SAMPLES TABLE 4.4-18.

<u>S</u>. . .

Parameter	Number of Samples	Mean X Recovery	X Relative Standard _b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria ^C (% Recovery)	EPA-CLP Expected Recovery (X Recovery)
Method 8270							
Acid Fraction							
d, -Phenol	7	64	20	0	0	38-90	24-113
2 ² Fluorophenol	7	44	44	0	0	6-82	25-121
2,4,6-Tribromophenol	7	101	27	0	0	47-155	19-122
Base Fraction							
dNitrobenzene	7	11	26	0	0	37-117	23-120
2 ² Fluorobiphenyl	7	106	13	0	0	78-134	30-115
d ₁₄ -Terphenol	7	58	13	0	0	43-73	18-137
Method 8240							
d ₁ 1,1-Dichloroethane	4	16	7.5	0	0	76-106	70-121
d _o Toluene	4	97	4.9	0	0	87-107	81-117
Bromofluorobenzene	4	92	14.6	0	0	63-121	74-121
^a The number of routine samples and field duplicates considered in ev blanks field blanks and matrix snike analyses were not included '	iamples and fi nd matri≖ spi	eld duplics ke analyses	ites considere were not inc	ed in evaluat: Juded)	and field duplicates considered in evaluating surrogate recovery. it snike analyses were not included)		(NOTE: Laboratory

This is the range of % recovery within which >90% of the surrogate blanks, field blanks, and matrix spike analyses were not included.) Relative Standard Deviation of mean % recovery. The mean % recovery ± 2 standard deviations (N-1). This is the range of % recovery within which >90% of the surroga spike recovery results lie. EPA-GLP range of expected surrogate spike recovery results for low/medium organic level soil samples by Method 8270. υQ

P

RADIAN



organics were more widely varied, ranging from 58 to 101 mean percent recovery for six surrogate compounds. All recoveries were within the <u>+</u>2 standard deviation limit and agreed well with the EPA Contract Laboratory Program (CLP) expected recovery limits. Surrogate compound recoveries for Method 8240 analyses ranged from 91 to 97 mean percent recovery. All recoveries were within the EPA-CLP expected recovery limits.

Soil samples analyzed for acid and base/neutral extractable organics by GC/MS Method 8270 were also spiked with target analytes (matrix spikes). Table 4.4-19 summarizes the Method 8270 analytical results for soil matrixspiked samples. These results show that parameter recoveries in the sample matrices tested were within the expected ranges. For compounds that have CLP expected recovery criteria, recoveries were also within the expected limits.

The quality assurance review of these data noted that several matrix spike reports for Method 608 (pesticides and PCBs) were present in the data volume (Appendices A-2 and A-3). The results were all reported as "ND". An inquiry revealed that the analyses for pesticides and PCBs by GC/MS had been included as part of the Method 625 analytical run for each sample. Each pesticide and some PCBs were calibrated for, but none were included in the matrix spikes. The correct result for these matrix spike analyses is "ND". No other QC data exist to support Method 608 by GC/MS results.

Matrix effects were not a problem in the analyses of soil for metals, except for selenium, and to a lesser extent, cadmium. The results of matrix spike analyses for the eight metals in soil samples ranged from 72 to 102 mean percent recovery. The mean percent recovery was outside the acceptance criteria of 75-125 percent for both cadmium and selenium. Sample results accompanied by spiked sample recovery data that indicate matrix interference are flagged, but results for similar samples should also be evaluated in light of the spike recovery data. Table 4.4-20 summarizes the matrix-spiked sample recoveries for metals analyses of soil samples. でしていていると Ì

Parameter	Number of Samples	Mean 7 Recovery	X Relative Standard _b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limita	Acceptance Criteria (X Recovery)	EPA-CLP Expected Recovery (2 Recovery)
Acid Fraction							
2.4.6-Trichlorophenol	7	108	7.9	0	0	91-125	
4-Chloro-3-methylphenol	2	100	6.9	0	0	80-120	
2-Chlorophenol	2	92	6.2	0	0	81-103	25-102
2.4-Dichlorophenol	2	66	10	0	0	79-119	
2,4-Dimethylphenol	2	63	67	0	0	D-148	
2-Nitrophenol	2	78	17	0	0	51-104	
4-Nit rophenol	2	100	22	0	0	56-143	11-114
2,4-Dinitrophenol	7	31	11	0	0	23-38	
2-Methyl-4,6-dinitrophenol	2	98	19	0	0	61-135	
Pentachlorophenol	7	121	19	0	0	76-166	17-109
Phenol	2	76	0.0	0	0	76-76	26–90
Base Fraction							
Acenaphthene	2	103	18	0	0	66-140	
Benzidine	6	35	135	0	0	D-129	
1.2.4-Trichlorobenzene	7	101	1.4	0	0	98-104	38-107
Hexachlorobenzene	7	103	24	0	0	53-152	
Hexachloroethane	2	291	96	0	0	D-851	
Bis(2-chloroethyl)ether	7	85	12	0	0	63-106	
2-Chloronaphthalene	2	102	17	0	0	66-137	
1.2-Dichlorobenzene	2	92	3.0	0	0	86-98	
1,3-Dichlorobenzene	2	94	3.7	0	0	86-101	
1,4-Dichlorobenzene	2	86	5.7	0	0	76-95	28-104
3,3-Dichlorobenzidine	2	167	48	0	0	7-326	
2,4-Dinitrotoluene	2	101	3.5	0	0	93-108	28-89
2,6-Dinitrotoluene	2	122	7.5	0	0	103-140	
Fluor an thene	2	126	18	0	0	81-171	
4-Chlorophenyl phenyl ether	7	110	17	0	0	73-147	26-103
N-Nitrosodimethylamine	2	58	62	0	0	D-130	
N-Nitrosodiphenylamine	2	116	30	0	0	46-185	
W Without a second and a fear							

BARRAN PRESSING AND PRESSING MARKED

KULLER STREET PERSON

LECTOR DESCENSION PRESSOR

Secret manual

(Continued)

R	TABLE 4.4-19. (Continued)

(Continued)
5
7
4
4
TABLE

RADIAN

Parameter	Number of Samples ^B	Mean X Recovery	Relative Standard _b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria ^C (% Recovery)	EFA-CLP Expected _d Recovery (% Recovery)
Bis(2-ethylhexyl)phthalate	2	105	61	0	0	D-232	
Butyl benzyl phthalate	7	33	56	0	0	D-70	
Di-butyl phthalate	2	102	33	0	0	34-170	
Di-n-octyl phthalate	2	115	58	0	0	D-249	
Dietyl phthalate	2	92	37	0	0	24-160	
Dimethyl phthalate	2	76	85	0	0	D-204	
Benzo(a) anthracene	2	108	26	0	0	52-163	
Benzo(a) pyrene	2	113	41	0	0	20-206	
Benzo(b)fluoranthene	2	123	45	0	0	11-234	
Benzo(k)fluoranthene	2	103	25	0	0	50-155	
Chrysene	2	109	38	0	0	27-191	
Acenaphthl yene	2	98	15	0	0	68-127	31-137
Anthracene	2	67	112	0	0	D-217	
4-Bromophenyl phenyl ether	2	107	36	0	0	29-184	
Bis(2-chloroisopropyl)ether	2	98	0.7	0	0	66-96	
Bis(2-chloroethoxy)methane	2	87	9.8	0	0	70-104	
Hexachlorobut adiene	2	107	0.7	0	0	105-108	
Hexachlorocyclopentadiene	2	59	141	0	0	D-226	
Isosphorone	2	105	26	0	0	49-160	
Naphthalene	2	87	5.6	0	0	77-96	
Nit robenzene	2	87	1.6	0	0	84-90	
Benzo(ghi)perylene	2	106	27	0	0	49-163	
Fluorene	2	96	19	0	0	59-136	
Phenan threne	2	106	10	0	0	84-127	
Dibenzo(a,h)anthracene	2	110	24	0	0	57-162	
Indeno(1,2,3-cd)pyrene	7	106	25	0	0	53-158	
Pyrene	7	104	44	0	0	13-195	35-142

^V Relative Standard Deviation of mean % recovery.
^C The mean recovery ± 2 standard deviations (N-1). This is the range of % recovery within which >90% of the reported matrix spike recovery results lie.
^d EPA-CLP range of expected range of recovery for low/medium organic level soil samples by Method 625.

a contraction of the

1 10000000000

Marcen marcene marce

SUMMARY OF MATRIX-SPIKED SAMPLE RECOVERIES FOR METALS ANALYSES OF SOIL SAMPLES TABLE 4.4-20.

Parameter	Number of Samples ^a	Mean % Recovery	% Relative Standard b Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria ^C
Silver	5	82	19	-	0	75-125
Arsenic	5	89	12	1	0	75-125
Barium	S	102	38	0	1	75-125
Cadmium	5	72	17	¢,	0	75-125
Chromium	7	85	6.6	0	0	75-125
Mercury	ę	93	12	0	0	75-125
Lead	4	87	13	0	0	75-125
Selenium	7	74	40	ę	0	75-125

mean % recovery.

for indicated parameters. ^URelative Standard Deviation of ^CAcceptable range of % recovery ale determined treested salated beleene bedened beterre breeze beschool beschool beschool beschool breezer bree



In addition to matrix spikes, QC check samples (QCCS) were analyzed to monitor analytical accuracy in the absence of matrix effects. Table 4.4-21 summarizes the QCCS recoveries for metals analyses of soil samples. The results show mean percent recoveries for all eight metals of interest within the acceptance criterion of 90-110 percent recovery. QCCS results were low (76 mean percent recovery) for silver analyses on 23 March, but overall measurement bias was not significant for silver or any other metal of interest. Based on QCCS results, bias estimates for the eight target metals ranged from 94<u>+</u>10 percent for silver to 102<u>+</u>3 percent for cadmium.

4.4.3.3 Duplicate Samples and Duplicate Analyses

Duplicate analysis results for volatile organic analyses of soil samples yielded limited analytical precision data since most compounds were not detected. Positive values were reported for two sets of duplicate analyses by Method 8010. The mean relative standard deviation for 1,1,1-trichloroethane in the two duplicate analysis pairs was 8 percent. Those results are presented in Table 4.4-22.

No duplicate soil samples were collected and analyzed for metals, but duplicate analyses were performed for approximately 10 percent of the field samples. Table 4.4-23 summarizes the results for duplicate analyses when at least one of the duplicate results is greater than the limit of quantitation. The limit of reliable quantitation for metals analyses is taken as five times the limit of detection. Values between the limit of detection and limit of quantitation are semi-quantitative, since uncertainty at this level approaches 100 percent. Analytical variability for metals in soil samples ranged from zero to 12 percent RSD, averaging four percent RSD for the eight metals analyzed. TABLE 4.4-21. SUMMARY OF QCCS RECOVERIES FOR METALS ANALYSES OF SOIL SAMPLES

Parameter	Number of Samples	Mean % Recovery	% Relative Standard Deviation	Number Below Acceptance Limits	Number Above Acceptance Limits	Acceptance Criteria	
Silver	15	94	10.2	£	o	90-110	
Arsenic	8	105	8.7	0	e	90-110	
Barium	15	101	3.0	0	0	90-110	
Cadmium	15	102	2.6	0	0	90-110	
Chromium	17	101	3.0	0	0	90-110	
Mercury	13	26	5.0	0	0	80-120	
Lead	10	67	5.2	2	0	90-110	
Selenium	12	97	6.7	2	0	90-110	
^a The number ^b Relative St	The number of spiked sample analyses performed Relative Standard Deviation of mean % recovery.	le analyses p on of mean %	^a The number of spiked sample analyses performed on routine samples. ^b Relative Standard Deviation of mean % recovery.	utine samples.			

recovery for indicated parameters. 8 cAcceptable range of <u>ka sekata kanana sanan sanan sekata sekata kenata kenata kenata kenata kenata kenata kenata kenata berata bera</u>

RADIAN

-04

RADIAN

ANALYSES	
ORGANIC	
FOR	
SAMPLES	
SOIL	
FOR	
ANAL YSES	
DUPLICATE ANALYSES FOR SOIL SAMPLES FOR ORGANIC ANALYSES	
TABLE 4.4-22.	

Parameter	Routine Sample ID	Routine Value (ug/kg)	Dupiicate Value (ug/kg)	mean Concentration (ug/kg)	Standard Deviation ^a	kelative Standard _b Deviation
Method 8010						
1.1.1-Trichloroethane	86-05-072-01	20	21.	20.5	0.71	3.5
1,1,1-Trichloroethane	86-05-072-02	21	18	19.5	2.1	11

م,

Standard Deviation of mean concentration. Relative Standard Deviation of % replicate analyses.

4-291

TABLE 4.4-23. SUMMARY OF PRECISION ESTIMATES BASED ON DUPLICATE SOIL SAMPLE ANALYSES FOR METALS

Parameter	Number of Analytical Duplicate Pairs	Analytical Variability ^a
Silver	4	12
Arsenic	1	3.2
Barium	5	5.7
Cadmium	1	0.6
Chromium	5	8.5
Mercury	1	2.7
Lead	1	0.0
Selenium	1	1.9

لاندنانه

^aPooled relative standard deviation for replicate analyses of individual samples reported above quantitation limit; represents analytical variability independent of sampling variability.

RADIAN

4.4.4 <u>QA/QC Results for Oil and Grease and Total Hydrocarbons</u> Analyses in Water and Soil Samples by IR

QC procedures associated with analyses for oil and grease and for hydrocarbons by infrared spectroscopy included analysis of

- Quality control check samples (QCCS);
- o Duplicate analyses; and
- o Analysis of duplicate samples.

Results for these QC tests appear in Appendix A and are discussed below. No problems were identified.

4.4.4.1 QCCS Results

QCCS results showed acceptable accuracy and repeatability for IR analyses. The results of QCCS analyses for oil and grease averaged 96 percent recovery, ranging from 89 to 99 percent. Hydrocarbon fuel QCCS results averaged 104 percent recovery, ranging from 92 to 118 percent recovery. The RSD for QCCS results was approximately 8 percent for hydrocarbon analyses and 5 percent for oil and grease analyses.

4.4.4.2 Duplicate Analyses

Duplicate analyses were performed on aliquots split from one sample before sample preparation. Oil and grease duplicate analyses showed a variahility of 21 percent (about 15 percent RSD); the hydrocarbon duplicate results were identical.

4.4.4.3 Duplicate Samples

Duplicate samples collected and analyzed for oil and grease and fuel hydrocarbons were free of contamination, but provided little precision information. The only parameters detected in the duplicate sample pairs were oil



and grease, reported at 1 mg/L in three of the samples in two duplicate pairs; the fourth value was below the detection limit of 1 mg/L. Duplicate sample results were also obtained for temperature, pH and conductivity; all were identical for each of the duplicate pairs.

4.4.5 <u>Radioactivity Analyses</u>

Standard QA/QC methods for radioanalytical work include the use of:

- Check sources and standard sources;
- o Background and blank samples;
- Replicate samples and counts;
- o Spiked samples; and
- o Special samples and counts.

Results for these procedures are discussed below. Detailed radioanalytical QA/QC results appear in Appendix A.

According to Nuclear Regulatory Commission (NRC) regulations, the maximum concentration for quantities of unknown mixtures of radionuclides is 3 x 10^{-8} uCi/ml or 3 x 10^{-8} uCi/g for solids (i.e., 30 pCi/L or pCi/kg). Gross alpha and gross beta counts were less than 6 and 8 pCi/L, respectively, in liquid samples. Alpha and beta radiation levels in solid samples were much higher, reported at <12,000 pCi/kg gross alpha and <24,000 pCi/kg gross beta, but no blank soil radioactivity was subtracted. Natural radioactivity in soils may be this high, and without knowing the endogenous level, anthropogenic contamination is not an acceptable conclusion. For cesium-137, the concentration limit in an unrestricted area is 20,000 pCi/kg for soluble species and 40,000 pCi/kg for insoluble species. Cesium-137 gamma radiation was less than 10 pCi/kg in liquid samples and less than 50 pCi/kg in solid samples. Nuclear fission by-product contamination or radioactive fallout would be clearly evidenced by the presence of cesium-137, which is a stable, strong indicator of nonnatural radioactivity.

RADIAN

4.4.5.1 Check Sources and Standard Solutions

National Bureau of Standards (NBS)-traceable check sources and standard solutions are used to verify that the instrument is operating properly. Counting rates for check sources that fall outside of statistically valid ranges indicate an instrument malfunction, contamination, or improper instrument settings that need correction. Standard solutions are also used to determine efficiency factors for unique geometries and energies. The results for analysis of americium-241 and strontium-90 stock standard solutions were within acceptable limits for gross alpha and gross beta radioactivities, as were results for gamma ray standard cesium-137 in both teflon jars and Marinelli beakers.

4.4.5.2 Background and Blank Samples

Cosmic radiation and natural radioactivity associated with a detector result in a positive background rate that is usually subtracted from the measured sample activity. Increases in the background rate from historical values can indicate contamination of the detector. For samples containing natural radioactivity in addition to the radioactivity of interest, blank samples can be used for the same purpose. (Blank samples are also used in determining unknown sample net rates.) Deionized water blanks showed less than 0.4 pCi/L gross alpha and less than 0.7 pCi/L gross beta radiation. Deionized water blanks showed less than 100 pCi/kg gamma radiation for samples in teflon jars and less than 5.1 pCi/L in Marinelli beakers.

4.4.5.3 Replicate Samples and Counts

Radiochemistry and sample preparation procedures can result in variations in the activities of aliquots of the same sample. Not all of these may be corrected by chemical yield determinations. Electronic instrumentation also can cause variation in measured sample activities over time. For these reasons, replicate samples and sample counts are performed on some of the samples within an analytical batch. Replicate activities should agree within



statistically determined limits. The analytical results for two duplicate sample pairs for gross alpha and gross beta radiation demonstrated acceptable replicability, with an average RSD of 8 percent for gross alpha determinations and 27 percent for gross beta. Duplicate counts performed on individual samples also demonstrated acceptable replicability, with an average RSD for duplicate gamma counts of 5 percent. Duplicate alpha and beta counts were also performed, but the radiation was too near the background level to allow estimation of analytical variability.

4.4.5.4 Spiked Samples

The ability of a detector system to accurately measure the activity of a sample should not vary with a small amount of additional "spiked" radioactivity. The detection "recovery" of the spike activity (after the subtraction of the normal sample activity) should approach 100 percent. Non-uniform dispersion of an alpha spike within the sample matrix can result in under- or over-correction due to absorption of the alpha particle within the sample Spiked beta activity is less susceptible to the within-mass absorption. mass. Samples were spiked with americium-241 for alpha activity and with strontium-90 for beta activity. Spike recoveries were 53 and 86 percent for alpha and beta activity, respectively, in a liquid sample, and 450 and 130 percent for alpha and beta activity in a solid sample. Recovery values for the liquid samples were reasonable, considering that the spike added was corrected by a factor of 15 before being used in the recovery equation. The recovery values for the solid sample are expected to be large, though closer to 100 percent for the higher energy beta activity than for alpha activity. The large alpha value is due to adsorption of the spiking solution on the surface of the sample particles, resulting in an overcorrection for adsorption.

4.4.5.5 Special Samples and Counts

EPA unknown samples and samples prepared by unique radiochemical techniques are used at various times to determine the overall precision and bias of laboratory procedures. Special samples are also used to define alpha



and beta sorption by the sample matrix. Other samples are used to monitor the energy discrimination windows in an alpha/beta counter. Analysis of an EPA interlaboratory unknown standard for cesium-137 demonstrated good performance, with a theoretical "recovery" of 112 percent, well within the EPA acceptability limit for the sample.

4.4.6 <u>Second Column Confirmation for GC Analyses</u>

As part of the quality assurance review of these data a random sample of EPA Methods 601 and 602 analyses of water samples where organics had been reported was conducted. Table 4.4-24 summarizes the results of this review. Chromatograms for both the first column and second column analyses were obtained from the laboratory and compared. Each of the six sets of analytical results was found to have been correctly assessed and reported by the laboratory. While all of the second column results were not compared during this review, this sample population can be considered to be representative of the rest of Methods 601 and 602 analyses. SECOND COLUMN CONFIRMATION RESULTS FOR METHODS 601 AND 602 TABLE 4.4-24.

		Firs Result	<u>First Column</u> 11t Analytical	Second Column Analytical	Confirmation
Monitor Well	Comp ound	(ug/L)	Date	Date	(Yes/No)
Method 601					
P-4	trans-1,2-Dichloroethylene	328	1/31/86	2/3/86	No
HM-7	trans-1.2-Dichloroethvlene		2/14/86	2/17/86	No
L-MH	Vinvl Chloride	55.6	2/14/86	2/17/86	Yes
HM7	Methylene Chloride		2/14/86	2/17/86	Yes
HM-7	1,1-Dichloroethane	4.23	2/14/86	2/17/86	Yes
HM-7	1,2-Dichlorobenzene		2/14/86	2/17/86	Yes
P-22	Vinyl Chloride	20.4	4/15/86	4/16/86	Yes
P-22	Trichloroethylene	46.3	4/15/86	4/16/86	Yes
HM-83	Trichloroethylene	7.42	8/ 26/86	8/26/86	Yes
HM-83	Tetrachloroethylene	0.10	8/26/86	8/26/86	Yes
Method 602					
HM-340	Toluene	125	2/28/86	3/4/86	Yes
HM-7	Benzene	60.7	2/14/86	2/15/86	Yes
HM-7	Ethyl Benzene	585	2/14/86	2/15/86	Yes
HM-7	1,2-Dichlorobenzene	44.0	2/14/86	2/15/86	Yes

ANNAN TARANI KANAN KANAN KANAN RABARI TARANI DEREM BUANNE KANAN KANAN KANAN

~\$^\$~\$~\$~\$~\$~\$~\$~\$~\$~\$

RADIAN



5.0 ALTERNATIVE MEASURES

This section discusses all major options for further IRP activities at AF Plant 4. It is preceded by a discussion of general considerations for further Phase II actions at AF Plant 4. The alternatives were developed after a thorough review of the data compiled from all investigations at AF Plant 4, including the current IRP Phase II Stage 1 study.

The recommendations developed in Section 6.0 are drawn directly from the alternative measures discussed in this section. The categories of alternative Phase II measures to be considered include:

- o Continued monitoring of existing wells;
- o Soil gas surveys;
- o Soil sampling and analysis;
- Installation of additional monitor wells;
- Records search/field reconnaissance; and
- o No further activities.

For those sites anticipated to require Phase IV Remedial Actions, a general, conceptual discussion of remedial options is provided. Since General Dynamics is initiating complete, plant-wide remedial action planning at the present time, the purpose of the following is to provide a limited overview, not to anticipate the results of the detailed planning that will take place.

5.1 General Considerations

Site conditions at AF Plant 4 have been extensively studied and are, in general, well known. General Dynamics has carried out extensive hydrogeologic investigations and has also initiated remedial action at many of the sites. This Phase II Stage 1 investigation was designed to complement those past and ongoing actions. Since the site investigation and remediation work sponsored by General Dynamics has also continued throughout this investigation, a detailed discussion of alternative measures is neither possible nor



appropriate. In lieu of such a discussion, this section focuses on those sites where ambiguities remain or where significant findings of this study do not overlap those of General Dynamics' activities.

In anticipation of the categorization of sites in Section 6, alternative measures for sites in a given category are discussed together. As part of the IRP program, existing sites are to be grouped as:

0	Category 1	No further IRP activities (including remedial action) required;
0	Category 2	Additional Phase II effort required; or
0	Category 3	Ready for IRP Phase IV actions.

There are eight Category 1 sites, one Category 2 site and twelve Category 3 sites (three of which comprise Zone 1).

5.2 Category 1 Sites

The following sites were investigated during this and previous studies and found to currently contain little or no hazardous material. On the basis of this finding, no further action is considered or recommended. For all listed sites, sufficient monitoring points exist.

o Site 17, Former Fuel Storage Site;

- o Site 2, Landfill 2;
- o Site 4, Landfill 4;
- o Site 6, Fire Department Training Area 3;
- o Site 7, Fire Department Training Area 4;
- o Site 18, Solvent Lines;
- o Site 19, Nuclear Aerospace Research Facility Area; and
- o Jet Engine Test Stand, Building 21.



Individual wells at the various sites, however, should continue to be considered for the plant-wide monitoring program discussed in Section 4.

5.3 Category 2 Site

Site 15, Fuel Saturation Area 2, was the only site identified for Category 2 during this investigation.

5.3.1 Additional Phase II Activities at FSA 2

Site 15, Fuel Saturation Area 2, was studied during this program, but insufficient data exist to fully characterize the nature and extent of contamination. Additional Phase II investigations are therefore recommended.

The analytical results of soil samples recovered indicate the presence of fuel contamination at this site. However, since the existing upper zone monitor wells appear to be upgradient of the site, the existence and extent of groundwater contamination remain unknown. Alternatives for this site include:

- Execution of an extensive program of soil borings and sample analysis to determine the extent of fuel contamination;
- Performance of a soil gas survey to delineate the maximum areal
 extent of fuel contamination; or
- Installation of up to two groundwater monitor wells downgradient (west and north) of the site, to determine the character of the upper zone groundwater exiting the site.

Either soil borings or a soil gas survey program would delineate the maximum area affected. Soil borings have the added benefit of determining geologic materials at depth, as well as contaminant concentration variations



with depth. Even though the site is currently paved, the soil gas survey would likely be faster and less expensive. However, until the migration of materials away from the site is confirmed, additional site investigations appear premature. Installation of two monitor wells in the upper zone, at or near the location of the borings made for this study, is recommended.

5.3.2 Remedial Alternatives at FSA 2

Fuel Saturation Area 2 was identified as requiring additional Phase II monitoring; therefore, a requirement for any remedial action cannot yet be fully defined. Based on the limited data available, it appears that excavation and redisposal of a limited additional amount of contaminated soil would eliminate any immediate threat to the environment. This alternative would have no continuing operation and maintenance (O&M) requirements, and only standard excavation safety practices would be needed. Further investigation is needed to fully define the extent of contamination associated with this site and to verify the need for any remedial action.

5.4 <u>Category 3 Sites</u>

The following sites were investigated during this and previous studies and found to contain varying levels of hazardous materials. However, all these sites currently have sufficient monitoring points and existing data which fully describe the nature of the problem. Many are the subject of current or past remedial action. Further, active remedial action planning is ongoing at AF Plant 4. Therefore, these existing, well-characterized sites should be released for Phase IV planning, as appropriate.

0	Site 1, Landfill 1;
0	Site 3, Landfill 3;
0	Site 12, Chrome Pit 3;
o	Zone 1 (Sites 13, Die Pits; 11, Chrome Pit 2; and 8, FDTA 5)
ο	Site 20, Wastewater Collection Basins;



- o Site 16, Fuel Saturation Area 3;
- o Site 9, Fire Department Training Area 6;
- o Site 10, Chrome Pit 1;
- o Site 5, Fire Department Training Area 2; and
- o Site 14, Fuel Saturation Area 1.

In addition to considering these sites for Phase IV planning, individual wells at the various sites should continue to be considered for the plant-wide monitoring program discussed in Section 4.

5.5 Remedial Alternatives

The following subsections discuss available remedial options for each of the Category 3 sites studied at AF Plant 4.

Sites 1, 3, 12, 20, 16, 9, 10, 5 and 14, as well as Zone 1 (Sites 13, 11, and 8) are recommended for release for Phase IV planning. This section presents a brief discussion of remedial action alternatives at each of the sites. Each alternative will be discussed with respect to engineering considerations, environmental effects, reliability and implementability, operation and maintenance requirements, off-site disposal needs, and safety considerations.

Two options have been identified at most of the sites. For those sites where some action has already been taken (e.g., excavation) only one additional alternative may be discussed. The no-action alternative is available for each site but has not been discussed in detail for every location. At some sites, groundwater extraction and treatment is identified as a possible option. Even though the option is discussed site by site, the actual implementation of any extraction alternative needs to consider the entire area's groundwater flow characteristics. If groundwater is to be extracted, a regional extraction field that addresses widespread contamination of the upper zone should be designed and implemented. Selection of specific treatment



technologies will depend on the nature of the contaminated groundwater (kinds and amounts of contaminants to be removed) and on an assessment of treatability by various methods.

5.5.1 <u>Site 1 - Landfill 1</u>

Alternative 1 - Further Excavation

Even though the waste oil pits, suspected to be the main source of contamination, were excavated in 1983, other parts of the landfill were used for disposal of drums and bulk liquids. Under this alternative, these other areas would be identified, excavated, tested and disposed of off site in an appropriate, approved disposal facility. Because of the proximity of the landfill to drainageways leading to Meandering Road Creek, special care would need to be taken to prevent the introduction of contaminated run-off and soils into the creek during excevation. Temporary erosion control procedures would also be required after backfilling with clean soil until repaving of the employee parking lot had been accomplished.

Special safety measures such as supplied air and skin protection, would be required since contaminated soils would be excavated. This alternative would require no O&M after reconstruction of the employee parking lot.

This alternative could reliably remove the source(s) of contamination, but would not address contamination that has migrated beneath or off the site. Also, groundwater contamination upgradient of the landfill (HM-53, HM-20) would not be addressed by this alternative.

Alternative 2 - Water Extraction and Treatment

Implementation of this alternative would entail the installation of extraction (and possibly injection) wells. The wells and well field would be designed to extract contaminated groundwater from directly beneath and







surrounding the landfill. Injection wells may be of benefit to help mobilize contaminants for more efficient collection through the extraction wells. Groundwater quality data from this investigation have shown the highest levels of contaminants upgradient (east) of the landfill at HM-20 and HM-63. The extracted water would be treated before discharge or introduction to the wastewater treatment system. The proper treatment of extracted contaminated water would prevent further environmental degradation. Treatment for the organic compounds present could include carbon adsorption or possibly biological degradation in an existing domestic wastewater treatment facility.

This alternative would demand reasonably extensive O&M activity for the duration of the pumping and treatment. Only standard drilling safety procedures, including skin and respiratory protection, would be required.

With proper design of the well field, this alternative would reliably extract contaminated groundwater in the vicinity of the landfill.

5.5.2 <u>Site 3 - Landfill 3</u>

Alternative 1 - Excavation

This alternative would involve the excavation and redisposal of contaminated soil at an approved off-site facility. Extensive effort would be required to prevent contaminated run-off and soils from entering adjacent Meandering Road Creek during excavation. Erosion-control measures would also be required after backfilling with clean soil until revegetation has occurred.

Special safety measures such as supplied air and skin protection, would be required, especially in the vicinity of well HM-38, since contaminated soils would be excavated using this alternative. This alternative would require no O&M after revegetation was completed.

This alternative could reliably remove the source of contamination, but would not address contamination that had migrated beneath or off the site.

Alternative 2 - Water Extraction and Treatment

Implementation of this alternative would entail the installation of extraction (and possibly injection) wells, as well as an interceptor trench between the landfill and Meandering Road Creek. Since alluvium and fill materials are generally less than 10 feet thick, an interceptor trench would probably be more cost effective than extraction wells for collecting contaminated groundwater. The trench would have a membrane placed at its downgradient wall, collection pipes and sumps installed, and be backfilled.

The water collected by the trench or extracted by the well would be treated before discharge or introduction to the wastewater treatment system. The proper treatment of the extracted contaminated water would prevent further negative environmental effects. This treatment could include a process to precipitate the metals in the groundwater and carbon adsorption to remove the organic compounds present. Provisions to prevent sediment introduction to Meandering Road Creek during trench installation would be required.

This alternative would demand reasonably extensive O&M activity for the duration of the pumping and treatment. Only standard construction and drilling safety procedures, including skin and respiratory protection, would be required.

With proper design of the well field and interceptor trench, this alternative could reliably extract contaminated groundwater in the vicinity of the landfill. However, the presence of thin and low-permeability soils would result in low withdrawal rates, thereby limiting the effectiveness of this alternative.

5.5.3 Site 12 - Chrome Pit 3

Water Extraction and Treatment

Since previous excavation removed all contaminated soils, only an alternative to address existing groundwater contamination is presented. The

principal groundwater contaminant at Chrome Pit 3 is TCE. Implementation of this alternative would entail the installation of extraction (and possibly injection) wells. The wells and well field would be designed to extract contaminated groundwater from beneath and in the vicinity of the pit. The extracted water would be treated before discharge or introduction to the wastewater treatment system. The proper treatment of extracted contaminated water would prevent further negative environmental impacts. This treatment could include a process to precipitate metals in the groundwater and carbon adsorption to remove the organic compounds present.

This alternative would demand reasonably extensive O&M activity for the duration of the pumping and treatment. Only standard drilling safety procedures, including skin and respiratory protection, would be required.

With proper design of the well field, this alternative would reliably extract contaminated groundwater in the vicinity of the pit.

5.5.4 Zone 1 (Site 13, Die Pits; Site 11, Chrome Pit 2; Site 8, FDTA 5)

Alternative 1 - Excavation

Soil testing is recommended to define the extent of contaminated soils. After the extent of contamination is determined this alternative would entail excavating the soil and disposing of it off site in an appropriate, approved disposal facility.

Since the ground surface in Zone 1 is flat and paved, containment of run-off and sediment during excavation and repaving should be relatively simple. Special safety measures may be required, depending on the levels of contamination encountered. These measures may include supplied air and skin protection. This alternative would require no O&M after repaving is completed.

This alternative could reliably remove the contaminated soils to the depth of excavation, but would not address contamination at greater depths.

Alternative 2 - Water Extraction and Treatment

Implementation of this alternative would entail the installation of extraction (and possibly injection) wells. The wells and well field would be designed to extract contaminated groundwater from beneath and in the vicinity of the pits and FDTA. The extracted water would be treated before discharge or introduction to the wastewater treatment system. The proper treatment of extracted contaminated water would prevent further negative environmental effects. The treatment could include a process to precipitate metals in the groundwater and carbon adsorption to remove the organic compounds present.

This alternative would demand reasonably extensive O&M activity for the duration of the pumping and treatment. Only standard drilling safety procedures, including skin and respiratory protection, would be required.

With proper design of the well field, this alternative would reliably extract contaminated groundwater in the vicinity of the pits and FDTA 5.

5.5.5 Site 20 - Wastewater Collection Basins

Water Extraction and Treatment

Since results of Phase II investigations indicate that only groundwater contamination is associated with this site, only one alternative is discussed. The basins and/or leaking drains need to be sealed. Further, a spill prevention program is needed to eliminate the opportunity for inappropriate chemicals to be introduced to the basins. Since this site is very close to and shares hydrogeologic conditions with Zone 1 sites, any remedial alternative would need to be evaluated with companion studies at Zone 1.



Implementation of this alternative would entail the installation of extraction (and possibly injection) wells. The wells and well field would be designed to extract contaminated groundwater from beneath and in the vicinity of the basins. The extracted water would be treated before discharge or introduction to the wastewater treatment system. The proper treatment of extracted contaminated water would prevent further negative environmental effects. The treatment could include a process to precipitate metals in the groundwater and carbon adsorption to remove the organic compounds present.

This alternative would have reasonably high O&M demands for the duration of the pumping and treatment. Only standard drilling safety procedures, including skin and respiratory protection, would be required.

With proper design of the well field, this alternative would reliably extract contaminated groundwater in the vicinity of the basins.

5.5.6 Site 16, Fuel Saturation Area 3

Water Extraction and Treatment

Remedial action alternatives for this site have been developed by Intellus, under contract to General Dynamics. The action presently defined by Intellus consists of the installation of extraction and injection wells to remove contaminated groundwater. The recovered groundwater would then be treated before discharge or use as make-up water.

The O&M requirements for this action would be relatively extensive. Only standard drilling safety measures, including skin and respiratory protection, would be needed.

This method of addressing contaminated groundwater uses existing technology and, if the well field is properly designed, should alleviate contamination.

5.5.7 <u>Site 9, Fire Department Training Area 6</u>

Alternative 1 - Excavation

Even though excavation has previously been conducted at this site, Phase II Stage 1 activities have found evidence of contaminated soil.

This alternative would entail further excavation and off-site disposal of contaminated soils at an appropriate, approved facility. and the second received become

لالالالالالكاليك

1.1.1.1.1

The extent of safety measures required will depend on the actual levels of contamination present, but the low concentrations of volatiles found indicate that extraordinary measures will probably not be needed. This alternative would require no 0&M after revegetation is complete.

This alternative will not mitigate any effects beyond the depth of excavation.

Alternative 2 - Landfarming

This alternative would involve landfarming the near-surface soils to enhance biological degradation of the organic compounds present. Nutrients would be added and the site tilled and irrigated periodically until organic contaminant levels have been reduced to an acceptable level. However, since soil is thin to nonexistent at this site, landfarming would probably not be an effective remedial alternative.

Alternative 3 - Long-term Monitoring

le Tenne State Contra Contr A 1997 A Martin Martin Martin Contra Contr

Upper zone materials do not occur at this site, so no shallow groundwater is present to monitor. Paluxy monitor well P-3 is appropriately located to monitor the impact of the site, but since it is screened throughout the thickness of the aquifer, the lack of positive analytical results for samples from this well is not sufficient evidence to rule out contamination at



the site. Possibly, contaminants occurring in the upper Paluxy are not being withdrawn by the well, which would draw the majority of its water from the more productive middle and lower Paluxy. The existing Paluxy monitor well will be sufficient to monitor any massive release of contaminants, but would probably not be sufficient to monitor small releases. In light of the past remedial action, which removed most of the contaminants that might have been present, the existing Paluxy well is marginally sufficient to serve as the continuing monitor well for the site.

5.5.8 Site 10 - Chrome Pit 1

Water Extraction and Treatment

Because the actual disposal pit is beneath Building No. 181, the contamination resulting from activities at this site can only be addressed by extracting contaminated groundwater and treating it before discharge.

The wells and well field would be designed to capture groundwater in the vicinity of the pit. The design of any extraction system would need to consider remedial actions contemplated at Zone 1 and the Wastewater Collection Basins. The extracted water would be treated before discharge or introduction to the wastewater treatment system. The proper treatment of extracted contaminated water would prevent further negative environmental impacts. Phase II activities identified organic contamination (i.e., trichloroethylene). This contamination could be removed using a carbon adsorption process.

This alternative would demand reasonably extensive O&M activity for the duration of pumping and treatment. Only standard drilling safety procedures, including skin and respiratory protection, would be required.

With proper design of the well field this alternative would reliably extract contaminated groundwater in the vicinity of the pit.

5.5.9 Site 5 - Fire Department Training Area 2

Alternative 1 - Excavation

If further soil testing indicates that near surface soils still contain high levels of contamination, these soils would be excavated to eliminate the source of groundwater contamination. The excavated soils would be disposed of off site at an approved facility.

This alternative would have no O&M requirements after repaying of the employee parking lot was complete. No special safety measures beyond normal excavation practices would be needed. Excavation will reliably remove contamination from the actual areas where soils is removed, but it will not benefit other areas.

Alternative 2 - Water Extraction and Treatment

Implementation of this alternative would entail the installation of extraction (and possibly injection) wells. The wells and well field would be designed to extract contaminated groundwater from beneath and in the vicinity of the FDTA.

The extracted water would be treated before discharge or introduction to the wastewater treatment system. The proper treatment of extracted contaminated water would prevent further negative environmental effects. The organic compounds present in the groundwater could be removed using a carbon adsorption process.

This alternative would demand reasonably extensive O&M activity for the duration of the pumping and treatment. Only standard drilling safety procedures, including skin and respiratory protection, would be required.

With proper design of the well field, this alternative would reliably extract contaminated groundwater in the vicinity of the FDTA.

5.5.10 Site 14 - Fuel Saturation Area 1

Water Extraction and Treatment

Intellus has developed remedial action alternatives for this site. The action presently defined by Intellus consists of the installation of extraction and injection wells and interceptor trenches to remove contaminated groundwater. The recovered groundwater would then be treated before discharge or use as make-up water.

The O&M requirements would be relatively extensive but only standard drilling and excavation safety measures, including skin and respiratory pro-tection, would be needed.

This method of addressing contaminated groundwater uses existing technology and, if the well field and trenches are properly designed and placed, should alleviate near-surface groundwater contamination.



h

[This page intentionally left blank.]

RADIAN

6.0 RECOMMENDATIONS

This section is a consolidated discussion of all major recommendations for further IRP activities at AF Plant 4. Consideration of remedial action alternatives was discussed in Section 5, but no recommedations were made concerning preferred remedial actions.

6.1 <u>Category 1 Sites</u>

The following sites were investigated during this and previous studies and found to currently contain little or no hazardous material. On the basis of this finding, no further action is considered or recommended for:

- o Site 17, Former Fuel Storage Site;
- o Site 2, Landfill 2;

a na na na na na na na n

- o Site 4, Landfill 4;
- o Site 6, Fire Department Training Area 3;
- o Site 7, Fire Department Training Area 4;
- o Site 18, Solvent Lines;
- o NARF Area; and
- o Jet Engine Test Stand, Building 21.

Individual wells at the various sites should continue to be considered for a plant-wide monitoring program.

6.2 Category 2 Site

Site 15, Fuel Saturation Area 2, was studied during this program but insufficient data exist to fully characterize the nature and extent of contamination. Additional Phase II investigations will be required. The following actions are recommended:

Installation of two monitor wells in the upper zone,
 downgradient (west and north) of the site, to determine the
 character of upper zone groundwater exiting the site.

RAPIAN

6.3 <u>Category 3 Sites</u>

The following sites were investigated during this and previous studies and found to contain varying levels of hazardous materials. All of these well-characterized sites should be released for Phase IV planning, as appropriate. Since General Dynamics is in the process of detailed planning of remedial actions on a plant-wide basis, specific recommendations for remedial action are not presented here. General categories of remedial technologies that should be considered in the planning process are listed by site:

o S:	ite	1,	Land	fi11	1
------	-----	----	------	------	---

- further excavation
- water extraction and treatment;
- o Site 3, Landfill 3
 - excavation,
 - water extraction and treatment;
- o Site 12, Chrome Pit 3
 - water extraction and treatment;
- o Zone 1 (Sites 13, 11 and 8)
 - excavation,
 - water extraction and treatment;
- o Site 20, Wastewater Collection Basin
 - water extraction and treatment;
- o Site 16, Fuel Saturation Area 3
 - water extraction and treatment;
- o Site 9, Fire Department Training Area 6
 - excavation,
 - landfarming.
 - long-term monitoring;
- o Site 10, Chrome Pit 1
 - water extraction and treatment;
- o Site 5, Fire Department Training Area 2
 - excavation,
 - water extraction and treatment; and



- o Site 14, Fuel Saturation Area 1
 - water extraction and treatment

Individual wells at the various sites should also continue to be considered for a plant-wide monitoring program.

